Copyright is owned by the Author of the thesis. Permission is given for a copy to be downloaded by an individual for the purpose of research and private study only. The thesis may not be reproduced elsewhere without the permission of the Author.

Responding to the Call to Care: Women's Experience of Breastfeeding in New Zealand

A thesis presented in fulfilment of the requirements

for the degree of

Doctor of Philosophy in Nursing at Massey University, Wellington, New Zealand.

Karen Sharee McBride-Henry

2004



Private Box 756 Wellington New Zealand T 64 4 801 5799 F 64 4 801 2692 www.massey.ac.nz

SCHOOL OF HEALTH SCIENCES

CANDIDATE'S DECLARATION

This is to certify that the research carried out for my Doctoral thesis entitled Responding to the call to care: Women's experiences of breastfeeding in New Zealand in the School of Health Sciences, Massey University, Wellington, New Zealand is my own work and that the thesis material has not been used in part or in whole for any other qualification.

Candidate's Name Karen Mithride - Henry
Signature Statebook
Date 5/8/04



Private Box 756 Wellington New Zealand T 64 4 801 5799 F 64 4 801 2692 www.massey.ac.nz

SCHOOL OF HEALTH SCIENCES

SUPERVISOR'S DECLARATION

This is to certify that the research carried out for the Doctoral thesis entitled Responding to the call to care: Women's experience of breastfeeding in New Zealand was done by Karen Sharee McBride-Henry in the School of Health Sciences Massey University, Wellington, New Zealand. The thesis material has not been used in part or in whole for any other qualification, and I confirm that the candidate has pursued the course of study in accordance with the requirements of the Massey University regulations.

Supervisor's Name Gillian White Ph)

Signature g. E. White

Date 5/8/04



Private Box 756 Wellington **New Zealand** T 64 4 801 5799 F 64 4 801 2692 www.massey.ac.nz

SCHOOL OF HEALTH SCIENCES

CERTIFICATE OF REGULATORY COMPLIANCE

This is to certify that the research carried out in the Doctoral Thesis entitled

Responding to the call to care: Women's experiences of breastfeeding in New Zealand in the School of Health Sciences at Massey University, New Zealand:

- (a) is the original work of the candidate, except as indicated by appropriate attribution in the text and/or in the acknowledgements;
- (b) that the text, excluding appendices/annexes, does not exceed 100,000 words;

that all the ethical requirements applicable to this study have been complied with as required by Massey University, other organisations and/or committees which had a particular association with this study, and relevant legislation.

Please insert Ethical Authorisation code(s) here: MUAHEC 99/086

Signature: All Market Signature: Signature:

Date:

Abstract

New Zealand breastfeeding experts have long contended that New Zealand does not have a breastfeeding culture, as demonstrated by anecdotal evidence suggesting that women find breastfeeding difficult to initiate and sustain. A review of the literature indicates that, in New Zealand, breastfeeding knowledge falls within the domain of health care professionals, which marginalises women's own experiential knowledge about breastfeeding. Therefore, this study explores the experience of breastfeeding for women in New Zealand. A reflective lifeworld research methodology underpins this study, allowing the participants' narratives to be explored without the use of pre-existing theoretical frameworks that may close down on aspects of the interpretive analysis. Nineteen women were interviewed for this study, all of whom were New Zealanders who were either breastfeeding at the time of the interviews, or had breastfed within the last two years. Many of the participants had breastfed more than one child.

What emerges as the central thesis of this study is that breastfeeding is a priori to unique embodied experiences. A number of sub-themes, which further explicate this central thesis, include: the silencing of the reality of breastfeeding within the public domain, the pervasive influence of society, or 'the they', through the accepted frameworks by which breastfeeding women interpret their individual breastfeeding experiences, and breastfeeding as a means of facilitating close relationships between women and their infants. The findings of this study will assist health care professionals working alongside breastfeeding women, as it offers fresh understandings of what it is to be a breastfeeding woman. It is important that health care professionals lay aside their previously-held perceptions about breastfeeding, and pay careful attention to individual women's experiences prior to planning interventions. If health care professionals value women's embodied breastfeeding narratives, women will be supported to articulate their breastfeeding experiences, thereby increasing women's confidence in their embodied breastfeeding knowledge and capabilities.

Acknowledgements

I am so grateful for the wisdom, experience and courage of the women who shared their life stories. They have touched my life, which is forever changed because they were open to talking about their lives. It is important that they know that, because of all they taught me, I was able to breastfeed my son.

My wonderful husband, Norm, has astounded me with all the ways he has managed to support me whilst I spent hours locked away with my computer, not the least of which was editing my final draft. I am grateful that he, a devout physicists, was open-minded enough to accept my phenomenological turning.

Matthew, my little gift, has taught me an incredible amount over the last 16 months. He has extended my horizon, and there is no going back.

I also need to acknowledge my family's contribution to this project. They have offered me endless encouragement, and bolstered my confidence during challenging periods. My mother always told me I should never settle for anything other than my best – I hope I have lived up to her expectations. I am so grateful for the family that my Mum and Dad built, which has provided me with such amazing and unceasing support.

I now appreciate the immensity of the commitment that Associate Professor Gill White and Associate Professor Cheryl Benn made when they agreed to supervise this process. I am grateful for the wise counsel they have both provided, which has shaped my thinking and writing.

The Journey

As a child thoughts bigger than words sought me

Criticised: My time, my place prevented thoughts from finding me

Fidelity's acceptance of my openness, my turning

Illuminates the breath that will lift me

My soul echoes with the voice of excitement

For the language that my body has always longed to speak is at last –

Coming to me

Karen McBride-Henry 16 June 1999

Table of Contents

CHAPT	ER ONE: INTRODUCTION AND BACKGROUND TO THE STUDY	1
1.1	INTRODUCTION	1
1.2	THE FOCUS OF THIS STUDY	
1.3	THE HEALTH BENEFITS OF BREASTFEEDING	
1.4	Women's Experience of Breastfeeding	
1.5	MY JOURNEY AS THE RESEARCHER	
1.6	BREASTFEEDING, A PRIORI TO NEW UNDERSTANDINGS	
1.7	OVERVIEW OF THE CHAPTERS	
1.8	CONCLUSION	
1.0	CONCLUSION	12
CHAPT	ER TWO: THE NEW ZEALAND CONTEXT	13
2.1	INTRODUCTION	13
2.2	BREASTFEEDING PRACTICES IN NEW ZEALAND PRE-1900	13
2.3	1900-1920	
2.4	1920-1950	
2.5	1950-1980	
2.6	THE 1980S	
2.6		
2.7	1990-2002	
2.7		20
	eastfeeding	25
	'.2 The Baby Friendly Hospital Initiative	26
	7.3 2002-2003	
2.8	SUMMATIVE THOUGHTS	
2.0	CONCLUSION	
2.9	CONCLUSION	
CHAPT	ER THREE: EXAMINING THE EXPERIENCE OF BREASTFEEDING	33
3.1	INTRODUCTION	33
3.2	THE INFLUENCE OF CULTURE ON BREASTFEEDING	
3.3	MAKING THE DECISION TO BREASTFEED	
3.4	INITIAL BREASTFEEDING EXPERIENCES	
3.5	EXPERIENCING BREASTFEEDING DIFFICULTIES	
	THE INFLUENCE OF SOCIAL SUPPORT FOR BREASTFEEDING WOMEN	
3.6	PROLONGED BREASTFEEDING	
3.7	THE ACT OF 'SUCCESSFUL' BREASTFEEDING	
3.8		
3.9	EXPLORING SENSUALITY, SEXUALITY, AND BREASTFEEDING	
3.10	BREASTFEEDING AND INTIMACY	
3.11	THE PROCESS OF WEANING	
3.12	BREASTFEEDING IN THE PRESENCE OF OTHER	
3.13	BREASTFEEDING AND EMPLOYMENT	
3.14	SUMMATIVE THOUGHTS	65
CHAPT	ER FOUR: INTERPRETIVE LIFEWORLD RESEARCH	67
4.1	INTRODUCTION	67
4.2	HISTORY OF PHILOSOPHICAL THOUGHT	
4.3	PHILOSOPHICAL UNDERPINNINGS OF PHENOMENOLOGY	/4

404		
4.3.1	Husserl: The Father of Phenomenology	75
4.4 HE	RMENEUTICS	78
4.4.1	Heidegger	80
4.4.2	Gadamer	
4.4.3	Merleau-Ponty	87
	PLICATION OF THE METHODOLOGY TO THE CURRENT STUDY	
4.5.1	Situating this Study	
4.5.2	Openness to the Phenomenon	
4.5.3	The Influence of History on Understanding	
4.5.4	The Hermeneutic Circle	
	DNCLUSION	
4.0 00	710E001011	
CHAPTER	FIVE: THE RESEARCH JOURNEY	94
	TRODUCTION	
	JIDING PRINCIPLES FOR THE DEVELOPMENT OF THE RESEARCH PROCESS	
	TENDING TO ETHICAL CONSIDERATIONS	
	E RESEARCH PARTICIPANTS	
	ATHERING PARTICIPANTS' NARRATIVES	
5.5.1	Continuing the Conversation	
5.5.2	Transcribing the Participants' Narratives	
5.5.3	Deciding to Stop Gathering Narratives	101
5.6 DA	TA ANALYSIS	
5.6.1	Initial Thematic Analysis	102
5.6.2	Constant Dialogue Between Participants and Philosophy	
5.6.3	Writing and Re-writing	103
5.6.4	Sharing Interpretations	
5.7 AT	TENDING TO ACCURACY OF INTERPRETATIONS	105
5.8 Su		
0.0	IMMARY	105
CHAPTER	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY'	107
CHAPTER 6.1 INT	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY'	1 07
CHAPTER 6.1 INT	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY'	107
CHAPTER 6.1 INT	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	107107107107
CHAPTER 6.1 INT 6.2 CC 6.2.1 6.2.2	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' TRODUCTION DMING TO UNDERSTAND BREASTS AND BREASTFEEDING The 'Always-alreadyness' of Breastfeeding Breastfeeding Women 'With-Other'	107107107107107
CHAPTER 6.1 INT 6.2 CC 6.2.1 6.2.2	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	107107107107107
CHAPTER 6.1 INT 6.2 CC 6.2.1 6.2.2 6.2.3	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION DMING TO UNDERSTAND BREASTS AND BREASTFEEDING The 'Always-alreadyness' of Breastfeeding Breastfeeding Women 'With-Other' Understanding the Breast as Functional HE THEY'	107107107107107109113
CHAPTER 6.1 INT 6.2 CC 6.2.1 6.2.2 6.2.3	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' TRODUCTION DMING TO UNDERSTAND BREASTS AND BREASTFEEDING The 'Always-alreadyness' of Breastfeeding Breastfeeding Women 'With-Other' Understanding the Breast as Functional	107107107107107109113
CHAPTER 6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3 TH	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	107107107107109113115116
6.1 INT 6.2 CC 6.2.1 6.2.2 6.2.3 6.3 'TH 6.3.1	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	107107107107109113115116
6.1 INT 6.2 CC 6.2.1 6.2.2 6.2.3 6.3 'TH 6.3.1 6.3.2	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	
CHAPTER 6.1 INT 6.2 CC 6.2.1 6.2.2 6.2.3 6.3 'TH 6.3.1 6.3.2 6.3.3	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	
6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3.1 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	
6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3.1 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	107107107107109115116123126129
6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3.1 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	107107107107109115116123126129
6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3.1 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	107107107107109115116123126129
6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3 'TH 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.4 SU	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION DMING TO UNDERSTAND BREASTS AND BREASTFEEDING	
6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3 'TH 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.4 SU	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' TRODUCTION	
6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3 'TH 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.4 SU CHAPTER 7.1 INT	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION DMING TO UNDERSTAND BREASTS AND BREASTFEEDING	
6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3 'TH 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.4 SU CHAPTER 7.1 INT	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' TRODUCTION DMING TO UNDERSTAND BREASTS AND BREASTFEEDING The 'Always-alreadyness' of Breastfeeding Breastfeeding Women 'With-Other' Understanding the Breast as Functional	107107107109115116123126135135
6.1 INT 6.2 CO 6.2.1 6.2.2 6.2.3 6.3 'TH 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.4 SU CHAPTER 7.1 INT 7.2 CA	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' TRODUCTION DMING TO UNDERSTAND BREASTS AND BREASTFEEDING The 'Always-alreadyness' of Breastfeeding	107107107109113116123126135135
CHAPTER 6.1 INT 6.2 CC 6.2.1 6.2.2 6.2.3 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.4 SU CHAPTER 7.1 INT 7.2 CA 7.2.1	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' TRODUCTION	107107107109113116123126135136136136136
CHAPTER 6.1 INT 6.2 CG 6.2.1 6.2.2 6.2.3 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.4 SU CHAPTER 7.1 INT 7.2 CA 7.2.1 7.2.2 7.2.3	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	107107107109113116123126135136136136136
CHAPTER 6.1 INT 6.2 CG 6.2.1 6.2.2 6.2.3 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.4 SU CHAPTER 7.1 INT 7.2 CA 7.2.1 7.2.2 7.2.3 7.2.4	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION MING TO UNDERSTAND BREASTS AND BREASTFEEDING The 'Always-alreadyness' of Breastfeeding Breastfeeding Women 'With-Other' Understanding the Breast as Functional HE THEY' Mirroring and the Call to Averageness Breastfeeding as Struggling in Silence Concernfully Uncovering the 'They-Self' When 'The They' Holds Sway Interpretations as Formed by 'The They' MMARY SEVEN: INHERITED UNDERSTANDINGS: THE BREAST AS OBJECTRODUCTION RETESIAN DUALISM The Body as an Object Understanding Objects Ready-to-Hand	
CHAPTER 6.1 INT 6.2 CG 6.2.1 6.2.2 6.2.3 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.4 SU CHAPTER 7.1 INT 7.2 CA 7.2.1 7.2.2 7.2.3 7.2.4 7.2.5	SIX: BREASTFEEDING WOMEN'S EXPERIENCE OF 'THE THEY' FRODUCTION	107107107109115116123126135136136136136136136136

CHAPTI	ER EIGHT: THE EMBODIED EXPERIENCE OF BREASTFEEDING	148
8.1	INTRODUCTION	148
8.2	PHILOSOPHICAL UNDERPINNINGS	149
8.3	A HIDDEN STORY	150
8.4	FIRST PICTURES	
8.5	BREASTFEEDING AS AN EXPERIENCE WITH EMBARRASSMENT	158
8.6	BREASTFEEDING AS AN EXPERIENCE WITH PAIN	160
8.7	BREASTFEEDING AS AN EXPERIENCE WITH TIREDNESS	170
8.8	BEING ANXIOUS	173
8.8.	1 Anxiety and the Fear of Failing	175
8.8.	2 Guilt	178
8.8.	3 Emotionally Crippling	181
8.9	CONCLUSION	186
CHAPT	ER NINE: BREASTFEEDING AS CONCERNFUL SOLICITUDE	188
9.1	INTRODUCTION	188
9.1	ONTOLOGICAL UNDERSTANDING OF CARE	
9.3	RESPONDING TO THE CALL TO CARE: COMING TO KNOW AND UNDERSTAND OTHER	
9.3		
9.3.		
9.4	BEING CLOSE	
9.4		
9.4.	· ·	
9.5	BREASTFEEDING AS A PRIVILEGED RELATIONSHIP	
9.6	SUMMARY	
CHAPT	ER TEN: DISCUSSION	218
10.1	INTRODUCTION	219
10.1	REVISITING THE AIM AND METHODOLOGY OF THE STUDY	
10.2	COMING-TO-KNOW: CREATING BREASTFEEDING KNOWLEDGE	
10.3	THE SHROUDING OF BREASTFEEDING IN SILENCE	
10.5	RE-CONCEPTUALISING THE OBJECTIFICATION OF BREASTS AND EMBODIMENT	
10.6	BREASTFEEDING REQUIRES PERSEVERANCE	
10.7	PROLONGING BREASTFEEDING: BREASTFEEDING AS RELATIONSHIP	
10.8	'THE THEY'S' INFLUENCE ON BREASTFEEDING BEHAVIOUR	
10.9	ALTERNATIVE INTERPRETATIONS FOR WOMEN'S BREASTFEEDING EXPERIENCES	
10.10		
10.11	IMPLICATIONS FOR HEALTH CARE PROFESSIONALS WORKING ALONGSIDE	
BREAS	TFEEDING	
	WOMEN	235
10.12	LIMITATIONS OF THIS RESEARCH	237
10.13	FUTURE DIRECTIONS FOR RESEARCH	238
10.14	Conclusion	240
CHAPTI	ER ELEVEN: THE CONCLUSION	242
11.1		
	REVISITING THE THESIS	
	A REVIEW OF THE CHAPTERS	
11.4	CONCLUSION	245
DEFEE		0.4-
KEFEKI	INCES	247

APPENDIX I:	INFORMATION SHEET	275
APPENDIX II:	CONSENT FORM	278
APPENDIX III:	TYPIST'S CONFIDENTIALITY CONTRACT	280

Chapter One

Introduction and Background to the Study

1.1 Introduction

The health benefits of breastfeeding for both mother and infant have been widely recognised, with research demonstrating that breastfeeding offers infants and young children the optimal source of nutrition, paving the way for a healthy start to life, while at the same time contributing to the mother's health (American Academy of Pediatrics, 1997; Australian College of Paediatrics, 1998; Riordan & Auerbach, 1993). Furthermore, research into the economic benefits of breastfeeding has highlighted the financial contribution breastfeeding women make to society through lower health care costs (Galtry, 1997; Waring, 2000). Despite all of these benefits, statistics indicate that the majority of New Zealand women wean their infants before six months of age (Ministry of Health, 2002); therefore, an investigation into what women in New Zealand experience when they breastfeed is warranted.

This chapter offers an introduction to this study into women's experience of breastfeeding in New Zealand. It outlines the biomedical literature that demonstrates the health benefits of breastfeeding, and locates alongside this the research that has examined women's experience of breastfeeding. To contextualise the researcher's personal journey, an account of how breastfeeding emerged as significant is presented. This chapter also provides a brief introduction to the methodology of reflective lifeworld research, which has facilitated this study. The chapter concludes with an overview of the research project.

1.2 The Focus of this Study

Scientific research has created a particular type of knowledge around breastfeeding; however, this knowledge often contradicts women's experience

of breastfeeding (Britton, 1997). It has been argued that women's experiences of breastfeeding within New Zealand have been silenced, and there has been a call to examine these experiences to facilitate a change in this society's perceptions of breastfeeding (Beasley & Trlin, 1998; Dignam, 2001; Watson, 2001). Furthermore, Dignam calls for research that explores the "psychological, emotional and interpersonal dynamics of breastfeeding" (MacLennan, 2001, p. 13).

Research indicates that women are well versed in the biomedical significance of breastfeeding for their infants, and are also aware of the economic benefits of breastfeeding (Earle, 2002; Galtry, 1997; Kessler, Gielen, Diener-West, & Paign, 1995; Meyer & de Oliveira, 2003; Wagner, Hulsey, Southgate, & Annibale, 2002; Waring, 2000). However, this knowledge appears to have limited influence on whether an individual woman will continue breastfeeding beyond the first few months. This is reflected in the national breastfeeding rates: forty nine percent of women wean their infants before three months of age (Ministry of Health, 2002), despite global (World Health Organisation & UNICEF, 1990; World Health Organization, 2002; World Health Organisation & UNICEF, 1989) and national policies (Ministry of Health, 1997; Ministry of Health, 1999a; Ministry of Health, 1999b; Ministry of Health, 2001; Ministry of Health, 2002) aimed at protecting, promoting and supporting breastfeeding.

Global and national initiatives often fail to take into account the lived experience of breastfeeding, essentially ignoring the cultural environment in which women breastfeed. Examining the meaning that breastfeeding has for women will allow the emergence of fresh understandings, thereby enabling health care workers to promote breastfeeding in a manner that takes into account the world in which women find themselves. Therefore, this research project asks the question: what is women's experience of breastfeeding in New Zealand?

1.3 The Health Benefits of Breastfeeding

Research has demonstrated that breastfeeding offers both women and their infants significant and long-term health benefits (Australian College of Paediatrics, 1997). There is strong evidence that, in developed nations, breastfed infants are less likely to suffer from a number of acute and chronic

illnesses; if they do, their illnesses are frequently less severe than for nonbreastfed infants. These include diarrhoea (Bhandari et al., 2003), otitis media (Duncan, Holberg, Wright, Martinez, & Taussig, 1993), lower respiratory infections (Bachrach, Schwarz, & Bachrach, 2003; Chulada, Arbes, Dunson, & Zeldin, 2003), urinary tract infections (Hanson et al., 2002), Haemophilus influenzae type B (Silfverdal et al., 2002), and bacteraemia (American Academy of Pediatrics, 1997; Heikkila & Saris, 2003). Breastfed infants are less likely to suffer from sudden infant death syndrome (Alm et al., 2002), and there is evidence that breastfeeding may provide protection from chronic illness such as insulin-dependent diabetes mellitus (Ravelli, van der Meulen, Osmond, Barker, & Bleker, 2000; Young et al., 2002), lymphoma (Davis, 1998), leukaemia (Infante-Rivard, Fortier, & Olsen, 2000), asthma (Chulada et al., 2003; Sears et al., 2002), atopic-related reactions (Kull, Wickman, Lilja, Nordvall, & Pershagen, 2002; van Odijk et al., 2003), Crohn's disease (Corrao et al., 1998; Silfverdal et al., 2002), high blood cholesterol levels (Plancoulaine et al., 2000; Ravelli et al., 2000), obesity (Gillman, 2002; Toschke et al., 2002), celiac disease (Ivarsson, Hernell, Stenlund, & Ake Persson, 2002), and ulcerative colitis (Corrao et al., 1998). Breastfeeding has also been linked to enhanced cognitive development (Dewey, Cohen, Brown, & Landa Rivera, 2001; Mortensen, Michaelsen, Sanders, & Reinisch, 2002).

Research also indicates a number of benefits for women who breastfeed. In the initial postpartum period, breastfeeding results in the increased secretion of oxytocin, decreasing the risk of haemorrhage (Riordan & Auerbach, 1993). Women are also more likely to return to their pre-pregnancy weight if they breastfeed (Dewey et al., 2001). Lactational amenorrhoea and delayed ovulation mean that women are less likely to become pregnant and they also lose less menstrual blood in the months following birth (Simondon, Delaunay, Diallo, Elguero, & Simondon, 2003). The long-term benefits of breastfeeding for women are also significant, and include enhanced bone re-mineralization (Paton et al., 2003; Polatti, Capuzzo, Viazzo, Colleoni, & Klersy, 1999), reduced risk of suffering hip fractures in older age (Cumming & Klinebergm, 1993), and reduced risk of developing breast and ovarian cancer (Collaborative Group on Hormonal Factors in Breast Cancer, 2002; Rosenblatt & Thomas, 1993).

In light of these health benefits, breastfeeding offers significant consequences for public health. As a result, organisations such as the United Nations

International Children's Fund (UNICEF) and the World Health Organisation (WHO) actively promote strategies that encourage breastfeeding. However, as the previously-mentioned statistics demonstrate, many New Zealand women do not breastfeed their infants beyond the first few months. The reasons for this trend begin to emerge through examining the research that explores the experience of breastfeeding in New Zealand.

1.4 Women's Experience of Breastfeeding

Research demonstrates that women gain knowledge about breastfeeding from their culture (Beasley & Trlin, 1998; Ryan & Grace, 2001). Examination of historical accounts of infant feeding in New Zealand also indicates that the past plays a major role in how breastfeeding is currently understood (Kedgley, 1996; Ryan, 1999a). This culturally-mediated awareness provides women with understandings of breasts and breastfeeding, dictating acceptable breastfeeding practices and influencing decisions about infant feeding.

Women's decisions to breastfeed are typically made during pregnancy (Aldridge & Stevenson, 2000; Arora, McJunkin, Wehrer, & Kuhn, 2000; Earle, 2000; Heath, Reeves Tuutle, Simons, Cleghorn, & Parnell, 2002), and are significantly influenced by the knowledge that breastfeeding provides infants with optimum nutrition, whilst also attending to their emotional wellbeing. These beliefs mean that women express a strong desire to breastfeed prior to giving birth.

However, women report that it can be more difficult to establish breastfeeding than they anticipated (Benn, Budge, Anderson, & Wright, 2002; Benson, 2000; Mozingo, Davis, Droppleman, & Merideth, 2000). They cite a lack of support from health care professionals and inconsistent advice as factors that contribute to negative breastfeeding experiences in the initial postpartum period (Benn et al., 2002; Cronin, 2003). The technical skill needed to effectively latch an infant on to the breast is reported to contribute to women's stress in the beginning phases of breastfeeding (Cronin, 2003). Perceptions of insufficient milk and unsettled infants also act as strong deterrents to breastfeeding (Beasley, 1996a; Beasley et al., 1998; Heath et al., 2002; Hill, Humenick, & West, 1994; Vogel & Mitchell, 1998a). However, the greatest difficulty faced by women who are establishing breastfeeding is pain, which is frequently attributable to sore and

cracked nipples (Benson, 2000; Mozingo et al., 2000; Page, Lockwood, & Guest, 2003; Schmied & Barclay, 1999; Schmied & Lupton, 2001), and may continue for months (Schmied & Barclay, 1999).

It is clear that many women wean their infants before they had intended to (Blyth et al., 2002; Mozingo et al., 2000; Vogel & Mitchell, 1998b), and it would appear that the major reason is a perception of insufficient milk (Blyth et al., 2002; Hill, 1992; Hill et al., 1994; Mozingo et al., 2000; Vogel & Mitchell, 1998b). Researchers have indicated that women often question their ability to produce enough milk to sustain their infants, especially if they are unsettled (Blyth et al., 2002; Heath et al., 2002). Other reasons cited for premature weaning include painful breastfeeding, unrealistic expectations, infant behaviours, a lack of knowledge, and the obligations of paid employment (Maclean, 1990; McLeod, Pullon, & Basire, 1998; Vogel & Mitchell, 1998b).

If a woman has confidence that she is capable of breastfeeding she is more likely to successfully establish breastfeeding (Blyth et al., 2002; Dennis, 1999; Ozturk Ertem, Kaynak, Kaynak, Ulukol, & Baskan Gulnar, 2001). As the length of time a woman breastfeeds increases, her confidence in her ability continues to grow (Blyth et al., 2002). Researchers often employ the term 'successful breastfeeding' to describe women who breastfeed for relatively long periods (Henderson, Stamp, & Pincombe, 2001; Inch & Fisher, 2000; Jones, 2001; Labbok, 2000). Women who are more likely to be successful at breastfeeding are well educated, older, and of European descent (Ford & Labbok, 1990; Heath et al., 2002; Ruowei & Grummer-Strawn, 2002). Women are also more likely to breastfeed if they are either not in paid employment, or are employed in a professional position (Carlson Gielen, Faden, O'Campo, Brown, & Paige, 1991; Hills-Bonczyk, Avery, Savik, Potter, & Duckett, 1993; Ryan & Martinez, 1989).

Women who continue to breastfeed their infants describe how breastfeeding can facilitate an intimate and close relationship with their infants (Bottorff, 1990; Dignam, 2001; Schmied & Lupton, 2001). Breastfeeding means that women feel a connection with their infants, and as a result women experience satisfaction, which supports their resolve to continue breastfeeding (Dignam, 2001; Schmied & Lupton, 2001).

1.5 My Journey as the Researcher

I was in high school during the late 1980s when the discourse of 'women's right to choice' was highly valued. The philosophy of choice, I believed, was the 'birth rite' of all women. The freedom to choose one's life path was a message and philosophy that I embraced as a young woman in the middle of defining who I was as a person. This belief affected the way I viewed the issue of breastfeeding: I believed that women had a choice as to whether they breastfed or not, and I did not understand why women chose not to.

I completed my undergraduate degree and subsequently registered as a nurse in the early 1990s. My degree programme did not contain much information about the importance of breastfeeding, nor did it provide me with the skills to support women who were breastfeeding.

I was aware that most women in New Zealand initiated breastfeeding, but that many did not continue to breastfeed. Within my clinical practice as a Child and Family Health Nurse, I also observed a preference within the acute care system for quantifiable methods of measuring infant nutrition, which is difficult to achieve if an infant is breastfed. This preference on the part of practitioners meant that breastfeeding women were frequently offered bottles of artificial formula¹ to feed to sick infants, so their intake could be measured.

As a nurse who resolutely believes in health promotion, I struggled to resolve the discrepancies between information provided by the WHO (World Health Organisation & UNICEF, 1989; World Health Organisation & UNICEF, 1990) and the everyday practice of health care practitioners in the acute care setting. This, coupled with the awareness that I lacked the skills and knowledge to advocate for, and support, the breastfeeding mothers whom I worked alongside, left me in a vulnerable position. These concerns led me to seek opportunities to learn more about the issues surrounding breastfeeding.

While studying towards a Master of Nursing degree, I enrolled in an elective paper on breastfeeding, which focused primarily on the social consequences of breastfeeding. I learned that breastfeeding is a multi-factorial issue and that

¹ Throughout this document, the phrase "artificial formula" is used to describe any milk-based product routinely offered to infants that is intended to replace human milk.

historical, social, cultural and political discourses all influence the manner in which breasts and breastfeeding are interpreted.

During the course of my study I came to position myself as a feminist, in part because of feminist writings that significantly influenced my understanding of women's health issues. I came to understand that women's bodies are frequently problematised and considered faulty (Carter, 1995; Greer, 1999), which has influenced my interpretation of research addressing women's health.

Following the completion of the breastfeeding paper I commenced a research project that focused on breastfeeding women's perceptions of infant care routines (McBride, 1999). I had hoped to gain insight into why breastfeeding women chose to use routines to manage the care of their infants, and I suspected that the need to reconcile breastfeeding routines with lifestyle prompted women to choose not to breastfeed. Instead, I discovered that a woman's choice of whether or not to use a structured routine does not influence her breastfeeding decisions. This left me asking the question: why, in light of the benefits of breastfeeding for both women and infants, were some women choosing not to breastfeed?

This led me to challenge my previously-held understandings about breastfeeding and choice, and I realised that I had no understanding of the lived experience of breastfeeding. Furthermore, the insight I needed was not fully available in the literature that examined the issue of breastfeeding. This desire to know what breastfeeding was like for women led me to embark on this doctoral study.

To investigate the fundamental question posed in this study, a specific methodology that valued the meaning that lifeworld experience held was paramount; therefore, reflective lifeworld research was chosen as the methodology to guide this research. This methodology is primarily based on the philosophical works of Heidegger (1962), Gadamer (1989) and Merleau-Ponty (1981). Reflective lifeworld research focuses on describing and interpreting lived experiences. It is a methodology that enables a researcher to examine phenomena in their entirety, and does not require previously-designed theoretical frameworks to prescribe interpretive directions.

Four years into this study, following the analysis of the findings, I gave birth to my first child. My son was premature and, due to severe reflux, was a challenge to breastfeed. During this period I became increasingly aware of the gift the women who had participated in this study had given to me, a novice breastfeeder. The information I had gained from their stories fuelled my desire to persevere, and gave me confidence that I was capable of rising to the challenge of breastfeeding. I was prepared for the difficulties that the establishment of breastfeeding held, and willing to endure a great deal of pain for the relationship that can be born through breastfeeding. Supported by the shared knowledge of the women who participated in this study, I am still breastfeeding my son a year later.

Another personal insight emerged when I was preparing the literature review nine months after the birth of my son. At that time I returned to Bottorff's (1990) article on persisting with breastfeeding. I marvelled at the wisdom and insight that emerged through her study, finding my own breastfeeding story mirrored in the narratives of the women who participated. Persisting with breastfeeding means adjusting to the unpredictable, the ever-changing ebb and flow of your infant's needs. It also necessitates giving birth to a new sense of self, which for me involved incorporating my baby as an integral part of me, a concept that is also discussed by Beasley (1996a) in her book about breastfeeding for the first time. Had I not readjusted my sense of self I would have fought the constant demands my son placed on me. It was important that I gave myself the freedom to be what I was called to be, and fight societal forces that attempted to bridle my breastfeeding philosophy.

Ultimately, this research has had a tremendous effect on me, both professionally and personally. However, had I not been influenced by the context in which I found myself as a nurse, the issue of breastfeeding would not have risen to the fore and held significance for me. Heidegger (1962) calls us to conversation. He calls to us to interpret his philosophy by way of conversation with that which we encounter in the world in which we find ourselves. This leads to differing ways of understanding the lifeworld. This research is my attempt to converse with the philosophical developments of Heidegger (1962), Gadamer (1989) and Merleau-Ponty (1981), to facilitate an understanding of the lifeworld of a woman who breastfeeds. It is my hope that this conversation will call you to thinking; that it will call you to engage with the 'taken-for-granted' in

a way you might otherwise not have. This, as with any interpretation, is dynamic and never finished – your own interpretations, as a result of engaging in this interpretation, will hopefully extend your own lifeworld horizon, facilitating a different way of perceiving and understanding the lifeworld in which we exist.

1.6 Breastfeeding, A Priori to New Understandings

Some experiences result in transforming one's life perspective, causing one to extend their horizon, giving birth to new and exciting possibilities that would otherwise never have existed. This "establishment of a new dimension of experience is the setting forth of a priori" (Merleau-Ponty, 1962, p. 30). Becoming a breastfeeding woman is a priori, as it irrevocably alters a woman's perspective and facilitates the emergence of expanded horizons; thus, breastfeeding provides a portal for new ways to understand and interpret, altering a woman's fundamental way of 'being-in-the-world'.

Breastfeeding provides women with access to new experiential knowledge, and becomes a priori to understanding embodiment, close relationships and 'the they'. The embodied understandings gained by the women in this study can only be birthed through experiencing breastfeeding. The role of experiential knowledge "opens up distinctly differentiated lifeworlds of perceptions, concerns and understandings" (Benner, 2000, p. 8). The experiential learning and emotions of breastfeeding women open up a differentiated lifeworld of awareness and concernful caring, concepts that are further explicated in the following pages.

1.7 Overview of the Chapters

Chapter Two of this study explores the history of infant feeding in New Zealand, outlining how women have historically interpreted breastfeeding in accordance with biomedical discourses. It also locates the role of both global and national policies to promote and support breastfeeding, due to its importance for public health.

In Chapter Three the literature that describes the experience of breastfeeding is examined. The literature review is not presented in a traditional manner, which might include a critical examination of methodologies and research locations; instead, this review takes an approach consistent with reflective lifeworld research and focuses on the meaning that the experience of breastfeeding has for women. This review demonstrates that health care professionals interpret breastfeeding as problematic, and that intervention is needed by professionals if breastfeeding is to be achievable. Researchers also paint a picture of women as incapable of knowing and articulating their breastfeeding experiences. An argument for re-examining the issue of breastfeeding, placing the emphasis on women's perspectives and knowledge of breastfeeding, is offered.

Chapter Four discusses the methodology of reflective lifeworld research, which offers the researcher the ability to re-examine the experience of breastfeeding afresh without any pre-conceived frameworks to limit their interpretations. As a part of this discussion, the reader is introduced to the principles that have guided the research process. Chapter Five then describes the process of reflective lifeworld research as it played itself out in the context of this study. The study's design, participant recruitment, and ethical considerations are also highlighted.

Chapters Six through Nine present the findings of this study. Particular emphasis is placed on the participants' narratives, as it is my contention that the language used by women to describe breastfeeding is often silenced, and as a result their knowledge is not visible in much of the research that has explored breastfeeding. Therefore, in these chapters I have focused on rendering the participants' narratives, deferring my interpretations of them until Chapter Ten.

The sixth chapter of the study begins by describing how women come to understand breasts and breastfeeding through dwelling with others. It also highlights that women interpret their breastfeeding experience in accordance with the socially-prescribed images of the perfect breastfeeding mother. Such images frequently confine women to a certain way of behaving, causing them to hide from public scrutiny if they do not mirror the perfect breastfeeder. Failure to measure up to externally-prescribed edicts means that breastfeeding women suffer in silence from emotions such as guilt.

Chapter Seven discusses how the participants came to understand breastfeeding through the inherited language of the body as object. The objectification of women's bodies means women have come to understand through language that their breasts are objects that need to be managed and tamed to be effective. The participants spoke about their breasts and breastfeeding by employing medical terms, and these accounts are explored using Heidegger's (1962) concepts of objects. However, it became apparent that a different language was required if women's experience of breastfeeding was to emerge in its entirety.

Chapter Eight demonstrates that a language that meaningfully describes breastfeeding emerged as the participants let the medicalised 'body-as-object' language fall away. What remained was a language that described breastfeeding as an embodied act, and the participants' narratives are discussed in light of Merleau-Ponty's (1981) writings on embodiment.

The last findings chapter, Chapter Nine, describes how women need to know their infants in a close way if breastfeeding is to be prolonged. The participants describe how over time they came to understand that breastfeeding offers the ultimate source of comfort for their infants. The closeness and exclusive nature of the relationship that breastfeeding offers was valued by those participants who continued to breastfeed. This chapter focuses primarily on how 'breastfeeding as a technical function' falls away and 'breastfeeding as relationship' emerges as significant for women who breastfeed over time.

Chapter Ten offers a discussion of the findings, describing how the reality of breastfeeding is silenced in New Zealand. Women who struggle to measure up to the perfect breastfeeding mother hide themselves from the public domain and reinforce unachievable ideals about breastfeeding. However, the emergence of a language that values embodied knowledge offers a path to illuminating breastfeeding in a meaningful manner. The implication of this knowledge for health care professionals is explored in the context of the findings. The limitations of this study are discussed, as is the need for future research projects.

1.8 Conclusion

When one gazes upon a mother breastfeeding her baby, one might surmise about the meaning breastfeeding has for that woman. It might mean providing her baby with needed nutritional requirements, or perhaps for her it means closeness. However, for that woman it may also symbolise a triumph over pain, or possibly physical weariness. "There is no way of understanding them [or the experience of breastfeeding] other than by looking at them, but to the beholder they say what they mean" (Merleau-Ponty, 1962, p. 321). This study attends to the multifaceted meanings that breastfeeding holds for women who breastfeed their infants, and seeks to understand what it is to breastfeed a baby in New Zealand. It begins with an examination of how the history of infant feeding in New Zealand has affected current understandings of breastfeeding.

Chapter Two

The New Zealand Context

2.1 Introduction

In this chapter, the history of infant feeding practices in New Zealand is explored through periods ranging from pre-1900 to the present, to place this study in a historical context. Prior to European colonisation in the nineteenth century, Maori history was primarily passed on through spoken word, and little has been written about the infant feeding practices of New Zealand's indigenous people. Therefore, this chapter focuses on the period from the late nineteenth century to the beginning of the twenty-first century, for which written history is available.

2.2 Breastfeeding Practices in New Zealand Pre-1900

Life in New Zealand prior to 1900 was pre-industrial. There were no mass-produced commercial goods, and although artificial infant formula had been produced in Europe since the 1840s, it did not become available in Australia until the 1880s, and not in New Zealand until the 1900s (Apple, 1994). English immigrants tried to establish lifestyles similar to those they had previously led. At that time in England, mothers were encouraged to follow the example of Queen Victoria and breastfeed their children, shouldering the responsibility of caring for them; mothers who chose not to breastfeed were viewed as selfish (Yalom, 1997). These Victorian values informed the manner in which New Zealand immigrants viewed breastfeeding.

After birthing, women of European decent were typically encouraged to 'lie-in' to assist in the establishment of breastfeeding. Often a neighbour or midwife would assist the new mother by living-in with her for a few weeks, and it is reported that Maori women commonly assisted in teaching immigrants to become lay midwives (Kedgley, 1996). There was a clear sense of community

responsibility for the nurturing of infants, and women's sphere of knowledge and expertise incorporated infant feeding (Kedgley, 1996).

Unfortunately, another defining characteristic of the lives of women in pre-1900 New Zealand was a relatively high infant mortality rate. In 1898, 19,000 infants of European descent were born in New Zealand, with the death rate being 79 per 1000 live births prior to one year of age (Davidson, 1984). Although this rate was approximately half that of England, infant deaths would certainly have been very common by today's standards. Maori birth and mortality rates were not documented reliably until 1930 (Davidson, 1984), although it has been estimated that the death rate was one in five infants before the age of one year prior to European colonisation (Pool, 1991; Pool & Bedford, 1996). Most deaths, of both European and Maori infants, have been attributed to infectious diseases, in particular gastroenteritis resulting from inadequate sanitation (Davidson, 1984).

Towards the end of the nineteenth century the medical profession had taken the first steps towards medicalising infant health and feeding, in effect taking over part of the responsibility for overseeing the physical growth and nutrition of infants (Apple, 1994; Baumslag & Michels, 1995; Davidson, 1984). It was believed by many physicians that specialised knowledge of infants and infant feeding was required if the infant mortality rate was to be reduced (Apple, 1994; Baumslag & Michels, 1995; Ryan & Grace, 2001). This trend was illustrated with the publication of the first book addressing infant feeding, by Dr Saunders in 1892 (Davidson, 1984). Saunders strongly advocated for breastfeeding, but suggested a recipe for artificial formula to be used in the event that a mother was unable to breastfeed (Davidson, 1984).

In England and the United States the medical fraternity and commercial manufacturers joined forces in the production and marketing of artificial formula (Baumslag & Michels, 1995). According to Baumslag (1995), their partnership was portrayed as intelligent men finally stepping in and helping comparatively uneducated women to raise healthy infants. This value system began having an effect on New Zealand culture around the beginning of the twentieth century (Apple, 1994; Davidson, 1984).

2.3 1900-1920

During the early twentieth century, New Zealand's infant mortality rate continued to concern government officials, which in part led to the founding of the Department of Health. The estimated mortality rate for European infants was between 62 and 100 per 1000 live births during the first decade of the twentieth century (Davidson, 1984; Dow, 1999). During the same period, it has been suggested that approximately half of all Maori infants died within their first year of life, which represents a substantial increase over the estimated precolonisation mortality rate (Dow, 1999). At that time, it was noted that infants fed on artificial formula were more likely to succumb to illness than breastfed infants; therefore, doctors advocated once again for breastfeeding, while ardently debating the best formula for producing an artificial milk (Beckman, 1977; Davidson, 1984). Doctors began monitoring infants through regular surveillance in their medical clinics, prescribing particular infant care regimes including advice on formula mixtures (Davidson, 1984).

During this period Dr Truby King was the medical superintendent of Seacliff Mental Hospital, where he studied science-based farming methods. The application of scientific methods to animal farming produced impressive results, and Dr King took a particular interest in dairy farming and milk production. This work profoundly influenced his views of infant health and feeding, and he began a publicity campaign that focused on educating parents about infant health. To prove his claims, he cared for sickly infants in his own home, demonstrating that unwell children could survive on 'humanised' cows milk (Davidson, 1984; Kedgley, 1996).

In 1907, King founded The Royal New Zealand Society for the Health of Women and Children, now commonly known as the Plunket Society after the Society's patroness, Lady Plunket. The organisation opened hospitals dedicated to caring for sick and premature infants, and training mothers in childcare. The Plunket Society also commissioned nurses to visit the homes of new mothers to monitor infant growth and development. Plunket nurses provided instruction on mother-craft, which attended to infant nutrition and regimens for both breastfed and formula-fed infants. For example, infants were to be fed at four-hourly intervals with no more than five feeds per day, and overnight feeding was expressly prohibited, a protocol that stemmed from King's research into dairy

farming. These rules remained in place for 40 years, mirroring King's belief in a standardised, scientific approach to infant feeding (Apple, 1994; Davidson, 1984; Kedgley, 1996).

Women's initial breastfeeding practices were also influenced by the formal registration of midwives in New Zealand in 1904, which coincided with the opening of the country's first maternity hospitals. The first of the St Helens Hospitals opened in Dunedin in 1905, and had a significant influence on the training of midwives (Mein Smith, 1986). Registered midwives commonly advised new mothers on breastfeeding during the initial postpartum period, encouraging women to demand-feed their infants and suggesting special diets intended to increase milk supply (Kedgley, 1996). However, after this initial period, Plunket nurses assumed the task of advising women on issues such as infant feeding.

Plunket nurses based their infant feeding advice on a perceived need for strict scheduling of infant care. This approach was difficult for breastfeeding mothers to follow, and often resulted in weaning to artificial formula to adhere to the required schedule. Despite the negative effect on breastfeeding, the mortality rate for infants of European descent did drop significantly following the Plunket Society's formation: the death rate fell from 73 deaths per 1000 live births in 1907 to 48 deaths per 1000 live births in 1917 (Apple, 1994; Davidson, 1984). However, it has been suggested that this was due primarily to improvements in sanitation and the end of the influenza epidemic in 1907 (Kedgley, 1996).

Because they lived primarily in rural areas, Maori infants fell outside the Plunket Society's sphere of influence, which was essentially urban (Davidson, 1984). No Maori births or deaths were registered until after 1920, which effectively made the plight of Maori infants invisible (Pool, 1991). However, it has been estimated that, prior to 1920, approximately half of the Maori population was wiped out, and that approximately half of all Maori children died before the age of four (Davidson, 1984; Pool, 1991). The majority of the deaths have been attributed to infectious diseases such as tuberculosis, malnutrition, and birthing difficulties (Dow, 1999; Durie, 2001; Pool, 1991; Ramsden, 2001). However, it was widely held among European settlers that the high infant mortality rate amongst Maori was the result of parental ignorance and incompetence (Dow, 1999).

When the Department of Public Health was established in 1901, a Maori Medical Officer, Dr Maui Pomare, was appointed to oversee the development of public health care for Maori. Dr Pomare attended the American Missionary College in Chicago, graduating from medical school in 1899 (Cody, 1953). He was reportedly a strong leader who worked closely with Maori elders to address issues affecting Maori health (Durie, 1998; Durie, 2001). He also focused on issues that indirectly influenced health status amongst Maori such as cultural values, housing and economics (Durie, 2001).

Although Pomare was a strong advocate for Maori children and believed a complete change was needed in the way that health care was provided for Maori (Davidson, 1984), at that time there was little support from people of European descent for re-orienting health services for Maori, some believing instead that the race should be left to die out (Pool, 1991; Ramsden, 2001). Pomare spent eight years pleading with his colleagues in the Department of Health for the financing of proposed reforms, without much success, and finally resigned in 1909 to pursue a career in politics (Cody, 1953; Davidson, 1984; Dow, 1999; Durie, 2001). He was eventually appointed the Minister of Health and served in that position from 1923 to 1926 (Dow, 1999), during which time he improved access to education, quality housing and employment (Cody, 1953; Durie, 2001). Other significant advancements that were made through Pomare's efforts were the introduction of community health workers and the enhanced recruitment of Maori nurses. Both of these groups were strongly encouraged to work in collaboration with Maori communities to improve the health status of Maori (Dow, 1999; Durie, 1998; Durie, 2001).

As noted above, by 1917 the mortality rate for European infants had fallen to 48 per 1000 live births; in contrast, by the 1920s the mortality rate for Maori ranged between 107 and 160 per 1000 live births (Dow, 1999; Pool, 1991). Although these figures represent an improvement in the plight of Maori infants, they highlight the continuing disparity between the health of Maori infants and those of European descent.

The overarching message of the early 1900s was that success at raising and feeding healthy infants required specialty knowledge that was the domain of health care professionals, whether they were registered midwives, doctors or Plunket nurses. In effect, women had lost some of their control over childcare, as their knowledge was no longer valued in relation to either child rearing or

infant feeding (Beckman, 1977; Davidson, 1984; Kedgley, 1996; Ryan & Grace, 2001). However, with the emergence of the suffrage movement in the early part of the century, women had become politically active, lobbying not just for the right to vote but also to achieve autonomy as both citizens and women. This new-found sense of self-determination sufficiently influenced infant feeding practices that it became unfashionable for middle-class women to breastfeed; mothers who chose not to breastfeed were expressing their personal right to freedom (Kedgley, 1996).

2.4 1920-1950

The use of artificial formula became widely accepted during the 1920s and, although many women continued to breastfeed, formula feeding was increasingly common. The acceptance of artificial formula assisted the development of industry in New Zealand, and in 1927 the Karitane Products Society was formed. The organisation coined the term 'humanised milk' as an aid in marketing the formula developed by King (Davidson, 1984), which proved to be an effective means of portraying formula as an acceptable substitute for breastmilk. Proceeds from this company were used to support The Royal New Zealand Society for the Health of Women and Children (Davidson, 1984), and contributions are still made by Karitane Products Society to the Royal New Zealand Plunket Society.

During the 1920s a dramatic shift in birthing practices took place. The perceived need on the part of medical practitioners to remove contamination from childbirth, as a means of reducing mortality rates for both infants and mothers, led to a rapid increase in the number of births occurring in hospitals (Kedgley, 1996: Mein Smith, 1986). In 1927, 60 percent of births in New Zealand took place in a maternity hospital; by 1936, this rate had increased to over 80 percent (Apple, 1994). During a woman's fourteen-day stay in the maternity hospital, feeding regimes were highly structured, which significantly influenced breastfeeding initiation. Women were allowed only limited access to their infants, feeding times were restricted, night feeds were prohibited, and infants were weighed after feeds to assess the amount consumed (Apple, 1994; Kedgley, 1996).

The shift to birthing in maternity hospitals contributed to the normalisation of birthing, mother care, and infant feeding as medicalised practices (Ryan & Grace, 2001). Once a mother and child were discharged home, a Plunket nurse would take over the monitoring and instruction of the new mother, reinforcing the notion that women were incapable of caring for their infants without professional assistance (Kedgley, 1996). The re-conceptualisation of motherhood as a medicalised scientific practice was, therefore, solidified during this era.

The 1930s had also seen serious debate about the composition of artificial formula. A number of medical practitioners critiqued King's recommended formula, claiming it contributed to anaemia and other vitamin deficiencies (Davidson, 1984). However, by the 1940s the medical community had accepted that artificial formula was of equal nutritional value to breastmilk (Ryan & Grace, 2001). A number of justifications for the use of formula rather than breastmilk were suggested during this era, including inadequate nutritional status for breastmilk, having fair skin, flat nipples, or a chronic illness. Infants also came under scrutiny in relation to their ability to breastfeed. At that time it was recommended that infants that were either large or small, had jaundice, or had a physical defect should not be breastfed (Ryan & Grace, 2001). Meanwhile, pressure from health care professionals to adhere to strict routines for both bottle-fed and breastfed infants also continued (Dignam, 2001).

Events of the 1940s had a significant influence on the role of women within the community. During the Second World War many men left New Zealand to fight in Europe or the South Pacific, leaving women to carry on much of the essential work outside of the home. During the period from 1940 to 1945 no breastfeeding statistics were kept; however, it has been suggested that approximately 70% of women initiated breastfeeding, with 44% of infants still being breastfed at three months and 21% at six months (Deem, 1945-1946, as cited in Davidson, 1984).

The 1940s were characterised by women's compliance with the mandates of the scientific and medical communities in relation to infant feeding (Ryan & Grace, 2001). Ryan (1999b) describes how women embraced a 'not for me' stance in relation to breastfeeding; breastfeeding was perceived as untenable because of potential difficulties and breastmilk's questionable quality. The practices adhered to in maternity hospitals continued to undermine women's

ability to breastfeed. These practices included complementary feeds, extremely limited access to infants during the first three days, and continued adherence to four-hourly routines (Kedgley, 1996). It has also been documented that when women were attempting to initiate breastfeeding, both they and their infants were handled roughly by nursing staff, further discouraging women to breastfeed (Kedgley, 1996). Such practices continue to the present day (Benn et al., 2002; Vogel & Mitchell, 1998a). To solve the difficulties of breastfeeding, which were often caused by hospital policies, women were offered the benefits of technology in the form of artificial formula (Kedgley, 1996; Ryan, 1999b).

Ryan and Grace (2001) described the scientific, technological, and medical establishments' influence over infant feeding practices during this era in the following manner: "It is difficult to illustrate the dominance of this knowledge source because it is so taken-for-granted, all-pervasive, and incorporated into every day understandings and language" (p. 492). This ubiquitous ownership of infant feeding practices continued throughout the 1950s and into the 1960s (Apple, 1994; Davidson, 1984; Dignam, 2001; Kedgley, 1996; Ryan & Grace, 2001).

2.5 1950-1980

In 1952, a group of women concerned about their birthing experiences in hospitals formed the Parents' Centre to facilitate antenatal education for women. The women involved with Parents' Centre were politically active, lobbying for women's rights during childbirth and actively encouraging women's unrestricted access to their newborn infants (Kedgley, 1996). Parents' Centre became a place where women could meet and discuss their experiences with birthing and mothering. Women were encouraged to listen to their instincts when caring for their infants, and the use of strict routines was challenged as it prevented women from responding to their infant's psychological needs (Kedgley, 1996). The establishment of community networks such as the Parents' Centre foreshadowed a major change in the way that women valued their own knowledge, which ultimately led to the feminist movement of the 1960s. Women's ways of knowing became valued in this setting, and women began to question the hegemonic sway held by the medical fraternity. A shift in

the ideology of infant feeding was thus initiated by feminists and the associated feminist movement (Carter, 1995; Ryan & Grace, 2001; Vares, 1992).

Despite changing perceptions amongst women, only 47 percent of infants were breastfed by the end of the 1960s (Davidson, 1984). In contrast, approximately 87% of infants were breastfed during the 1920s (Royal New Zealand Plunket Society, 1995). The drop in breastfeeding rates has been attributed in part to the practices of health care professionals in relation to infant feeding (Kedgley, 1996), as discussed in the previous section. However, social pressure also played a role in reducing breastfeeding rates. During this era, Western societies viewed the breast as a sexual item, rather than a functional part of a woman (Carter, 1995; Yalom, 1997). As a result, women felt embarrassed to 'expose' their breasts whilst breastfeeding, effectively relegating breastfeeding to the private sphere. Breastfeeding was, therefore, viewed as a perverse, indecent practice. These socially-prescribed understandings effectively left women who chose to breastfeed their infants isolated from public places (Kedgley, 1996).

The breastfeeding support group La Leche League emerged during the mid-1960s, with the express purpose of supporting and valuing women's knowledge of breastfeeding and infant care (Gordon, 1998; Kedgley, 1996). La Leche League groups grew rapidly in New Zealand, especially during the 1970s. It has been interpreted as primarily a white middle-class organisation; however, it has been noted that the rise in the number of women choosing to breastfeed during the 1970s and 1980s mirrored the growth of the number of La Leche League groups (Gordon, 1998), suggesting their influence assisted a general change in attitude.

The women involved in La Leche League groups were vehement in their desire to re-position breastfeeding expertise within the realm of women's knowledge rather than that of health care professionals (Gordon, 1998). The La Leche League had radical ideas about breastfeeding at that time, and completely contradicted existing hospital practices of restricted access and regimented scheduling of infant feeding (Apple, 1994; Gordon, 1998; Kedgley, 1996). Its philosophy embraced a mother's right to actively participate in breastfeeding at birth, unlimited access to the infant whilst in hospital, breastmilk as the best source of nutrition during an infant's first six months of life, the provision of sensitive care oriented to the infant's needs, and that the timing of weaning be

directed by the infant (Gordon, 1998). La Leche League's success is reported to have stemmed from the organisation's foundational belief that women who are mothers are the experts in breastfeeding (Palmer, 1993). Because of the work of such organisations during the 1970s, women began to oppose the notion of strict infant routines, focusing instead on responsive mothering of infants (Ryan & Grace, 2001).

The resurgence of women-centred knowledge of breastfeeding freed women to make informed choices about whether or not to breastfeed their infants. During this decade, increasing numbers of women were initiating breastfeeding, and continuing to feed their infants for longer periods (Ford, Wild, Mitchell, & Tuohy, 1995; Msuya, Harding, Robinson, & McKenzie-Parnell, 1990; Royal New Zealand Plunket Society, 1995). It has been suggested that this was the result of the feminist movement as well as the work of organisations such as Parents' Centre and La Leche League (Carter, 1995; Kedgley, 1996; Palmer, 1993; Ryan & Grace, 2001).

Ryan and Grace (2001, p. 493) describe how, during this era, women created a language about breastfeeding that included scientific-sounding depictions of breastfeeding, such as 'infant-led feeding' and an infant being 'positioned at the breast'. This re-framing of breastfeeding assisted women to once again normalise breastfeeding. Health care professionals adopted this new breastfeeding language, incorporating it into research studies that supported what women had been asserting about the beneficial nature of breastfeeding (Ryan, 1999a). As a result, the catch phrase 'breast is best' became popular during the 1970s as research results supporting the superiority of breastmilk began to emerge (Anyon, 1976; Fergusson, Horwood, Shannon, & Taylor, 1978; Hood, Faed, Silva, & Buckfield, 1978; Kedgley, 1996; Kerr, 1981; Roberts, 1980; Ryan & Grace, 2001).

2.6 The 1980s

The research conducted during the 1970s clearly demonstrated the superiority of women's milk, and breastfeeding, over bottle feeding with artificial formula. These results, coupled with reports from health care professionals working in developing countries, raised wide-spread concern about the behaviour of

companies producing artificial formula (Baumslag & Michels, 1995). In an effort to increase sales, companies distributing formula in developing countries were employing tactics such as radio advertising and the employment of 'milk nurses' who would visit new mothers in hospitals. Women began to believe that formula was superior to breastmilk, with tragic consequences –infant morbidity and mortality rose dramatically (Baumslag & Michels, 1995), due primarily to the lack of clean water with which to make formula, an inability to sterilize used bottles or correctly make up the formula, and a lack of refrigeration. A number of individual groups from countries such as the United States, Australia, Canada, and New Zealand, banded together to form the International Baby Food Action Network (IBFAN), which actively publicised the plight of infants in developing countries. Their efforts led to a consumer boycott of Nestle products in 1981 (Baumslag & Michels, 1995; Van Esterik, 1989).

New Zealand breastfeeding rates continued to rise in the 1980s, with 78 percent of infants still being at least partially breastfed at three months of age (Ford et al., 1995; Royal New Zealand Plunket Society, 1995). This trend has been attributed to societal influences such as the positive portrayal of breastfeeding by the media, health care professionals ceasing to actively promote artificial formula, and the activities of breastfeeding support groups (Msuya et al., 1990). However, breastfeeding rates had plateaued by the end of the decade (Msuya et al., 1990).

During the 1980s, a number of New Zealand researchers addressed issues surrounding infant feeding (Davies, 1989; Flight & Adam, 1986; Ford, Schluter, & Wild, 1996; Gunn, 1984; Msuya et al., 1990; Perry & Trlin, 1985; Trlin & Perry, 1982), focusing primarily on the demographic trends of infant feeding. Their results highlighted many factors that influenced breastfeeding and bottle feeding rates, such as socio-demographic influences (Davies, 1989; Flight & Adam, 1986), ethnicity (Flight & Adam, 1986; Gunn, 1984; Perry & Trlin, 1985), health care professionals (Davies, 1989; Gunn, 1984; Trlin & Perry, 1982), insufficient milk due to complementary feeding practices (Davies, 1989; Gunn, 1984), and the availability of support for breastfeeding women (Flight & Adam, 1986; Gunn, 1984; Perry & Trlin, 1985; Trlin & Perry, 1982). It is interesting to note that the recommendations from the authors of these studies focused primarily on the role of the health care professional in increasing breastfeeding rates. However, a few studies acknowledge the role that social networks, including family

members and La Leche League, had on women's breastfeeding success (Davies, 1989; Trlin & Perry, 1982).

2.6.1 The International Code for the Marketing of Breast-milk Substitutes

The WHO and UNICEF responded to pressure from IBFAN and health care professionals to develop breastfeeding policies that would protect infants in developing countries. In 1979, WHO and UNICEF met to discuss the regulation of breastmilk substitutes (Baumslag, 1995; Van Esterik, 1995; Williams, 2000). In 1981, they jointly issued the *International Code for the Marketing of Breast-Milk Substitutes*, which called for a ban on the distribution of free artificial formula, a ban on advertising of breastmilk substitutes in health care centres, the removal of any idealisation of artificial formula, and the addition of labelling on formula tins clearly stating that breastmilk is the safest and best source of nutrition for infants (World Health Organisation & UNICEF, 1981). New Zealand was among the 118 countries who signed the Code (Baumslag & Michels, 1995; Galtry, 1998).

Lobbying groups objected that the Code had been developed in conjunction with the industry involved in the making and marketing of artificial formula (Baumslag & Michels, 1995; Van Esterik, 1989), interpreting it as a minimal effort to curtail the activities of the artificial formula industry (Van Esterik, 1989). These fears proved to be well-founded, as it soon became clear how difficult it would be to force companies to comply with the Code. Unless a country's government was willing to pass legislation to enforce the Code, individual companies were left to regulate their own activities in relation to the promotion of their products (Baumslag & Michels, 1995; Van Esterik, 1989; Williams, 2000).

In 1983, the Breast Milk Substitutes Monitoring committee was formed in New Zealand specifically to monitor compliance with the Code. Unfortunately, the committee had no authority, and was disbanded due to a lack of funding in 1990 (Dignam, 2001; New Zealand Breastfeeding Authority, 2000). By the end of the 1980s, there was growing global awareness that the Code was failing to deliver change in the promotion of breastfeeding, prompting the WHO and UNICEF to summon international policy makers to revisit the issue of breastfeeding

promotion (Baumslag & Michels, 1995). This eventually led to the development of the *Innocenti Declaration on the Protection, Promotion and Support of Breastfeeding.*

2.7 1990-2002

2.7.1 Innocenti Declaration on the Protection, Promotion and Support of Breastfeeding

This latest WHO/UNICEF initiative led to a meeting, titled "Breastfeeding in the 1990's: A global initiative", at the international assembly of August 1990, during which the *Innocenti Declaration on the Protection, Promotion and Support of Breastfeeding* was designed (World Health Organisation & UNICEF, 1990). The declaration states that breastfeeding provides infants with the ideal food needed for growth and development, reduces the incidence of infant morbidity and mortality, results in enhanced health for women, and is economically beneficial. It stipulates that the minimum age that all infants should be exclusively breastfed is four to six months, and that breastfeeding should continue for at least the first two years of age. The declaration also states that "all women should be enabled to practice exclusive breastfeeding" (World Health Organisation & UNICEF, 1990, p. 1).

The declaration also outlines targets for individual countries to achieve, including the appointment of a national breastfeeding coordinator, the establishment of a multi-sector national breastfeeding committee, ensuring hospitals support the 'Ten steps to successful breastfeeding', compliance with the Code, and legislation to protect breastfeeding women. It was hoped that the *Innocenti Declaration* would pressure governments to implement policies that would support women to breastfeed.

In 1990 the New Zealand Department of Health signalled their support for the *Innocenti Declaration*. Responding to pressure from professional groups such as The Lactation Consultants Association (Vogel & Mitchell, 1998c) and lay groups including La Leche League (Gordon, 1998), the Department of Health convened a meeting in 1991 to re-consider the Code's place within New

Zealand. In 1992, the department developed a 'breastfeeding kit' to educate health care professionals working for the Royal New Zealand Plunket Society, and within the hospital system, about the Code and the *Innocenti Declaration*. However, a national breastfeeding coordinator has never been appointed. Furthermore, legislative policy requiring companies to comply with the Code has not been developed and, to date, compliance with the Code remains voluntary. A committee to monitor the implementation of the Code was eventually established in 1998; however, as was the case with the 1983 Breast Milk Substitutes Monitoring committee, it lacks the authority to enforce the Code.

New Zealand has had only limited success with the *Innocenti Declaration*, which has been attributed to the lack of a co-ordinated effort on its behalf (Beasley & Trlin, 1998). It has also been suggested that there has been a lack of commitment to, and financial support for, implementation of the Code in New Zealand (Dignam, 2001).

However, it became evident early on that New Zealand was not the only country having difficulty ensuring that the Code and the *Innocenti Declaration* were adhered to. This prompted a group of 17 international organisations to form the World Alliance for Breastfeeding Action (WABA) in 1991 (Baumslag & Michels, 1995). WABA supports breastfeeding through World Breastfeeding Week, which is held each year and through collaboration with UNICEF. In 1992, the theme for World Breastfeeding Week was the Baby Friendly Hospital Initiative (BFHI). New Zealand took part in celebrating the 2002 World Breastfeeding Week, which assisted in raising the profile of both breastfeeding and the BFHI in this country (Heritage, 2003).

2.7.2 The Baby Friendly Hospital Initiative

The fact that most women give birth to their babies within a hospital setting led the WHO and UNICEF to develop guidelines for encouraging breastfeeding in maternity facilities, resulting in the *Ten steps to successful breastfeeding* (World Health Organisation & UNICEF, 1989). These steps outline how health care professionals should go about ensuring that breastfeeding is protected and promoted within the hospital setting. Each of the ten steps is supported by

research that highlights how it can support women to breastfeed². It was hoped that the implementation of these steps would revolutionize practices, especially in maternity hospitals and associated facilities.

The guidelines form the basis of the Baby Friendly Hospital Initiative (BFHI), which was launched in 1991 by the WHO and UNICEF. The BFHI was established with the aim of empowering women throughout the world to breastfeed, and eliminating the provision of artificial formula within the hospital setting. The New Zealand College of Midwives included the *Ten steps to successful breastfeeding* in their publication *Protecting, promoting and supporting breastfeeding* (New Zealand College of Midwives, 1992). The *Ten steps to successful breastfeeding* was also adopted by the Royal New Zealand Plunket Society, which, in 1996, published their own breastfeeding policy, adapting the document to better fit the services offered by the community-based organisation.

By the mid-1990s, New Zealand government agencies were beginning to respond to pressure from lobby groups, and in 1994 the Ministry of Health convened a meeting of interested parties to discuss how the BFHI might be implemented. It was concluded that the BFHI should serve as a vehicle through which to increase breastfeeding rates in New Zealand (Vogel & Mitchell, 1998c). The Ministry oversaw the formation of the BFHI taskforce, which became known as the New Zealand Breastfeeding Initiative Taskforce (NZBIT), and reported to the Ministry as well as to the Public Health Commission.

That year the Public Health Commission acknowledged the importance of breastfeeding, and outlined targets for breastfeeding rates, which included increasing the number of exclusively breastfed infants at three months of age to 75% (Public Health Commission, 1994). In 1995, the Public Health Commission also specifically mentioned the BFHI in the publications *Food and nutrition*

² For further information about the research that formed the basis for the *Ten Steps to Successful Breastfeeding* please refer to the following authors: Step one (Ellis, 1992); Step two (Auerbach, 1988; Becker, 1992); Step three (Rentschler, 1991); Step four (Kuriniji & Shiono, 1991); Step five (Slavin & Harvey, 1981); Step six (Glover & Sandilands, 1990); Step seven (Keefe, 1986); Step eight (De Carvalho, Robertson, Friedman, & Klaus, 1983); Step nine (Newman, 1990); Step ten (Kyenkya-Isabirye & Malgalheas, 1990).

guidelines for healthy infants and toddlers (Public Health Commission, 1995a; 1995b), and Food and nutrition guidelines for healthy breastfeeding women (Public Health Commission, 1995a; 1995b). However, no funding was made available for the operation of the NZBIT (New Zealand Breastfeeding Authority, 2000; Vogel & Mitchell, 1998c). Following the 1997 New Zealand Lactation Consultants Association Conference, the Lactation Consultants' Association, La Leche League New Zealand, and the College of Midwives joined forces to submit a proposal to the Health Funding Authority to implement BFHI (Vogel & Mitchell, 1998c). Their proposal was not funded.

In 1998, the Ministry of Health published *Progress on Health Targets*, which noted that the goals set for breastfeeding rates had not been achieved (Ministry of Health, 1998). The report highlighted changes to global policy as implemented by the WHO (World Health Organisation & UNICEF, 1990), and offered an assessment of why the goals had not been met; however, this discussion was limited because of a lack of available research that explored the reasons why women were not breastfeeding, with only one study being cited by way of explanation (Basire, Pullon, & McLeod, 1997).

That same year saw the establishment of the New Zealand Breastfeeding Authority (NZBA), a coalition of 30 stakeholder organisations that included midwives, La Leche League representatives, medical practitioners, dieticians, and representatives from the Royal New Zealand Plunket Society. The NZBA was granted funding in 1999 by the Health Funding Authority to establish the BFHI within New Zealand. The funding contract provided for the establishment of structures to support the implementation of BFHI within New Zealand, and initiate training for BFHI assessors (New Zealand Breastfeeding Authority, 2000).

It became evident when assessing breastfeeding rates that there was a need to streamline the manner in which national health targets for breastfeeding were measured. Therefore, the Ministry of Health commissioned a document that reviewed, and provided recommendations for, national breastfeeding definitions (Coubrough, 1999). The report compiled research and expert opinion on how breastfeeding rates could be measured against precise definitions, for the purpose of accurately assessing New Zealand's success at meeting its targets.

The year 2000 saw the establishment of a NZBA Committee and an Implementation Advisory Group to oversee implementation of the BFHI in New Zealand. A national co-ordinator was appointed to manage the initiative and, by the end of the year, 17 trainers had completed the NZBA's BFHI training course (Stufkins, 2000).

A survey of New Zealand hospitals was undertaken in 2000 to assess compliance with the *Ten steps to successful breastfeeding* (Pownall, 2000). The results demonstrated that few hospitals had an organisation-wide breastfeeding policy, and it was noted that few hospitals actually had education programmes that informed staff about breastfeeding. The study also highlighted that hospitals were struggling to comply with the BFHI because of the inflexible nature of the ten steps, the inconsistencies inherent in the assessment process, and the financial commitment needed to the meet the initiative requirements. Probably the most significant difficulty for hospitals wishing to implement the BFHI was that independent maternity care providers did not have reporting mechanisms by which their practice could be monitored.

In 2001 the Ministry of Health released a toolkit for the District Health Boards (DHBs), called *Improve Nutrition* (Ministry of Health, 2001). The toolkit directs DHBs to support their maternity facilities to attain BFHI accreditation; however, the way in which this is achieved is left to the discretion of the DHBs.

2.7.3 2002-2003

In 2002, the World Health Assembly convened in Geneva with the goal of formulating a global strategy for infant and young child nutrition and feeding. The Assembly adopted a resolution stating that infants should be exclusively breastfed for six months, with continued breastfeeding for up to two years or beyond (World Health Organization, 2002). As a matter of urgency, the Assembly called for governments to "formulate, implement, monitor, and evaluate a comprehensive policy on infant and young child feeding" (World Health Organization, 2002, p. 13). The World Health Assembly recommendation was endorsed by the WHO, which has since adopted the global strategy.

The New Zealand Ministry of Health has yet to establish a policy to implement the World Health Assembly's resolution (Ministry of Health, 2002), although it has published *Breastfeeding: A guide to action* in 2002. This document highlights that there have been no significant changes to breastfeeding rates in New Zealand over the past ten years, and presents a plan to increase these rates in 2002/2003. It specifies seven goals, which include the establishment of a national breastfeeding coordinator, the achievement of Baby Friendly Hospitals throughout New Zealand, an increased focus on rendering culturally-appropriate breastfeeding support, enhanced support of breastfeeding promotion, improved access to antenatal education, and the provision of quality care during the postpartum period (Ministry of Health, 2002).

At the time of writing, some 13 years since the inception of the BFHI, only six of New Zealand's 85 maternity facilities have attained 'Baby Friendly Hospital Status' (Heritage, 2003). This contrasts sharply with the success of many developing nations such as Kenya, which has 232 facilities with BFHI status, Nigeria with 1147, and the Philippines with 1047 (UNICEF, 2002). Although its not clear what breastfeeding definitions have been employed, nor how strict the assessment processes and criteria for approval are, the number of approved hospitals in these developing countries does play to the spirit of the BFHI concept. The New Zealand Government has been hesitant to make the necessary changes at a policy level and pursue strategies to support breastfeeding women. Strategies that have been suggested have been primarily population-based, without emphasis on understanding the personal meaning of breastfeeding for breastfeeding women.

In 2002, Minister of Health Annette King noted that a woman's decision to breastfeed is influenced by "social norms and by the beliefs and values of women and their significant others" (Ministry of Health, 2002, p. iii). The goals recommended to increase breastfeeding rates take into account a woman's need for support from society in the form of BFHI and community strategies. However, until these societal factors and women's beliefs and values regarding breastfeeding are addressed, the suggested strategies will not alter the current demographic trends.

2.8 Summative Thoughts

To date, government-supported breastfeeding strategies have been based on biomedical research, essentially ignoring the lived experiences of breastfeeding women. Improving our understanding of the experience of breastfeeding may lead to strategies that work for women, rather than broad population-based strategies into which women must fit to receive support from well-meaning government officials and health care professionals. This is especially true for women of minority cultures whose needs may differ considerably from those of the prevailing majority.

History has demonstrated that health care professionals have essentially established and owned the language that describes both breastfeeding and breastfeeding success. The manner in which breastfeeding is articulated holds sway over how breastfeeding as a concept is understood, and significantly influences how policies at both national and international levels are developed. However, the language that exists around breastfeeding is confined by boundaries set up by those external to an individual breastfeeding dyad; as a result, this language dismisses women's understandings and individual definitions that characterise success.

It can, therefore, be argued that the language that currently contributes to the construction of breastfeeding is a language of exclusion, because it essentially ignores the language that describes the lived experience of breastfeeding. It is paramount that a language that accommodates women's experience of breastfeeding emerge, as this would facilitate meaningful understandings of the lived experience of breastfeeding women. There is also a need to embrace women's individual definitions of breastfeeding success. If this is not achieved, breastfeeding goals and definitions of success will continue to be confined to arbitrary categories that only meet the needs of policy makers.

Well-meaning health care professionals may be aware of the *Ten Steps*, but unless they have the insight and ability to meld them with women's needs and current research, breastfeeding statistics will not improve. Health care professionals need to speak a language that reflects an understanding of the breastfeeding experience, not a language that simply reflects policy. There is a need to learn a language that can accommodate experiential knowledge and

lifeworld-based research, as well as biomedical research. Only when research and experience become fused in a language that accommodates both forms of knowledge will breastfeeding be positively enhanced, promoted, and supported. A woman's reactions to the community environment, her body, and her infant have more bearing on her desire to continue breastfeeding than any policy, no matter what the source.

2.9 Conclusion

Prior to 1900, women owned and shared breastfeeding knowledge, thereby supporting one another to breastfeed. As biomedical knowledge of human lactation emerged as the dominant discourse, and artificially-contrived replacements were proposed, women's knowledge of breastfeeding was usurped. Breastfeeding was re-interpreted as a purely biological act, leaving behind generations of women's knowledge that defined breastfeeding as an art.

Research has focused on biomedical understandings of breastfeeding, and this has formed the basis for international, as well as national, strategies to enhance, support and promote breastfeeding. A woman may well have achieved all *Ten steps to successful breastfeeding*, but her experience of breastfeeding exists outside of the realm of international policy or hospital procedures. A woman's personal experience of breastfeeding her infant will hold critical and meaningful sway over her decision to continue to breastfeed, regardless of external policy or strategy. And it is her experience that will dictate whether or not she will exclusively breastfeed her infant for the first six months of his or her life and beyond, as well as what she will tell other women. There is, therefore, a need to explore research that examines and illuminates the various facets of women's experiences of breastfeeding; it is this research that is presented in the following chapter.

Chapter Three

Examining the Experience of Breastfeeding

3.1 Introduction

The previous chapter examined the historical and contextual considerations surrounding breastfeeding in New Zealand, and we now turn our attention to the literature that deals with the act of breastfeeding. There exists a large body of literature that explains breastfeeding in terms of physiological processes³, locating breastfeeding as a mechanical, technical act within a socio-political context. It is from this biomedical framework that the majority of government-supported policies and strategies to promote breastfeeding have been derived. I have argued that a deductive approach to breastfeeding research is needed to ensure that the lived experience of breastfeeding is adequately considered. To promote breastfeeding and provide better support for breastfeeding women, an understanding of their experience of breastfeeding is essential; therefore, this chapter examines the academic literature that deals with experiential accounts of breastfeeding.

3.2 The Influence of Culture on Breastfeeding

Understandings of breasts and breastfeeding emerge within the context of community, which can be understood on different levels ranging from a broad global perspective to an individual's own social circle. These understandings are built within popular culture, which is conveyed to us through entities such as the media and art, and are coupled to the beliefs of previous generations. Anthropologists (Bryant Merrill, 1987; Long, 2003; Makhlouf Obermeyer & Castle, 1997; Morse, 1985; Paine & Dorea, 2001), feminists (Bartlett, 2002; Blum, 1993; Carter, 1995; Fellow, 1994; Galtry, 1997; Law, 2000; Meyer & de Oliveira, 2003; Ryan, 1999a; Schmied & Barclay, 1999; Schmied & Lupton,

³ For more information see Chapter One, Section 1.3 and Chapter Two, Section 2.7.2.

2001; Vares, 1992; Yalom, 1997), and sociologists (McIntosh, 1985; Meyer & de Oliveira, 2003; Murphy, 1999) have explored societal understandings of breastfeeding and how they influence individual interpretations of breastfeeding.

Examples of the influence of the wider New Zealand society, along with international influences, on breastfeeding were explored in Chapter Two. Chapter Three focuses on the experience of breastfeeding as described by women who have breastfed, and the manner in which their experiences are reflected in the literature. It must be acknowledged that each individual breastfeeding woman is embedded within her society's culture, the effects of which are explored in Chapter Five.

Research has highlighted that there are differing types of historically-embedded knowledge that affect a woman's breastfeeding experience (Ryan & Grace, 2001). In Ryan's (2001) exploration of women's knowledge of breastfeeding over the period from 1945 to 1990, participants in her study articulated the ways in which they gathered knowledge about breastfeeding. These include instinctual knowledge, culturally-absorbed knowledge, experiential knowledge, and shared knowledge. Women also discussed how breastfeeding knowledge was acquired through their interactions with health care professionals, which assisted in forming their biomedical understandings of breastmilk.

These different ways of knowing inform the way women approach and manage breastfeeding. If women reside in communities that do not have a breastfeeding culture they are significantly hampered in their attempts to breastfeed, as they will not have access to culturally-absorbed or shared breastfeeding knowledge. It has been convincingly argued that New Zealand does not have a breastfeeding culture (Beasley, 2002; Beasley & Heritage, 1998/1999; Ryan, 1999a; Watson, 2001).

Although lactation is essentially a physiological process, breasts and breastfeeding are interpreted against a backdrop of culturally-mediated ideologies. New Zealand breastfeeding literature highlights how culture significantly influences women's decision to breastfeed, suggesting that breastfeeding is supported in theory but not in practice (Beasley, 1991; Beasley, 1996b; Beasley & Heritage, 1998/1999; Dignam, 2001; Ryan, 1999a; Thurtle, 1997; Watson, 2001).

Beasley and Heritage (1998/1999) describe how women's ability to breastfeed in New Zealand has been hindered by the historical promotion of structured infant feeding times, emphasis on a woman's right to choose the method of infant feeding, and the sexualisation of breasts. Ingrained cultural beliefs portray breastfeeding as an inconvenient feeding method that disadvantages women by 'tying' them to their infants.

Ryan (1999a) discusses how culturally-mediated discourses have constructed understandings of breastfeeding within New Zealand. During the period from the 1940s to the 1960s, women espoused a discourse of breastfeeding being "not for me", arguing that artificial formula was scientifically prepared and a suitable replacement for breastmilk. During the 1970s and 1980s the discourse espoused by women changed to one that valued the "natural earth mother - breast is best", in which breastfeeding was viewed as a right to be exercised. This discourse emerged as prevalent at a time when research studies were being published extolling the virtues of breastfeeding (Anyon, 1976; Hood et al., 1978; Roberts, 1980).

Another discourse that Ryan (1999a) claims has consistently played a role in women's breastfeeding discussions is that of "I'II give it a go." This position allows a woman the freedom to stop breastfeeding if she doesn't think it's working for her, highlighting the common belief that women's bodies may fail at breastfeeding. All of these discourses are still operating in New Zealand to varying degrees, confirming the thoughts of New Zealand-based breastfeeding experts who report that women are confined by a culture that medicalises breastfeeding, and portrays breastfeeding as problematic (Beasley, 1991; Beasley, 1996b; Beasley & Heritage, 1998/1999; Dignam, 2001; Ryan, 1999a; Thurtle, 1997; Watson, 2001).

New Zealand is legally considered a bi-cultural country, comprising the indigenous Maori race, the *Tangata Whenua* – the people of the land, and those of British descent. Quantitative New Zealand research studies contain breastfeeding statistics describing which people groups are, or are not, breastfeeding (Essex, Smale, & Geddis, 1995; Flight & Adam, 1986; Ford et al., 1995; Heath et al., 2002; Msuya et al., 1990); however, only a small number of studies have explored the values and beliefs about breastfeeding from a Maori perspective (Ellison-Loschmann, 1997; Ellison-Loschmann, 1998; Rimene, Hassan, & Broughton, 1998).

Historically, Maori people have valued women for their ability to nurture through breastfeeding. Rimene, Hassan, and Broughton (1998) describe how Maori women are considered to have the same responsibility as the land in sustaining life, through their ability to give birth and breastfeed. This role has afforded Maori women a sacred respect, referred to as *mana wahine*, that is solely the right of women, especially after they have given life to the next generation. However, in recent decades, dramatic social change, leading to urbanization (Pool, 1991; Pool & Bedford, 1996) and the loss of cultural identity, has significantly changed the life of Maori in New Zealand (Durie, 2001). These changes have altered the way Maori women see their role in the community, affecting the way women approach breastfeeding infants (Ellison-Loschmann, 1997).

Maori women typically have children at a younger age than their European-descent counterparts (Ministry of Health, 2001). This, coupled with the fact that many Maori women now live in urban environments away from the support structures present in extended family environments, does affect women's experience of breastfeeding. Women frequently find themselves isolated from the people traditionally responsible for nurturing them as new mothers, contributing to relatively low breastfeeding rates among Maori. In 1998, 39% of Maori infants were fully breastfed at three months as opposed to 55% of their European counterparts (Ministry of Health, 2001). However, for those that do breastfeed, their close connections with mothers and grandmothers are a source of invaluable support when they are in the process of establishing breastfeeding (Ellison-Loschmann, 1997; Ellison-Loschmann, 1998; Rimene et al., 1998).

The women who participated in Ellison-Loschmann's (1998) study all highlighted initial difficulties in establishing breastfeeding, including physical factors such as sore nipples, as well as a lack of support from health care professionals in the hospital setting. It is clear from other studies that these difficulties are common amongst both Maori and European-descent New Zealanders (Benn et al., 2002; Britton, 1997; Ellison-Loschmann, 1997; Hauck, Langton, & Coyle, 2002; Hill et al., 1994; Hood et al., 1978; McLeod et al., 1998; Rimene et al., 1998); however, mainstream maternity providers may be particularly prone to neglecting the cultural needs of Maori women. Areas that are often overlooked include the need for *whanau* (family) support, the need for

mana wahine to be observed, as well as the observance of tapu (ceremonial restriction or the sacred) and noa (free from ceremonial restriction), which differ between tribal groups (Rimene et al., 1998).

New mothers discussed the benefit of positive role models and support from Maori health care professionals working in the community in establishing and continuing with breastfeeding (Ellison-Loschmann, 1997; Ellison-Loschmann, 1998). Women's decisions to continue to breastfeed over time were influenced by a need to resume their own lives, as breastfeeding did not always afford them the freedom they desired. Negotiating the balance between freedom and the needs of their infants is a challenging task. Juggling the demands of multiple roles, such as that of being a partner or the need to return to paid employment, has also been cited as another significant challenge to continued breastfeeding for Maori women⁴ (Rimene et al., 1998).

Although New Zealand is legally a bi-cultural nation, it is home to a number of other people groups⁵, notably from Pacific Island nations including Samoa, Fiji, Cook Islands, and Tonga. While these distinct cultural groups all provide differing perspectives on breastfeeding, one value appears common to them all - that Maori and Pacific Island women understand breastfeeding, and breastmilk, to be a gift for their young (Abel, Park, Tipene-Leach, Finau, & Lennan, 2001; Rimene et al., 1998; Ryan, 1999a). Grandmothers support their daughters with food to assist in nourishing a new mother, and in doing so provide a gift for their grandchild. Grandmothers also pass on experiential knowledge about breastfeeding, and in doing so support the new mother to continue to provide nourishment for their offspring. It has been suggested that this ideology differs considerably from that of New Zealand women of European descent, who can view breastfeeding as a choice they make according to their lifestyle needs (Beasley & Heritage, 1998/1999). However, participants who were primarily of European descent in two different qualitative studies (Bottorff, 1990; Dignam, 2001) also described how breastfeeding, when successful, was considered to be a gift that encompassed emotional and physical comfort, as well as nutrition for their infants.

⁴ The issue of breastfeeding and employment is further discussed in Section 3.13.

There is comparatively little published research that examines the experience of breastfeeding for Pacific Island women. Dodd (1999) describes an interview with two Pacific Island women, who were also health care professionals, that raises some commonly held beliefs among Pacific Island cultures about breastfeeding. One such held belief is that colostrum is viewed as harmful and discarded, with substitutes such as water frequently offered to infants until a woman's mature milk comes in. This belief was also discussed by Morse (1985) in her research into Fijian women's infant feeding practices. Pacific Island cultures advocate the provision of specific foods to new breastfeeding mothers, as well as the avoidance of certain foods that are believed to affect the milk's quality. Another issue affecting the continuation of breastfeeding arises when a woman's breasts become less full and feel softer, which occurs between two and four months postpartum; at this time, a woman might believe that her milk has 'flown away', necessitating the need for the introduction of artificial formula. Able and colleagues (2001) discuss how this results in women being anxious about their ability to feed their infants, leading to premature weaning.

Issues that affect Pacific Island women's breastfeeding in New Zealand are similar to those experienced by others in this country. The medicalisation of breastfeeding has eroded their confidence in their body's ability to breastfeed without professional help (Abel et al., 2001; Dodd, 1999). The need to return to paid employment also impacts upon the ability of women to sustain breastfeeding. And being part of an immigrant population, Pacific Island women may lack the extended family or community support traditionally available. However, the ongoing support of women's own mothers continues to be a defining part of a successful breastfeeding experience (Dodd, 1999).

Morse's (1985) research into breastfeeding practices in Fiji suggests that Fijian women use a combination of breast and bottle feeding, which allows them to continue to fulfil their various roles within the community. For Fijian women, a strong sense of community means that they receive the support of their extended families and cultural community when they breastfeed (Morse, 1985). No research studies have been conducted to assess how the breastfeeding

⁵ Other people groups in New Zealand include those of Asian and Indian descent; however, there is no published research exploring the breastfeeding experiences of these people groups in New Zealand.

experience is constructed in the New Zealand context where extended contact with communities or tribes may be disrupted.

In a landmark New Zealand study, Able and colleagues (Abel et al., 2001) conducted 37 focus groups to examine the differing culturally-based infant care beliefs of Maori, Tongan, Samoan, Cook Islands, Niuean and European caregivers in this country. This study clearly illustrates that these differing groups all believe breastfeeding is ideal for infants, because it is a healthy, cheap and convenient food source, and also provides for the infant's emotional needs. The study demonstrated that Pacific Island parents who have good family support do not rely on professional organisations for breastfeeding support; instead, the mother or mother-in-law are considered to be the principal source of breastfeeding advice. The majority of Pacific Island women participating in the study had endeavoured to breastfeed; however, there was general consensus that breastfeeding was difficult to establish, because of pain associated with cracked nipples and engorgement. The early introduction of complementary feeding, both artificial formula and food, was also a common theme for the Pacific Island participants, as it was believed that women were not always capable of producing enough milk to satisfy their infants. The authors of this study highlighted the tension that exists for New Zealand-born Pacific Island women, who must balance their traditionally-held beliefs with those of the prevalent Westernized culture, in which a biomedical approach to breastfeeding is valued.

3.3 Making the Decision to Breastfeed

It has been documented that women make the decision to breastfeed whilst they are pregnant, although their decisions are mediated by social and cultural ideologies (Arora et al., 2000; Baumslag & Michels, 1995; Earle, 2000; Heath et al., 2002; Ozturk Ertem et al., 2001; Schmied & Lupton, 2001). The primary reason women cite for deciding to initiate breastfeeding is that it is a natural process that provides the best source of nutrition for the baby's physiological and emotional health (Arora et al., 2000; Earle, 2002; Kessler et al., 1995; McIntosh, 1985; Murphy, 1999; Schmied & Lupton, 2001). Women who espouse these beliefs report a strong commitment to breastfeeding as a result (Schmied & Lupton, 2001).

The need to negotiate busy life schedules whilst managing complex relationships with significant others, as well as a desire to support the family's sense of well-being, all factor into a woman's decision about breastfeeding (Carter, 1995; Earle, 2000; Gigliotti, 1995; Lothian, 1994; Schmied & Lupton, 2001). Because of this, if a woman's lifestyle does not support breastfeeding and she has a baby that is experiencing difficulties at the breast, she may decide that breastfeeding is not for her and subsequently wean her infant (Carter, 1995; Ferguson, 1998; Gigliotti, 1995; Hill et al., 1994; Lothian, 1994).

Significant relationships in women's lives play a role in their decisions about breastfeeding, with significant others typically described as either the baby's father or the women's mothers (Arora et al., 2000; Chalmers, Ransome, & Herman, 1987; Cronin, 2003; Earle, 2000; Earle, 2002; Kessler et al., 1995). Women may choose to initiate breastfeeding, but the opinions of significant others impact upon a woman's confidence to continue breastfeeding (Kessler, 1995). This suggests that breastfeeding initiation is not dependent solely upon cognitive processes, but occurs in conjunction with social factors. Research also indicates that if a woman was breastfed herself she is more likely to breastfeed her own infant (Isabella & Isabella, 1994; Kessler et al., 1995; Williams, Innis, & Vogel, 1996). This correlation has been attributed to the beliefs and breastfeeding support offered by women's mothers. The issues surrounding social support and breastfeeding are explored in greater depth in section 3.6.

The length of time a woman intends to breastfeed her infant has been discussed in the New Zealand literature. Research supports the notion that women approach breastfeeding with an attitude of 'lets see how it goes', believing that if it proves to be problematic they will stop (Ryan, 1999a; Vogel & Mitchell, 1998b). Vogel and Mitchell (1998b) discovered that women do not typically set out to breastfeed for long periods; instead, they 'grow into it'. However, the decision to continue to breastfeed is mediated by many factors such as family demands, return to paid employment, social expectations, and the introduction of solids into the infant's diet.

Murphy (1999) explored women's depictions of their breastfeeding decisions within a framework of sociological deviance. Her study demonstrates that women's intention to either bottle feed or breastfeed their infants is understood to be a moral decision that, once made, dictates whether or not a woman is a

good mother, or indeed a good woman. Women who choose not to breastfeed are understood to be selfishly placing their own needs above those of their infant, particularly in light of the beneficial physiological nature of breastfeeding. When recounting decisions to formula feed antenatally, the women Murphy interviewed had what were described as 'well-rehearsed' depictions to defend why they were choosing not to breastfeed.

Women have expressed various rationales for not breastfeeding for personal as well as external reasons (Murphy, 1999). Personal issues included embarrassment at feeding in front of others, the need for self-fulfilment and returning to paid employment. External reasons for not breastfeeding included a desire to involve the infant's father in care, a belief in the superiority of formula, and that the opinion of significant others held weight over that of health care professionals. These reasons for choosing not to breastfeed have also been highlighted in other studies (Carter, 1995; Cronin, 2003; Earle, 2000; Earle, 2002).

3.4 Initial Breastfeeding Experiences

Initial experiences with breastfeeding can leave a lasting impression on a woman, and may influence future decisions about breastfeeding subsequent children (Mozingo et al., 2000). A quantitative study (Tarkka, Paunonen, & Laippala, 1998) describes how women who were provided with individualised support and education about breastfeeding in the initial postpartum period were more likely to successfully establish and maintain breastfeeding. Tarkka, Paunonen and Laippala (1998) interviewed women who had given birth in hospital to healthy, full-term infants. They discovered that women who feel positive about their initial breastfeeding experiences are more likely to continue breastfeeding following discharge from tertiary maternity services. For some women, initial breastfeeding experiences are described as positive and rewarding; unfortunately, other women recount their experiences employing terms such as disappointing, hard, exhausting and stressful (Benn et al., 2002; Benson, 2000; Britton, 1997; Mozingo et al., 2000).

Some women report that health care professionals would forcefully handle their breasts and the baby to facilitate the initial breastfeeds. This behaviour on the

part of health care professionals was interpreted by the women as uncaring, and was perceived to demonstrate a lack of understanding of both women's and infant's needs (Benn et al., 2002; Mozingo et al., 2000; Vogel & Mitchell, 1998a). Women have also cited lengthy delays in getting a response from health care professionals to requests for assistance, reporting waiting periods of up to three hours before getting assistance (Vogel & Mitchell, 1998a).

Inconsistent advice from health care professionals is cited by women as a difficulty that vandalises the initial breastfeeding experience (Benn et al., 2002; Britton, 1997; Cronin, 2003; Hood et al., 1978; Mozingo et al., 2000; Vogel & Mitchell, 1998a). This problem has been identified as an important issue for a considerable time (see Hood, 1978 #455). Along with the issue of inconsistent advice, women have also discussed how health care professionals frequently offer what has been described as 'standardised advice', that is not tailored to their individual circumstances (Hauck et al., 2002). This lack of individualised advice leaves women feeling isolated in their attempts to resolve breastfeeding difficulties.

Women participating in research have identified a number of commonly occurring impediments to establishing breastfeeding, including difficulty in positioning or latching the infant to the breast, painful or cracked nipples, perceptions of insufficient milk supply, engorgement, unsettled infants and, as noted above, inconsistent advice from health care professionals (Beasley et al., 1998; Beasley & Gordon, 1998/1999; Benn et al., 2002; Britton, 1997; Hauck et al., 2002; Hill et al., 1994; Hood et al., 1978; McLeod et al., 1998; Mozingo et al., 2000; Scott, Landers, Hughes, & Binns, 2001; Vogel & Mitchell, 1998a; Vogel & Mitchell, 1998b). Furthermore, some women have noted that, following the birth of a second baby, they were left with little support from health care professionals when attempting to establish breastfeeding, despite struggling, presumably because they were expected to be fully capable through prior experience (Benn et al., 2002; McLeod et al., 1998).

The difficulties encountered during the process of initiating breastfeeding have led women to conclude that breastfeeding is not as easy as they were led to believe by others (Benn et al., 2002; Benson, 2000; Britton, 1997; Hauck et al., 2002; Hill et al., 1994; Mozingo et al., 2000; Vogel & Mitchell, 1998a). Women relay that they felt ill-prepared for breastfeeding and that advice provided by other women and antenatal educators did not necessarily assist them in

establishing breastfeeding (Britton, 1997; Hauck et al., 2002; Hill et al., 1994; Mozingo et al., 2000; Vogel & Mitchell, 1998a). The gulf between idealised expectations and the reality of breastfeeding was, for some, extremely difficult to reconcile, resulting in disillusionment and the early cessation of breastfeeding (Mozingo et al., 2000). The follow section explores in greater depth some of the difficulties women face whilst breastfeeding.

3.5 Experiencing Breastfeeding Difficulties

While problems experienced during the initial breastfeeding period, such as engorgement, are often quickly resolved, some breastfeeding problems cause women significant distress for some time. Schmied and Lupton (2001) note that the majority of the women who participated in their study experienced difficulty breastfeeding, especially in the early weeks.

Research has demonstrated that women are frequently surprised by how difficult it can be to establish breastfeeding (Hauck et al., 2002; Schmied & Barclay, 1999; Vares, 1992), and find themselves unprepared to deal with ongoing problems. The struggle to reconcile pro-breastfeeding discourses with the reality of breastfeeding can be extremely difficult for some women (Mozingo et al., 2000; Schmied & Barclay, 1999). Some women who struggle to resolve breastfeeding difficulties end up navigating their way through a minefield of insecurity and guilt, which results in them questioning their ability to mother (Hauck et al., 2002; Mozingo et al., 2000). These issues are often compounded by the exhaustion that accompanies mothering a young infant.

Two of the most frequently cited difficulties with breastfeeding are sore and cracked nipples, which can be a source of discomfort for a number of months (Benson, 2000; Blair, Cadwell, Turner-Maffei, & Brimdyr, 2003; Carter, 1995; Chapman, Macey, Keegan, Borum, & Bennett, 1985; Davies, 1989; Hood et al., 1978; Mozingo et al., 2000; Page et al., 2003; Salmon, 1994; Schmied & Lupton, 2001; Vares, 1992; Vogel & Mitchell, 1998b). A participant in one study described breastfeeding as "very painful, very, very painful" (Benson, 2000, p. 24); others employ words such as 'excruciating' to describe their experiences (Vogel & Mitchell, 1998b). Schmied and Barclay (1999, p. 330) discussed how

the women who participated in their study described breastfeeding as a painful and violent act in which infants would attack them, resulting in 'bodily mutilation'.

It is interesting to note that sore nipples, especially for first time mothers, along with difficulties in establishing breastfeeding have been documented as problems for hundreds of years (Salmon, 1994). Yet, as outlined above, many women are surprised to find how difficult and painful breastfeeding can be to establish; that breastfeeding can be painful initially is silenced in our society. A common response from health care professionals is that painful nipples are the result of a woman's failure to establish a 'good' latch between the infant and breast. This way of framing a woman's pain can trivialize the distress she's experiencing and undermine her confidence in her ability to successfully breastfeed. Few women would want to admit they were unable to provide their infant with what is the best source of nutrition because they could not master what is essentially a mechanical skill.

Feelings of tiredness and exhaustion are commonly discussed by breastfeeding women (Beasley & Gordon, 1998/1999; Carter, 1995; Davies, 1989; Earle, 2002; Hill et al., 1994; Isabella & Isabella, 1994; Mozingo et al., 2000), especially during periods of infant growth. For adolescent mothers, tiredness associated with breastfeeding can make feeding their infants with artificial formula seem attractive, as it allows them to share infant feeding responsibilities (Benson, 2000). Feelings of fatigue and tiredness have also been highlighted as commonly experienced symptoms of postnatal depression (PND) (Bozoky & Corwin, 2001).

Breastfeeding women often gauge their performance on the behavioural messages transmitted by their infants. An unsettled infant that feeds frequently, especially overnight, and has poor weight gain sends a powerful message to women struggling to breastfeed. Women's interpretations of their infants' behaviour are meaningful because they offer women evidence by which they can measure their ability to breastfeed (Leff, Gagne, & Jefferis, 1994). This can lead women to conclude that either the quantity or quality of their milk is inadequate to sustain their infant (Arora et al., 2000; Beasley et al., 1998; Beasley & Gordon, 1998/1999; Chapman et al., 1985; Davies, 1989; Hill, 1992; Hill et al., 1994; Hochhausen, 1991; McLeod et al., 1998; Mozingo et al., 2000; Vogel & Mitchell, 1998b). Women who do not believe that their milk supply is sufficient to meet the needs of their baby frequently report suffering from

feelings of guilt (Essex et al., 1995; Hochhausen, 1991; Vogel & Mitchell, 1998b).

This phenomenon has been labelled by researchers as 'insufficient milk syndrome', and is frequently cited as the principal reason that women cease to breastfeed (Blyth et al., 2002; Essex et al., 1995; Gunn, 1984; Heath et al., 2002; Hill, 1992; Hill & Humenick, 1989; Hillervik-Lindquist, 1992; Hood et al., 1978; Ingram, Johnson, & Greenwood, 2002; Isabella & Isabella, 1994; Morse, Ewing, Gamble, & Donahue, 1992). Whether insufficient milk syndrome is real or imagined, even the perception of insufficient milk increases the likelihood that a woman will cease to breastfeed her infant before her peers who do not question their milk supply (Blyth et al., 2002; Heath et al., 2002).

The literature notes a number of physiological events that create ongoing discomfort for breastfeeding women, including engorgement, sore breasts and leaking milk (Bottorff & Morse, 1990; Carter, 1995; Hill et al., 1994; Mozingo et al., 2000; Schmied & Barclay, 1999; Schmied & Lupton, 2001; Tarkka et al., 1998). The 'always thereness' of breastmilk can, therefore, be problematic for some women. The need to engage in activities such as expressing milk to maintain personal comfort, even when away from their infants, means that breastfeeding women can never escape the demands of infant feeding (Bottorff & Morse, 1990).

Studies have also explored women's experience of the let-down reflex (Britton, 1997; Schmied & Lupton, 2001). Participants describe how their body signals a let-down of breastmilk by using phrases like 'tingly', 'tightening', and 'pins and needles sensation' (Britton, 1997, p 180). Women's descriptions of the let-down reflex emphasise that they feel powerless to control this particular bodily function, which may leave them leaking milk in awkward social circumstances. These studies highlight how women frequently feel unprepared for the presence, or absence, of a let-down sensation, which once again leads them to question their body's ability to produce milk at appropriate times.

The constant demands for milk from an infant, coupled with the demands of breasts filled with milk, lead some women to describe their lives whilst breastfeeding as being 'on hold' (Earle, 2002; Schmied & Barclay, 1999; Schmied & Lupton, 2001). Indeed some women characterise their dislike for breastfeeding by using terms like being 'sucked dry' (Schmied & Lupton, 2001,

p. 243) to illustrate the unrelenting nature of breastfeeding, which drives women to desire a sense of self that is separate from their infant.

For women with more than one child, the need to manage multiple roles and tasks factors into their decisions about breastfeeding (Arora et al., 2000; Carter, 1995; McBride, 1999; Vogel & Mitchell, 1998b). Women have discussed the social pressure to be visibly producing results, which is not fulfilled by breastfeeding a baby, at least not in the short term (Vogel & Mitchell, 1998b). Women need to fit breastfeeding around the demands of caring for other family members, which often results in the need for routines to maintain a sense of balance within the family. The maintenance of a routine can become difficult if an infant is unsettled or requires frequent feeding (McBride, 1999). The need to return to paid employment also poses significant challenges for women who choose to continue breastfeeding.

Breastfeeding problems that have persisted over some time can result in women feeling frustrated, isolated and guilty (Hauck et al., 2002; McLeod et al., 1998). This issue may be compounded if breastfeeding difficulties cause a woman to wean her infant earlier than she planned, which can lead to feelings of self doubt, failure, anger, and guilt (Earle, 2002; Hochhausen, 1991; McLeod et al., 1998; Mozingo et al., 2000). However, for some women a sense of relief characterises their decision to wean their infants, as it signals the end of what was for them an extremely stressful period (Gigliotti, 1995; Mozingo et al., 2000).

Women who have struggled to overcome breastfeeding problems talk about the need to find hope that the problems will be resolved, taking heart in even small signs of improvement (Bottorff, 1990; Hauck et al., 2002). Hope provides women with the encouragement to persist with breastfeeding. Hope that breastfeeding will get easier frequently comes through interaction with others with knowledge of breastfeeding, which is why women who receive emotional support and practical help from both individual social networks, as well as health care professionals, are more likely to report that they are coping well with breastfeeding (Tarkka et al., 1998).

3.6 The Influence of Social Support for Breastfeeding Women

The impact that women's social networks have on continued breastfeeding has been well documented (Arora et al., 2000; Chalmers et al., 1987; Hauck et al., 2002; Humphreys, Thompson, & Miner, 1998; Schafer, Vogel, Viegas, & Hausafus, 1998). Research indicates that women's social networks have more influence over their decision to breastfeed that any other factor (Humphreys et al., 1998).

According to the research, husbands⁶ also play a vital role in determining whether or not women breastfeed. If a woman perceives that her husband is ambivalent or not supportive of breastfeeding, she is more likely to opt for artificial formula (Arora, et al., 2000; Charlmers, 1987; Cronin, 2003; Davies, 1989; Hauck, 2002; Kessler, et al., 1995; Isabella & Isabella, 1994; McLeod, et al., 1998; Scott, et al., 2001). Husbands with positive attitudes to breastfeeding provide both emotional and financial support for women, enabling women to continue to breastfeed. In contrast, a husband's withdrawal of support for their spouse's continued breastfeeding will result in the infant being weaned shortly thereafter (Morse & Harrison, 1987).

Women's own mothers also significantly influence breastfeeding, as they provide empathetic support, as well as information about breastfeeding (Chalmers, Ransome & Herman, 1987; Isabella & Isabella, 1994; Kessler et al., 1995; Scott, et al., 2001; Vogel & Mitchell, 1998b). Chalmer's (1987) research suggests that it is women's own mothers who provide the greatest level of both practical and emotional support for breastfeeding women. However, the helpfulness of a woman's mother may not contribute to breastfeeding success, especially if she has not breastfed her children (Isabella & Isabella, 1994).

Peer support also contributes to positive breastfeeding outcomes, as this type of support assists in removing some of the isolation women experience when

⁶ Researchers interviewed women's husbands, not de-facto partners. Additional research is needed to assess the role of de-facto partners on women's breastfeeding experience.

attempting to establish breastfeeding (Dennis, 2002; Hauck et al., 2002; Humphreys et al., 1998; McLeod et al., 1998; Schafer et al., 1998). Peers allow women to observe others who are also struggling with breastfeeding, leading to solidarity that provides them with the encouragement to continue. Formalised peer support counsellors have also been demonstrated to enhance the duration of breastfeeding through the sharing of information and the provision of encouragement, which builds women's confidence in their ability to breastfeed (Dennis, 2002; Schafer et al., 1998).

La Leche League⁷ also provides woman-to-woman breastfeeding support through their meetings and phone contact, which has been described as a beneficial source of support within New Zealand (McLeod et al., 1998). This organisation was formed to facilitate women's knowledge of breastfeeding, and is currently active within New Zealand. It has been contended that La Leche League was formed primarily to overcome the lack of family support for breastfeeding women (Bryant Merrill, 1987). The women involved in La Leche League formed a structure that essentially mirrors that of an extended family: it has leaders, women with experience of breastfeeding, and new mothers who want to breastfeed, all of whom meet together with the goal of supporting one another to breastfeed.

While health care professionals do affect women's perceptions of available breastfeeding support, overseas research suggests that professionals play a minimal role in influencing breastfeeding outcomes, especially following discharge from maternity facilities (Chalmers, 1987; Davies, 1989; Humphreys, 1998). However, research conducted in New Zealand highlights how unique provisions within the health care system in this country play pivotal roles in supporting women to breastfeed (Ellison-Loschmann, 1998; McLeod et al., 1998; Vogel & Mitchell, 1998b). Midwives are funded to provide home visits for the first six weeks postpartum, and Plunket Family Centres frequently employ Lactation Consultants who can provide care at an individual level. Women have highlighted the importance of encouragement offered by health care professionals in relation to their breastfeeding success. Similarly, recognition of women's breastfeeding achievements by health care professionals provides an important signal to women that they are successful mothers (McLeod et al., 1998; Vogel & Mitchell, 1998b).

3.7 Prolonged Breastfeeding

Women's sense of confidence in their ability to breastfeed is emerging as a strong indicator of prolonged breastfeeding (Blyth et al., 2002; Dennis, 1999; Hill & Humenick, 1996; Ozturk Ertem et al., 2001). A Breastfeeding Self-Efficacy theory was developed by Dennis (1999) to predict which women have confidence that they are able breastfeed their infants. Dennis reports that breastfeeding self-efficacy is mediated by four antecedent factors: performance accomplishments (or past experiences), vicarious experiences, verbal persuasion, and a woman's physiological and affective states (p. 196-197). Women's internal dialogues affect their ability to breastfeed, as confidence in one's ability will most likely produce the desired breastfeeding behaviour. Dennis also asserts that people tend to pursue activities they believe they can be successful at, which directly affects the effort and persistence women bring to breastfeeding. Finally, women's breastfeeding ability is influenced by their emotional reactions to breastfeeding. If they encounter difficulties, a strong belief that they can overcome them will most likely enable them to continue to breastfeed.

Blyth (2002) has demonstrated through testing the Breastfeeding Self-Efficacy theory that women with high self-efficacy were significantly more likely to be breastfeeding at one week postpartum than those who had low self-efficacy. Women's confidence in their ability to breastfeed was enhanced the longer they continued; consequently, women with high confidence levels breastfed for significantly longer (Blyth et al., 2002).

Women who choose to breastfeed for periods of a year or longer report that the primary reason for continuing is that breastfeeding facilitates a special time between a woman and her baby (Hills-Bonczyk et al., 1994). Participants in this study have explained that the unique and close bond that is formed through breastfeeding encourages them to continue beyond twelve months. Furthermore, they have reported that they wished to allow their infant to initiate weaning when they were ready. It is noteworthy that both Blyth (2002) and Hills-Bonczyk (1994) state that for women to continue breastfeeding beyond one year they must display considerable strength to overcome the social

⁷ For more information about La Leche League please see Chapter Two, Section 2.5.

pressure they encounter, as New Zealand society portrays prolonged breastfeeding as unacceptable.

In an original and influential article, Bottorff (1990) explores what it is to persist with breastfeeding despite the difficulties women may encounter. Bottorff employed a phenomenological approach to examine women's narratives of persistence in breastfeeding. The women who participated in this study spoke of the importance of persisting through difficulties and that this was the key to continued breastfeeding. Bottorff highlights that it takes time for women to adjust to the ever-changing and unpredictable nature of breastfeeding, which pre-empts an altered sense of self. Therefore, the need for encouragement from others who understand what persevering with breastfeeding means is crucial for breastfeeding women.

Women who persist with breastfeeding are defined by the belief that breastfeeding is the best gift that they can give their infants, and this conviction encourages women to continue to offer their bodies to their infants despite the pain they might be experiencing. A woman's level of commitment to breastfeeding is what will ultimately facilitate continued breastfeeding; however, this choice does mean giving up some of her freedom (Bottorff, 1990).

Being constantly in demand, both physically and emotionally, results in confusion for some women over their bodily boundaries, and pre-empts a desire to re-gain a sense of self-autonomy and determination. In Vares's (1992) study breastfeeding women discussed feeling at times as if their infant was devouring them, because it seemed that all of their infant's needs were met through breastfeeding. Other women reported that they were able to resolve this dilemma by consciously making the choice to be available to breastfeed, and through this decision they gained a sense of self-determination otherwise missing from the relationship between a breastfeeding mother and her child.

3.8 The Act of 'Successful' Breastfeeding

Successful breastfeeding can be defined in many ways, and extensive research has been conducted to predict which women will be 'successful' at various stages of the breastfeeding process. For example, research exploring what

facilitates successful breastfeeding initiation (Heath et al., 2002; Kessler et al., 1995; Matthews, 1991) has demonstrated that successful initiation is enhanced when a woman's intention has been to breastfeed (Heath et al., 2002). Women's exposure to social modelling of breastfeeding, which increases women's self-efficacy, also enhances successful breastfeeding initiation (Kessler et al., 1995). Research into the initial breastfeeding experiences indicates that women report higher levels of satisfaction with breastfeeding if they perceive that the infant responds positively (Matthews, 1991).

Research has also demonstrated that women who were breastfeeding exclusively at one month postpartum display greater overall satisfaction with motherhood when their infants are nine months of age (Isabella & Isabella, 1994, p. 93). It has been suggested that successful breastfeeding allows women to interact with their infants in a positive manner, which in turn supports the development of an emotionally satisfying relationship for breastfeeding women. In part, this has been attributed to women learning to respond to their infant's cues, which in turn elicits a positive response from the infant, and ultimately validates women's ability to mother. The concept that women learn to respond to their infant's cues has also been supported by women participating in Dignam's (2001) study.

It has been argued that successful breastfeeding has been traditionally understood to be the responsibility of the brain, which when working effectively achieves the desired breastfeeding outcome (Bartlett, 2002). Not only does the brain regulate hormonal responses, but it also initiates emotional responses, which are understood to either support or interfere with the physiological process of breastfeeding. Bartlett (2002) proposes that, if one holds to this line of thinking, breastfeeding is solely mediated by a woman's will to breastfeed, to the exclusion of the other factors discussed above. To mediate additional understandings of breastfeeding, Bartlett (2002) suggests that we should accept that women's bodies and breasts possess their own intelligence; when breastfeeding women learn to understand and articulate this bodily knowledge, then they are more likely to breastfeed successfully. This demonstrates how traditional definitions of breastfeeding success need to be critiqued and reframed, allowing new understandings of breastfeeding success to emerge.

Traditionally, women who have breastfed have been judged within a framework of success or failure. The use of the term 'successful' in relation to

breastfeeding has previously been critiqued by Harrison and colleagues (1985), who found that researchers and lay writers have defined and measured breastfeeding success in multiple ways, including duration (which can rate success anywhere from initiating to twelve months), prenatal variables, infant health, exclusively breastfeeding, relational achievement, and the management of breastfeeding within the context of the family (Harrison, Morse, & Prowse, 1985).

These measures of breastfeeding success are still evident in the current literature where success is measured by duration (Essex et al., 1995; Riordan & Auerbach, 1993; Schafer et al., 1998), correct technique (Henderson et al., 2001; Righard, 1998), health benefits (Australian College of Paediatrics, 1998; Ford & Labbok, 1990; Jones, 2001), and a lack of breastfeeding problems (Chapman et al., 1985; Inch & Fisher, 2000; Righard, 1998; Williams, 1997). However, definitions of breastfeeding success change across research studies, and also are mediated by cultural understandings of what constitutes successful breastfeeding (Coubrough, 1999; Hunter, 1998; Labbok, 2000; Labbok & Krasovec, 1990).

It is important to note that, to have the ability to attain breastfeeding success, it must be possible to fail at breastfeeding. Breastfeeding failure has typically been defined as breastfeeding cessation (Hunter, 1998; Laufer, 1990; Lethbridge, McClurg, Henrikson, & Wall, 1992). Laufer (1990) has developed a model to typify how breastfeeding failure occurs, which is described in the following quote:

The complex of maternal emotions and inadequate support, ignorance about routine breastfeeding management, awkward interactions with the baby, inappropriate supplementation, poor positioning, and poorly timed feedings lead not only to impaired production of oxytocin and prolactin but is also associated with increasingly ineffective technique, a poor milk supply, and poorly conditioned letdown reflex. This, in turn, leads to inadequate drainage, nipple pain, and an unsatisfied baby with an unacceptable weight gain pattern. The mother's resultant feelings of incompetence and failure both contribute to and feed on her everdiminishing self-esteem. She becomes increasingly awkward and ineffective... The baby does not learn how to nurse well, gets little milk,

and requires supplementation. Breastfeeding 'failure' has occurred (p. 40).

Laufer goes on to explain that breastfeeding success occurs when women either meet or exceed their breastfeeding expectations.

As evidenced above, measures of success are frequently created for the purpose of research and have little relevance for the individual breastfeeding dyad. However, it is clear from the literature that women do measure themselves against their own internal definitions of breastfeeding success, which are mediated through knowledge of society's accepted interpretations. It has been argued that women should individually define what breastfeeding success is for them, not be held accountable to arbitrarily-assigned measures of success that leave them open to critique and feelings of guilt and shame (Auerbach, 1994; Gigliotti, 1995; Hunter, 1998).

Even though women frequently develop culturally-mediated definitions of success, research that has delved into women's own definitions of breastfeeding success offers differing understandings of what constitutes success (Dignam, 2001; Leff et al., 1994). One qualitative study of 26 women highlights how they defined successful breastfeeding according to five categories: infant health, infant satisfaction, maternal enjoyment, attainment of desired maternal role, and lifestyle compatibility (Leff et al., 1994). These categories are not mutually exclusive, but rather describe various factors that contribute to how success is rated by the participants. Of particular interest is that women did not rate success on duration of breastfeeding; rather, they rated it according to how well they achieved their ideals in the above-mentioned categories. The primary message to emerge from this study is that women believe that breastfeeding is successful when they are working in harmony with their infants. This concept is supported by Dignam's (2001) study into breastfeeding and intimacy, which demonstrates that women who have the opportunity to experience intimacy with their infants whilst breastfeeding consider themselves successful.

3.9 Exploring Sensuality, Sexuality, and Breastfeeding

Western cultures frequently objectify the breast as a sexual object (Morse, 1989; Yalom, 1997), essentially denying other existential interpretations of the breast, such as the breast as functional. The failure of society to allow for multiple interpretations of the breast can result in women experiencing difficulty adjusting to breastfeeding. Breastfeeding frequently requires women to reconceptualise how they have understood their bodies, a concept that is explored extensively in the breastfeeding literature (Avery, Duckett, & Frantzich, 2000; Beasley, 1991; Beasley, 1996a; Beasley & Heritage, 1998/1999; Bentovim, 1976; Dignam, 1995; Dignam, 1998; McIntosh, 1985; Morse, 1989; Schmied & Lupton, 2001; Traina, 2000; Vares, 1992).

A certain amount of ambiguity exists between the terms sexuality and sensuality when exploring the issue of breastfeeding. Traditionally, researchers have employed the term 'sexuality' to explore women's understandings of their breasts, breastfeeding, and other adult relationships (Avery, et al., 2000; Morse, 1989; Traina, 2000; Yalom, 1997). However, feminist writers have tended to use the term 'sensual' to describe women's pleasurable bodily-based experiences, which captures feminist discourses surrounding breastfeeding (Carter, 1995; Schmied & Lupton, 2001; Vares, 1992). As a result, these two words, sensual and sexual, tend to be used interchangeably within the literature that explores women's bodily experience of breastfeeding.

Quantitative studies have explored how breastfeeding impacts upon women's experience of sexuality (Avery et al., 2000; Hillervik-Lindquist, 1992). Women have noted that breastfeeding did not alter their sexual relationships significantly; however, due to factors such as the need to wear a nursing bra and leaking whilst aroused, libido was reported to be diminished (Avery et al., 2000; Hillervik-Lindquist, 1992). One study highlighted how breastfeeding can be a pleasurable experience for women, although the majority of women do not experience sexual arousal whilst breastfeeding (Avery et al., 2000). Despite these issues, adjustments in sexual relationships for breastfeeding women following the birth of a baby do not differ significantly from those experienced by non-breastfeeding women, with all women citing fatigue as the major contributor to alterations in their sexual practices following birth (Maushart, 1999; Traina, 2000).

Societal perceptions on the sexual nature of women's breasts do, however, affect women's experience of breastfeeding. Bryant Merril (1987) notes that, faced with the need to satisfy both their partner's and their breastfed infant's needs, women may minimise any competition by favouring their partner. Women have also discussed how they struggled to maintain their breasts as a sexual asset whilst breastfeeding, due to such things as restrictive clothing and milk let-down whilst aroused. To continue to breastfeed, a woman must consider the needs of her infant to be more important than the sexual needs of her partner (McLeod et al., 1998).

Morse (1989) examined women's struggle to reconcile the functional breast with their understanding of the breast as sexual. One participant who had a ten month old son reported that her mother-in-law relayed that she was "afraid that her grandson is turning into a sexual pervert." Another participant, when reflecting on her own experience of her breasts whilst breastfeeding her infant, stated that she thought to herself "Euch, those are for your husband!" (Morse & Harrison, 1987, p. 229) Conceptualising the breast as a sexual object clearly inhibits breastfeeding.

Female sexuality is understood primarily in the context of adult relationships; however, a woman's sexuality can be understood to extend past this boundary to incorporate maternal sexuality. The silencing of maternal sexuality closes down upon language that explores women's experiences of pregnancy, childbirth and breastfeeding. For example, both suckling and sexual stimulation produce nipple erection and milk ejection, because the hormones that are released by a woman's body during breastfeeding are the same as those released during sexual climax (Riordan & Auerbach, 1993). It has been argued, however, that discussions about women's sexual experiences have been confined to the boundaries of adult relationships to maintain and ensure men's access to women, thereby lessening the threat to men (Morse, 1989; Traina, 2000).

The prevailing hegemonic discourse surrounding mothering is that mothers are asexual (Maushart, 1999), and that any deviations from this should be considered perverse, which essentially silences women's experience of sensual feelings whilst breastfeeding. Because of this, Traina (2000) questions whether there is a need to create a language that accepts the various forms of sensuality as sexual, thereby legitimising women's experience of breastfeeding as sexual.

However, it would be impossible to change cultural perceptions about motherhood and sexuality without the reconstruction of societal understandings of women, which would in turn affect what society considers morally acceptable behaviour (Morse, 1989; Traina, 2000).

To cope with the experience of breastfeeding in an acceptable manner within society, breastfeeding women frequently re-define their breasts as both sexual and functional (Dignam, 1998); essentially, they are re-framed as a commodity with a dual purpose. One study participant stated "We're using them [the breasts] during the day for something totally different from what they are used for at night" (Dignam, 1998, p. 82). This re-framing allows women to embrace both their roles as a breastfeeding mother, and as a sexual partner, in a society where breasts are primarily considered to be sexual.

3.10 Breastfeeding and Intimacy

To cultivate a language that describes the sensual nature and closeness that breastfeeding brings for some women, researchers have employed the term intimacy (Bottorff, 1990; Dignam, 1995; Dignam, 1998; Dignam, 2001; Schmied & Lupton, 2001). Although intimacy is most frequently associated with a sexual relationship, it can occur on many levels from friendship to a breastfeeding dyad. Intimacy is a safe and acceptable place from which to view a woman's experience of sensuality within the context of breastfeeding. This re-framing of the breastfeeding experience, which includes intimacy as an acceptable and inherent part, allows a previously-silenced part of the breastfeeding experience to be explored.

In a grounded theory study exploring the intimacy experienced by women during breastfeeding, Dignam (2001) included the phrase 'linking as one'. The women who participated in her study highlighted how intimacy through breastfeeding evolved through a process that began with 'being with' their infants. The process continued with the participants coming to know themselves as breastfeeding women, which included redefining how they understood their breasts and, therefore, their body. Finally, the journey involved coming to know their babies, with participants describing how understanding their infant's cues and body language assisted the journey towards intimacy. It is the place where

a woman knows herself, and knows her baby, that 'linking as one' occurs and intimacy through breastfeeding is achieved.

The women who participated in Dignam's (2001) study felt that breastfeeding entailed gifting their infant their milk, as well as comfort and physical contact. Dignam describes how the act of breastfeeding during an intimate moment is "mutually exclusive and mutually satisfying to both participants" (p. 246). A woman's experience of intimacy does not occur at every breastfeeding encounter; however, during moments when both the mother and baby are relaxed intimacy occurs. It is these intimate moments that foster a woman's sense of satisfaction with breastfeeding and provide her with the motivation to continue. It is acknowledged by Dignam that not all women experience intimacy with their infants whilst breastfeeding, and she asserts that this contributes to breastfeeding cessation. However, an understanding of what contributes to breastfeeding intimacy may allow health care professionals to facilitate it, supporting rather than undermining breastfeeding by doing so.

Dignam's findings are supported by Bottorff (1990) and Schmied and Lupton (2001), who also describe breastfeeding as an intimate practice. These researchers describe how women experience breastfeeding as an act that supports harmonious intimacy between a woman and her infant, the ultimate gift given by a mother to her infant. Women who participated in these studies articulated that they felt connected with their infants because of breastfeeding, and this interconnectedness persisted throughout the time they breastfed their infants. A woman's experience of breastfeeding intimacy was also unavoidably sensual and pleasurable, due to the nature of the physical contact and the emotional dependence of the infant on its mother (Schmied & Lupton, 2001).

Some of the women who participated in Schmied and Lupton's (2001) study discussed how they strongly desired to experience intimacy and harmony within their breastfeeding relationship with their children. This desire fuelled their commitment to persevere with breastfeeding despite difficulties. Those women who continued to breastfeed stated that they were 'glad' that they had because it afforded them the opportunity to experience an intimate relationship with their infant.

The idea of breastfeeding and intimacy appears elsewhere briefly in the breastfeeding literature; however, with the exception of the research into

breastfeeding intimacy described above, the primary lens through which intimacy is interpreted is women's struggle to re-conceptualise the breast as non-sexual. A woman's need to re-conceptualise the breast as functional versus sexual is exposed when women discuss their experiences of breastfeeding in public places.

3.11 The Process of Weaning

The process of weaning can be initiated by either the breastfeeding mother or the infant. Many of the reasons why women initiate weaning earlier than planned emerge from the above discussions on breastfeeding experience. As highlighted previously, women choose to initiate breastfeeding because it is by far the best source of nutrition for infants (see section 3.3). However, women's knowledge of, and commitment to, this ideal is not always sufficient to enable them to continue to breastfeed in the face of difficulties, with many women opting to wean earlier than planned (Blyth et al., 2002; Chezem, Montgomery, & Fortman, 1997; Gunn, 1984; Matulonis Williams & Morse, 1989; Morse & Harrison, 1987; Mozingo et al., 2000; Vogel & Mitchell, 1998b; Williams et al., 1996).

It has been demonstrated that women who lack confidence that they can breastfeed are more likely to wean their infant before two months postpartum (Blyth et al., 2002; Hill & Humenick, 1996; Ozturk Ertem et al., 2001). One study reveals that, if a woman believes her infant likes artificial formula more than breastmilk, she is also more likely to wean within two weeks of the infant's birth (Ozturk Ertem et al., 2001). This study suggests that neither breastfeeding difficulties nor perceptions of insufficient milk are directly related to premature weaning. However, women's confidence in their decision to breastfeed was predictive of longer periods of breastfeeding.

Maclean (1990) describes how the women who participated in her study weaned their infants earlier than planned for a variety of reasons, including a desire to re-claim their previous lifestyle, the physical discomfort of breastfeeding, infant responses to the breast, un-met or unrealistic expectations, a lack of knowledge about the breastfeeding process, medical advice, and the resumption of paid employment. These findings are supported

by other research studies (McLeod et al., 1998; Vogel & Mitchell, 1998b). However, simply itemising reasons for premature weaning fails to convey the emotional impact that early weaning has on a woman (see discussion in section 3.4 & 3.5). It also ignores the social context in which weaning occurs, which frequently mediates the length of time women breastfeed their infants.

In a qualitative study that examined the social coercion that exists around infant weaning, Morse (1987) found that during the initial breastfeeding period, women's social networks are supportive; however, by the time the infant is between nine and ten months of age people actively start encouraging women to wean. For the women who participated in Morse's study, this frequently meant that continuing to breastfeed required a withdrawal from social situations that drew critique of their infant feeding method. If a woman was breastfeeding when her infant was 12 months old, the infant's grandparents began to comment about the physically draining nature of breastfeeding for a woman. It is interesting to note that once husbands were no longer supportive of breastfeeding most women weaned within weeks.

Western cultures espouse strong discourses that dictate the expected duration of breastfeeding (Baumslag & Michels, 1995; Hills-Bonczyk et al., 1994; Morse & Harrison, 1987; Vares, 1992). Dignam (2001) suggests that women wean their infants prematurely because breastfeeding is a socially restricted practice that is often viewed as unacceptable. Coupled with this are perceptions of sexuality and partner support of breastfeeding, which can also significantly influence a woman's decision to wean. Murphy (1999) highlights how even breastfeeding women will verbally distance themselves from prolonged breastfeeding. Women report that prolonged breastfeeding indicates an inappropriate relationship between a mother and child, claiming that a child will become too dependent upon their mother. This idea was also discussed briefly by the women in Vares's (1992) study, who reported that they understood that society considered prolonged breastfeeding to be inappropriate.

Whenever a woman decides to wean, she engages in a process of letting go. If she has struggled to overcome breastfeeding difficulties and feels a sense of success in her accomplishment, weaning can signal a big adjustment, a letting go of something that was 'hard won' (Bottorff, 1990; Maclean, 1990; Vares, 1992). Weaning signals the end of an era, and the beginning of a new phase in a woman's and her offspring's life. As difficult as breastfeeding can be to

establish, many women have noted that they have experienced feelings of loss and pain at their last breastfeed (Bottorff, 1990; Maclean, 1990; McLeod et al., 1998; Vares, 1992).

3.12 Breastfeeding in the Presence of Other

New Zealand's Victorian legacy has had a significant influence on breastfeeding, and the mere fact that breastfeeding around others is an issue that warrants investigation gives credence to history's influence on the present. The sexualisation of the breast has left women with the need to cover this part of their bodies, significantly affecting how women experience breastfeeding. Even the commonly used phrase 'breastfeeding in public' alludes to the unacceptability of breastfeeding in the presence of others.

Understanding women's experience of breastfeeding around others provides valuable insight into how, as a culture, we interpret breastfeeding. Literature that examines this issue highlights the impact that societal depictions of the breast as a sexual commodity have on breastfeeding (Abel et al., 2001; Beasley, 1996a; Carter, 1995; Earle, 2002; Ellison-Loschmann, 1998; McLeod et al., 1998). Women who breastfeed in public frequently experience difficulty in navigating societal expectations of modesty whilst providing for the needs of their babies.

Breastfeeding in public can be an uncomfortable experience, as women often have heightened awareness of social attitudes and responses when breastfeeding in the presence of others (Benson, 2000; Carter, 1995; Dignam, 1998; Heath et al., 2002; McIntosh, 1985; Murphy, 1999; Sheeshka et al., 2001). Some women have even felt that they needed to overcome their own perception that breastfeeding in public is a deviant behaviour (Heath et al., 2002; Sheeshka et al., 2001). Research demonstrates that women who are embarrassed to breastfeed in any situation find breastfeeding less enjoyable and fulfilling, and as a result are more likely to wean early or opt for artificial formula (Heath et al., 2002; McIntosh, 1985). Research also demonstrates that some women's partners find breastfeeding in public embarrassing and offensive, opinions that negatively affect women's decisions about

breastfeeding (Earle, 2002; McIntosh, 1985; Murphy, 1999; Shepherd, Power, & Carter, 2000).

To minimise their discomfort over breastfeeding in public, women will try to find places where they can feed their infants whilst maintaining their modesty, which is usually achieved by either behaving in a discrete manner or securing a visually private place (Beasley, 1996a; Dignam, 1998; McLeod et al., 1998; Murphy, 1999). Women have stated that it is their commitment to breastfeeding that motivates them to breastfeed in public when their infants require feeding (Heath et al., 2002; Sheeshka et al., 2001).

Despite the pressure to keep the act of breastfeeding hidden, women have expressed strong resistance to resorting to feeding their infants in places like public bathrooms (McLeod et al., 1998; Vares, 1992). The mere fact that women would ever consider breastfeeding in a public toilet is an offensive proposition; no other population group, no matter how unacceptable they were perceived to be to society, would ever be forced to consider eating a meal in a toilet. However, for women who have difficulty breastfeeding around others this is sometimes considered to be an option. It is not surprising that some women comment on how grateful they are for facilities provided specifically for breastfeeding women (McLeod et al., 1998); however, if a mother feeding her infant in public was viewed with acceptance within our society, such narratives from breastfeeding women need not emerge when considering the experience of breastfeeding.

The desire to challenge societal mores by the act of breastfeeding in public raises awareness of the prejudices surrounding breastfeeding and, therefore, factors into some women's decision to purposely breastfeed in front of others (Ellison-Loschmann, 1998; Vares, 1992). Breastfeeding in the presence of others provides society with a visual statement about their sense of femininity, and also the importance of breastfeeding in nurturing their infants (Ellison-Loschmann, 1998; McLeod et al., 1998; Vares, 1992).

It is interesting to note that some of the teenage mothers who participated in Ellison-Loschmann's (1998) New Zealand-based research reported that the needs of their infants took precedence over the needs of others when they considered breastfeeding in public. These beliefs are also highlighted by Beasley (1996a). This has led some researchers to propose that young women

are less confined by definitions of socially acceptable behaviour (Beasley, 1996a; Ellison-Loschmann, 1998). However, the findings of a different New Zealand study that examined the experience of breastfeeding for adolescent women contradicts this view, with participants claiming that they felt conspicuous, embarrassed, and had a heightened sense of self-awareness when breastfeeding around others (Benson, 2000).

In a more recent New Zealand-based study, Abel and colleagues (2001) found that women and their partners did not cite breastfeeding in public as an issue. Although the study did not explore why this change may be occurring, it could indicate changes in the perceptions of both breastfeeding women and society at large regarding breastfeeding in public places. A recent Australian study also noted that women did not indicate that breastfeeding in public influenced their decisions to not breastfeed, supporting the notion that a shift in societal values is occurring (Schmied & Barclay, 1999). Nevertheless, it is clear that women who are embarrassed to breastfeed in public are likely to breastfeed for shorter periods than those who are not (Heath et al., 2002; McIntosh, 1985).

3.13 Breastfeeding and Employment

Research demonstrates that if women return to paid employment before one year postpartum they face considerable challenges if they wish to continue breastfeeding (Arora et al., 2000; Galtry, 1998; Vogel & Mitchell, 1998b). It has also been demonstrated that women engaged in paid employment are significantly more likely to feed their infants for shorter periods of time than those who are not in paid employment (Carlson Gielen et al., 1991; Hills-Bonczyk et al., 1993; Ryan & Martinez, 1989). The economic benefits of breastfeeding are undeniable; not only do women save the cost of artificial formula, women that breastfeed contribute to reducing the financial cost of the nation's health care (Galtry, 1997; Waring, 2000). These benefits are often invisible, and women do return to paid employment motivated by factors such as financial need and the preservation of previously-established careers.

Over the past twenty years, women in New Zealand have increasingly returned to paid employment following the birth of a baby (Callister, Podmore, Galtry, & Sawicka, 1995). However, they do not have access to paid breastfeeding

breaks (Galtry, 1998), which has a significant impact upon women who wish to return to paid employment whilst breastfeeding. This has been borne out through research, which indicates that New Zealand women frequently decide to wean their infants between six weeks and three months because they are returning to paid employment (Essex et al., 1995).⁸

Women who choose to combine breastfeeding and paid employment face a variety of challenges (Chezem et al., 1997; Hills-Bonczyk et al., 1993; Hochhausen, 1991; Ryan & Martinez, 1989). On a daily basis they must deal with practical issues such as securing resources for expressing and storing of milk, and coping with leaking milk at inappropriate times (Auerbach, 1984; Hochhausen, 1991; Ryan & Martinez, 1989). Role overload resulting in fatigue is also a major factor cited by women who are in paid employment. Having to address these issues on a daily basis means that some women decide to wean despite intentions to continue to breastfeed (Carlson Gielen et al., 1991; Hochhausen, 1991; Maclean, 1990; Ryan & Martinez, 1989). Some studies have determined that the main reason for women weaning their infant is directly related to their return to paid employment, especially if a woman is employed in a full time capacity (Chezem et al., 1997; Matulonis Williams & Morse, 1989). Other studies have noted that returning to work is a primary consideration when opting to artificially feed an infant (Arora et al., 2000; Maclean, 1990; Morse, Bottorff, & Boman, 1989). This appears to be true regardless of whether women are engaged in part-time or full-time employment (Morse et al., 1989).

The consequences of weaning earlier than planned for employed women are overarching feelings of guilt (Hochhausen, 1991; Maclean, 1990). Chezem and colleague's (1997) study reveals that women who do not manage to fulfil their breastfeeding goals whilst employed suffer from feelings of sadness, depression and guilt more than their counterparts, who may have weaned because of other maternal or infant issues.

⁸ In 2002, the New Zealand Government made a new provision for women in paid employment who choose to take parental leave; women now receive funding for the first six weeks following the birth of their infants. This funding will increase to cover the first 13 weeks postpartum in December 2004. Research is needed to assess if this positively impacts on women's experience of breastfeeding; of particular interest are women who are in relatively low-paid positions.

Galtry (1998) points out that for women to successfully combine breastfeeding and paid employment, they need both employer and collegial support. As there is no legal requirement for paid breastfeeding breaks, women need support from their employers to take the time needed to either express milk or feed their infants, and also have access to milk storage facilities as needed. Colleagues also need to be respectful of women's need for privacy if expressing milk in the work place. Women in professional positions frequently have access to private office space, which is a luxury often not afforded to their counterparts in lower-paid positions (Callister et al., 1995; Galtry, 1998).

Research that has examined the profile of women who successfully continue to breastfeed whilst engaged in paid employment describe women who work part-time in professional positions, are well educated, over 25 years of age, and earn a good income (Carlson Gielen et al., 1991; Hills-Bonczyk et al., 1993; Ryan & Martinez, 1989; Williams et al., 1996). Despite the picture this demographic profile paints, women engaged in paid employment breastfeed for shorter duration than their unpaid counterparts (Hills-Bonczyk et al., 1993; Ryan & Martinez, 1989). However, women are significantly more likely to continue to breastfeed their infants whilst employed if they are able to feed their infants during the working day. Unfortunately very few employed women have this luxury (Morse et al., 1989)

Women who manage to successfully combine breastfeeding and paid employment report a sense of satisfaction with their accomplishment (Auerbach, 1984; Hills-Bonczyk et al., 1993; Maclean, 1990). The research participants relay that the difficulties they had to overcome to continue to breastfeed were worth the effort, as they were able to continue to offer their infants a unique gift that they otherwise would never have received.

⁹ To facilitate women breastfeeding whilst engaging in paid employment, many suggestions have been put forth by researchers; however, a full exploration of these issues is beyond the scope of this discussion. Please refer to the sources indicated for additional information (Auerbach, 1984; Callister et al., 1995; Carlson Gielen et al., 1991; Chalmers et al., 1987; Chezem et al., 1997; Galtry, 1997; Galtry, 1998; Hills-Bonczyk et al., 1993; Hochhausen, 1991; Morse et al., 1989).

3.14 Summative Thoughts

This chapter has explored the literature that has been constructed around experiential accounts of breastfeeding. Historical, cultural and societal understandings mediate individual women's interpretations of their breastfeeding experience, and within this context, substantive patterns emerge when exploring this literature. One such pattern is that people within society construct cultural understandings of breasts and breastfeeding, and it is these constructs that mediate women's decisions about their infant feeding practices. Women's confidence in their ability to breastfeed can be determined by myriad factors such as internal dialogues and emotional responses to the situations in which they find themselves. Women's social networks also play a major role in supporting women to breastfeed.

However, the pattern that emerges as overwhelmingly dominant is 'breastfeeding as problematic' – a pattern containing a subversive dialogue that is quietly evident in the available research on breastfeeding, to the effect that health care professionals are considered to own the solutions to women's breastfeeding problems. This is demonstrated in the vast amount of literature authored by health care professionals who examine women's negative breastfeeding experiences, then go on to propose solutions to the problems women encounter.

Auerbach (1994, p. 70) offers her critique of the impact of research on women's perceptions of their breastfeeding experience when she states "if we characterise their experiences based on methodologically relevant definitions that are not necessarily reflective of women's experiences or their own views, we do a serious disservice to ourselves, our research science colleagues, the readers of our own research reports, and the women we profess to assist". Auerbach's comments highlight the effect that the framing of breastfeeding within research has on breastfeeding women and the society in which they reside.

Within the literature, women are often portrayed as ignorant, without knowledge of either their bodies or breastfeeding. This is evidenced by the phrases such as 'women struggled to put it into words' (for example see Schmied & Lupton, 2001). Such comments highlight how silenced women are in relation to bodily

knowledge. Researchers' discussions render women as incapable of competently articulating bodily functions and experiences; instead, they give words to women's experience, thereby taking ownership of what is not theirs.

It is important that researchers examine the experience of breastfeeding with a stance of openness. There is a need to return to the phenomenon afresh, examining what it is to be a breastfeeding woman without preconceived ideas about how it is defined and understood in the current research environment. Reflective lifeworld research provides a methodology for examining a phenomenon without the need for a pre-defined theoretical framework to guide the analysis and control the emergence of findings. It is through research that examines breastfeeding without the prejudices of previously-held understandings that what it is to be a women who is breastfeeding can emerge, as it exists for those who are experiencing it. To fully expound on the currency that reflective lifeworld research has for women's experience of breastfeeding, the following chapter examines the philosophical approach taken to understand the phenomenon under study – what it is to be a breastfeeding woman in New Zealand.

Chapter Four

Interpretive Lifeworld Research

4.1 Introduction

The term lifeworld stems from Husserl's philosophical exploration of phenomena, and describes the world as it is experienced everyday¹⁰. The philosophical contributions of Husserl (1967), Heidegger (1962), Gadamer (1989) and Merleau-Ponty (1981) provide a foundation for the inquirer seeking an enhanced understanding of the lifeworld. To better understand interpretive lifeworld research¹¹, on which this present work is based, it is useful to contextualise its philosophical underpinnings within the development of Western philosophy.

Traditionally, the phases of Western philosophy are set chronologically within their distinct periods of development: Ancient, Medieval and Modern Philosophy (Magee, 1987). A chronological approach allows the inquirer to understand and appreciate how the development of philosophical thought occurred through a process of calling forth new understandings, which in turn quickens further advances. The following examination of the history of philosophy, from its origins through to the relatively recent emergence of phenomenology and hermeneutics, is set out accordingly.

The principles that have guided this research into women's experiences of breastfeeding are explored in the context of this historical review. The work of the philosophers Heidegger, Gadamer, and Merleau-Ponty have emphasised the need to attend to phenomena with a stance of openness if our understandings of being are to be enhanced. It is also recognised that history has a significant influence on understanding, and the journey to understanding must necessarily involve an exploration of what has gone before. Finally, the

¹⁰ Husserl's conception of the lifeworld is discussed in section 4.3.1.

¹¹ The philosophy of interpretive lifeworld research is discussed in detail in sections 4.4.1 through 4.4.3.

principle of the hermeneutic circle has played a crucial role in this research. This principle tells us that the whole and the parts of a phenomenon have value and must be attended to for understandings to emerge. These guiding principles are traced through the history of philosophical thought in the following sections of this chapter.

4.2 History of Philosophical Thought

Western philosophy traces its origins to the Ancient Greek philosophers of the sixth century BC, whose thoughts affected the manner in which life during that period was interpreted (Kenny, 1998). Traditionally, classical Greek cultural beliefs were expressed in myths (Parker & Gibbs, 1998), which communicated "important lessons about a person's place and value within society and articulated the values, beliefs and laws of conduct of the culture they represented" (Parker & Gibbs, 1998, p. 147). In contrast, the philosophers of the Greek era, such as Pythagorus (582-500 B.C.), Anaximander (611-549 B.C.), Xenophanes (570-474 B.C.), Parmenides¹² (540-470 B.C.), and Heraclitus (540-475 B.C.), concerned themselves with the origin of the universe, developing theories about water, air, fire and earth to explain the composition and genesis of the world (Cavendish, 1964; Needham, 1999). They questioned beliefs and values that were not based in reason, logic and the observation of naturally-occurring phenomena, and thus challenged the classical Greeks' methods and ways of knowing.

During the fifth century B.C., professional educators, known as Sophists (which comes from the Greek word *sophia* meaning wisdom), became popular (Cavendish, 1964). They attempted to teach young men how to make rational decisions to facilitate successful lives. To achieve this, the sophists focused on the development of persuasive linguistic skills and the critical examination of taken-for-granted beliefs about the gods, morals and the State, all the while espousing the primacy of man. Protagorus (480-411 B.C.), the most famous of the Sophists, concerned himself with humanity through his focus on the primacy of education and the meaning of human essence (Needham, 1999). As a result

¹² It is interesting to note that Parmenides is the first recorded philosopher who was concerned with 'Being' (Kenny, 1998), a concept that is explored in section 4.3.1.

of the Sophists' extreme views of humanity, a philosophical transformation was ushered in, paving the way for Socrates (470-399 B.C.) and Plato (427-347 B.C.) (Cavendish, 1964; Kenny, 1998)

Socrates was most well known for examining the issues of ethics and politics (Huby, 1964), and was especially concerned with issues surrounding justice, morality and human existence (Magee, 1987; Needham, 1999). He believed that uncovering knowledge would result in virtue and happiness, and spent much of his life in dialogue with young men so that the knowledge generated through their discussions would provide them with the keys to virtue, happiness and morality (Kenny, 1998; Magee, 1987). Unfortunately, Socrates' teachings were not recorded in writing, but rather were passed on orally, so little is actually known about his philosophy other than inconsistent reports from other philosophers of the time, such as Plato, and Xenophon (Kenny, 1998; Huby, 1964).

After Socrates' death, Plato, one of his most noted followers, founded the 'Academy', a school for those interested in learning about issues of morality, metaphysics and mathematics (Kenny, 1998). He believed the everyday world that we experience is a façade, but that knowledge is real because it is enduring and beyond error, and, therefore, must be grounded in something outside of the everyday world (Huby, 1964). For man to be able to have knowledge of, and express, such things as happiness or justice, albeit imperfectly, happiness and justice must exist somewhere outside of the everyday world, in a universally applicable, unchangeable, perfect form (Ironside, 1997; Huby, 1964; Magee, 1987). Plato's ideas about knowledge represent the first attempt at understanding epistemology¹³ (Ironside, 1997; Magee, 1987).

Aristotle (384-322 BC) was the most distinguished of Plato's students (Magee, 1987; O'Connor, 1964). He was concerned with human experience (Magee, 1987) and disagreed with Plato's rationalistic dismissal and distrust of human senses as being 'unreal'. In his approach to the senses, he was "empirical, cautious, and anxious to consult all relevant facts and opinions before making up his mind" (O'Connor, 1964, p. 47).

¹³ Epistemology is the theory of the basis and methods of knowledge; the study of how knowledge is formed (Ironside, 1997; Palmer, 1969).

Aristotle believed that rationalism and logic provided a method to progress philosophical understandings (O'Connor, 1964). He developed a logical process based on an examination of the totality of what is known about a phenomenon (Magee, 1987), and the belief that if foundational premises about the phenomenon are valid and true, then the conclusions drawn logically from them must also be valid and true (Kenny, 1998; O'Connor, 1964). This theory of formal logic has been described as the science of deductive proof (O'Connor, 1964), and provides a means of explaining phenomena in greater structure and depth (Magee, 1987). Amongst Aristotle's writings were works that centred on epistemology and the nature of Being, which were known as the 'first philosophy', later becoming known as 'metaphysics' (Kenny, 1998; Rutty, 1998). Although fraught with problems, Aristotle's theories provided some legitimate basis for research that seeks to understand phenomena (Magee, 1987; O'Connor, 1964).

Development of what has become known as Medieval Philosophy spanned approximately one thousand years, from the fall of the Roman Empire during the fifth century to the Renaissance of the fifteenth century (Russell, 1996). During this period, commonly referred to as the Dark Ages, the church exercised dominion over philosophical development, and philosophers who did not adhere to the theocentric view were considered heretics, subject to imprisonment or execution. The philosophical work of this era focussed on the open pursuit of thought confirming the existence of the Christian God. There was a generalised dismissal of astronomy, music and mathematics; instead, Medieval philosophers espoused that God enveloped knowledge and eternal truth, and that this God could never be fully known.

Medieval philosophers strove to distinguish between knowledge that was divinely inspired and knowledge that was developed through human reasoning. The works of Plato and Aristotle were critically examined in an attempt to reconcile them with a monotheistic worldview. St. Augustine (354-430) and St. Aquinas (1225-1274) problematised the concepts of realism¹⁴ and idealism¹⁵, claiming that, in the quest for knowledge, faith versus reason was pivotal

¹⁴ Realism is the belief that physical entities have structures that exist independent of the human mind.

¹⁵ Idealism is the belief that physical entities only exist because the human mind perceives that they do.

(Kenny, 1998; Magee, 1987; Markus, 1964). However, this era is not associated with significant advancement in philosophical thought, as the main focus for its philosophers was the desire to reconcile Greek philosophy with Christianity (Magee, 1987).

The arrival of the Renaissance, towards the close of the Medieval era, ushered in an awakening of philosophical thought. The re-emergence of the Greek language encouraged philosophers to return to the writings of the early Greek philosophers, which led to a re-interpretation of the early Greeks' original works. More (1478-1535) offered his *Utopia*, which had similarities to Plato's *Republic*, as a critique of the current political and social environment, and Pico della Mirandola (1463-1494) argued for the supremacy of human nature. Because of such writings, humanistic philosophy came to the fore (Kenny, 1998).

In the early sixteenth century, Luther (1483-1546) challenged the blanket acceptance of Catholicism, and in the years following, strong debate about religion significantly affected the dominant philosophical discourses. Eventually, due to the complete lack of consensus amongst philosophers, scepticism gained favour (Kenny, 1998). Philosophers, such as Montaigne (1533-1592), who adhered to scepticism, argued that one could never be sure that he or she had acquired genuine knowledge due to the deceptiveness of the senses, the prevalent belief being that there would never be a method by which knowledge could be firmly and universally established (Ironside, 1997; Kenny 1998; Magee, 1987).

One of the important advances of the sixteenth century was that the science of physics began to be conceived as separate from the philosophy of nature. This followed from arguments claiming that observation and theory development took precedence over the conceptual analysis that was traditionally conducted by philosophers. This division of thought was subsequently pre-empted by Bacon (1561-1626), who categorised philosophy into differing branches, making distinctions between natural theology, natural philosophy (which included the natural sciences), and human philosophy. The major shift in the division of physics and philosophy during this era was appropriated by the philosopher Bruno (1548-1600) and the scientist Galileo (1564-1642), who believed that physics was advanced through the formalisation of hypotheses, observation and theory (Kenny, 1998). Although the demarcation was not clear until the eighteenth century, this era marked the beginning of the separation of

philosophy and physics, and ultimately philosophy and science (Kenny, 1998; Magee, 1987).

The predominance of scepticism during the sixteenth century promulgated the belief that there was no certain method by which knowledge could be determined. This position concerned Descartes (1596-1650), who became fixated on discovering a method through which absolute knowledge could be ascertained (Magee, 1987). To achieve this, he emulated the sceptics, and developed a method that commenced with a process of hyperbolic, or exaggerated, doubting of the entity under study (Burnham & Fieser, 2001; Magee, 1987; Sorrell, 1987). To undertake this process of hyperbolic doubting, Descartes believed that he must be capable of thinking; therefore, he is a thinking thing. This led him to coin the phrase 'I think, therefore I am' (Kenny, 1998; Magee, 1987; Sorrell, 1987). He argued that he could conceive himself without a body, but conception is not possible without a mind that is conscious of occurrences and thoughts; thus, the mind is the subject and the body is an object, an idea which has come to be known as Cartesian dualism. He proposed that people gain knowledge because they are subjects capable of scrutinising objects.

In keeping with the philosophical position that man's mind is a subject that is capable of knowing objects, Descartes claimed that the world was knowable as an object. This was an important advancement for the natural sciences, as it suggested that truth could be discovered through the science of observation (Burnham & Fieser, 2001; Leder, 1984; Sorrell, 1987). This desire for absolute truth led to the development of theories, concepts and thus knowledge (Magee, 1987). As a result of the contributions he made to philosophy through his work on epistemology, Descartes became known as the inaugurator of the era of Modern Philosophy (Kenny, 1998; Magee, 1987).

Descartes paved the way for rationalists such as Spinoza (1632-1677) and Leibniz (1646-1716), who believed that knowledge of reality stemmed from one's ability to use their mind (Magee, 1987). Rationalists adhered to the belief that the mind has an inherent capability to discern truth about existence and the earth (Cooper, 1999). These philosophers, using the concept of Cartesian dualism, believed that the world was understandable because their philosophy recognised that knowledge comes through observation and experience. For example, Spinoza believed that the reality of the world, which existed as an

amalgamated substance, could be fully explained through the employment of a systematic geometrical method (Kenny, 1998; Magee, 1987; Nidditch, 1964). Leibniz claimed that entities within the world are ordered and complex, and must be reducible to parts to facilitate analysis. Upon completion of analysis, what is left are simple indivisible phenomena; therefore, the 'whole' can be understood through knowledge of its various 'parts' (Kenny, 1998; Magee, 1987). And as such, rationalists held that science and the scientific method, as developed by philosophers, would provide knowledge and, thus, 'truth' about the world (Cooper, 1999).

In contrast to the rationalist position, the empiricists Hobbs (1588-1679), Locke (1632-1704), Berkeley (1685-1753) and Hume (1711-1776) adhered to the belief that all knowledge has a foundation in human experience (Cooper, 1999). Hobbs argued against Descartes' philosophy of Cartesian dualism, contending that experience was based on human senses, and thus knowledge is gained from an embodied experience (Cooper, 1999; Ironside, 1997; Kenny, 1998). Locke examined the underpinnings of human epistemology, and to achieve this he endeavoured to outline categories of human experience (Kenny, 1998; Magee, 1987; O'Connor, 1964). An important part of the epistemology of Locke is his belief that humans are born with a 'blank slate' upon which ideas are written, resulting in knowledge; however, to gain knowledge, individuals must be able think for themselves (Magee, 1987). Berkeley was also concerned with understanding epistemology through examining what it is to exist. He believed that existence was revealed through perceptions of existence. To exist, one must be able to acknowledge that he or she does exist, and this ability to perceive one's own existence forms the basis for knowledge acquisition (Kenny, 1998; Magee, 1987; Thomson, 1964).

Hume's philosophical thesis influenced the thinking of the German philosopher Kant (1724-1804) (Magee, 1987), who wrote *The Critique of Pure Reason* in 1781, in which he set out to offer a scientific basis for philosophy (Kant, 1999; Kenny, 1998; Magee, 1987). Kant wished to elevate metaphysics to the status of being *a priori* to knowledge. He outlined how not all knowledge arises from experience and, therefore, there must be an *a priori* knowledge, including such things as mathematics and physics that exist before experiential awareness. Kant called this *a priori* knowledge 'transcendental metaphysics' (Kenny, 1998).

The German philosopher Hegel (1770-1831) believed that ideas and concepts were historical and constantly evolving, thus facilitating the appropriation of new knowledge (Magee, 1987). In his book *The Phenomenology of Spirit*, the 'spirit' was driven to actualise its utmost potential, which it achieved through the accumulation of knowledge throughout the historical development of thought (Kenny, 1998). Hegel emphasised knowledge through the process of the dialectic: the resolution of the contradictions between the thesis and antithesis to ultimately achieve a greater level of synthesis (Ironside, 1997; Kenny, 1998; Magee, 1987).

It is at this point in philosophical development that the concept of phenomenology emerged as a distinct theme within the tradition of philosophy, primarily as a result of the philosophical thesis of Husserl (1859-1938). The following section focuses specifically on the philosophical development of phenomenology.

4.3 Philosophical Underpinnings of Phenomenology

The term 'phenomenology' is derived from the Greek words *phainomena*, meaning 'to show itself' or 'to appear', and *logos*, which means 'reason', 'ground' or 'saying' (Heidegger, 1962; Inwood, 1999). Heidegger (1889-1976) defined phenomenology as that which shows itself (1962). It "can be understood as an object, a 'thing' or a 'part' of the world, as it presents itself to, or, as it is experienced by, a subject" (Dahlberg, Drew, & Nystrom, 2001, p. 45).

Phenomenology emerged as a concept in the work of Brentano (1838-1916) (MacIntyre, 1964). An existentialist, he employed the term in opposition to the British empiricists and the predominant reductionist philosophies that claimed truth could be discovered if entities were reduced to the lowest common denominator (Dahlberg et al., 2001; Ironside, 1997; MacIntyre, 1964). He believed that natural science was the method of true philosophy (Spiegelberg, 1969), and focused on examining the structure of such things as believing, love and hatred to define his method of philosophical analysis, a method that mirrored those used in the natural sciences (Spiegelberg, 1969). It has been suggested that Brentano wished to facilitate the dominance of Cartesian

philosophy, because he claimed that one cannot doubt their inner experiences (MacIntyre, 1964).

Husserl went on to more fully develop Brentano's philosophy and thus became a predominant figure in the development of phenomenology. He proffered a strong critique of popular reductionist philosophies, arguing for experience-based knowledge (Dahlberg et al., 2001; Magee, 1987).

4.3.1 Husserl: The Father of Phenomenology

Husserl, known as the father of phenomenology (Gadamer, 1994), was a German philosopher who lectured at Freiberg University at the beginning of the twentieth century. He desired to turn philosophy into a rigorous science (Palmer, 1969), and his ultimate goal was to develop a philosophical approach to both the natural and human sciences. To fulfil his goal he focused on epistemology, in keeping with the philosophical tradition.

In his quest to uncover epistemological truth, Husserl espoused that 'conscious awareness' is the one foundational premise upon which knowledge can be established (Magee, 1987). Consciousness is directed out into the world from us to an object. The object projects itself back to the consciousness, and in doing so, the object presents itself as something – as meaning (Dahlberg et al., 2001; Husserl, 1967; Magee, 1987). He believed that for an object to exist, someone must perceive that it does. A person is conscious of an object that exists in nature; therefore, meaning is formed, resulting in knowledge. "Husserl points out that there can be absolutely no doubt whatsoever that the objects of our consciousness do exist as objects of consciousness for us" (Magee, 1987, p. 254).

Husserl believed that what is in the 'world' is learned through 'consciousnesses, and consciousness in turn becomes the source of a person's reality. He described this consciousness as a "uni-dimensional sequence of mental events" (Swearingen, 1977, p. 50). These two entities, the world and consciousness, are interrelated and inseparable (Moustakas, 1994; Spiegelberg, 1969), and to describe them he employed the terms 'embodiment' and 'lifeworld' (Polit & Hungler, 1995; Spiegelberg, 1969).

Husserl believed that the lifeworld was the central theme of phenomenology (Dahlberg et al., 2001; Husserl, 1967). Lifeworld describes "the world as it is experienced; it is a phenomenological concept because phenomenology is concerned with the human experience of phenomena, that is, the objects and events of everyday life" (Dahlberg & Drew, 1997, p. 303). Husserl desired to describe the lifeworld and thereby construct knowledge. Gadamer (1989) described Husserl's lifeworld as the "whole in which we live as historical creatures" (p. 247). The lifeworld is a communal world in which we live with others (Gadamer, 1989), and the task of phenomenology is to return to the everyday taken-for-grantedness of the lifeworld and examine experiences (Gadamer, 1989; Koch, 1994). Husserl attempted, through the development of phenomenology, to provide a way by which the structure of the lifeworld could be described and understood (Fjelland & Gjengedal, 1994).

A crucial part of phenomenology, according to Husserl, is the need for subjective openness that recognises the importance of 'returning to the self' to uncover the essential nature and meaning of phenomena (Moustakas, 1994). However, we dwell primarily in the 'natural attitude,' which describes "the everyday immersion in one's existence and experience in which we take for granted that the world is as we perceive it, and that others experience the world as we do. In the natural attitude we do not critically reflect on our immediate action and response to the world... we just are" (Dahlberg et al., 2001, p. 45-46). Whilst in the natural attitude we are unable to engage in critical analysis, and so we are unable to uncover the essence of phenomena.

A key concept in phenomenology is that of 'intentionality' (Husserl, 1970). Husserl utilised this term to more fully expound and describe the natural attitude (Dahlberg et al., 2001). Intentionality is our most basic, or primordial, mode of being, in which we unthinkingly navigate our way through everyday life (Husserl, 1967; Magee, 1987; Spiegelberg, 1969). We engage with our world uncritically using our pre-formed meanings of objects. Husserl (1967) espoused that these pre-formed meanings were 'intentional content' upon which the meanings of objects were inscribed, and thus consciousness was directed accordingly. Intentionally, our consciousness is directed towards an entity because it has meaning for us. For example, if a person wishes to write a note, they are conscious that a pen is needed to write the note, even if one does not exist within their perceptual field at that moment. When they see the pen, they are

conscious of it, pick it up, and commence writing the note; however, when writing, they are not critically engaged in thinking about the pen, but instead are focused on the meaning of the words being written.

It is the recognition of intentionality, and the conscious decision to overcome it that allows a philosopher or researcher to engage in 'epoche.' Epoche is the process of setting aside one's pre-conceived ideas about the world, allowing a person to approach a subject with open receptiveness (Moustakas, 1994). To achieve epoche, a person must engage in 'reduction', which is the process of identifying one's judgements, biases and prejudices regarding phenomena. Once reduction is completed one 'brackets' his or her beliefs and judgements. Bracketing is the process of putting aside previously-held understandings about a phenomenon so that preconceptions about a phenomenon are not subjectively reflected in his or her description of the experience (Parahoo, 1997). Through bracketing one's pre-conceptions, a person is able to engage in the process of 'intuiting' – looking at phenomena without any presuppositions or biases. It allows an individual to become absorbed in the phenomenon, to peel away the layers of experience and gain a fresh understanding of the 'essence' or true reality of them (Baker, Wuest, & Stern, 1992; Dahlberg et al., 2001; Husserl, 1967; Spiegelberg, 1969).

Husserl was concerned with why things present themselves in a form that veils their true essence (Husserl, 1967; Merleau-Ponty, 1981; Spiegelberg, 1969). He determined that all phenomena have meaning in both a historical and future context, and it is only when phenomena are understood in these contexts that they are truly known (Dahlberg et al., 2001; Merleau-Ponty, 1981). "Meaning is never finally complete but is always expandable, limited only by our readiness to enlarge our understanding" (Dahlberg et al., 2001, p. 59).

Husserl named Heidegger his successor at Freiberg University upon his retirement in 1929; however, the two had very different agendas for the development of philosophical thought. Husserl idealised that philosophy could be the basis for all science, and believed that Heidegger would further develop his phenomenological philosophy. However, while Husserl's philosophy of subjective consciousness epitomized the Cartesian tradition, Heidegger sought a more meaningful and adequate way to understand our relation to objects (Magee, 1987; Spiegelberg, 1969).

It has been contended that, due to Heidegger's departure from Husserl's original intent, he actually contributed to the corruption of pure phenomenology. Heidegger believed that Husserl "remained captive to the tradition he inadequately explored" (Inwood, 1999, p. 161) and he highlighted his concerns about traditional philosophical approaches to epistemology, in the form of Cartesian ideologies, to free himself from such beliefs (Heidegger, 1962; Inwood, 1999; Magee, 1987). Therefore, these two philosophers focused on the development of different spheres of thought – Husserl on epistemology and Heidegger on ontology¹⁶.

Despite Heidegger's unwillingness to accept that human consciousness held the key to understanding all phenomena, he used the concepts expressed in Husserl's phenomenological method to underpin, open-up, and expound on the being of humanity (Ironside, 1997; Palmer, 1969). In doing so, he dramatically altered both phenomenology and the phenomenological method. To emphasise this, Heidegger employed the term 'hermeneutic' to describe his philosophical project, signifying his departure from a scientific approach to phenomenology, in contrast to Husserl's desire for philosophy to become a rigorous science (Palmer, 1969). Whereas Husserl focused on describing phenomena as they appeared to the consciousness (Dahlberg et al., 2001; Koch, 1994), Heidegger asserted that phenomenology was historical, "a creative recovery of the past, a form of interpretation" (Palmer, 1969, p. 126).

4.4 Hermeneutics

The Greek word *hermeneutik* means "a method of interpretation" or "the doctrine or study of interpretation" (Inwood, 1999, p 87). It has its origins in the words for philology (historical linguistics) and theology (the study of gods), and the Greek word *Hermenuein*, meaning 'to interpret' (Inwood, 1999). The ultimate goal of hermeneutics is to achieve a deeper understanding of phenomena (Palmer, 1969).

Schleiermacher (1768-1834) desired to develop a science that understood language, introducing the term hermeneutics to describe a systematic

¹⁶ Ontology is the science and study of being (Ironside, 1997; Palmer, 1969)

procedure for interpreting both theory and understanding (Dahlberg et al., 2001; Palmer, 1969). He also believed that it could be extended to the domain of the human sciences, in that hermeneutics could be employed to understand such things as art and history (Dahlberg et al., 2001; Inwood, 1999). He believed that hermeneutics involved the "reconstruction and re-experience" of the thought processes of an author, as the original intent of the author was needed to fully understand the text (Ironside, 1997, p. 85). To achieve this, Schleiermacher sought to access the principles or processes by which a formalised objective method for interpretation could be developed.

An important aspect of Schleiermacher's formalised interpretative method was the need to attend to the context in which the original author was writing. To understand a text one must understand the author's culture and life circumstances (Dahlberg et al., 2001; Gadamer, 1989); one must attend to the parts (words) and the whole (context) of that text (Annells, 1996; Dahlberg et al., 2001) – a concept that later became known as the hermeneutic circle (Annells, 1996).

Another philosopher who contributed to the development of hermeneutics was Dilthey (1833-1911), a contemporary of Schleiermacher, who continued to focus on the development of objective hermeneutics with the hope of discovering a method by which to uncover apodictic knowledge (Ironside, 1997; Palmer, 1969). However, Dilthey was opposed to the reductionist ideologies of his time, and sought to develop a method of uncovering meaning that would be more appropriate for human sciences (Dahlberg et al., 2001). In the process, he redefined hermeneutics to include the interpretation of written and spoken text, so that the focus of understanding was not just on an author's intent, thereby introducing hermeneutics as a fundamental way by which human sciences could be understood (Dahlberg et al., 2001; Ironside, 1997).

Dilthey was stimulated by a quest for understanding life (Heidegger, 1962). He conceived that man had the power of understanding, and personally sought to understand "life from out of life itself" (Palmer, 1969, 124). "Starting from 'life' itself as a whole, he tried to understand its 'experiences' in their structural and developmental inter-connections" (Heidegger, 1962, p. 72). Dilthey espoused that all humanity has life, or lived, experiences, and through these common structures others' experience of life is understandable (Dahlberg et al., 2001; Ironside, 1997).

The philosophies of Schleiermacher and Dilthey provided the foundation for hermeneutics, and their ideas influenced and informed Heidegger's understanding of it (Dahlberg et al., 2001; Gadamer, 1994; Palmer, 1969). He subsequently interpreted phenomenology as hermeneutical, and as such he sought to use it to extricate "the meaning of a text as a whole, a text obscured by past interpretations of it" (Inwood, 1999, p. 161).

Heidegger applied Schleiermacher's and Dilthey's understandings of hermeneutics and life, and in doing so constructed a hermeneutic of ontology (Inwood, 1999; Palmer, 1969). His understandings of hermeneutics provide a foundation for understanding human existence, by linking the theories of lived experience, ontology and phenomenology (Palmer, 1969). He "points to a ground for hermeneutics not in subjectivity but in the facticity of world and in the historicality of understanding..." by embedding the essence of hermeneutics in "the ontological power of understanding and interpretation which renders possible the disclosure of being of things and ultimately the potentialities of Being's own being" (Palmer, 1969, p. 130 & 137).

4.4.1 Heidegger

In *Being and Time*, Heidegger disputed the focus of traditional philosophy that sought to understand epistemology, believing instead that philosophy should focus on understanding ontology (Ironside, 1997; Palmer, 1969). Ontology embraces such issues as the nature of existence and the structure of reality, providing the basis for epistemology. In other words, we must exist before we can gain knowledge. Therefore, Heidegger believed that the path to understanding humanity does not lie in examination of how we gain knowledge; rather, it lies in inquiry into the nature of being.

Heidegger (1962) examined the ontological and proposed that we understand the world through being. He focused his arguments around the understanding of a Being which is there, otherwise understood as: the being there of existence (Dahlberg et al., 2001), the being of humans or the person who has this being (Inwood, 1999). This has also been defined as a being capable of understanding (Guignon, 1983). Heidegger named this type of being, which exists and is capable of understanding, 'dasein'. He positioned his beliefs in binary opposition to Cartesian dualism and argued instead that dasein is

embodied; we exist as whole human beings, not as a mind, body or spirit that is capable of separation. Typically, we are not cognisant of our bodies, as our bodies are transparent to us unless something shows itself as problematic, such as in the case of illness (Inwood, 1997).

Dasein is always situated and engaged in a world; it is impossible to understand dasein apart from the world; therefore, Heidegger proposed that dasein is a 'being-in-the-world' (Heidegger, 1962; Magee, 1987). The term being-in-the-world describes the enduring connection between humanity and the world (Dahlberg et al., 2001). The world is a priori to all understanding and meaning, as the world houses culturally-mediated ideas, languages, and common practices, as well as meaningful relationships (Heidegger, 1962; Palmer, 1969). Meanings are formed and entities are understood because dasein has, as its mode, that of being-in-the-world; it is, therefore, constituted by, and also constituting, the world in which it is situated (Leonard, 1994). The world is not necessarily visible to us as it is the taken-for-granted backdrop upon which existence is played out.

It has been contended that Heidegger used the expression being-in-the-world instead of the term lifeworld, which was employed by Husserl (Dahlberg et al., 2001); however, for Heidegger, the word world meant more than just 'the everyday world about us', as described by Husserl; it also embraced the concept of the familiar world as a workplace, a place that is known, and in which one experiences and builds experience. The familiar world exists within a world that is unfamiliar; however, when the unfamiliar world is accessed, that too becomes the familiar world. The world exists in "indefinitely expanding circles of decreasing familiarity" (Inwood, 1997, 247).

The world is *a priori* and, as such, *dasein* does not choose the world in which it is found; instead, *dasein* is 'thrown' into a world, and inhabits the world in a position of 'thrownness' (Heidegger, 1962). *Dasein's* thrownness mediates understanding of such things as language and culture. For example, one can understand that a fork is for eating only if one exists in a 'world' in which forks are used for eating. Therefore, *dasein's* interpretive freedom is mediated by the world into which it is thrown (Heidegger, 1962; Leonard, 1994).

Dasein is always interpreting the world into which it is thrown. Heidegger (1962) proposes three foundational concepts that form the basis of the ability to

interpret and, thus, understand: 'fore-having', 'fore-sight', and 'fore-conception'. When one turns his or her attention to an object with a previously-formed understanding of what the object is, the object is approached with the forehaving of understanding. Before attention can be focused on an object, a person must be able to see the object in advance, which is fore-sight. One must also be able to grasp and conceive an understanding of an object prior to approaching the object, which is the result of fore-conception. "Whenever something is interpreted as something, the interpretation will be founded essentially upon fore-having, fore-sight, and fore-conception. An interpretation is never a presuppositionless apprehending of something presented to us" (Heidegger, 1962, p. 193). Fore-having, fore-sight and fore-conception exhibit themselves in commonly-formed meanings that are developed as a result of our thrownness in a world where these meanings already exist. These commonlyformed meanings provide the basis for interpretations of objects that appear in the world. If one does not obtain fore-having, fore-sight, and fore-conception, one is incapable of understanding the object one approaches, and no interpretation can occur.

Dasein can focus attention on objects because dasein's primordial and essential mode of being within the world is that of focus or 'care'. This primordial mode of care provides dasein with an ontological basis for understanding, and thereby provides the ability to navigate the world into which it is thrown (Heidegger, 1962). Through care we are comported to that which we are concerned with, and thereby we come to understand being through the everyday things that capture our focus or care (Heidegger, 1962; Inwood, 1999). In accordance with our thrownness, our care is directed towards entities that call for our attention, and thus through care meanings are formed (Heidegger, 1962; Leonard, 1994). There is no uniformity to what calls out for our attention – it is particular to each individual being.

Dasein exists in the world 'with-other' – with-other people as well as with-other objects – and through the structure of care, as outlined by Heidegger, dasein is able to be mindful of and connect with-other (Heidegger, 1962). When encountering objects, dasein's attention is directed by 'concern' about the object; concern manoeuvres and uses objects to achieve desired outcomes (Heidegger, 1962). However, the care with which dasein attends to other people differs from the type of care, or concern, that facilitates the manipulation

of objects: *Dasein* attends to other people with 'solicitude' (Heidegger, 1962). In its most authentic state, solicitude can facilitate another reaching their ownmost potential for being.

Just as Heidegger employed the phrase being-with-other to describe how we are with others in the world, the others are also with-other (Dahlberg et al., 2001; Heidegger, 1962). As a result of being-with-others and the 'others-being-with-others', commonly-formed meanings and understandings emerge through the interactions of the others or 'the they'. The understandings formed by 'the they' produce a set of expected ways of behaving, and thus provide a framework for interpreting which practices are acceptable and which are not. Thus, 'the they' can hold sway over one's decisions and ways of being within the world in which one exists (Heidegger, 1962).

Although traditional scientific thought has conceived time as linear, with properties or experiences perceived as encapsulated events (Leonard, 1994), Heidegger (1962) proposed a new way of conceptualising time. He claimed that the linear view of time fails to embrace the manner in which a being will interpret the future in light of previous experience. Our expectations and experiences, which are formulated by our past, will essentially shape our interpretations of the future. Therefore, the past is always ahead of us (Heidegger, 1962).

Heidegger desired to highlight how humanity had become estranged from its essential being, claiming that being's ownmost potentiality had been lost through misunderstandings and concealment of being (Cooper, 1999; Wood & Giddings, 2001). Heidegger (1962) employed the terms authentic and inauthentic to describe how *dasein* becomes estranged from its essential being. *Dasein* is authentic if it is owned by truth, and 'being' is the priority. To be authentic means making one's own decisions separate from 'the they'. Conversely, if *dasein* submerges its interpretive capabilities amongst 'the they' it is being inauthentic (Heidegger, 1962; Inwood, 1999). The call to authenticity always exists for *dasein*, and while it may respond to the call in part, *dasein* struggles to find its essential being (Inwood, 1999).

Heidegger (1962) wrote extensively on how being is understood only superficially through such things as the scientific understandings of the world (Cooper, 1999). He highlighted how being's essential modes appear through letting the meaning of entities, such as events, time, equipment and being-with-

others, emerge. Heidegger's exposition of being is a call to thinking (Cooper, 1999), a call to meditate upon being, a call to uncover being's ownmost potential.

4.4.2 Gadamer

A contemporary of Heidegger, Gadamer (1900-2002) wrote *Truth and Method*, in which he laid out a highly critical account of positivism. He claimed that any scientific method designed to understand being would be pointless, because understanding existence would never occur through the use of a particular positivist method. Instead, he suggested that understanding must occur through being 'open' to the world. If we wish to see and understand the 'otherness' of something, we must have an open mind when approaching phenomena.

Gadamer believed that it was possible to uncover the otherness of phenomena, and to achieve this he turned to hermeneutics. In attending to the philosophical basis of hermeneutics, Gadamer problematised Schleiermacher's belief that historical reconstruction facilitated understanding of an author's intent and indeed of the author himself. Instead, Gadamer argued that such reconstructions were impossible because any attempt to do so fails to grasp the historical context in which the writing occurred, thus doing a disservice to the author's original meaning: "a hermeneutics that regarded understanding as reconstructing the original would be no more than handling on a dead meaning" (Gadamer, 1989, p. 167). All literature, artwork, and events that have gone before are tradition and are only ever interpreted via presently-held conceptions.

Both history and tradition provide the foundation for the present; therefore, any understandings or meanings emerge accordingly (Gadamer, 1989). It is upon this premise that Gadamer builds his exposition of hermeneutics. He believed that awareness of this historical consciousness is central to the human sciences because "understanding is, essentially, a historically effected event" (p. 300).

Historical consciousness is primarily unrecognised and not under conscious control; however, it is the mechanism that directs one's attention to points of significance. At the same time, the significance of any particular experience will highlight what is important to the individual involved. This rising up of

significance results in certain aspects of the phenomenon being veiled, thereby blocking the whole truth about the phenomenon. Gadamer (1989) states that "we should learn to understand ourselves better and recognise that in all understanding, whether we are expressly aware of it or not, the efficacy of history is at work" (p. 301). To be aware of one's own historical consciousness, one must adopt a reflective attitude toward one's own selfhood, current situation and any inherited traditions. One "understands itself in terms of its own history"; therefore, "historical consciousness is a mode of self-knowledge" (Gadamer, 1989, p. 235). It is acknowledged that complete knowledge of one's historicity is unattainable because one's vision is limited by the situation in which one exists.

In part, our vision is limited because our historical consciousness facilitates the creation of 'prejudices' (Gadamer, 1989). Such prejudices govern and limit ones ability to understand, as they close down upon one's ability to attend to another's meaning. Through acknowledging a prejudice, its existence is called into question, and thus, fresh understandings previously hidden by the exposed prejudice can emerge (Gadamer, 1989). A prejudice will function unnoticed unless it is provoked through being called into question. Through a process of seeking to understand the historical consciousness, and the prejudices created as a result, a researcher comes to appreciate how his or her own horizon may close down upon the ability to be open to new understandings. Prejudices "represent that beyond which it is impossible to see" (Gadamer, 1989, p. 307).

Gadamer (1987) explored how one's own horizon influenced understandings of the world. This concept, first employed by Husserl (1967), acknowledges that one's horizon incorporates a personal history that, combined with the horizon of tradition, provides the canvas for understanding. "The horizon of the past, out of which all human life lives and which exists in the form of tradition, is always in motion" (Gadamer, 1989, p. 304). One's thrownness — both the past and present context in which one occurs — constitutes a particular horizon, and thus, an interpretation of the world that in turn gives rise to an individual's understanding. A horizon might be narrow or wide, and the significance of an issue varies according to its position within a horizon.

This philosophy might be interpreted as reinforcing solipsistic¹⁷ ideals; however Gadamer (1998) believed that language, which never occurs in a vacuum, stems from tradition. It is language that is the fundamental mode of being-in-the-world. Because of this, individuals interpret their horizons through language. Understanding occurs when individuals experience a 'fusion of horizons'; this transpires when one fuses his or her horizon with another's horizon through the use of language (Annells, 1996; Gadamer, 1989). When a fusion of horizons occurs between two individuals, understanding develops, a common language is created, and a new community is formed; the emergent, and jointly held, horizon becomes the antecedent for a transformation where individuals are irreversibly altered because of the new understanding that has occurred (Gadamer, 1989). The portal of language provides a way to understand another; through the dialogical process we gain the ability to meld and constitute our individual horizons; therefore, we constitute, and are coconstituted by, each other's worlds.

Gadamer believed that those engaged in hermeneutic research must acquire a horizon that enables them to question the lifeworld, since without it, significant questions can never be called forth. The failure to ask significant questions results in an inability to truly understand. Gadamer (1989), therefore, claims that true understanding emerges through a dialectical process of questioning and considering others' horizonal position. This questioning does not take the path of arguing, but of thinking: "for in this process what is said is continually transformed into the uttermost possibilities of its rightness and truth, and overcomes all opposition that tries to limit its validity" (Gadamer, 1987, p. 367).

Ultimately, Gadamer (1989) criticised strict adherence to any one method as a means of understanding existence. This position has implications for phenomenological researchers in relation to the development of the research method. If a researcher adheres to a previously-developed and prescribed method, openness to the phenomenon under study is potentially compromised and access to true understanding is closed down upon. "What the tool of method does not achieve must – and really can – be achieved by a discipline of questioning and inquiring, a discipline that guarantees truth" (Gadamer, 1989, p. 491).

¹⁷ Solipsism is a branch of philosophy that espouses that no comprehension exists other

4.4.3 Merleau-Ponty

Merleau-Ponty (1907-1961) was a prominent French philosopher who explored theories of human perception. He argued that the existent theories on this topic failed to fully explain bodily perception, because they were confined by the inherited Cartesian tradition that conceived the body as an object. Bodily perception, due to its complexity, could never be explained through a Cartesian philosophy that promulgated a rational scientific method; therefore, he argued that one must call forth a language that precedes science and Cartesian dualism.

The position for Merleau-Ponty's examination of bodily perception is the lifeworld; he believed the lifeworld preceded all knowledge because it is pre-reflective (Dahlberg et al., 2001). The lifeworld is the starting point for all experience and understanding. It is through this pre-reflective lived world that any bodily perceptions are experienced; as a result, our bodies belong to our lifeworld, and because of this belonging, our bodies become the portals by which experience comes to us. Merleau-Ponty (1981) stated that "I am the absolute source, my existence does not stem from my antecedents, from my physical and social environment; instead it moves out towards them and sustains them, for I alone bring into being for myself..." (p. ix). The body and the world are bound together, and through this union subjectivity is conceived; as subjectivity is born through existence, human existence exists as a body, and this body is one with the world (Merleau-Ponty, 1981).

Merleau-Ponty (1981) set forth to redefine the manner in which our existence is interpreted. Traditional interpretations of the body, as a subject that is known by an object (the brain), inadequately articulated the complexity of how a body perceives and operates within the world. He described how we exist as embodied; we exist as a body capable of both physical movements and non-physical reasoning, not just a mind that thinks and exists separately from a body, which moves, as commanded, around a physical environment. "The union of soul and body is not an amalgamation between two mutually external terms, subject and object, brought about by arbitrary decree. It is enacted at every instant in the movement of existence" (Merleau-Ponty, 1981, p. 89). The

human body is always a lived body and it is this lived body that provides humanity with meaning.

In arguing for embodied understanding, Merleau-Ponty (1961) demonstrated how one's body is experienced differently from other objects. One's body is always present. One cannot observe their body as they do a chair that can be examined at a distance or from differing positions. One is incapable of seeing his or her face or back without using a mirror. It is impossible to be without one's body because it is always already present. One cannot turn away from their body. The "body is constantly perceived" (Merleau-Ponty, 1961, p. 90). One's body is the portal by which one gains access to sensual experience through the world; our knowledge of the world and all experience comes to us through our bodies. It is, therefore, understood that one is the body because they are jointly physical and mental; it is impossible to sustain either without the other (Merleau-Ponty, 1981; Priest, 2000).

Merleau-Ponty (1981) described how understanding emerges as one's body facilitates experiences, and thus the body provides the ability to function within the world. Understanding, and ultimately meaning, come through being a body "which rises towards the world" (p. 75). Our bodies are, therefore, the primary receptacle for information from the external world, and it is from this position of being-in-the-world that we gain experience (Merleau-Ponty, 1981). We are a body that exists intrinsically bound to the world; through our bodily being-in-the-world, the experienced world is sustained and constituted because through this interplay systems of meaning are formed. These systems are supported by, and sustain, the world and allow for the development of complex social structures that have resulted in conceptualising notions such as that of the objective body.

As with Husserl (1967) and Gadamer (1989), Merleau-Ponty (1981) employed the word horizon to explain how one's experiential history constitutes the present. Experience causes the living present to open upon a past horizon, as well as potential future horizons that have not yet been lived. These experiential horizons also grant one access to social horizons that have been constituted by societal history. Combined, these differing horizons form a collective horizon that is carried forth by one's existence and forms the basis for interpretations.

Merleau-Ponty (1981) facilitated a full and complete re-conception of how to understand the human body through his philosophy of perception. The starting place for his philosophy is the pre-reflective lifeworld, or phenomenal field. He explains how we are embodied and that experience of the world comes to us through our embodiment. This experience of the world facilitates the formation of interpretations and meaning, and ultimately creates horizons. It is through this process that complex social structures develop and are sustained through the embodied body that always rises towards the world.

4.5 Application of the Methodology to the Current Study

4.5.1 Situating this Study

This inquiry is situated in the philosophical tradition of those that sought to understand the lifeworld. Throughout its course, I have been immersed in the works of lifeworld philosophers such as Husserl (1967), Heidegger (1962), Gadamer (1989) and Merleau-Ponty (1981), along with the interpretations of lifeworld philosophies belonging to a number of modern philosophers and nursing researchers. This research represents my interpretations of Husserl, Heidegger, Gadamer, and Merleau-Ponty's work, and in some respects will differ from their original intent; however, they have been dialogical partners in this journey, assisting in my attempt to understand what it is to be a woman who breastfeeds in New Zealand. These philosophers have co-constituted my understandings and interpretations of being, as have the women who participated in this study.

The writings offered by the above-mentioned philosophers give the lifeworld researcher direction on how one might examine the experiences of humans who reside within a world of customary understandings. It is a world that a researcher is also embedded in experientially, which highlights the need for a researcher to be aware of the meanings and understandings that are brought to any phenomenon under study.

The philosophical work of those that strove to understand the lifeworld has provided principles by which phenomena can be explored, analysed, and

understood. These principles guide and assist an interpretive lifeworld researcher when seeking to understand and call forth the essential meaning of a phenomenon. The fundamental principle that has guided this research into the experience of being a breastfeeding woman is the need to approach the phenomenon with openness. Openness comes from understanding how history has influenced previously-held interpretations of this phenomenon. Holding to the principle that it is the combination of many parts that constitutes the whole of a phenomenon is also important when attempting to engage a full understanding. Finally, there is the need to ensure that all interpretations are firmly established in, and accurately reflect, the participants' narratives so no departures from their original intent occur.

The following discussion explores the foundational principles that have governed this research project, and the way I have approached the phenomenon of breastfeeding within New Zealand. It is the principles of openness, historicity, and the hermeneutic circle that have given rise to the process for this research ¹⁸.

4.5.2 Openness to the Phenomenon

Traditional methods of inquiry have limited the questions we can ask as we attempt to uncover and enhance our understanding of what it is to be a human (Leonard, 1994). Such inquiry does not lend itself to quantitative methods, as "dasein's shared way of behaving... contains an understanding of being that must be studied as an interpretation" (Dreyfus, 1991, p. 19). However, interpretive phenomenology has created a way forward through the concept of openness. Openness has been described as "a perspective free of unexamined assumptions" (Dahlberg & Drew, 1997, p. 304), and thereby a means of attending to the phenomenon as it presents itself (Dahlberg et al., 2001; Gadamer 1989). Although the researcher is not able to achieve this completely, openness is constantly strived for – calling into question one's historically-effected consciousness, and prejudices, and vigilantly assessing how these preunderstandings affect one's interpretations of the narratives (Gadamer, 1989). "The important thing is to be aware of one's own bias, so that the text can

¹⁸ The specific process undertaken for this research is elaborated on in Chapter Five.

present itself in all its otherness and thus assert its own truth against one's own fore-meanings" (Gadamer, 1989, p. 269).

Openness involves waiting for the "phenomenon to reveal its own complexity rather than imposing an external structure on it, such as the dogmatic use of theories or models" (Dahlberg et al., 2001, p. 111). The researcher must seek to become the servant of the text, through being open and listening to the text; it is in that place that the researcher will hear what the text is saying (Palmer, 1969). There is a need to engage in a process of waiting, and of willingly submitting oneself to openness, for it is in this place that understandings come (Gadamer, 1989; Smythe, 1998). It is through approaching the phenomenon with such a stance that the 'otherness' of the phenomenon can emerge (Gadamer, 1989; Smythe, 1998). Therefore, the guiding principle of this study has been openness to the participants' narratives, so that the phenomenon of what it is to be a woman breastfeeding in New Zealand could emerge, providing a deeper understanding of what it is to be in that place.

4.5.3 The Influence of History on Understanding

One cannot separate oneself from the world in which he or she exists; therefore, all of one's interpretations will emerge from the pre-understandings that they hold of their world. It is important when examining any phenomenon that one attends to the historical narratives that surround it. One cannot examine the future, nor seek to uncover pre-understandings, without first dissecting what has gone before, as traditions, values, and historical discourses all affect how we interpret the present (Gadamer, 1989; Heidegger, 1962). By attending to the historical influences, pre-understandings can emerge and be acknowledged (Smythe, 1998). All interpretations will emerge through one's historical context because it is this that forms the basis of currently-held meanings and understandings.

This research project examines how breastfeeding has been traditionally understood in New Zealand through a review of historical descriptions (see Chapter Two). As a New Zealander, I have been immersed in the country's culture and history, and my understandings of breastfeeding will be affected accordingly. It is acknowledged that one can never completely 'free' oneself from the influence of history on one's understandings. One can make clear their

'horizon' through careful articulation of their world view, but one can never divorce oneself from one's pre-understandings, culture and traditions (Gadamer, 1989). Throughout this study, I have been constantly mindful of the need to examine that which may be veiled by my prejudices and pre-understandings, calling into question all interpretations accordingly.

4.5.4 The Hermeneutic Circle

Schleiermacher, when developing his formalised interpretative method, discussed the importance of attending to both the whole as well as the parts of a text. When reading a part of an interpretation, one must always be mindful of the context in which the interpretation occurs (Dahlberg et al., 2001; Gadamer, 1989). There is a constant flux between the whole and the parts that must occur for understanding to emerge, which describes the hermeneutic circle.

One's understanding of a phenomenon is never complete. Understandings change with time as they are always contextualised in ever-evolving cultures and traditions (Gadamer, 1989). "Each understanding is taken back to all previous understandings, and moves forward to all new understandings. No one understanding stays static or fixed" (Smythe, 1998, p. 94). Therefore, this interpretation is a call to the reader to engage in the hermeneutic circle. It is a call for the reader to commence their own dialogue with this interpretation as well as the women's narratives as they appear; it is a call to re-interpret the words held here, in accordance with the time and place in which you, the reader, are situated.

4.6 Conclusion

Chapter four has explored the philosophical underpinnings of this study, which is situated within the tradition of lifeworld research. It began by reviewing the development of Western philosophical thought, then explored the philosophical traditions of phenomenology and hermeneutics. The writings of Husserl, Heidegger, Gadamer, and Merleau-Ponty were examined in greater depth, as it is their work that provides the foundation for the philosophical framework of this research; their work has provided understandings of the lifeworld, and how one

attends to phenomena with the openness needed to achieve understanding. They have acted as philosophical dialogical partners in this journey of uncovering the experience of what it means to be a breastfeeding woman in New Zealand.

In this study, the phenomenon of breastfeeding is examined through a stance of openness, which allows the otherness of the phenomenon to emerge. Examination of the historical context in which a phenomenon occurs is critical to achieving an understanding of it (Gadamer, 1989); therefore, the historical context of breastfeeding in New Zealand was explored in Chapter Two. Finally, the hermeneutic circle provides the context in which understandings and meaning emerge. This philosophical position provides the foundation for this research, and has informed the method by which this study has been undertaken. This method, or 'frame', is discussed in detail in Chapter Five.

Chapter Five

The Research Journey

5.1 Introduction

This chapter details the process undertaken for this research, beginning with an exploration of the principles that have guided the development of the research method. The ethical considerations of this research and the process of recruiting participants are also discussed, as is the process by which the interviews have been analysed. Finally, this chapter concludes with a discussion of the trustworthiness of this interpretive account exploring the phenomenon of breastfeeding.

In attending to the issues of method, I am aware of the tension that exists for the interpretive lifeworld researcher. Adherence to method is considered to be antithetical, as method can close down upon potential interpretations (Gadamer, 1989). Commitment to a method that requires strict adherence to a prescribed, rigid process prohibits the spontaneity required to fully examine and, thus, understand a phenomenon. A rigid process may prevent a researcher from utilising a particular lens or perspective that does not adhere to the theoretical underpinnings of said method. Therefore, nurse researchers who have undertaken lifeworld research have suggested employing a narrative approach to addressing the issue of the process used for the research (Ironside, 1997; Smythe, 1998). This narrative account begins by exploring the philosophical underpinnings that have guided the development of the interpretive process.

5.2 Guiding Principles for the Development of the Research Process

In *Truth and Method*, Gadamer (1989) offers a strong critique of method-based research, claiming that rigid adherence to a particular method closes down upon the emergence of a full understanding of phenomena. Instead, he suggests that

the development of a research method be based on the following principles: the need to attend to the phenomenon with openness, the importance of being aware of how one's historical effected consciousness impacts upon the interpretations that are made when examining a phenomenon, and the need to search for the otherness of the phenomenon under study, meaning that attention must be paid to the understandings that emerge as distinct from any that were previously held (Dahlberg et al., 2002; Gadamer, 1989; Heidegger, 1962; Merleau-Ponty, 1981). These principles do not operate independently, but rather act in a circular, inter-related manner, each informing an understanding of the others. It is these principles that inform the process undertaken for this research project.

A concept that has also guided this research is the need to restrain the use of theory in the initial phase of analysis (Dahlberg et al., 2001; Gadamer, 1989). Adherence to this concept assists in limiting the effect that previous theories might have on subsequent interpretations.

In responding to these four principles I have been vigilant in my attempts to be open to the phenomenon of breastfeeding as it has emerged through the participants' narratives. To achieve this, I have adopted a stance of questioning – critically assessing my pre-understandings and making these explicit (see Chapter One, Section 1.5), and acknowledging the role that history has played in my understandings of breastfeeding as a New Zealander (see Chapter Two). I have immersed myself in the narratives to allow the meaning within the text to emerge through the voice of my own pre-understandings, constantly returning to the text in an attempt to be owned by the text, and in doing so allowing the text to speak. I have also engaged in a continuous process of questioning any interpretations that have emerged. Openness cannot be forced; it involves a process of being quiet and allowing interpretations to come, no matter how long it might take.

Attending to one's historical effected consciousness is a difficult task. It involves mindful examination of one's beliefs, values, culture and history.

Gadamer (1989) discusses how awareness of historical effected consciousness is the first step in understanding its impact on one's horizon of understanding. Throughout this research journey I have kept a journal in which I have written my reflections, my analysis of which allows me to make my pre-understandings and personal horizon increasingly transparent. The important prejudices that

emerged, and that I had to lay aside to be open to the phenomenon of breastfeeding, are outlined in Chapter One. It must be acknowledged that it is impossible to remove oneself from the world of history, language, and culture in which they exist; however, awareness of these entities is the first step in uncovering that which might prevent me from fully understanding the phenomenon of breastfeeding.

Attending to the otherness of the phenomenon of breastfeeding required me to constantly examine what pre-understandings were interrupting the emergence of new understandings. I had to be constantly mindful of what was known about breastfeeding, and search for those aspects that had previously remained unexplored. This seems somewhat paradoxical in light of the need to limit the use of theory when seeking to understand a phenomenon. However, this principle encourages the researcher to attend to what is not known about a phenomenon, or equally, turn to what is known and examine it afresh, which may lead to new understandings.

The principle that required me to restrain the use of theory during the active researching phase was an extremely difficult one to adhere to. The initial phase of a research project requires a justification of the project itself through identification of gaps in the published literature, which may have influenced the way I understood the phenomenon of women's experience of breastfeeding. However, once the research proposal was completed, I made a conscious decision to step away from the literature and focus instead on the narratives as offered by the participants. The literature was examined in depth only after the interpretations had been completed.

As highlighted in the previous paragraphs, these four tenets have guided the development of the method and also the examination of the phenomenon of breastfeeding. Being vigilant to the development of a process that would incorporate these four principles as an integral part of the research design was of utmost importance to me as a lifeworld researcher. The following section examines the development of this research process.

5.3 Attending to Ethical Considerations

Attention to ethical management of this study was an integral aspect of the research process. Ethics approval was acquired through Massey University Human Ethics Committee, and the Royal New Zealand Plunket Society Ethics Committee.

All the women who participated in this study were self-selected, and were made aware of the study through advertisements placed in Auckland Plunket Area Clinics. Plunket clinics were chosen for this purpose because The Royal New Zealand Plunket Society provides approximately 93 percent of well child health services within New Zealand. Plunket Nurses were notified of the study through informal presentations given in various Auckland Region Plunket offices; however, the Plunket Nurses themselves were not required to promote the research. Women who were interested in participating in the study contacted the researcher directly.

Each participant received both an information sheet (Appendix I) and a consent form (Appendix II). The information sheet discussed the aim of the research and provided information about the study. It also explained that I was a child health nurse, but did not state the name of my employer. The consent form addressed voluntary participation, the right to confidentiality, any potential harm that might result from being involved in the research, and information about how the results would be made available to the participants. Completed consent forms were collected from each participant at the start of their interview, and stored separately from the transcribed narratives. Each participant nominated the time and venue for a single individual interview, with all participants in the study electing to be interviewed in their home.

The narratives were recorded, and tapes, transcripts and other computer-generated data were stored in either secured files or under lock and key.

Access to these files was limited to the researcher and the research supervisors. A typist transcribed tape-recorded narratives and signed a confidentiality agreement (Appendix III). Once the narratives were transcribed, all identifying data was removed, interview numbers were assigned, and each paragraph was numbered. The chapters that explore the interpretations contain

portions of narratives, which are identifiable only through a pseudonym, interview number, and paragraph number (e.g., Amy. Int. 1. Para. 45).

The transcripts were returned to each participant for validation of the narrative. Each participant was free to withdraw from the study at any time until she returned the transcript, or verbally acknowledged its validity, and verbally agreed that her transcript could be included in the analysis. All participants gave either their verbal or written consent after reading their transcript. The ongoing mandate that the participants' confidentiality was to be maintained continues to be an important part of the research process.

It was important that the participants knew that I was a registered nurse, but that my role as a researcher would preclude me from offering a professional service; the request to offer health care advice did not arise during the process of gathering the participants' narratives. Two participants became distressed during their interviews. In both cases, they were asked at that point if they wanted to discontinue the interview, and both elected to continue. At the close of the interview I offered them the name of a family counsellor to provide them with support as needed. Neither participant wished to discuss their experience with another person. Instead, they responded that telling their stories was cathartic for them, and had provided them with the opportunity to speak about their experience. One other participant, as a result of the interview, decided to contact a medical practitioner to discuss postnatal depression, as she recognised the need to seek help to resolve her breastfeeding experience.

5.4 The Research Participants

The nineteen women who participated in this study were all born in New Zealand, with ages ranging from mid-twenties to late thirties, and had experienced the phenomenon of breastfeeding an infant. They were predominantly of European descent, with one participant identifying herself as Maori, and another as Samoan. All of the participants' specific cultural values and experiences were explored with the participants.

Ten of the participants had given birth once and subsequently breastfed their child. Seven had given birth to, and breastfed, two children each. One woman

had three children whose ages ranged from five months to three years, while another woman had four children aged two to seven years. Both of these women had breastfed all their children.

Nine of the participants were not breastfeeding at the time of the interview, but had breastfed within the last three to 32 months. Of those women who had weaned their children at the time of the interviews, the length of time spent breastfeeding had been between three weeks and 22 months. With one exception, all of the participants had partners.

Many of the participants were on maternity leave; however, there was a range of occupations that the women had either held before giving birth, or were actively engaged in at the time of the interviews. These included dental surgeon, child-care worker, fast-food worker, registered nurse, office manager, bar manager, early-childhood teacher, home manager, town planner, actress, business manager, radiographer, laboratory technician, and insurance manager.

The participants' educational backgrounds were not formally elicited during the interviews, although some women did discuss their education. For example, one was in the final year of an undergraduate degree programme, four had completed undergraduate degrees, and three reported having Master degrees.

5.5 Gathering Participants' Narratives

As noted above, the interviews all took place in the women's homes and were recorded on audiotape. The average interview lasted approximately 90 minutes, with the longest taking three hours. Each tape was listened to within two hours of completion of the interview, which assisted me to crystallise each individual interview encounter in my mind. This also allowed me to pay attention to the embodied nature of the interview, including such things as emotional reactions and non-verbal responses that had transpired during the interview.

The interviews took the form of a conversation between the women and myself. I did not go into the interview with a list of structured questions, but typically

started the interview by asking the women to speak about their first memory of breastfeeding; in most cases their stories of breastfeeding flowed easily from that point. I would ask questions if an aspect of what was being said needed clarification; however, the interviews focused on that which emerged as significant about the breastfeeding experience for each individual woman.

During the interviews I had an overwhelming sense that the women were gifting me a great treasure. These women, who did not know me personally, opened up their lives and horizons; they blessed me by sharing the intimate experiences that they had had with their children. Throughout the process of 'doing' the research I have been constantly mindful of the value of the women's narratives; this knowledge has made me strive to represent the experience of breastfeeding with clarity and trustworthiness.

5.5.1 Continuing the Conversation

As the interviews progressed, insights into the phenomenon of breastfeeding that I gained through conversations with the participants were brought to subsequent interviews; as a result, each interview became a continuing conversation between the women that participated in the study. The conversation that was carried into the subsequent interviews provided the opportunity to explore issues in greater depth, leading to enhanced understandings of various parts of the whole breastfeeding experience. For example, one participant talked about breastfeeding being an experience of tiredness. During subsequent interviews with other participants, I would ask if they too had found breastfeeding to be tiring. Their responses would frequently build on and develop the picture of what tiredness was within the context of the experience of breastfeeding.

5.5.2 Transcribing the Participants' Narratives

Following a review of each completed interview, the audiotapes were given to the transcriber. The validity of the written transcription was then checked by comparing it to the interview tapes. At that time I corrected any discrepancies between the interview tapes and the transcripts. After corrections had been made, the transcripts were returned to the participants for verification of

accuracy. At that time, the women could make any changes or corrections of their own, after which they returned the transcripts. Three women returned their transcripts, each with only grammatical changes. After two weeks I contacted the women who had not returned their transcripts and requested verbal consent for the transcripts to be included in the study. All the women who had been interviewed agreed to let their interview be a part of this research.

5.5.3 Deciding to Stop Gathering Narratives

The decision to stop gathering narratives was difficult. It is impossible to provide an interpretation that would represent the whole population of breastfeeding women, nor is this the goal of any qualitative research study. The lifeworld research methodology embraces the belief that people exist in their individual history, culture and world; therefore, the researcher only seeks to increase understanding about the phenomenon under investigation.

At the time the interviews were completed, the amount, and depth, of the information contained in the narratives was overwhelming, with each story shared resulting in pages of script. There was also the growing awareness within me of a desire to do 'justice' to the stories gifted to me by these women. It was important to not neglect the 'part' offered by the individual whose narrative might potentially become subsumed by the 'whole'.

Although no interpretation that is brought forth will ever be complete, this study offers a place to begin to understand the experience of breastfeeding. Smythe (1998) discussed experiencing the tension of not knowing whether she had gathered enough data for her doctoral thesis. She wrote: "I knew it was enough, while at the same time I knew it never would be enough" (p. 115). I acknowledge that this tension was clearly evident when I came to completion of the process of collecting data.

5.6 Data Analysis

It is difficult to articulate the process of data analysis – it is a difficult and timeconsuming process of calling forth interpretations. The interpretations that emerged through undertaking this study came through a process of thematic analysis, constant questioning, and engaging in the dialogical conversation between the philosophy, the participants, the narratives and myself as a researcher. Finally, the process became one of writing and re-writing interpretations. The steps are outlined in greater detail in the following sections.

5.6.1 Initial Thematic Analysis

Repeatedly reading the narratives led to the content of the transcripts being organised into 85 distinct themes, based on the language used by the participants to describe their experiences. The development of these themes allowed me to explore specific parts of the participants' narratives, and their inter-relatedness allowed the themes to be organised into three groupings. At this point I returned to each narrative and re-immersed myself in the whole, constantly mindful of both the parts and the whole of the phenomenon under study.

5.6.2 Constant Dialogue Between Participants and Philosophy

The process of lifeworld research is grounded in the need for a dialogical relationship between the participants, the narratives, the philosophy and the researcher. From the project's conception I have embraced the dialogical process; it has provided the path by which I gained understanding of lifeworld research as well as the phenomenon of breastfeeding. The journey began with a dialogue with the philosophy, and continued through conversations with colleagues and research supervisors. It then proceeded to become a dialogue with the women who gifted their understandings of breastfeeding in the form of narratives. Finally, it became a conversation with the literature on breastfeeding, and thus new understandings and meanings about what it is to be a woman who breastfeeds emerged.

The intimate dialogue between myself, the philosophy, and the participants' narratives constituted the interpretive phase of the journey; it was, at times, a difficult path to navigate that required isolation and retreat from others. Interpretation requires that the researcher maintain the dialogue through constant questioning of the narratives, philosophy, and any emergent

interpretations. It also involved a great deal of time spent reading, reflecting, journaling and thinking. During this time I was embedded in the process of calling forth hidden meanings, attending to the otherness of the phenomenon, whilst maintaining a dialogue that sought to clarify and identify my own pre-understandings and horizon.

The interpretations laid out in subsequent chapters have emerged from this process of constant questioning, and examination of the phenomenon of breastfeeding. They are my interpretations. The journey now continues with you the reader. As you read my interpretations you will begin your own dialogue, you may engage in a process of questioning my interpretations, and through this process new interpretations will emerge – they will be your interpretations. Through this dialogue new understandings will emerge and horizons will be shaped.

5.6.3 Writing and Re-writing

Finally, this project has evolved through the writing and re-writing of interpretations. The writing began after I had devoted considerable time to the process of dialogical questioning. The act of writing once again triggered the dialogical process, which in turn assisted the process of refining, clarifying and honing in on the phenomenon of breastfeeding.

5.6.4 Sharing Interpretations

As the interpretations neared completion I began to seek opportunities to share the results with others. The sharing occurred through returning to one of the participants, presentations to a number of community and academic groups, the research supervisors, and discussions with other health professionals.

Returning interpretive accounts to participants has been a contentious issue amongst qualitative researchers. Some researchers claim that returning to participants should be an integral aspect of the research process (Bergum, 1991; Dahlberg et al., 2001; Krasner, 1997; Leipert, 1996; Meighan, Davis, Thomas, & Droppleman, 1999; Smythe, 1998; Walters, 1995). This solidifies the significance of the participants' contributions and verifies that any

interpretations are reflective of the actual experience under investigation. Other researchers claim that any interpretations made are the researcher's alone (Geanellos, 1998; Ironside, 1997; Reinharz, 1983). The process of interpretation is strongly influenced by the particular world view held by the researcher and, therefore, any other person's interpretations would differ significantly.

For the purposes of this research one participant demonstrated a strong willingness to read the interpretations, and she did so. Because I had not experienced the phenomenon of breastfeeding until the project was nearing its completion, it was important that at least one of the participants had the opportunity to respond to the interpretations. The research was also presented to a community group who had an interest in breastfeeding.

I am privileged to have two research supervisors who are both midwives, and have overseen the project from its conception. They have engaged in the dialogical process with me, calling for clarification and noticing important aspects that I had left unsaid.

Sharing the research interpretations with health professionals occurred in both formal and informal settings. I have had the opportunity to present the research to postgraduate nursing students on numerous occasions. Members of the faculty at the University where I am employed have informally discussed the research project with me, and I have also made a formal presentation at another New Zealand University. These audiences were made up of both registered nurses and midwives.

A group of fellow phenomenologists, who are either midwives or nurses, have also assisted my journey into understanding the phenomenon of breastfeeding. They have asked important questions and provided valuable feedback throughout the journey, and in doing so, have assisted in cultivating my understandings of lifeworld research and breastfeeding.

5.7 Attending to Accuracy of Interpretations

Walters (1995) believes that any interpretation made by a lifeworld researcher is tentative. She suggests that the researcher must provide enough information to ensure that the processes undertaken to arrive at interpretations are evident. This is achieved through illustrating interpretations by the inclusion of interview narrative. Koch (1994) suggests that rigour is evident in the research account when a clear audit trail is evident. Such audit trails allow for trustworthiness to be established.

Dahlberg, Drew and Nystrom (2001, p. 203) suggest that the accuracy of any lifeworld account needs to be assessed through the reader engaging in a dialogical process with the research. They recommend that a series of five questions should be asked of the research. The first involves questioning the use of data and the process by which the data has been gathered. Next, the reader needs to ask what questions have been asked to uncover the phenomenon. Has the researcher practised openness? How has the relevance of the interpretations been verified? Finally, the reader must ask "what influence has the researcher's pre-understandings had on the interpretation?"; which includes examining whether the researcher has sought to expose pre-understandings through critical reflections.

Throughout this chapter I have sought to demonstrate how I have answered these questions. All the interpretations offered in this research project are tentative. The interpretations presented are mine; they are influenced by my pre-understandings, and grounded in my world as a woman, a breastfeeding mother, and a health professional, despite my diligent attempts to be open to the phenomenon as it emerges. I continue to be overwhelmed with the knowledge that there is no end to the process of understanding or interpretation.

5.8 Summary

This chapter has examined the 'happenings' of this research project. It has highlighted the tensions that exist for a lifeworld researcher, as it is critical that the process allows the phenomenon under study to emerge in all its otherness. Therefore, the philosophical principles that guided the choice of method were

examined. The process of the research and the manner in which any interpretations emerged were also discussed. Finally, the way in which I have approached the issues of trustworthiness and accuracy of the interpretations has been addressed. The following chapters offer the interpretations that have emerged from attending to the phenomenon of breastfeeding.

Chapter Six

Breastfeeding Women's Experience of 'The They'

6.1 Introduction

A unique aspect of women's experience with breastfeeding is coming to an understanding of what it is to exist in a world that contains culturally-mediated interpretations of lifeworld experiences. This chapter examines the 'world' in which women come to know what it is to be a breastfeeding woman. It explores how women come to understand the breast, and how these understandings are mediated by 'the they' or the public sphere. It is through 'the they' that women's interpretations about breastfeeding are formed, and thus women measure their success according to the standards of 'the they's' view of what it is to be 'average'. It will be established that these notions of averageness hold women to ransom as they attempt to conform to the contrived, ideal breastfeeding mother. This chapter begins with an exploration of how women come to know about breastfeeding, highlighting the cultural frameworks by which women interpret their breastfeeding experiences.

6.2 Coming to Understand Breasts and Breastfeeding

6.2.1 The 'Always-alreadyness' of Breastfeeding

We have all been 'thrown' into a world that we share with others (Benner, 1994; Dreyfus, 1991; Heidegger, 1962). We come to understand and gather meaning through 'being-with-others'; thus, *dasein*¹⁹ is revealed through the world shared 'with-others' (Inwood, 1999). People agree to act and judge in particular ways,

¹⁹ Heidegger used the term *dasein* to describe a being that exists and is capable of understanding or to describe the being there of existence (Dahlberg et al., 2001; Heidegger, 1962; Inwood, 1999). Refer to Chapter Four, section 4.4.1 for additional information.

and these agreed ways of behaving form public sources of meaning (Dreyfus, 1991; Heidegger, 1962; Olafson, 1987). An individual's understandings stem from previously-formed collective, public sources of meaning, which are taken up through 'being-with-others'. According to Dreyfus (1991, p. 144), by the time an individual has "dasein in them, they are 'always already' socialised". Therefore, to gain an understanding of the meaning that breastfeeding holds for women, it is important to examine the world of 'being-with-others'.

The narratives used in this study include many examples of women's experience of 'always already' being socialised into a world that contains previously-formed collective sources of meaning. The following excerpts grant insight into how society forms public sources of meaning in relation to breastfeeding.

When an infant is young, he or she learns that events are meaningful. Meanings are taken up through being-in-the-world and being-with-others, a process dictated by the world in which we are thrown. One could describe meaning as being 'always-already' present. If the world in which *dasein* is 'thrown' is a world in which 'other' supports breastfeeding through the formation of agreed ways of acting, *dasein* will understand breastfeeding as a part of that world. This is supported by research, which has demonstrated that those who have been breastfed are more likely to breastfeed than those who were not (Isabella & Isabella, 1994; Kessler et al., 1995; Scott et al., 2001). Essentially, *dasein* will engage and act in accordance with its thrownness, and submerge its beliefs in consensus with the 'others'. For example, a woman will embrace the cultural ideologies of the group of people with whom she dwells.

I guess it's 28 years of being told, or knowing that breastfeeding is better for the baby than bottle-feeding. If you were brainwashed for the next 28 years that bottle-feeding is great and better for the baby, then that's what you'd believe.

(Cathy. Int. 18. Para. 69).

I hadn't done a lot of reading about breastfeeding, but you do pick up the knowledge that it's best for your baby and that as a teenager he's likely to be more intelligent because of having been breastfed. That kind of thing seeps into knowledge, without necessarily making a conscious effort to find it out.

(Sam. Int. 12. Para. 40).

Alice painted an apt description of the young in New Zealand as they are 'always already' being socialised. As a child, one learns to take up meaning from an event or situation through being-with-others; children, therefore, learn to act and judge by observing others as they engage with the 'world', and thus modify their way-of-being accordingly.

One of the other women in the Plunket group has had another child and it's so funny and sweet to see her little girl pretending to breastfeed her little doll and also to bottle feed it because the mother is doing both. Little girls pick up what they're supposed to do when they grow up, if they see other women breastfeeding it's - 'yes, that's what you do', if they see other women bottle feeding they think 'that's what you do'. (Alice. Int. 5. Para. 130).

The 'always-already' taking up of meaning results from being thrown into a world, which provides us with a backdrop for understanding phenomena; however, these understandings are mediated by the actions of others.

6.2.2 Breastfeeding Women 'With-Other'

Understandings are not produced in a vacuum; instead they are formed as one interacts with an entity, or 'with-other' (Heidegger, 1962). It is in this interaction that interpretations are made and understandings are formed. This sharing of understanding takes place through media such as books, and languaging 'with-other.' Abby spoke of how she sought out common understandings as a way of knowing about breastfeeding, by engaging in languaging 'with-other' through written and spoken word.

When I found out I was pregnant I read as much as I could so I was aware. It doesn't mean just because you've read about it that you're going to know when it happens. It's just easier. I talked to ladies at work and they said 'I've heard about this book and that book'. I also got information from antenatal classes and the antenatal clinic and also from my doctor. Whatever I found I read.

(Abby. Int. 2. Para. 16-29).

As the women in this study engaged 'with-other,' understandings about the importance of breastfeeding emerged – a commonly-cited aspect being the benefit of the immunological properties of breastmilk for the infant. Amy offered a picture of how new mothers come to learn about breastfeeding, demonstrating how understandings emerge; they blossom as one dwells with a phenomenon.

It's a combination of the nutrition, providing him with the best possible nutrition, and also the immunological aspects of it have been really important to me. By breastfeeding him I knew that I'm giving him really good protection against bugs and infection and on top of that is the building of the relationship. The building of the relationship isn't something I thought about until later on. I looked back at it retrospectively and I thought the breastfeeding relationship has built our relationship whereas initially to me it was always the nutrition and the immunological factors that were important in breastfeeding him. (Amy. Int. 1. Para. 50).

Common understandings about breastfeeding have developed over generations. In New Zealand, breastfeeding has come to be primarily understood as the optimal food source for infants²⁰, sometimes to the exclusion of other understandings, such as breastfeeding facilitating the relationship between a woman and her infant. As the boundaries of what is known are explored and new meanings evolve, women's acceptance of the previously-established understandings are unpacked, and further developed in light of the 'newly-formed' meanings.

Kate told the following story of how generational understandings of breastfeeding have evolved. These evolved understandings provided the illumination needed to more fully interpret the experiences that Kate's mother and grandmother had in relation to breastfeeding, and also provided Kate with guidance and the courage to persevere with feeding her child.

My grandmother, I was speaking to her this morning, she's 82 and she's going 'Are you keeping up with him? Have you got enough milk for him?' My girlfriend who read a bit of that La Leche literature was saying

²⁰ See Chapter Two for more information about how breastfeeding is understood in New Zealand society.

to me that you won't ever run out and I can't imagine that nature would let me run out. I do believe that the quality, I know from expressing, that my milk is totally different at different times of day. The look of it is totally different. So the generations from my 82 year old grandmother, who has stories about an old lady taking my uncle, because he was a colicky baby, and feeding him milk and sugar and when he came back he was fine. God it makes everyone's toes curl. To my mum, who when I was pregnant and I heard all these negative things and all this pressure and mum just said 'I never had any problems breastfeeding', she got mastitis, which, I haven't had. She just said she so enjoyed the experience. She said breastfeeding was really special and she was very positive about the whole thing so I did tend to sort of think well, I can't see why I should have any problems, I suppose was my approach to it. (Kate. Int. 3. Para. 52).

'Being-with-others' means that common understandings are generated in relation to what it is to be a mother, an aspect of which is uncovered through sharing the reproduction of life 'with-others'. Eva relays how the meaning of what it is to be a mother came from dwelling 'with-other'.

I do feel quite proud that I have given birth and have a little baby. So I walk around and feel really happy about that especially when I see other women with prams. It's a really lovely feeling like we've all done something quite similar to each other, and all wanted to obviously bring a new life into the world and share that life with others and create a new little being for society.

(Eva. Int. 19. Para. 111).

A Samoan woman, Mai, who participated in this study revealed how the 'other' in her 'world' led to a shared understanding that breastfeeding comes naturally for women of her ethnicity. This realisation came to her, and gave meaning, through 'being-with' European cultures; thus, some taken-for-granted meanings emerged through contrast 'with-other.'

I'm not criticizing European cultures, it's not a downfall for them or anything, but for Pacific Island mothers' breastfeeding comes naturally (laugh). After a few days we're, you know... [able to breastfeed]. (Mai. Int. 4. Para. 72). In the following narratives, participants Sue and Eva reveal the meanings about breastfeeding that they had gathered through 'being-with' health care professionals. These meanings highlight that breastfeeding is the best nutrition for a baby, that all women can breastfeed, and that all women must breastfeed. Eva and Sue interpreted these as pressure to breastfeed, which came at a time when their experience of breastfeeding was at odds with what 'the they' espoused breastfeeding to be.

I assumed that breastfeeding was a natural thing and that it would happen, and if you choose formula... it was a choice, for whatever reason they chose not to breastfeed. You're told about the benefits of breastfeeding all the time i.e.: less chance of SIDS, the perfect food results in a good digestive system, complete food, formula equals poison, increased immunity with breastmilk was also a biggie along with the allergy thing. It never occurred to me that it was a difficult thing, even with the antenatal classes that wasn't really suggested. So doing the right thing was breastfeeding your child. And like I said I assumed it would just be easy and it wasn't.

(Sue. Int. 6. Para. 12).

There's two sides to it, one is the head part where you've been told constantly all through your pregnancy, at antenatal classes and by your midwives, that you must breastfeed, and that everybody can breastfeed. [There's] no reason that you can't breastfeed, breastfeeding is best for the baby and that's it that's what you're going to do. You've got this pressure to do that. The second thing is the heart thing, which is probably more powerful. That you so badly want to be able to feed your own baby, you have these breasts that are full of milk, and ready to feed and you're the only person that can do it for her in the whole world; and that's what you're here for, like your sole purpose. It's the one special thing - other people can look after your baby, they can change her nappy and give her bottles and cuddle her and sing her songs but no one else can breastfeed.

(Eva. Int. 19. Para. 9).

It is interesting to note that the messages received from the health care professionals paled in comparison to the internal pressure that Eva placed upon herself to breastfeed. Eva's own 'thrownness' led her to the interpretation that

her sole purpose was to feed her infant. She believed she was the only one who could nourish her daughter, with her breastmilk. This is reflective of a cultural belief amongst New Zealander's of European descent, which can be traced back to Victorian England, that a woman should take the responsibility of breastfeeding her own children (Kedgley, 1996; Yalom, 1997). If a woman did not assume her breastfeeding responsibility, she was considered to be selfish and unfit to mother (Yalom, 1997).

As highlighted above, 'being-with' others allows interpretations about the role of women's breasts to emerge and become culturally established, which includes the role of women whilst breastfeeding. The following section highlights another culturally-established rendering – that a woman's breasts are integrally sexual.

6.2.3 Understanding the Breast as Functional

An integral part of understanding the 'world' into which a woman of New Zealand descent is 'thrown' is the awareness that the breast is interpreted as sexual. The perception of the breast as sexual has a long history in Westem culture (Maushart, 1999; Morse, 1989; Yalom, 1997), which has silenced the understanding that breasts can exist as lactating breasts; thus, this knowledge has been confined to a private, unknown, unrealised world. In the 'thrownness' of mothering the women who participated in this study experienced a transition that enabled them to embrace the concept of breasts as a functional part of the body. The following excerpts highlight how women initially came to know their breasts as sexual, a perception that was altered by their experience of breastfeeding.

We're mammals, that's why we've got breasts. We've turned them into these sexual icons and, to be fair, when you're dating of course that's all they are. You don't think of them as being anything vaguely useful other than being attractive on the female.

(Kate. Int. 3. Para. 27).

I was aware that breasts were used for breastfeeding but because I hadn't had any personal experience of that I hadn't really made the transition in my mind. I hadn't thought about it - breasts are everywhere, if you look in all the magazines they're there and they're usually half

uncovered, and they're always used for glamour or to turn men and women on or to look sexy.

(Eva. Int. 19. Para. 125).

Another culturally-mediated interpretation of the breast includes acceptance of the breast as functional, as capable of providing nutrition for an infant. For women in New Zealand, the socialisation process leads to the acceptance of general understandings of breastfeeding, and there appear to be definite boundaries on a breast as functional. One such boundary dictates that there is an acceptable length of time for the breast to be functional. Sally spoke of the general understanding that a woman who breastfeeds for a significant length of time must subscribe to fringe Bohemian ideals, a view supported by other participants.

I don't know where I've picked this up from - it's just been a general understanding that women that breastfeed for a long period of time tend to be supposedly your free-age, new-age hippies.

(Sally. Int. 11. Para. 78).

'The-other' let women know their disapproval of unacceptable behaviours, such as prolonged breastfeeding, in covert ways. In the following example, Amy was made aware of 'the-others' disapproval through critical facial expressions being directed towards her. Thus, Amy gained the understanding that feeding an infant of such age was not an agreed-upon way of behaving 'with-other.'

We were in Christchurch at the airport waiting for a plane. He needed a feed and there's nowhere really private to go. I don't have a problem with feeding in public so I sat down to feed him in public, in a seat in the airport and all these people were walking past all the time and all they could do was look. Their expressions, at least my impression of what they were looking at, was 'what is that really big child doing breastfeeding at that age'. He wasn't one year then, he was about 11 1/2 months - but that was the way that I felt about it. I felt exposed, uncomfortable. That was the only time I ever felt uncomfortable [breastfeeding]. It was kind of strange, a societal pressure. (Amy. Int. 1. Para. 10).

Dasein is 'thrown' into a 'world' in which meanings exist, which are 'alwaysalready' present and taken-up through 'being-with-others'. It is these taken-up meanings that form the basis for interpreting experience; the commonly-held meanings and interpretations that represent cultural standards are owned by 'the they'.

6.3 'The They'

Women understand breastfeeding in accordance with the understandings that are created through being 'with-others'. Heidegger (1962) claims that the authors of commonly-held understandings are 'the other' or 'the they.' I am in the 'world' with-others and these others are also with-others (Inwood, 1999).

In this inconspicuousness and unascertainability, the real dictatorship of the 'they' is unfolded. We take pleasure and enjoy ourselves as *they* [or man] take pleasure; we read, see, and judge about literature and art as *they* see and judge; likewise we shrink back from the 'great mass' as *they* shrink back; we find 'shocking' what *they* find shocking. The 'they', which is nothing definite, and which all are, though not as the sum, prescribes the kind of Being of everydayness (Heidegger, 1962, p. 127).

Thus, it is 'the they' that provides the framework for interpretation, by establishing boundaries for the acceptable ways of existing. To conform to these boundaries, one constantly engages in attempts to mirror 'the they', doing no better or no worse than 'the they' (Inwood, 1999).

One of the primary influences of 'the they' can be found in one of 'the-they's' existential aspects – that of the call to averageness (Heidegger, 1962), which dictates what is success or failure, what is possible or impossible. As a direct result of this dictatorship of 'the they', or this call to averageness, a process of "leveling-down' of all possibilities of Being" occurs (Heidegger, 1962, p.128). The ability of *dasein* to achieve possibilities is closed down upon in the call to averageness; thus, 'leveling-down' prevents the emergence of modes of being that are distinct from the sphere of 'the they.'

As a result of engagement with 'the they', "dasein's everyday possibilities of being are for the others [the they] to dispose of as they please" (Heidegger, 1962, p. 127). The possibilities of being are 'taken-over' through 'being-with', which occurs as one is submerged into 'being-with-the-they.' Dasein has become "entirely subsumed under this essentially anonymous public identity" (Olafson, 1998, p. 35). Thus dasein is freed from responsibility, as interpretive choices are given over to 'the they'; dasein's giving over of responsibility means that "it has not yet come to itself and can constantly be in this world without having to come to itself" (Inwood, 1999, p. 213). This process of 'giving over' reveals how dasein comes to be inauthentic, as responsibility for choices and decisions are essentially deferred to 'the they' (Inwood, 1997). However, one can exist in apparent conformity with the expectations of 'the they', but at the same time be authentic, provided one has thoughtfully chosen to express his or her individuality in that manner (Inwood, 1997; Olafson, 1998).

6.3.1 Mirroring and the Call to Averageness

The women in this study were constantly engaged in the act of mirroring; constantly striving to fulfil the requirements of the 'perfect mother,' as dictated by 'the they's' interpretation. Eva is highly educated with a successful career, having obtained a certain level of standing within her field of expertise; despite that, she still spoke of the image of the perfect mother and how this image held sway over her. She described how women strive to mirror the image of the perfect mother, noting that she was 'constantly disappointed' that she did not 'measure up'.

The perfect mother can do everything; she can breastfeed while holding down a full-time professional job, earning truck loads of money, can manage a household, drives the perfect car, has great skin every day, lovely shiny hair, and goes to the gym, eats the perfect diet, and does all this and can still be having a good sex life, and be a gourmet chef (laugh). It's just expectation that we build up in ourselves to do with breastfeeding, natural birth, perfect career and the perfect marriage. I don't know why we do it to ourselves because we're constantly disappointed.

(Eva. Int. 19. Para. 168).

The narratives offer many examples of how the drive to be the perfect mother was an aspect of the participants' experience as breastfeeding mothers. Failure to achieve this image meant they were failing not only themselves, but also their infants, family and society.

In the following story, Sam tells of how, in persevering with breastfeeding, she strove to mirror the 'other.' She relates how she gauged her success at breastfeeding in relation to 'other', and how the knowledge that she had achieved a certain length of time breastfeeding meant that she could wean and not feel like she failed 'the they' or her child. By engaging in this action of mirroring, Sam inadvertently closes down on the possibilities that her experience of breastfeeding might hold.

I remember thinking, and saying to my partner, that I didn't want to be the only one at our first antenatal get together to be bottle-feeding. I couldn't think of anything worse in the world than being in a group of a dozen other women and being the only one that couldn't breastfeed. It's a silly pressure that you put on yourself but to me it seems like that would be the worst thing. (Participant had tears in her eyes). But sometimes I wonder if it was really worth it. (Participant was crying). I know that it is in terms of the nutritional value and all of that kind of thing, but the stress that I put myself through, I don't know whether it was worth it really.

We're at ten weeks now; I don't plan to give up in the near future but if something happened and I did that would be fine. I have a good friend who breastfed her first baby for three months and is at the moment feeding her second baby and plans to feed her for three months. That makes me feel better about it as well that we're nearly at three months and if I can get to that then I've kind-of done as well as she did. I don't know why that makes me feel better, but it does.

(Sam. Int. 12. Para. 54, 56 & 80).

Pat describes how pressure to be the perfect breastfeeding mother causes women to shy away from situations that leave them open to criticism, if they perceive that they are not reflecting the perfect image. Her infant suffered from severe reflux, and she spoke of an experience with health care professionals in which she believed she was being scrutinized by 'the they.'

Mum didn't know what to do with me so she sent me off to the Family Centre but I didn't want to go there.... I guess in a way I felt a bit of a failure and I didn't want to expose myself to whatever criticism they were going to give me, not knowing what they were going to do. When I did get there [to the Family Centre] the little horror performed perfectly, (small laugh) so then it was back to 'what is it about me that wasn't right.' (Pat. Int. 7. Para. 62).

When being assessed, Pat's infant 'performed perfectly'; thus, she was left feeling insecure in her ability to mother her infant. She perceived that others were evaluating her ability to cope with parenting, which left her feeling exposed and inadequate. This resulted in her wishing that she could withdraw from those who may have been able to assist her with the difficulties of trying to breastfeed an infant with severe reflux.

Debbie provided a poignant example of a woman struggling to mirror 'the they's' ideal. She strove to breastfeed her son despite significant pain because of severely cracked nipples. Despite frequently expressing breastmilk and trying to persevere with the pain, she finally weaned her son when he was approximately eight weeks of age. Debbie's experience led her to resent her infant; she spoke of her belief that she had developed postnatal depression as a result of the physical pain caused by breastfeeding her son.

I didn't want to bottle-feed but I knew I would eventually have to do [so]. I couldn't carry on the way I was because I was depressed; it was just horrible. But because of this stigma about breastfeeding, everyone makes out that if you bottle-feed you're a bad parent that's why I struggled on for so long. We didn't bond until I started bottle-feeding him. It was the resentment of him hurting my body. (Debbie. Int. 15. Para. 48).

Despite her depression, she persisted with breastfeeding for eight weeks because she did not wish to be perceived as a bad parent, and confided that she continued to feel guilty for two months after she decided to wean her son.

Cathy discussed how the pressure to mirror the perfect breastfeeding mother was so great that, although she knew that adhering to the image was destructive, the need to conform still held sway. She believed that by not

breastfeeding she had failed as a mother, and strongly wished she could have the opportunity to breastfeed her next child without difficulty, and not have to endure the associated feelings of guilt. Cathy longed to flee the grasp of 'the they's' interpretations; however, she responded to the summoning to conform to the image of the perfect mother epitomized by 'the they', despite the pain and the negative emotional impact that her breastfeeding experience held for her.

If I had a third child I'd have an elective caesarian and put them [sic] straight on the bottle. That would be exactly what I would want to do because the first two goes have been awful. But why wouldn't I do that if I had a third? Why do I carry on doing it?.... I don't know. I don't know, because nobody actually says "you're a bad mother for bottle-feeding." I guess when I do see my other friends' feeding it makes it hard. But honestly with all my heart if I had another one I would want to feed them [sic], I would love it to work. I would love to have one baby, for it to be perfect, for her or him to latch on and for it to work. But to know if I had to go through this experience again... it would be awful. It would be awful.

(Cathy. Int. 18. Para. 45).

Cathy also provided a story of how her husband was influenced by the call to averageness, and the effect this had on her breastfeeding. Her husband struggled with the idea that she might deviate from the averageness of 'the they' and thus became the vehicle by which the 'voice' of 'the they' was carried. Her ability to make decisions about her breastfeeding was closed down upon in order that she fulfil the call to averageness.

One day we had a really good talk about it because I felt so hurt that he would put the baby above me, above his wife. He would rather me suffer in pain - I'd be biting on hammers, sticks, clothes, and balling my eyes out while she was latching on - and he would say "I think it's best for her for you to be feeding." And I would just be bawling my eyes out in pain and that really hurt that he would do that. He's happy that we've been able to carry on for a little bit longer so that she has had what they say is the best.

I don't know why I didn't give up because it's awful, it was absolutely awful and I should have because it's stressful enough what you have to

go through without having to be in pain feeding. It was stupid not to give it up because it was almost marriage-breakable material because I was so upset the whole time and just crying the whole time feeding. And causing stress between us because I was so down. And he's brought up thinking that breastfeeding is best for baby too. It was interesting because I said to him one day "I know why you don't want me to stop, it's because you don't want to tell your parents that I can't feed our children" and I think that, that was what it was. He was scared that they'd see me as a failure and I think that was what he was also more upset about. That pressure when his parents ask "how's the feeding going?" "Well you know she can't feed, the baby's on the bottle." (Cathy. Int. 18. Para. 49 & 54).

Jan provided another example of how the call to averageness provides barriers to experience. Jan was supportive of breastfeeding, and was involved in a voluntary organisation that assisted new mothers in the transition to motherhood; yet she offered the following narrative expressing a commonly-held meaning. Her sentiments provided an example of how 'the they' levels down and provides measures of averageness, and may offer an explanation of why a woman would discontinue breastfeeding prematurely, as she would hardly wish for her actions to be interpreted in this light.

I've heard lots of stories of people still breastfeeding children who are going to school. To each their own but I wouldn't like to see that in public. I think whatever you do in your own home is up to you but I don't think I would like to see a five year-old in public running up and opening your shirt and having a drink. I would just find that a little bit hard. I suppose because I consider five year-olds to be little people at that age. It's kind of like going to the side of the road and toileting at five. (Jan. Int. 13. Para. 74).

While Jan's narrative describes an extreme example that is not often seen in New Zealand, it is common for New Zealand mothers to feel pressure to discontinue breastfeeding a baby who is over one year of age. Kay was continuing to breastfeed her son, who was one year old at the time of the interview. She talked in-depth about the pressure she experienced from 'the they' to discontinue breastfeeding her son, describing two opposing views that hold sway over women's breastfeeding decisions: she spoke of the research

highlighting the importance of prolonged breastfeeding, and also spoke of how this knowledge is silenced in the face of commonly-formed meanings that dictate women's actions in relation to prolonging breastfeeding, especially if a woman is in the vulnerable position of being a first-time mother. Through such meanings the call to averageness is issued, and women's experiences are leveled down; their choices are closed down upon, which may lead them to wean their infant earlier than they wished.

Some women don't get a choice and it doesn't seem fair. I think women should get to choose but she needs help though, especially a first time mother who has no idea, like me. If for some reason research shows the longer you breastfeed the better, but people get pressured to stop earlier, I think that's wrong. I think it's the same with anything though, you don't get to choose. You should be able to choose what you want to do. You shouldn't be pressured. Like people I know that have been pressured to stop.... It's people in society. It's not the 'in' thing anymore. For the first few months everybody thinks you're doing a really great job but after that when they [the infant] get older they think you should stop. "They're old enough, they don't need it." That's wrong. The beginning and the end of it are the two things that annoy me most about it. Which is stupid really cause everything is fine in the middle. But you shouldn't be pressured.

(Kay. Int. 8. Para. 136-138).

Kay highlights the pressure in 'the beginning and the end' of a woman's breastfeeding experience. During the interview, she spoke of the societal demand to commence breastfeeding, even if breastfeeding had become a source of stress and trauma for the woman. If a woman chose not to breastfeed she became subject to the scrutiny of 'the they.' 'The they' have many voices, and women who could no longer breastfeed had no difficulty in heeding the message that they had committed a transgression by not complying with the call to averageness.

Jan had an incredible ordeal with breastfeeding her son, struggling through painful cracked nipples, mastitis, hospitalisation and surgery before choosing to wean her infant. She spoke of her experience with 'the they' after weaning her infant when he was approximately one month of age.

People just asked you all the time, if they saw you bring out a bottle this look came over their face like 'wow, you've decided' - our antenatal group was like that initially and Plunket group were like that, because I was the very first to bottle-feed. There was an assumption that it was my choice completely, which at the end of the day it probably was, but there was no understanding behind the reasons why we had done that. (Jan. Int. 13. Para. 24).

Eva provided an example of a woman who succeeded at breastfeeding despite the 'odds.' She struggled through tremendous pain, pain she likened to the pain of giving birthing, cracked nipples, five days in hospital because of mastitis, and still she managed to continue to breastfeed her daughter. She mindfully articulated how she could potentially contribute to the pressure 'the they' places on women to continue to breastfeed, but also celebrated the special place that breastfeeding has in her experience as a mother.

Now that I'm breastfeeding 'successfully' I suppose you might say, almost fully breastfeeding, and I've had so many problems and I have overcome them I'd like to think that everyone else would try as hard as I have. I've joined the club of people that says "anyone can do it" but I have to be careful because I could have easily given up and been on the other side of the fence. I don't want to put pressure on people.

Just looking at her growing and looking so healthy, she hasn't been sick at all, almost all the weight she's put on has been from me and it's such a satisfactory feeling. It's like the best feeling in the world. People will say how healthy and beautiful she looks and I think I'm actually growing her myself. She's thrived on breastmilk and those times of discomfort you don't even remember them really physically anymore. I'm not, you know, delighted with the pain but knowing that there's a higher purpose. (Eva. Int. 19. Para. 11 & 85).

The women who participated in this study attempted to mirror the publiclyespoused ideal breastfeeding mother, thereby responding to 'the they's' call to averageness. For those women who did not mirror the requisite image, breastfeeding was an experience of struggling in silence.

6.3.2 Breastfeeding as Struggling in Silence

The pressure to be silent, or act out something, was the hallmark of the women in this study who perceived that they were failing to mirror the image of the perfect breastfeeding mother. An aspect of this silence exhibited itself in the need to hide the difficulty they experienced with breastfeeding. Sam tells this story about how she is silent about her struggles with breastfeeding to all but those that may be having, or have had, a similar experience.

When I'm out I guess I feel the pressure more, and again it's probably a pressure I put on myself. Pressure to make it work. I don't want to have a screaming baby in public because he won't breastfeed properly. And that's never happened, but it's something I worry about. I'd probably feel like I'm failing in some way... There are some people that I'm happy to tell about my problems... but others I think do perceive that I've failed in some way if I tell them that things aren't going well. (Sam. Int. 12. Para. 39 & 66).

Eva also spoke of the pressure to remain silent about how difficult breastfeeding was, and provided these sentiments supporting Sam's experience of breastfeeding in public.

There's definitely pressure if you are out in public to make it look easy.

The image of the perfect breastfeeding thing and pressure to live up to it.

(Eva. Int. 19. Para. 39).

Women also concerned themselves with how their babies mirrored the image of the perfect infant, the achievement of which is directly attributable to a perfect mother. If their infant reflected a less than perfect image, women interpreted it as an indication of failure on their part. Sam gave birth to a healthy, eight-pound infant, who subsequently lost 11 percent of his birth weight within the first week post-birth. Sam struggled anxiously to establish breastfeeding, frequenting a local Family Centre to gain some reassurance that her breastmilk was sufficient to nurture her child.

Sam valiantly endeavoured to mirror the perfect mother who had the perfect infant. In her insecurity, she concernfully examined her infant and analysed 'the

they's' perceptions for signs or proof that she lacked the ability to be a 'good' mother, with enough breastmilk to adequately nurture her child.

People will say things like "Oh he doesn't look like an eight pound baby" or "He looks small for his age." I'm constantly comparing him with other babies of the same age, which I know you're not meant to do, but you just can't help yourself. I do, I always worry about his weight gain. I weigh him usually once a week to make sure it's OK and it has been OK. (Sam. Int. 12. Para. 36).

Mel, like Pat²¹ (Int. 7. Para.62), exerted significant effort to establish and maintain breastfeeding after she gave birth to an infant who suffered from severe reflux. During the first weeks after giving birth, she vigilantly visited her midwife, doctor, the local Plunket Nurse and Family Centre in an effort to establish breastfeeding. She finally weaned her daughter when she was six weeks of age. Mel reported that she agonised over the decision, but finally decided that she could no longer cope with the physical and emotional trauma of breastfeeding. She spoke in-depth of the difficulty of living with the failure to portray the image of the perfect mother with the perfect infant.

Failure because you haven't done it, and envy too because I was in a coffee group and only one of the other mothers had gone onto formula. The others were all breastfeeding and all their babies looked so good and I thought 'Oh god, why can't my daughter be feeding like that?' Helplessness too, 'why isn't it working for me? Pity - 'why isn't it working for me?' I can remember saying to my mother when I was holding her, 'I wonder if I can exchange her' like you exchange an item of clothing. That's dreadful, I didn't mean it. But it crossed my mind, 'these pair of shoes don't fit I wonder if I'll exchange them next week. This baby is not quite right I wonder if I can exchange it next week?' And I thought 'Oh my god, that's awful!' But it's what goes through your head; this baby's not fit to the mould of what it was meant to be so where's my one that is meant to be perfect? Where's my perfect baby that latches on and feeds perfectly and doesn't have reflux..

(Mel. Int. 17. Para. 18).

²¹ See section 6.3.1.

Emma's narrative illustrates how women, as breastfeeding mothers, give the responsibility of interpreting their individual experience over to 'the they.' Her story reflects a common sentiment among the women in this study, regarding the need to conform to the ideals espoused by 'the they.' Through accepting 'the they's' judgement of what constitutes a perfect breastfeeding mother, a woman may inadvertently experience guilt if she chooses to not conform to the esteemed ideals, even if she might be choosing the best option for her and her child. The ideals articulated by Emma show that what might be a good decision for a woman and her child can become a source of guilt, and possibly a sense of failure, if a woman adheres to 'the they's' judgement. Emma uses the term 'going against the grain' to describe a decision to not adhere to 'the they's' judgements – choosing to do the unexpected by not mirroring 'the they'. A woman who chooses not to heed the call to averageness will suffer the consequences; in this case, a woman will suffer from feelings of guilt if she does not breastfeed her infant.

I saw this ad on TV about women that leave their babies in daycare while they continue with their career. It said the day you give birth you give birth to guilt. You will always feel guilty if you make decisions that go against the 'grain'.

(Emma. Int. 14. Para. 234).

The following story, as told by Eva, reveals her perseverance, despite enormous discomfort from breastfeeding. Eva spoke of how she, at least in part, relinquished the responsibility to a health care practitioner for choosing whether to continue to breastfeed or wean her infant. This responsibility is not easily relinquished, but as Eva felt insecure about her ability to make the best decisions for her child, she gave over her interpretive choices believing 'the they' knew better.

Despite the mastitis, despite having cracked nipples that got infected, despite being in tears... It's really quite difficult. I feel quite sad for people who have tried but haven't been able to continue. But I also feel an admiration for women who totally know that they just don't want to breastfeed and they just go onto a bottle straight away. It's all about being comfortable in your decision. Maybe with the second or third child down the track it would be much easier to be clear in what you want. But with your first child it's all so new and different and you are so reliant

on the health people, because you don't have any experience of it yourself. So it's tricky and hard. (Eva. Int. 19. Para. 29).

It has been argued that New Zealand does not have a breastfeeding culture (Beasley & Trlin, 1998; Beasley & Heritage, 1998/1999; Watson, 2001); however, the participants' responses indicate that New Zealand society expects women to breastfeed, but women do not believe they have access to appropriate support so they can breastfeed their infants without difficulty.

Women will struggle in silence with breastfeeding if they perceive that they are failing to aspire to the ideal breastfeeding mother as espoused by 'the they'. There may come a time though when a woman will see through 'the they's' espoused image, but to do so she must concernfully uncover the 'they-self'.

6.3.3 Concernfully Uncovering the 'They-Self'

It comes down to the idyllic things through the media, magazines and TV where the perfect baby sleeps through the night and latches on instantly and smiles and goo-goos all day, hardly has any wet or dirty nappies. If you ask any other women they might say "yea that's the idea, but it's just not a reality." It doesn't happen. I don't know why. Like mother's looking glamorous and walking out of the hospital in size eight jeans after just giving birth, and having nice shiny hair, and make up on days after giving birth. It doesn't happen (laugh) to most of us anyway, not to me. I don't know why we do it to ourselves in society, why all these images go out when they're not truthful and they do put pressure on women to look a certain way

(Eva. Int. 19. Para. 87).

In some cases, the participants in this study attempted to understand and liberate themselves from the demands to submerge their created meanings in the public identity of 'the they.' Heidegger (1962, p. 317) suggests that it is in being concernful as we are 'with-others' that we understand 'the they.' This understanding develops out of listening to the call of one's conscience. Only as one engages in listening to this call is *dasein* summoned "from its lostness in 'the they" (Heidegger, 1962, p. 319).

Heidegger discusses how it is one's conscience that calls us into what is *ownmost* to us, describing this call as a "momentum of a push – of an abrupt arousal" that "reaches him who wants to be brought back" (Heidegger, 1962, p. 316). It is through this call that *dasein* is drawn from its hiding place, within 'the they', into what is most true, its *ownmost*. This section of narratives depicts women who have become concernful in their interaction 'with-others', enabling them to revise their interpretations of their experience as breastfeeding women, freeing themselves from the lostness of 'the they.'

In the following story, Kate describes how she heard the summoning of 'the they' but chose to flee the pressure to conform; instead, Kate navigated her own path that enabled her to see through the superficial facade presented by 'the they' and remain true to her conscience. She was still feeding her sevenmonth-old child at the time of the interview.

I didn't have an overwhelming fear of breastfeeding, and before, while I was pregnant I really talked myself into not being intimidated by being forced to breastfeed. There's a real sector out there that put the fear of God into you. People just browbeating you into persevering with cracked nipples and nipples falling off, and cracked. I thought I'm going to have a really good go at this, but if I can't do it, I can't do it. It was a really healthy attitude in hindsight because I didn't put any pressure on myself. But for the first few days it's a bit like breaking in a pair of running shoes. (Kate. Int. 3. Para. 10).

The following narrative was provided by Kay, a single parent with limited support who fell into a low socio-economic bracket, had limited education, and had experienced extremely negative reactions from close family members to her breastfeeding. Had she been assessed in light of predictive research findings, it might have been suggested that she would fail to breastfeed her child (Isabella & Isabella, 1994; Kessler et al., 1995). Despite these factors, she continued to breastfeed, believing strongly that it was the best thing for both herself and her son.

I want to do what I think is best for me and my son. I'm the one that's raising him. I'm the one that's there 24 hours, no one else is here helping me. So the way I see it is what right do they have to tell me what to do?

(Kay. Int. 8. Para. 111).

Sue struggled through insufficient milk syndrome for three months before she decided to cease trying to breastfeed her baby. During the interview she talked of the hours she spent letting her son 'graze' at the breast, as her son screamed constantly when not suckling at the breast. On one occasion during the first six weeks postpartum, she found urine crystals in his diaper, and yet the midwives encouraged her to persevere as her milk still had not 'come-in.' Throughout the time she breastfed, she had the growing awareness that she did not produce the quantity of milk her infant needed. Despite this knowing, she battled internally to defend her decision to discontinue breastfeeding as she struggled through the prevalent pro-breastfeeding pressure. It was only as she came to her own conclusions about breastfeeding that she gained the courage she needed to stop.

All the stuff that you get told about breastfeeding and it isn't actually right for everybody. You've got to wonder what it's founded on. I read that book by Adelia Ferguson²² and she had the excerpt from the WHO about why 'breast is best' and why they're pumping it so much; it's largely propaganda. They don't want third world countries going the formula route because babies do die due to the water supplies. We don't live in a third world country. Everything else is based around convenience in our day and age, convenience foods etc, except for this. You've got to go back to nature and do it the natural way. I think it's wrong. They're prejudiced against women, forcing women to do something that may not work. And there's no information to tell you that or what to do if it doesn't work. It's not fair... To be quite honest I think that people that breastfeed their babies exclusively are nuts. I'm angry in some respects that the downfalls of breastfeeding are not told to you at all. It's a real slap in the face, you think that there's something wrong with you because it's not happening for you. But I don't have feelings of guilt anymore, blow that. The most important thing is that he's happy. (Sue. Int. 6. Para. 80 & 100).

²² Adelia Ferguson authored a book called *Bottle babies: A New Zealand guide to guilt-free bottle feeding* (Ferguson, 1998).

Women who were able to concernfully uncover the 'they-self' were able to form interpretations based on their own embodied experience, thus freeing themselves from the tyranny of 'the they's' edicts. However, while some women were able to understand the inconsistencies within 'the they's' espoused ideals, they could not free themselves from striving to attain averageness – 'the they' held sway.

6.3.4 When 'The They' Holds Sway

At the time of her interview, Cathy was still persevering with breastfeeding despite numerous obstacles. Her daughter had reflux and refused to latch on for more than 10 to 20 seconds at a time; she also experienced severe pain, and described how she would bite on objects so she could cope with the pain of initially latching her daughter on her breast. Cathy's comments characterised the women in this study who could understand there were inconsistencies with the image portrayed by 'the they' in relation to breastfeeding and the reality of this experience for some women. Despite this acknowledgement, Cathy could not flee the summoning and sought to fulfil her obligation to 'the they.'

I would love, if I had a third one, to feed it but I would also like not even to try, to go straight onto the bottle and not have that stress. Why am I still persisting with expressing? I'd hate another woman to go through what I've gone through. And I'm stupid, I'm stupid to carry on, I'm not in pain like I used to be at all, but the hassle of having to express and be sore when she does come on and off, why don't I just bottle-feed her? I would love to say to another woman "just give it up, if it's not working just give it up, you're still OK, your children will be OK and you're not a bad mother." I don't think it makes any difference whatsoever in terms of bonding at all. I didn't bond with my son for the first month but now we've bonded heaps and he was on a bottle. You can actually look into their eyes more when you're giving them a bottle than when they're looking straight at your boob. They say skin to skin but I don't know what that is because the only skin to skin is his mouth to my nipple. They say that skin to skin contact is important but it's not like I take all her clothes off and all my clothes off to feed her so I don't know what skin to skin contact they're talking about is? When you're bottle-feeding them you can look them in the eye and talk to them but when they're

breastfeeding they're looking straight at your boob. So yea so I would just like for women to know that it is OK, she's still ok, she hasn't failed but I don't believe it myself.

(Cathy. Int. 18. Para. 45).

Cathy discussed the internal dialogue and decision making process she engaged in when thinking through her experience of breastfeeding. She stated that she did not think it made a difference whether she breastfed or not, but could not flee the summoning of 'the they.' She discussed the arguments used by health care professionals, anthropologists and sociologists to promote breastfeeding. She wrestled thoughtfully with these arguments in light of her embodied experience and her solicitude for women who may have similar breastfeeding experiences to her own.

Cathy went on to speak of breastfeeding using the term 'hellish' to describe her experience; however, she continued to try to breastfeed. She shared at length her feelings, her desperation to breastfeed, and how she believed she was failing as a mother because she couldn't fully breastfeed.

I have a friend and she gave up breastfeeding her daughter after two months and that led to her getting postnatal depression. But she felt that pressure to feed. So we talked about how hellish it is. To me it's just like hell on earth breastfeeding. I don't know who recommends it. Why people recommend it because it's...hell.

(Cathy. Int. 18. Para. 43).

Sam provided another example of how 'the they's' edicts hold sway over women as breastfeeding mothers. Women feel obliged to adhere to publicly-held beliefs, although they are able to accept that 'other', for good reason, may not. Thus, it might appear that 'other' is freed without consequence. However, Sam also, provides insight into how 'other' are also held sway by the same obligations and seeming struggle with the failure to portray the perfect image.

It's funny how I say, that you can be a great, successful mother, because when I look at women who bottle-feed their babies I don't think they've failed, and yet I apply that judgement differently to myself. I would have thought I had failed but I don't think other women have. And the women I know who did bottle-feed they must have put themselves

under that same pressure because they do get defensive and they do tell you why. One of them was 'I got mastitis,' and what ever, there's always a reason for it and I can completely understand where they're coming from. You do put yourself under pressure and you do feel that you need almost an excuse for it, if it didn't go well.

(Sam. Int. 12. Para. 1112-114).

The women who were held sway by 'the they' were unable to free themselves from the call to averageness emitted by 'the they'; in consequence, these women's experiences were interpreted in accordance with publicly-held sources of meaning.

6.3.5 Interpretations as Formed by 'The They'

The breastfeeding women in this study were certainly not immune to 'being-with-the-they'. Their language painted pictures of how their worlds were interpreted in accordance with 'the they', revealing how their experiences of breastfeeding hinged on 'the they's' interpretation of breastfeeding.

Eva was persevering with breastfeeding despite extremely painful cracks in her nipples – she practiced crystal gazing and deep breathing to facilitate coping with the pain of latching – and a five-day hospitalisation for mastitis. She spoke at length of her struggle to breastfeed and how, when her midwife presented her with the option of ceasing to breastfeed, she chose to continue as it was best for her infant. Three months into the experience the thoughts were coming to her that she had forgotten to listen to her embodied knowledge. This embodied knowledge had been silenced; she had been deafened by the information offered by 'the they.'

I've got this huge pile, full of information, almost all of this is on breastfeeding. I've got this huge clear file that's four inches thick and I've got six or seven books on breastfeeding. There's so much information that you've almost forgotten how to listen to your own body. You become reliant on other people to say its OK, such as midwives and doctors, or books or pamphlets, to say how things should be done. I also think that because the 'breast is best' slogan is told to you over and over again before giving birth you're so hyped into that mindset that you

actually need to hear someone professional say that "well actually bottle is OK too."

(Eva. Int. 19. Para. 16).

The strong messages offered by 'the they' rendered some participants incapable of making decisions that would promote the well being of both themselves and their infant. Mel, who provided the following narrative, spoke of her attempts to breastfeed her infant daughter. To ease the reflux, Mel would breastfeed with her baby in a sitting-up position, at the same time attempting to support her large breasts and splint her caesarean wound. She found this method of feeding impossible, but felt incapable of making the decision to stop breastfeeding.

I don't know why somebody didn't tell me in the beginning to give up. It's really horrible because people are saying to you "breastfeeding is the best thing for your baby", but it's emotionally crippling for you to have to give it up. Or with the Plunket nurse saying 'I'll support you no matter what', that is really nice but I didn't want that, I wanted someone to say "it's not working, go to formula it's a lot easier, don't berate yourself over this and just don't worry about it", but they don't. They try and convince you to do otherwise. It's really hard.

I was so disappointed to give up, that's probably why I persevered for so long with my daughter when it was getting quite awkward in the end. But yea, yes - failing her and failing me and failing what other people think of you too. People that don't have children or mothers that do and have breastfed successfully six, seven times, they are so opinionated on what you could be doing wrong or why you aren't breastfeeding. (Mel. Int. 17. Para. 2 & 24).

Sam had embraced the idea that the only entity capable of making her 'a good mother', was her ability to breastfeed. Thus she became lost in 'the they's' interpretation that anyone can 'nurture' a baby, but only a mother can breastfeed. As she struggled to establish breastfeeding, she began to question her ability to be a good mother, especially in light of her partner's ability to nurture their child. This resulted in her becoming insecure and feeling like a failure.

My partner is so good as a father and I wanted to be able to do something that he couldn't do. I wanted to be able to be the mother of this baby. And my partner's mother was saying, "if he had a pair of boobs he'd be happy." I thought 'great, thanks, so I do nothing.' That really upset me at the time and still upsets me now when I think about it. (Participant had tears in her eyes). It was like that is one thing that I can do to demonstrate that I am a mother, and a good mother that my partner can't do. It wasn't like I resented him being a good father, I'm really lucky, it's just that I wanted to be able to do the things that my body should be able to do.

(Sam. Int. 12. Para. 95-96).

Jan's sentiments echoed Sam's. 'The they's' created meanings leave women believing they are not good mothers if they cannot breastfeed. From the first breastfeed, shortly after giving birth to her first child, she and her son had difficulty. Jan reported that, by day ten, "he had basically eaten off my right nipple completely" (Int. 13. Para. 6). She went on to subsequently develop mastitis which resulted in hospitalisation and surgery. Despite all of these obstacles, she still struggled with the belief she was a bad parent because she could not continue to breastfeed. She spoke of the guilt that was so pervasive she cried for three to four months because she could not breastfeed. 'The they' reinforced these notions by being unable to accept her bottle feeding her son.

I think you have this expectation that you can do it and that it's relatively easy. People say "it's painful but you can get through it, if you keep persevering it will be fine." My midwife was fantastic, she was visiting me every day so I wasn't unsupported it was just that it wasn't going to work with him and myself. And I think I started thinking that it was all me, and that I should be able do it. People kept saying "just keep going and it will go OK, it will go away", but we should have stopped heaps earlier. We shouldn't have got to such a state where I couldn't stand up, I couldn't eat - I couldn't do anything. I was useless to everyone including my son. I felt terribly, terribly guilty about not being able to breastfeed. You get all this stuff shoved at you all the time about how breastfeeding is best and you almost feel like you're not a good parent if you can't.

It's a thought in the back of your mind all the time, that there's something not right. That there's something I should be doing but I'm not. It was a very emotional, teary thing for me that breastfeeding guilt. Tears were never far away for three to four months. I guess because it's like an unknown with that particular question 'am I really affecting my child in such a way so that when he goes to school he's going to be so different from everybody else?' He was also a relatively sickly child and that's because he wasn't breastfed. That he didn't get all the goodness he needed and became a sickly child and I've actually done that to him by not being able to breastfeed. So you have all this baggage that you carry around with you. And I think it was very in my face because people would talk to me about bottle-feeding a lot, and you'd go out and people would look at you bottle-feeding I mean that's your perception. And think 'Oh, she's got a wee, wee baby and it's being bottle-feed'

At the end of the day, when you're in the middle of being really sick, you're not sure and you don't have the ability to make the right decision sometimes, well I didn't. I got to the stage where I was told that I had to stop. And if I hadn't been told I would have kept going, and going, and getting sicker, and sicker. I was lucky, I suppose, that somebody said "this is the end of the road, we are going to stop now". And so maybe that helped because it wasn't my choice. Maybe if I had, had to make that decision I might not have got over the guilt as easily. (Jan. Int. 13. Para. 8, 56 & 58).

Jan gave over the interpretation of her breastfeeding experience to 'the they' – she relinquished the decision to wean her son to health professionals, leading her to believe that she did not own the decision, and was thus freed from responsibility. However, she still gave credence to 'the they's' interpretation that all good mothers breastfeed, succumbing to feelings of guilt that she resolved only when she accepted that she had not actually made the decision to wean her son.

It is not the purpose of this exposé to critically analyse the 'hidden' purpose of 'the they'; 'the they' is neither good nor bad (Inwood, 1999). Instead, the purpose is to highlight the manner in which women interpret their experience of breastfeeding. The call to comply with the practices of 'the they', the pull to averageness, the 'levelling-down' experienced by women at a vulnerable time in

their lives, are very powerful effects. "We cannot be average if what is averagely expected is beyond the reach of most of us" (Inwood, 1999, p. 213). Due to the vulnerability experienced during the transition to motherhood, women are at the mercy of 'the they'; thus, 'the they' powerfully influences women's interpretations of their breastfeeding experiences.

6.4 Summary

This chapter has laid out an interpretation of how 'the they' influences women's breastfeeding experiences. I have discussed how 'being-in-a-world' 'with-other' provides the foundation for commonly-formed meanings, that provide the framework for the interpretations made about women and their experiences with breastfeeding. At times, the participants chose to flee the summoning to submerge their created meanings in the public identity, and as a result had to learn to navigate their way through 'the they's' scrutiny. Women who failed to concernfully navigate 'the they's' inspection struggled in silence through an emotional minefield of guilt and a sense of failure. The following chapter explores how women have inherited a language that describes breasts and breastfeeding in an objectified manner.

Chapter Seven

Inherited Understandings: The Breast as Object

7.1 Introduction

This chapter explores how women have come to understand their bodies and, therefore, their breasts, as objects. The interpretation of the breast-as-object has emerged from the philosophical work of Descartes, who positioned the body as an object. Descartes' interpretation and associated language was long ago embraced by the medical fraternity, and has been handed down over centuries to describe human experience of our bodies; as a result, this language has become embedded in our understanding of the body (Inwood, 1999; Leder, 1984). The participants in this study relayed how their breasts were objects to be handled and managed, and therefore Heidegger's philosophical work on objects is used to explicate women's experience of breastfeeding.

7.2 Cartesian Dualism

During the Medieval era²³, people viewed bodies as mystical entities that contained the essence of life. Such beliefs had prevailed since the time of Aristotle and were problematised by Descartes, who, instead of focusing on the mystical quality of life, argued for a monotheistic approach to teleology, claiming that God was the source of all truth and purposefully-ordered existence (Leder, 1984). It is on this foundational premise that he based his philosophical quest for an epistemology that transcends historicity and subjective experience (Inwood, 1997; Inwood, 1999; Leonard, 1994).

One of the aims of Descartes' philosophy was to advance medicine through a mathematical and philosophical analysis of the body (Burnham & Fieser, 2001;

²³ The Medieval era spanned approximately one thousand years, from the fifth century to the fifteenth century (Russel, 1996).

Leder, 1984). He believed that the physical world is a mathematical world and that, to achieve this revisioning of the physical body, a systematic mathematical approach must be employed (Burnham & Fieser, 2001; Inwood, 1999).

According to Descartes, the first act of any systematic analysis must begin with *hyperbolic doubt*, through which all knowledge is exposed to critical philosophical examination (Sorrell, 1987). To distinguish and clarify knowledge, the philosopher must undertake hyperbolic doubting to uncover that which is indubitably true (Burnham & Fieser, 2001; Sorrell, 1987). An integral part of hyperbolic doubt involves examining any information received through the body with extreme caution. Descartes believed that such secondary knowledge, received through the senses, has the capacity to deceive the mind, thus blinding the examiner to truth (Burnham & Fieser, 2001; Leonard, 1994; Sorrell, 1987).

Ultimately, Descartes came to believe that he existed essentially as a 'thinking thing', and to describe this he employed the phrase *cogito ergo sum* - 'I think therefore I exist' (Burnham & Fieser, 2001; Dreyfus, 1991; Inwood, 1999; Sorrell, 1987). Descartes ameliorated the mind as the source of intelligence and life for humans and, although the mind and body are intrinsically linked, he relegated the body to the domain of *res extensa* (i.e., extended thing) (Burnham & Fieser, 2001; Dreyfus, 1991; Inwood, 1999; Leder, 1984; Leonard, 1994). By arguing that the mind completely controlled the body, Descartes relegated the human body to the realm of machine – a machine completely incapable of spontaneous mystical behaviour, intelligence or purposeful movement (Burnham & Fieser, 2001; Leonard, 1994). This understanding rendered the body as an object controlled by the human mind; thus, people gained knowledge through examining objects, which included the human body. This separation of mind and body has become known as Cartesian dualism (Inwood, 1999; Leder, 1984; Leonard, 1994; Sorrell, 1987).

Due to the prevalent belief in the seventeenth century that the body was the sacred source of life, doctors were prevented from dissecting or interfering with a body in any way (Leder, 1984). With Descartes' revisioning of the body as an object, medical practitioners were no longer subject to such social taboos – the body could now be operated on, dissected and analysed in accordance with mathematical and scientific paradigmatic ideals (Leder, 1984). Thus, Descartes had a significant impact on the medical establishment, which adopted the philosophical thesis of the body as an object (Inwood, 1999), and Cartesian

dualism emerged as the major paradigm for medical practitioners (Gadamer, 1996; Leder, 1984).

The medical discourses that exist in our society still embrace notions of Cartesian dualism (Gadow, 2000; Leder, 1984; Leonard, 1994). Due to their significant exposure to health care professionals, women in New Zealand gain access to medical discourses during pregnancy. Therefore, their interpretations of their birthing and breastfeeding experiences are strongly informed by these discourses (Gadow, 2000), as evidenced by the consistent appearance of notions of Cartesian dualism in the initial stages of the interviews described in this study.

7.2.1 The Body as an Object

Medical discourses that adhere to notions of Cartesian dualism have influenced the beliefs of the women who participated in this study. These notions were reflected when the women initially discussed their breastfeeding experiences using medical terms such as mastitis, antibiotics, and colostrum, and referring explicitly to their bodies as objects.

Eva discussed how her body was being used as 'nature intended', through giving birth and having breasts that produced milk. Eva's body was something she possessed, and she attributed some of her body's success to modern medicine's ability to manage it.

It does feel good to have used your body for the purpose that nature intended for it. And that applies to breasts - I've utilised what I've got. I feel pleased that everything worked as much as modern medicine allowed... and that my body produced a baby and I'm producing milk. (Eva. Int. 19. Para. 145).

Cathy spoke of delivering her second baby by caesarean section. Having had a particularly bad experience with breastfeeding her son, she was relieved to have a lactation consultant midwife who could put her little girl 'on' so that she was 'latched' properly. Cathy believed she was unable to get her first baby 'latched' onto the breast properly, and therefore having a health professional

present to manage her body meant that the initial breastfeeding experience with her daughter was successful.

The second day in hospital I had a caesarean but I was very fortunate because a lactation consultant was my midwife in the operation room. She came with me to the recovery room and put my little girl on [the breast] and she suckled for probably about half an hour and latched on really well.

(Cathy. Int. 18. Para. 6).

Women in this study were exposed to mechanical devices to help them manage their object bodies. Liz came to understand and possess a language about breasts and breastfeeding through being exposed to, and using, multiple devices: she expressed, she syringed, and she steamed.

I ended up expressing [colostrum], syringing colostrum out and syringing colostrum into her [my daughter's] mouth trying to get her to get some of the colostrum. So it was kind of squeezing in machines, the works; expressing machines and then one night we had the steam unit on. You know we 'steamed the nipples.'

(Liz. Int. 9. Para. 8).

Jan talked of her experience of her body producing milk for her first son, which included discussions about mastitis, homeopathy, antibiotics and a 'whole body' infection. Her narrative did not explore the experience of how it was to have a whole body that was infected; instead, she focused on treatments that were unsuccessful.

Once the milk came through it [the breast] didn't reduce in size so I was getting continual mastitis. We tried homeopathy and all sorts of things then we tried antibiotics but nothing worked and in the end my whole body became really infected.

(Jan. Int. 13. Para. 6).

Kate discussed her first experience with breastfeeding. She told how she struggled to comprehend that her body was producing anything that could nourish a baby. She spoke of how 'they' put him on the breast and that feeding was something that was happening to her body, not to her.

Breastfeeding - it was probably just after he [my son] was born when they actually put him on the breast, it was only colostrum obviously. It was one of the things I couldn't get my head around, like it's all happening to your body.

(Kate. Int. 3. Para. 6).

Kate went on to say that the infant had her functional, nurturing breasts on lease; in part, she knew and understood her body as an object that could be loaned to her infant, having previously shared it only with her partner. Whilst breastfeeding, the baby was in possession of the breasts.

What was yours as a couple is now a functioning organ, it's not even an organ, but it's a functioning part of your body that's nurturing your child. [My husband and I] we've got this big joke here that the baby's got them on lease, we will actually get them back into our relationship at some stage.

(Kate. Int. 3. Para. 60).

Mel reiterated this idea of breasts as objects that are capable of being shared, and felt that her breasts had ceased to be hers. Instead, they were an object that she could loan out to others.

You don't feel that your breasts are yours and you're certainly not sharing them with your husband (laugh). Sometimes it's like 'when will I get my body back? I'm not sharing it anymore'.

(Mel. Int. 17. Para. 38).

7.2.2 Understanding Objects

One could apply Heidegger's (1962) philosophy to these objective accounts of the body, interpreting the participants' narratives in light of his comments about equipment. It is important to note that Heidegger never intended for the human body to be seen as an object or res extensa; instead, his philosophical work was, in part, an attempt to destroy notions of Cartesian dualism. However, because of their inherited understandings, some of the women in this study came to know and understand their bodies within the context of a subject-object perspective. Therefore, an exploration of this abstraction is appropriate to fully

illustrate the participants' experience, because "in every case...interpretation is grounded in something we have in advance" (Heidegger, 1962, p. 191), and "an interpretation is never a presuppositionless apprehending of something presented to us" (Heidegger, 1962, p. 193). In this research, the women interpreted their experiences through the lens of the body as object.

The foundational concepts of how the participants came to understand and interpret breasts and breastfeeding were examined in Chapter Six. These concepts exhibit themselves in commonly-formed meanings that are developed as a result of our 'thrownness' in a world that already contains these commonly-formed meanings. Examples of this are found in the interpretations of the 'always-alreadyness' of breastfeeding, the breast as sexual, and breastfeeding as an intrinsic function of mothering, which were discussed previously. Such understandings form the basis for how these women interpret their bodies, breasts and breastfeeding. These commonly-formed meanings also form the basis for interpreting objects that appear in the world.

When interpreting the human body as *res extensa*, breasts are considered to be objects that can be manipulated, a concept that becomes increasingly evident when a woman desires to breastfeed. Within this framework, a woman needs to manipulate her object-breast to be able to breastfeed. To more fully explain how people experience objects or equipment, Heidegger (1962) discusses how objects are understood to be either 'ready-to-hand', 'present-at-hand' or 'unready-to-hand'.

7.2.3 Ready-to-Hand

'Ready-to-hand' describes how we proximally encounter entities or objects (Heidegger, 1962,p. 121). 'Ready-to-hand' occurs when one takes hold of an object and uses it. The object itself becomes transparent to us, and we focus on the outcome of what is being done rather than the object. All the women who participated in this study had breastfed; therefore, they all had the requisite bodily 'equipment' in the form of functioning breasts. To some extent the women understood and, therefore, encountered, their breasts as 'objects'. The following narratives emerged from the interviews with these women, some of whom spoke of the wonder that their bodies were able to nourish an infant. Those women who had become technically proficient at breastfeeding

exemplified the concept of 'ready-to-hand': the lactating 'breast' as an object became invisible, and what emerged as the focus for these women was that their breasts represented the nurturing or comforting of their baby.

Kay reported that the first time she breastfed the midwives put the baby to her breast and she was unable to remember the experience properly. Her body, her breast, was providing the milk to her baby without her being aware. Her breasts were 'ready-to-hand'.

I mean they put him on my breast, but I don't really remember it properly the first time because there's just so much emotions and everything happening so I really don't remember the couple of hours afterwards because you're on a high so to speak.

(Kay. Int. 8. Para. 22).

Kate spoke of how she could not comprehend that her body could produce milk that would nourish an infant. After she had given birth she saw visibly the effect that putting her infant on the breast had on her son; however, she denied any active involvement on her part.

It was outside my realms of comprehension, that I could even produce something that was going to feed the baby... but just putting him on the breast - that whole reflex action where he latched on... and here he was - instant calm and it was an instinct and it was right... it was totally passive on my part.

(Kate. Int. 3. Para. 6).

Liz discussed the fact that a woman's body will naturally produce milk after giving birth. As breasts are a 'ready-to-hand' part of a woman's body, she believed it encouraged being physically close to her baby, as much time was spent providing nourishment when her baby was very young.

Well it is something that your body simply does unless you stop it. I mean your milk does come in, it is a natural part of it [having a baby] especially in the initial days it's really the only contact that you have with your baby apart from cuddling her. Having the baby on the breast is really the time when you're sitting there, you're close to her, you're looking at her, and she's getting nourishment from you.

(Liz. Int. 9. Para 50).

Liz also spoke of how she could just 'pop' out her 'ready-to-hand' breasts and feed her daughter. She commented how simple breastfeeding was in contrast to feeding an infant with formula, because the breast was 'ready-to-hand'.

It's so much easier when you just have the breast and you can be at somebody's place and just pop them out and away you go, whereas having to carry formula, boiled water and sterile bottles and [having to] shake them up. it's a fiddle.

(Liz. Int. 9. Para. 18).

7.2.4 'Unready-to-hand'

The 'unready-to-hand' mode of equipment is exhibited when an object ceases to work, or becomes problematic in a functional sense, and thus the transparent nature of the equipment that exists in the 'ready-to-hand' mode is suspended. The equipment becomes the focus, not the outcome (Heidegger, 1962; Plager, 1984). The concept of the breast being 'unready-to-hand' was illustrated numerous times during the participants' interviews. Prior to their breastfeeding experiences, many of the women had only encountered their breasts as sexual equipment; when their sexual breast became also a lactating breast, they experienced their breasts in a new and different manner. Many described how their breasts were 'unready-to-hand' in their new mothering role. They discussed how they had to become technically proficient at 'using' their functional breasts, and thus their focus was the breast, and becoming proficient at breastfeeding, not the outcome of breastfeeding. This mode of 'unready-tohand' was especially dominant when a woman was struggling with the technical skill of handling the 'breastfeeding breast', which commonly occurred when the baby was less than three months of age.

Liz described how she had previously-held understandings of how the breastfeeding breast would be, and had expected her breasts to be 'ready-at-hand' when her infant was born. She was astonished that this was not the case, finding instead that her breasts were 'unready-to-hand', as her nipples were cracked and sore. She mused at her former naiveté.

I just always had the expectation that it would work and that it would be fine. And I was quite surprised by having these problems, they weren't little at the time, with the cracked nipple and the sore nipples, I really didn't expect them at all. I just expected him to start feeding and everything to be wonderful (laugh).

(Liz. Int. 9. Para. 4).

Eva wanted to latch her daughter on to her breast but struggled with 'unready-to-hand' nipples. She came to understand that breastfeeding is a skill both she and her baby had to learn. Through exposure to differing latching techniques, and information about nipple care and proper diets to ensure adequate supply, she gained knowledge on how to breastfeed. Eva had to purposefully practice a technique that would enable her baby to suckle and acquire milk from the breast.

I was faced with the situation where the baby was born and I wanted her to latch on but she couldn't really because my nipples weren't sticking out as much as they should have, it took quite a lot of work to get the technique right and to get the baby sucking them out... The first one and a half weeks we were learning about breastfeeding. The first thing I realised is that it doesn't come naturally at all. It's actually a skill that the baby has to learn and it's a skill that you have to learn. There are all sorts of different techniques that you get shown and different things work for different people, and then there's nipple management, how often and how long, your diet and all those sorts of things. It's tricky and it's really hard and you have to work at it and practice and practice and practice lots. It's only now, three months down the track, that I feel confident that we're doing it right, that she's getting enough, that she feels happy and I feel happy we're on demand feeding and it's really good.

(Eva. Int. 19. Para. 6 & 7).

Sam struggled to get her son latched onto her breast, trying many different techniques suggested by her midwife. She discussed how both she and her son struggled with breasts that were 'unready-to-hand', but two weeks after the birth they managed to make breastfeeding work.

We were doing all sorts of tricks like, expressing a little bit first and we were even dribbling water into his mouth so that he was getting

something to kind-of keep him satisfied. Every trick in the book we were using and then that morning, for some reason, he latched on fine and then continued to and he's been fine ever since... kind-of, I mean that we've both got better at it.

(Sam. Int. 12. Para. 8).

The first time that Lee fed her son she was completely focused on the technique rather than on the outcome of breastfeeding. She felt that she 'obsessed' over acquiring this new skill so that, in the first week of her son's life, the mechanism for feeding her son, her breasts, were 'unready-to-hand'. However, Lee went on to describe how, once she had mastered the act of breastfeeding, she ceased to think about the mechanical and was able to focus on those entities that lay beyond the need to provide her son with essential nutrition.

I was thinking the whole time really about the mechanics of it and whether I was breastfeeding properly, and that was probably the first week that he was alive. They became sort of obsessive thoughts — 'are you holding him properly, are the lips right, is the nipple drawn right back' you know all those things because it's all new to you... In the beginning, whenever you start a new job or task you start thinking about the mechanics and whether you're doing it right, and your mind gets fixed on all those sorts of things and you don't see outside of that but once you master it then you start thinking about other things. (Lee. Int. 10. Para. 18 & 19).

7.2.5 'Present-at-hand'

'Present-at-hand' describes entities that we experience by going through entities that we have proximally encountered (Heidegger, 1962). "Things that are 'present-at-hand' only become visible when those entities that are 'ready-to-hand' become transparent" (Inwood, 1999, p. 214). Once the women had become proficient in managing their 'functional breast' and 'breastfeeding', they were able to navigate their way through the proximally-encountered equipment and focus on those things that lay beyond. The women's focus dwelt with those entities that were more metaphysical in nature: the comforting nature of breastfeeding breasts for their infant, and being able to visibly see the infant's body grow.

Lee spoke of how enjoyable breastfeeding became for her once she became confident in providing for her baby's nutritional needs through breastfeeding. What emerged as 'present-at-hand' through breastfeeding was the chance to sit and be close to her infant. Breastfeeding afforded her the opportunity to develop her relationship with her son, a relationship that brought her satisfaction.

Once I was very confident with breastfeeding him, probably by the time he was about six weeks old, then I started to really enjoy it as a lovely time when you're holding your baby really close. Having a chance to sit down for a while and hold him and talk to him. You know [it is] a really pleasurable experience.

(Lee. Int. 10. Para. 20).

Liz reinforced the idea that, when she was not anxious about the technical act of using breasts to feed, she gained access to the 'present-at-hand' experience of breastfeeding as pleasurable. She spoke of how she enjoyed meeting her daughter's nutritional needs, whilst providing a relational connection between both mother and daughter through the physical touch involved with breastfeeding.

You are her [the baby's] source of nourishment. You're giving her the best food that she can have and you're giving her that contact with you, which is so lovely. To be at the breast is lovely for both her and me. (Liz. Int. 9. Para. 52).

Eva spoke of how pleasurable it was to see the benefits of breastfeeding. The breast in its 'present-at-hand' mode became physical growth and health for her daughter, a mode that emerged only as Eva became proficient at breastfeeding.

Just looking at her [my daughter] growing and looking so healthy, she hasn't been sick at all, almost all the weight she's put on has been from me and it's such a satisfactory feeling.

(Eva. Int. 19. Para. 11).

These narratives have examined how the body has been interpreted as *res extensa*, or extended thing, by the participants. This way of interpreting the body has been inherited and is exhibited through the language available in the

subject-object discourse. It has had a significant influence on the way in which we understand the human body, including breasts. Exploring the women's narratives through Heidegger's (1962) understanding of how people interpret objects has offered a depiction of how women can come to know their breasts and breastfeeding.

7.3 Summary

Concepts of how people understand and access equipment are useful when exploring women's narratives about breastfeeding. Because we have inherited an understanding of the body as an object, Heidegger's (1962) thesis on how humans approach objects has offered some insights into the being of a breastfeeding woman. However, this approach fails to take into account embodied²⁴ understandings of experience, and continues to relegate bodily-based intelligence as inferior to that of the mind. If women's experience of breastfeeding is to be more fully explored, a different perspective on how we experience our bodies is needed. Therefore, the following chapter explores how an embodied interpretation of women's breastfeeding experience allows the emergence of a meaningful understanding of the being of a breastfeeding woman.

²⁴ Embodiment describes how we exist and understand existence through a lived body capable of both physical movements and non-physical reasoning, through which we gain meaning (Merleau-Ponty, 1981). See Chapter Four, Section 4.4.3 for additional information.

Chapter Eight

The Embodied Experience of Breastfeeding

8.1 Introduction

The concept of Cartesian Dualism provides a lens through which to interpret the body as an object. As described in Chapter Seven, this way of understanding the body has informed women's understandings of their breastfeeding experience. However, breastfeeding is undeniably an embodied experience and, as such, the language embedded in the subject-object tradition is inadequate to describe its depth. When attending to embodied narratives, women's experience of breastfeeding becomes an experience of embarrassment, pain and tiredness. It becomes an experience with the fear of failing, and feelings of guilt if women perceive that they have 'failed' at breastfeeding. The experience of breastfeeding can ultimately leave a woman feeling emotionally crippled.

This chapter describes the embodied experience of breastfeeding, centering on the process of establishing breastfeeding. It begins by highlighting the philosophical basis for the concept of embodiment, and then proceeds to explore how embodied breastfeeding experiences are silenced in the everyday language available to describe breastfeeding. Women's initial breastfeeding experiences are then explored through narratives that describe breastfeeding as a painful and tiring process. Exploration of the narratives highlights how the women participating in this study were anxious about being successful at breastfeeding, and how they struggled with feelings of guilt if they were unable to successfully establish breastfeeding. This chapter concludes with the participants' discussions about the emotional impact of not being able to breastfeed.

8.2 Philosophical Underpinnings

It can be argued that interpreting women's experience through an object-subject lens is a superficial way of dealing with the depth of their experience of breastfeeding, as only some specific facets of the experience can be explored, thereby limiting the discussion and prohibiting the emergence of a fuller understanding of what it is to be a breastfeeding woman. Any interpretation that is limited by the subject-object lens neglects to expound on the experience of existence; in consequence, such interpretations close down upon the opportunity to gain a full understanding of the lifeworld. It is this failure that led philosophers such as Heidegger (1962), Gadamer (1989) and Merleau-Ponty (1981) to argue that the Cartesian approach to explaining existence inadequately articulates and describes the lifeworld, as it is from our embodiment that we experience the lifeworld (Robertson-Malt, 1999).

Heidegger desired to "reverse the Cartesian tradition by making the individual subject somehow dependent upon shared social practices" (Dreyfus, 1991, p. 14). Heidegger (1962) claimed that 'subject' and 'object' can only exist and be experienced through the contextualisation of 'being-in-the-world' and therefore it is only in acknowledging this ontological position that anything is capable of being understood. He further argued that being is existence, and that the source of meaning, or significance for being, replaces the individual subject (Dreyfus, 1991; Heidegger, 1962; Inwood, 1999). The Cartesian tradition claims that the source of meaning is always the mind; however, this belief is still grounded in subjective experience, and ultimately cannot be sustained in the light of the development of this ontological perspective (Dreyfus, 1991).

Gadamer (1996) argued that any objectification of our bodies "demands of all of us a violent estrangement from ourselves" (p.70). He claims that all the kinaesthetic phenomena that are experienced by the body require attention, as this is an essential task of our 'being-in-the-world'.

Merleau-Ponty (1981) argued for an embodied interpretation of human beings, espousing that he could not "understand the function of the living body except by enacting it myself, and except in so far as I am a body which rises towards the world" (Merleau-Ponty, 1981, p. 75). He believed that the body was not an 'amalgamation' of the subject and object; instead, the union is a constantly-

present part of existence. In the *Phenomenology of Perception* (Merleau-Ponty, 1981) he wrote: "We need to reawaken our experience of the world as it appears to us in so far as we are in the world through our body, and as far as we perceive the world with our body" (p. 206).

For Merleau-Ponty (1962), the lifeworld is a world that we have access to through our bodies. He stated that "it is through my body I go into the world" (Merleau-Ponty, 1962, p. 316), thus espousing that we do not have a body that can be utilised as an object; instead, we are a body. Because we have inherited a historical way of interpreting our body as an object, understanding the embodied body is a difficult undertaking (Dahlberg et al., 2001). To fully understand the lifeworld, it must be interpreted from an embodied perspective, because "we cannot step outside the body, instead, we experience it from both inside and outside simultaneously" (Dahlberg et al., 2001, p. 51). It is the body that is constantly perceived, always present, and moves towards the world in a meaningful manner (Merleau-Ponty, 1962, p. 92). The body grasps the world through intentionally navigating its way within the world, and thus the body gains access to the meaningful (Leder, 1984; Leonard, 1994). The embodied person dwells in a "culture and particular lifeworld that enable the cultural meanings to be inscribed on the body" (Benner, 2000, p. 8), and thus we come to understand the world.

As stated previously, the women who participated in this study began narrating their breastfeeding experience by employing the language embedded in the subject-object discourse. However, as they sought to explore and articulate the effects of their breastfeeding experience, it became evident that the subject-object language was inadequate to describe the depth of that experience. Instead, their descriptions turned to that of an embodied experience of lived breastfeeding. The examination of this embodied experience begins with an exploration of the hidden nature of the phenomenon.

8.3 A Hidden Story

Mothering is an experience often veiled in secrecy; neither the positive nor negative aspects of what it is to be a mother are often discussed between women who have given birth (Maushart, 1999; Swigart, 1998). Furthermore,

embodied knowledge of what it is to be a mother is not extended to women who are pregnant with their first child. The women who participated in this study commented that the embodied experience of breastfeeding, which can be viewed as an integral part of mothering, was also silenced. The embodied experience of breastfeeding is therefore a hidden experience as described by Mel in the following narrative. Prior to breastfeeding she had perceived that breastfeeding was an easy aspect of mothering; however, once she engaged in the act of breastfeeding she discovered that her assumptions had led to unrealistic expectations.

When you breastfeed you just assume 'I'm breastfeeding, what could go wrong?' They don't tell you that it's so hard... You couldn't even get them [the baby] latched on and no one tells you that you can experience that. No one prepares you.

(Mel. Int. 17. Para. 24).

Sue struggled with what she described as a lack of milk, which she could not resolve no matter how many hours she let her son 'graze' at the breast. She spoke passionately of the hidden stories about breastfeeding, and the associated pitfalls, which led her to believe she was, in some respects, inadequate.

I'm angry in some respects that the downfalls of breastfeeding are not told to you at all. It's a real slap in the face, you think that there's something wrong with you because it's [breastfeeding] not happening for you.

(Sue. Int. 6. Para. 80).

Later in the interview, Sue further reflected on the lack of information about the difficulties of breastfeeding, articulating how this can negatively affect one's experience with breastfeeding and limit opportunities to overcome complications that one may encounter.

There's no information to tell you what to do if it [breastfeeding] doesn't work. If it isn't for your mother or a friend who's had a similar problem you'd be totally at sea really. It's not fair. (Sue. Int. 6. Para. 100).

Eva contemplated why the experience of breastfeeding was hidden from the public domain. She spoke about how women do not wish to disclose their experiences due to a 'stigma' that surrounds the topic of birth and breastfeeding, and that the only acceptable place to voice one's experience with breastfeeding difficulties is with those who have already lived the experiences of birth and breastfeeding.

The question is really why don't people talk about the difficulty of breastfeeding? There is definitely a stigma attached to that and I only discuss those things with a very elite circle of people who are in exactly the same situation as I am, like other young mothers... I think of all the people that I know that have had babies not one of them talked about the pain of birth, not one of them, or of breastfeeding difficulties. I would have been a perfect candidate for listening but no one chose to share that information. I'm still trying to figure out why. (Eva. Int. 19. Para. 23).

This statement might reflect the difficulty women have in articulating their embodied experiences due to such things as a fear of being perceived as a failure. It might also be indicative of life experiences that are beyond our capacity to articulate because the language we possess is inadequate for portraying the essence of the phenomenon. Also, breastfeeding could be considered to be a rite of passage, which is similar to the manner in which giving birth is sometimes perceived (Maushart, 1999; Swigart, 1998). It is hoped that, through the thoughtful analysis of these women's experiences, insight might be gained into the lifeworld of a mother who chooses to breastfeed.

For the women who participated in this study, the embodied experience of breastfeeding involved encountering embarrassment, pain, and fatigue. The fear of failing to breastfeed was also a re-occurring theme in the women's narratives - if a woman weaned her infant prematurely, she experienced feelings of guilt, described by one participant as emotionally crippling. The embodied experience of breastfeeding begins when a woman gives birth and begins to breastfeed her infant, and is revealed in the participants' narratives about their first breastfeeding experiences.

8.4 First Pictures

The arrival of a newborn marks the beginning of a new era of embodied experience for a woman. It makes no difference whether this is her first or seventh child – each experience is unique. Therefore, her embodied experiences that result from her relationship with this new infant will provide her with new understandings of the lifeworld.

For many, a part of mothering a new baby is breastfeeding, and the first time a woman puts her baby to her breast is an experience that is often vividly remembered. These encounters are *embodied experiences*. The following interpretative account starts at the beginning and explores the participants' experiences as they emerge from this first picture of what it is to be a woman who breastfeeds. These first pictures provide a glimpse of the complex fabric that makes up the complete experience of breastfeeding. These first breastfeeds offered by women to their newborns provided the participants with an initial understanding of what it is to be a breastfeeding woman, but are not necessarily an indication of the path upon which their decision to breastfeed would take them.

Sue's first breastfeeding experience was an experience with newness. She managed, with some difficulty, to get her newborn son latched on to her newly lactating breast, and spoke of how the midwife made the experience positive for her as a new mother. This first picture reflects how, as a new mother, Sue struggled to acquire the embodied knowledge that enabled her to establish breastfeeding; however, she marvelled that her new baby appeared to have embodied knowledge about how to breastfeed.

It was just after he [my son] was born. The midwife at the hospital was delightful, she was great. And she handed my baby and I said 'Oh yes so we're supposed to feed him aren't we' and she said 'yes' and she showed me how, she was good. And she [the midwife] tried to show me lying down, which I found really difficult. She decided that I had floppy boobs so I had to hold my boob (laugh), which was really hard to. You know, you've got this newborn baby, you're a new mother and you're trying to hold your breast as well, and that whole process it was tricky but we succeeded and he latched on really well and the comment then

was that he [my son] was going to be a really good feeder because he was straight into it, he latched on and was aware.

(Sue. Int. 6. Para. 42).

In contrast to Sue's experience, Emma felt uncomfortable about her first breastfeeding experience. She was aware of being with a number of people and was concerned about her body being exposed. Emma was also mindful of being touched; although she understood that the midwife was trying to help, she wanted to reclaim her body before attempting to breastfeed her son.

My first memory would be - having a whole lot of people in the room and I was thinking how am I meant to hide it (laugh), I didn't want to flash it all off to my father. That was my biggest thing. Then having the midwife poking and prodding your breasts while you're trying to, you know [start to breastfeed]. That whole night was horrible. She [the midwife] was making sure that you were doing it right and stuff, which was really good, but you've been poked and prodded everywhere else then you've got to put up with that as well.

(Emma. Int. 14. Para. 10).

Sam's initial experience of breastfeeding involved a great deal of input from the hospital staff, some of which made her uncomfortable. She spoke of one particular event in which a midwife attempted to force her infant onto her breast, despite both Sam and her infant being upset at the time. Over the subsequent two-week period, Sam struggled to get her son to feed from her breast and was only successful after significant input from other health care professionals, and the use of multiple 'tricks'. At the end of the interview, Sam returned to her initial experience and spoke once again of how negative it had been for her as a new mother. She attended to the vulnerability of a new mother who might not know what was best, nor have the power to speak up against the actions of a health care professional.

We had quite a long labour and the midwife put him [my son] to the breast pretty much straight afterwards but he wasn't interested. Then we stayed in the hospital for the night and had no luck breastfeeding there either. And had actually had a fairly awful experience during the night with a midwife who was determined to get him to the breast, and he was screaming and I was really upset. I could tell, even though I'd

never done it before that this was not going to work... He [the baby] was just too upset to be able to latch on. She was grabbing his head and forcing him and he just screamed more and I just got more and more upset and she was getting really impatient with us. That was our first experience, which didn't really bode well for the future.

While we're speaking about health care professionals I do still blame almost that first awful midwife at the hospital because I think that made our first experience of breastfeeding so hellish. I will never know but I wouldn't be surprised if it scared him off it for the first little while, the first two weeks. It certainly had a hugely negative impact on me. If that one thing had never happened I wonder how much more smoothly the process might have gone... which is a shame because I wonder how many other women she has done that to and is still doing that to. Given that that was my first time I didn't know that it wasn't supposed to be like that.

(Sam. Int. 12. Para. 2, 4 & 122).

During her interview, Jan initially provided a very objective, medically-oriented summary account of her experience with breastfeeding, after which she offered the following description of her first night as a mother.

I had my son in the evening and went up to the ward and saw a nurse when we got there but then didn't see another nurse until six am. I didn't know whether I was supposed to wake him to feed him, when he's crying am I supposed to feed him? Am I supposed to feed or settle him or whatever? And if you pushed the buzzer no one came, and the same in the morning, I waited about an hour for someone to come. And again I didn't know...I assumed that you're supposed to breastfeed, but didn't really know how to go about it and so just tried, and didn't really get on very well. I thought 'Oh well I'll do this later I 'spose.'... [It was] terrible, terrible. I thought I was the hardest done by person in the whole world really, just because you want to get it right, it's just so new and you're very nervous, really, really nervous, and frightened that you're going to do something wrong, something that may affect your child. I found it awful.

(Jan. Int. 13. Paragraphs 18 & 20).

Prior to giving birth, Jan was employed as a regional manager of a large company, a position she described as being both very time intensive and stressful; hence, she is a woman well-placed to handle stressful circumstances. However, in the above narrative she clearly articulates the anxiety she experienced in her new role as a mother, her main concern being to provide the best for her son. Being unsure of what the 'best' was, she attempted to refer to those who would know best – the midwives. Instead of receiving the guidance she desired, her avenue for gaining support was closed down upon. Jan described her anxiety at being unable to gain support from health care professionals by calling on the words *terrible*, *nervous*, *frightened* and *awful*. Later in the interview, she offered the following statement reflecting on the embodied knowledge she had gained through her initial experiences as a breastfeeding mother.

So it's very hard for a new breastfeeding person... because you don't know which is the right way, or the best way, the best way for you. That's your first lesson with little people is that they're unpredictable and that there is no best way for lots of things. It's very hard for someone who has been in control before because it's so out of control and you don't have a yes or a no, or a right or a wrong. (Jan. Int. 13. Para. 65).

In contrast to Jan's struggle to gain confidence in mothering, Mai had her first child more than a decade before the arrival of her second. Mai was working as a registered nurse and was comfortable in her knowledge of caring for infants. However, this knowledge could not have prepared her for the birth of her young daughter at 25 weeks gestation because of intrauterine growth retardation. Her daughter weighed one pound at birth. Mai was determined that everything possible should be done to ensure her daughter's survival, and resolved to express breastmilk for her even though initially she was only being fed through a nasogastric tube. She spoke about the first time she sat down to use a pump to express her breastmilk.

I felt sad. I said to myself "this isn't how it's [breastfeeding] meant to be." Especially when my first baby, I managed to breastfeed straight away in the hospital. I wanted to put her [my daughter] straight onto my breast. (Mai. Int. 4. Para. 88).

Sometime later Mai had the opportunity to feed her daughter the way she intended to when she gave birth. In the following narrative, she describes how it was for her to finally be able to put her little girl to her breast for the first time.

I had been told by the doctors and the paediatrician told me that she [my daughter] is not going to make it. There we were all stacking these containers of breastmilk. For the first few days they had to hold off the breastmilk, glucose was basically what she was having for the first few days before they started feeding her with the nasogastric feedings. And finally I get the chance, I get to latch her onto my breasts... She was a tiny little thing, I was overwhelmed (laugh). My breast was larger than this little thing sucking away on them. It was such a lovely feeling from then on. The nurses didn't believe that she would be able to suck like that. We were encouraged not to have her too long on the breast because she had to conserve her energy to grow. It's hard work sucking away, you know this tiny little thing so... She wasn't even a kilo when we came there, but then I think she was just over a kilo when I thought I'll give you a go... Like what I was saying my breasts were bigger than her (laugh), from then on I just kept it [breastfeeding] up. They kept saying that she had to conserve her energy as much as possible in order to grow but I couldn't help it. I couldn't help it I just thought this was the best way to bond with my baby, to put her on the breast.

Well - first of all it was quite emotional. It was quite emotional, not only that, I was quite emotional thinking that 'Oh she [my daughter] is so young, so small but look at her sucking away.' She knew, she's tasted it, and [she thought] 'this is different from what I'm getting through my nostrils, through my nasogastric, it tastes the same, but it's a different way.' I was just amazed actually just looking at her, and also emotional as well looking at that tiny little thing sucking away. Mixed emotions actually. Well I was quite emotional and they [the nurses] said 'you'd better stop now' but I still wanted to carry on breastfeeding. It was lovely, it [was] hard to start off with but it was the best experience about (laugh).

(Mai. Int. 4. Para. 8, 10-12 & 90).

Mai had previously experienced breastfeeding, and understood it as starting immediately following birth. Having to wait to breastfeed meant setting aside

her previously-held understandings about how breastfeeding should be established. However, once she was able to bring her tiny daughter to her breast, Mai was overcome with emotion, in part due to of her daughter's strength and determination. Her initial experience of breastfeeding her daughter was a defining moment in her life; Mai understood and experienced breastfeeding as a powerful embodied experience that underpins the comingtogether of a mother and her infant both physically and relationally. It was, for Mai, an experience forged through perseverance, which she described as the 'best' of embodied experiences – a moment when her potential for being was revealed and experienced as full and rich.

Although women's initial embodied breastfeeding experiences are varied in nature, they all hold significance and meaning when attempting to understand what it is to be a breastfeeding woman. New experiences offer access to a world that will be forevermore differentiated; 'being' is revealed at the point at which one acquires access to new experiences. For the participants in this study, the birthing of a new type of 'differentiated being', attained through breastfeeding, meant that they felt uncomfortable and struggled with feelings of embarrassment.

8.5 Breastfeeding as an Experience with Embarrassment

A common theme in the participants' narratives is the struggle, at least initially, with embarrassment at breastfeeding in the presence of others. Having a health care professional touching their breasts in such an invasive manner, whilst being watched by family members or friends, was both daunting and distressing for these new mothers. Breastfeeding required women to behave in a way that was foreign, as their chosen way of being-in-the-world had not previously included exposing their breasts in such a public manner. During their initial breastfeeding experiences, women were exposed to different-ways-of-being that left them feeling uncomfortable and vulnerable. These feelings of embarrassment negatively coloured their breastfeeding experience, hindering their ability to initiate breastfeeding.

Emma spoke of being conscious of her body after giving birth, and in the presence of others she felt concerned about what was being exposed; her

chosen way of being-in-the-world precluded her from accepting 'other' touching and viewing her in a close and intimate way. Attempting to breastfeed a young infant for the first time, her concern was a calling to focus on nurturing her infant; instead, she felt called to attend to people who could potentially observe her body in a manner that was unfamiliar to her. She likened this experience to being on stage naked – she felt exposed and vulnerable whilst trying to master a new skill. This feeling extended beyond that initial experience in the hospital, through the first month of her baby's life.

Just getting cleaned up would have been nice. You're conscious of your body... whether you're still bleeding all over the show and what parts of you are hidden or what's covered. While you're giving birth you don't care what you're showing but, once the [baby is] out the modesty starts kicking in... You've never had to sit there completely topless in front of everyone before and try and carry on as normal. When you start, you can't, you've got to flop everything out and it's a major mission to get it in the baby's mouth, it's hardly discrete...

It [breastfeeding] just made me uncomfortable but I'm finding now that most people realise that and they'll turn away while you're trying to get yourself sorted... Especially that first month when you're trying to get to know how to do it and it's horrible to have someone sitting there watching you, it's like being on stage naked... I find men will walk out of the room, but other women will watch especially if they've had babies. (Emma. Int. 14. Para. 21-31).

This experience of breastfeeding exposure was also shared by Debbie, who spoke of how difficult it was to be observed while struggling to breastfeed. She believed that having another bear witness to her struggle allowed others insight into her inability to cope with breastfeeding. Her way of being-in-the-world did not allow her to have others witness her struggling, because it meant she was not achieving what she perceived to be an appropriate standard of parenting. Debbie wanted to veil from others her experience of breastfeeding her son as her struggle spoke of failure as a parent.

It was really embarrassing because he wouldn't latch on so it was taking me five to ten minutes to get him on each time. He'd always go on but that time, especially when you had someone there, which you do when you first have them because there's all these visitors, and you're trying to latch him on and hide yourself as well and he's screaming... My dad's wife, came in and helped me latch him on which was really good and I had a girlfriend there, who's a good friend, but it was hard having her watch me struggle so much especially because I didn't know what I was doing either, my baby and I didn't know what we were doing. So I found that really hard... I wanted everyone to go away because I was not coping. Everyone would just sit and watch you and even if I went to the bedroom I would have people follow me in and it was like 'go away' cause it was such a struggle... But not the struggling part. I guess it's the whole good parenting stigma - you can cope and that sort of thing. I wasn't coping at all because I burst into tears every time I latched him on, it was so sore because he had made such a mess of them. Each time I'd put him on I'd just about scream and it's hard to keep a smiley face in front of your visitors when you're doing that. (Debbie. Int. 15. Para. 33-39 & 46).

Debbie described her embodied experience with breastfeeding as involving not only embarrassment but also pain. The experience of breastfeeding as painful was a recurring theme throughout the participants' narratives.

Breastfeeding as an Experience with Pain 8.6

In a world without pain we feel light, and participation in life is effortless (Gadamer, 1996); however, for the majority of women who participated in this study, breastfeeding was an experience with pain in the form of engorgement, as well as sore and cracked nipples. According to Gadamer (1996, p. 75) pain alienates one from him/herself: "pain causes us to withdraw from all external experience of the world and turn us back upon ourselves". Breastfeeding pain meant that the women turned back in upon themselves, withdrawing from external social experiences as a result. Sometimes, because of the severity of the pain experienced, some women brought their breastfeeding experience to an end. For others, breastfeeding pain became something to conquer and overcome. Irrespective of the outcome, pain is always an embodied experience that has meaning, that forms the basis of interpretations offered by the women.

For some of the participants in this study, their first experience of breastfeeding pain was associated with engorgement, a process that most commonly occurs in the initial few days of breastfeeding. The breasts swell due to increased blood supply and the production of milk (Lawrence & Lawrence, 1999; Renfrew, Woolridge, & Ross-McGill, 2000). It is reported that engorgement is a time-limited event that is manageable by frequent infant feeding, manual expression of breastmilk and the application of ice packs (Lawrence & Lawrence, 1999). However, these descriptions of engorgement do not relay how it is to be the woman undergoing the embodied experience of engorgement. The following narratives provide some insight into this experience.

For Cathy, engorgement was described using the words *blown-up*, *balloon*, *uncomfortable*, *sore* and *awful*. She also spoke of how being engorged made latching her daughter onto her breast extremely difficult.

The third day was absolutely awful because you're blown up like a balloon (laugh) and they're underneath your arms, and they're meeting together in the middle, and they're just so sore and uncomfortable and you just feel awful. And, because they were so engorged she couldn't latch on. The Lactation Consultant came in and spoke to me for a while and showed me what to do but that was very painful because I was so engorged.

(Cathy. Int. 18. Para. 8).

Kate described her experience attempting to feed her infant when her breasts were engorged with the following narrative, in which she explains why her embodied experience of engorgement was the lowest moment of breastfeeding her son.

Trying to get a baby latched on an engorged breast - for god's sake - I heard someone say it's like trying to get your lips around a brick wall.

My milk kicked in when I got home [from hospital], now I was not at all prepared, I thought I was getting mastitis. I knew I would come home and you'd be all blown up but I wasn't prepared for the lumps or for it to be so sore. The first night home with the baby he was up all night basically. I must've been up just about every hour. He [my son] was suckling at one end and just pouring out the other, which apparently

does happen sometimes when the [breastmilk] goes from colostrum to milk. I didn't realise that, and the next morning I just thought I don't want to do this, I can't. I was staring down the next 20 years of looking after this child and it was just going to leave me feeling like I felt that next morning. I thought well why couldn't they keep me in hospital just one more night so I could just have someone there to say that this is normal, don't worry about this.

(Kate. Int. 3. Para. 44 & 50).

Pain was also a part of Kay's experience with breastfeeding. She relayed how on the fourth day after giving birth she wanted to stop breastfeeding, but then learned the best way of reducing the pain related to engorgement and blocked ducts was to feed her infant. She realised that not only was she responsible for feeding her baby, she was also responsible for healing herself.

I was in pain. I was in agony. I had blocked ducts as well and that's quite painful. And the best way to get rid of it is to get the baby to feed or by expressing, but it was so painful, expressing really hurt me. The baby actually made it better but you had to help yourself, the ducts are all backed up and you have to massage to try and clear them out, and it was really hard. But I was lucky, I didn't go through what a lot of women go through with cracked, sore, red, ripped nipples... I was lucky on that side of it.

(Kay. Int. 8. Para. 10).

Later in the interview, Kay returned to the pain she was experiencing on that fourth day post-delivery because of being engorged, and spoke of how difficult it was at the time to see past the moment, because her embodied experience was that of being emotionally overwhelmed, weary and in agony.

At that time all I could think about was the pain in myself. You [are] not thinking rationally, because everything that you've gone through it all catches up to you... all you can think about is right there and the pain and the agony... deep down I really didn't want to put him on the bottle. He didn't go on formula in the end and I'm really glad he didn't. (Kay. Int. 8. Para. 34).

In western society we have come to understand that pain can be managed and resolved by accessing medical assistance. However, the pain associated with breastfeeding is not always easily resolved, and primarily requires a woman to take action to heal herself as highlighted in the above narrative from Kay. Unfortunately, this knowledge is not always available to women; instead, they turn to that which they have previously understood to provide a solution – treatment by health care professionals. Gadamer (1996) points out that uncertainty about the duration of pain results in suffering, eventually leading to a change in character as a person attempts to cope. This is a concept that is revealed in women's narratives about pain that persists over time.

The nipple is the most sensitive part of the breast, and therefore is more susceptible to producing painful stimulus when the integrity of the skin is impaired (Lawrence & Lawrence, 1999). When a woman initiates breastfeeding it is not unusual to experience pain, especially if the infant's positioning at the breast is less than optimal (Page et al., 2003; Renfrew et al., 2000). All the women who participated in this study spoke of how, at some point, they experienced pain associated with breastfeeding, most commonly associated with sore and cracked nipples.

Lee had attended antenatal breastfeeding classes prior to giving birth, so that she would not have difficulty breastfeeding. She had gained an understanding that positioning was intimately linked with cracked nipples and the pain associated with breastfeeding; however, the first time she fed, she discovered that the information she had received was not correct, and thus she felt betrayed by her educators and friends. This was the first step in beginning to understand that she had to take ownership of healing herself.

Despite her cautious positioning, Lee developed significant cracks in her left nipple, leaving her stunned that her efforts at prevention had failed. She employed nipple shields to reduce the pain associated with latching, but was castigated by her midwife for this. Lee spoke of how she withdrew from the advice offered by the midwife, choosing instead to inwardly endure the pain using the only thing that gave her relief – a nipple shield. She discovered that she was capable of persevering through breastfeeding pain; when the cracks finally healed, she was able to enjoy the experience of breastfeeding her son, which she continued for 15 months.

I felt that I was quite well prepared but the first feed was very, very uncomfortable, and my nipples were really sore. I was really surprised how hard he [my son] sucked and how sore it was. I was surprised by that because I thought if you had the baby properly positioned it didn't hurt... From what I'd heard painful feeding only came through bad positioning. I had my friend and the midwife helping with the positioning and checking it and everything like that. They both said that the positioning was fine so I felt slightly cheated that it [breastfeeding] was sore.

My left nipple got really cracked really badly. I'd tried so hard to get good positioning and gone to the class and learnt all about it. Every time I fed him [my son], I thought all the things that they tell you; chest to chest, chin to breast and checked it so carefully. I think it [the nipple] got cracked through him being latched on and pulling away and shaking his head at the same time. The crack was really bad: it was about half way around the nipple, and a couple of millimetres in and it was weepy and horrible and so when he was about a week old I went and got some breast shields. And I found that I was feeding well with them again. I guess it was a combination of the breast shields, it wasn't so painful because it was really painful with him latching on with the cracked nipple, and also the milk sort of settling down to a normal supply after being engorged. So then I fed through the breast shield on the left side for about four weeks. I remember one day the midwife came around and I was feeding with the breast shield and she said "that has to go, the baby won't be getting enough milk". And I just looked at her and I just thought 'you are mad!' (Laugh). There was no way I was giving it away, it was just so painful feeding without it. I thought 'I can't cope without it and he was feeding fine with it and I feel fine and he was thriving, he wasn't losing weight or anything, so I'm not going to worry about it'.

Well gradually the crack healed. I remember the first time that I fed without the shield, it had healed up quite a bit but it gaped open that very first time. I looked at it and I was so disappointed, you know, because I thought it was healing really nicely and that I was going to be able to feed without the shield. The only hassle feeding with the shields is that it's a real pain putting them on because you have to use K-Y Jelly and

you've got to boil them after every feed. The baby would wake up and he'd be crying and all of a sudden you've got all this fiddle, so that was a pain. So I was really glad when the crack healed up and the soreness stopped. Even on the right side that wasn't cracked I found the nipple really tender especially the first, sort of, ten sucks of the feed would be very uncomfortable. I'd be, sort of, sitting there gritting my teeth. But as soon as that all settled down I was really happy with feeding. (Lee. Int. 10. Para. 12, 22 & 24).

Gadamer (1996) describes pain as an oppressive weight that forces us down, dragging us under. This sense of being dragged down by breastfeeding pain prevented some of the participants from continuing with breastfeeding their infant. Liz experienced a great deal of pain due to cracked nipples after a few days of feeding her new daughter. She spoke of pain and bleeding at each feed, and even her mother's coaching did not help her cope.

The nipples were still incredibly sore, both of them, and the left one in particular bled quite a lot, and that was horrible. [My daughter would] start to feed and I'd be in agony. Mum would say "Count to twenty" and I'd be up to 30, going "31, 32, 33" and then you'd look down and there'd be blood all through it [the breastmilk]. That was all a bit horrendous. (Liz. Int. 9. Para. 16).

As described above, feeding her daughter was a painful feat for Liz, made especially difficult because she had flat nipples and her daughter would not latch properly. Liz spoke of how fearful she was of the pain associated with latching. Breastfeeding pain was a deterrent to breastfeeding; it meant that life ceased to be the light and effortless experience it had previously been, and stopped her from enjoying contact with her new daughter. Liz struggled to overcome the pain by using nipple shields, and was eventually able to feed her daughter without them.

[The breasts] they're so hard. She [my daughter] just wasn't having a bar of it. And the [midwives] said it was a combination of my [flat] nipples and her latching. And if she'd been a better latcher it would have pulled it [the nipples] out and it would've been fine but the combination of the two meant that it was really an uphill battle... I was scared of pain, and it is, a slight barrier to [breastfeeding]. I found that latching on every

time I would be tense because I was prepared for a lot of pain. It was an ordeal, especially in the initial minute. You'd be sort of thinking 'Oh my god, oh my god this hurts' so I didn't really have a chance to enjoy the skin-to-skin contact. I was thrilled that I'd passed that hurdle because when I left the hospital I thought I would never be able to feed, and well I'd be expressing and it would be bottles and expressing and trauma. So there was a series of steps and another step was getting rid of the nipple shields and wow, fantastic, I've got rid of them and it's such a big leap forward and it's only been two weeks.

(Liz. Int. 9. Para. 16 & 18).

Cathy had weaned her previous child early and was determined to breastfeed her new daughter; however, she described her embodied experience of breastfeeding her daughter as excruciating – more painful than having mastitis. She came to understand breastfeeding as intolerable, but felt trapped by her desperation to breastfeed.

For Cathy, life with breastfeeding meant being forced to carry an oppressive weight that she was unable to cope with, and eventually her embodied experience of pain overwhelmed her desire to continue. When she felt she could no longer cope, she spoke about weaning to her husband, who suggested that she just needed to persevere. In the face of what she had already struggled through, her husband's response was crippling; Cathy's breastfeeding experience had stripped her of hope and undermined her ability to mother.

She [my daughter] was quite unsettled and I was in excruciating pain and I would have to have anything, a toy, or my clothes to bite on to try and stop, dull the pain before I would latch her on again... The day that I got it, mastitis is very, very painful but the thought of having to have her latched on to me was even more upsetting, more painful than the mastitis. So it was probably the third or fourth week I finally talked to my husband, you don't talk much in the first four weeks (laugh). And I finally talked to my husband I was just in tears about having to carry on feeding because I was in so much pain from her not latching on and coming off all the time. He wasn't too supportive. He really wanted me to feed her since I hadn't really fed our son. He thought maybe if I could keep going, because what's what he would do - be a brave hero - he was

saying that he'd be able to carry on feeding. So that was really hard not having that support from him. I [would] just like for him to say "Well look if it's too sore maybe you can stop." I didn't want to stop, I would have loved to carry on feeding but I wanted to know that I had that support from him if I did. By the fifth week I was still in so much pain... (Cathy. Int. 18. Para. 8).

Eva was an educated and articulate woman, who held a Master Degree and was a valued professional in the firm in which she was employed. Giving birth to her daughter was a completely new experience for her, and she read widely to gain knowledge about all aspects of mothering. Breastfeeding was an important issue for Eva, and she was committed to breastfeeding her baby; unfortunately, her reading did not prepare her for what was to come.

Eva's experience with breastfeeding was an embodied experience of pain. Her daughter was three months old at the time of the interview and Eva was still experiencing significant pain every time she fed her infant. Thus, she was able to describe her pain in great detail. Eva shared that she had flat nipples prior to breastfeeding, which made latching her daughter onto her breast difficult, but in persevering she managed to start breastfeeding her infant. However, she soon developed cracks in her nipples, followed by mastitis one week post-delivery, ending up in hospital on intravenous antibiotics.

Eva likened breastfeeding pain to the pain of giving birth and called on techniques normally associated with birthing to cope. She experienced pain as described by Gadamer (1996) – pain that caused her to feel alienated from herself; despite this, she found the strength to continue. Eva relays her experience of the pain associated with breastfeeding in the following paragraphs.

I got terrible cracks, terrible cracks [in my nipples], and that led to an infection, which led to a mastitis, which led to me going back into hospital after a week with huge engorgement, a huge infection and horrible, like really bad cracks. The whole thing was nightmarish and extremely painful, there would be days when she [my daughter] would latch on and I would be crying with pain, it was so bad and the midwives didn't really know what to do. Because they were really keen for me to breastfeed, they kept saying "breast is best and blah, blah, blah..." but

then they also had me, their client, sitting there in tears saying "it's so painful, I don't know if I can do it anymore." It was a dilemma for them how to deal with me and it was a dilemma for me as to whether to continue feeding or not. Except that right throughout I felt that I wanted to continue unless it got excruciatingly bad, which it almost did, but I kept going. It's a story of triumph over pain in a sense.

The first eight weeks were really, really, really hard. It [breastfeeding] takes practice and time, and I was in great pain as well. It was an interesting thing because once the pain of the previous feed wore off, it was a little bit like the pain of birth, it fades very quickly, so that by the time the next feed came around I was ready to feed again because I'd almost forgotten the pain from the last time. It is quite an interesting phenomenon for me. But it was never so painful that I didn't want to feed her the next time.

It [breastfeeding] felt like a really personal thing between the two of us that I wanted to work at, and sure it was painful... I just so badly, so badly wanted to be able do it for the baby, that I was able to work through the pain. I had techniques to get through it. I have a crystal here that a friend gave me and every time before I would latch her on I would take a moment to look at the crystal and take a big deep breath and then I'd pop the baby on and I'd breath through the pain, and that was every feed. And, imagine that you're feeding six times a day, that's quite a lot of feeding and crystal watching (laugh). I guess it's like giving birth, the benefit that you get at the end of the process outweighs the pain that you have during it.

There was one day, when it was a turning point, where I almost gave up but I had a fabulous midwife. She, [the baby], was about four weeks old. And I was in tears sitting here feeding, it was dreadful and she [the midwife] could see that I was mentally and physically exhausted, I had lost quite a bit of weight and she goes "Right, you have a choice here. Whatever choice you make you just have to be happy with it, and have to go with it and don't look back. You can choose to feed and put up with the pain or just accept that there is going to be some pain and be happy with that - or start weaning her off and go onto the bottle and

being happy with that." She gave me the weekend to think about it. She also gave me a mid-way option, which was to feed a little bit then after a certain time stop feeding, and give her a bottle to top her up. So therefore, I could have a rest and the baby could have as much food as she needed to feel not hungry, that was quite good. We did that for a week and I started to feel much better, a lot stronger, I'd had a rest and my nipples were healing slightly. I really appreciated the fact that she [midwife] didn't say "you have to do this." She gave me some options and wasn't judgmental about it, which is important because there's quite a lot of pressure on women to breastfeed and if they don't they're considered a failure. So what this woman was saying was "it's OK not to breastfeed, it's OK to bottle-feed, breast is ideal but it's OK to bottle-feed and you're not going to be damaging the child." Just somehow the lifting of that pressure at that very crucial time made me stop and think and made me realise that I did want to continue breastfeeding but that there was an option, if it did get ever too painful I could bottle-feed her. I got better, the nipples got a little bit better - they're still a little bit painful now, but everything took a turn for good.

The other thing that the midwife said on that day was if you're breastfeeding and not enjoying it, and you're not happy with it, it's like having someone over for dinner and not actually wanting them to be there. I thought that was a very good analogy because when you go to someone's house and you don't feel quite welcome it's a bit edgy and there's tension, you know that there's something under the surface and you don't feel happy and comfortable. I imagine that was what it must have been like for my daughter, latching on but feeling all that tension, because I was in agony. So she [the midwife] said if you do decide to breastfeed you have to be happy about it and go with it and that's what I decided to do. I decided I wanted to try and continue.

(Eva. Int. 19. Para. 6, 7, 9 & 12).

Eva persevered and was still exclusively breastfeeding her daughter when she was six months of age. She chose to attend a post-natal depression group and counselling as a result of her experience, and commented that these interventions greatly assisted her in dealing with her unresolved, negative feelings about the pain she experienced as a result of breastfeeding.

"Merleau-Ponty shows how any sensing already recognises a set of meanings, in so far as part of the perceptual field leaps to the foreground as the centre of significance" (Leder, 1984, p. 31). The experience of breastfeeding for the majority of women is an experience with pain (Blair et al., 2003; Hill et al., 1994; Page et al., 2003; Salmon, 1994). However, many of the participants in this study were disconcerted by their pain; it leapt to the foreground, alerting them that their experience was incongruous with their previously-held understandings about breastfeeding as a natural and easy process.

According to Merleau-Ponty (1981), pain always has an embodied origin that invariably has meaning and signifies an attitudinal position towards the world. For those participants who spoke of breastfeeding pain, it meant unexpected injury and assault – something to be endured so they could offer their infants the best available nutrition. Their narratives speak of the great lengths to which a woman will go, and the sacrifices she will make, for her infant. However, there comes a time when, for some women, the pain becomes too much to bear: "pain makes me give way... We all know the moment at which we decide no longer to endure pain... and when pain becomes intolerable" (Merleau-Ponty, 1981, p.441).

In addition to experiencing pain when they breastfed, many of the women in this study also experienced tiredness. Both their pain and their fatigue were significant deterrents to breastfeeding, at times undermining their determination to continue.

8.7 Breastfeeding as an Experience with Tiredness

New mothers frequently relay how tired they are. New babies have differing nutritional requirements from adults, which often results in an infant needing to be fed overnight. This is especially true for breastfed infants, as breastmilk is digested easily, usually being fully digested in one and a half hours (Lawrence & Lawrence, 1999). This means that an infant may wake to feed several times during the night, leading to a disruptive sleep pattern that can be viewed negatively by breastfeeding women. Tired, weary, and fatigued were words that the majority of participants used to describe their embodied experience of being a breastfeeding mother. Debbie struggled to survive the initial period of being a

new mother, and spoke of how sleeping was intrinsically linked to coping with her new role.

Yes, I think 95% of my entire problems were due to lack of sleep. I didn't sleep after he [my son] was born for about three days and at the end of that I'd just sleep for an hour or so and then I'd be up again. I think a lot of my problems were due to a lack of sleep with breastfeeding, because I wasn't coping when he wouldn't go straight on, I'd just burst into tears. Whereas if I'd had a good night's sleep I probably would've coped a lot better with it.

(Debbie. Int. 15. Para. 84).

At the time of the interview, Kate's baby was seven months old and, although she had recovered from the engorgement and resolved her transient feelings about the daunting nature of mothering, she went on to speak of how physically demanding breastfeeding is in terms of the embodied effort of making enough milk to feed a healthy infant.

[Breastfeeding] is demanding... he [my son] sleeps really well but there's an incredible sense of tiredness the whole time and all I can put it down to is the breastfeeding. By the end of the day I'm still really tired and I think it must be producing [breastmilk], he's growing and growing like a weed, just that whole production must take it out of you. It would be nice to have some energy back but it's such a drop in the ocean in the scheme of things. It's [breastfeeding] just so worth it... I think that a lot of people would not want the expense of formulas and mucking around with mixing them up and heating them up in the middle of the night where everything's the right temperature - it's all ready to go. (Kate. Int. 3. Para. 74 & 76).

Sally was breastfeeding her fourteen month old son at the time of the interview, and spoke of how family and friends had tried to convince her to cease breastfeeding to ease her tiredness. Despite her fatigue, breastfeeding helped to keep her going when she was feeling emotionally and physically drained. Sally was able to continue breastfeeding, as it was the one thing that kept her going through the tiredness.

Just before Christmas I got really tired, really, really tired. I've had glandular fever and I think it was a bit of a relapse. My mum was really concerned and she tried to convince me to wean him then for me, for my sake. There have been times when I've been exhausted, you get exhausted as a mother anyway, but I think I have been more exhausted breastfeeding because of the physical making of the milk – [friends and family] they've been concerned for me to stop. But when I'm really tired and down the only thing that keeps me going is actually breastfeeding. I need to breastfeed to buoy me up. So it would be a vicious circle if, I stopped breastfeeding I wouldn't be able to buoy myself up. It's kind-of a balancing game.

(Sally. Int. 11. Para. 104).

Just as pain has its origin within a person, and expresses an attitude about the world, so too does fatigue; thus, fatigue reveals its significance to the person experiencing weariness (Merleau-Ponty, 1981). For some of the women who participated in this study, tiredness forced a pause in their breastfeeding journey, mirroring Merleau-Ponty's (1981) comment that fatigue makes one break one's journey. It may be that for a moment a woman is so weary that she decides that breastfeeding is not for her, only to take up breastfeeding a moment later. For others fatigue spells the end of the road.

When Mel's son was ill she weaned him because she was so tired from constantly getting up and feeding him two times during the night. She described her unease about night-time when she desperately longed to sleep but knew it was not possible. She spoke knowingly about what it was to be mother-tired, relaying how important it is to know that other women experience similar feelings.

When he [my son] got strep throat, at nine months, I gave it [breastfeeding] up because I was getting up twice in the night and I was so tired, at least if I gave him a bottle I could get up once and my husband could get up the second time.

When you're a mother you can cope during the day, and the day is fine, but... when the night starts coming in you get butterflies in your stomach because you know it's your sleep time and this is when you want your baby to sleep. It would be so much nicer if a mother could stay awake

for forty eight hours, but it doesn't work that way. It was really nice to talk [to my friend] about it and it made her feel more relaxed and it made me realise that other people went through the same thing but you don't discuss those sorts of feelings with other people.

(Mel. Int. 17. Para. 40 & 44).

There comes a moment when a woman decides she can no longer withstand the fatigue, and as with pain, it becomes overpowering. "Tiredness does not halt my companion, because he likes the clamminess of his body, the heat of the road and sun, in short, because he likes to feel himself in the midst of things, to feel their rays converging upon him, to be the cynose of all this light, and an object of touch for the earth's crust. My own fatigue brings me to a halt because I dislike it, because I have chosen differently my manner of being in the world... I am free in relation to fatigue to precisely the extent I am free in relation to my being in the world, free to make my way by transforming it" (Merleau-Ponty, 1981, p. 441).

Being tired was a part of the embodied experience of breastfeeding, at times deterring women from continuing to breastfeed their babies. Another aspect of the embodied experience of breastfeeding that held significant meaning for the participants was that of being anxious.

8.8 Being Anxious

We exist as a body 'within-the-lifeworld'; therefore, all experiences are embodied (Dahlberg, et al., 2001; Merleau-Ponty, 1962). For Merleau-Ponty (1962) the lifeworld is that which we have access to through our bodies; all experience and sensing that occurs as a result of our bodies being in the lifeworld, such as anxiousness²⁵, as well as inauthentic²⁶ fear and guilt, are embodied experiences. The following section explores how women experience anxiety ontologically, as well as inauthentic fear and guilt on the ontic level. Prior to returning to the participants' narratives, it is worth exploring the

²⁵ Anxiety and anxiousness are used interchangeable in the following sections.

²⁶ Refer to Chapter Four, Section 4.4.1 for a discussion of Heidegger's (1962) understanding of *dasein's* experience of authentic and inauthentic states.

philosophical and theoretical understanding of how *dasein* exists as 'being anxious'.

Dasein exists in its primordial state in nothingness; in this state, dasein may not have formed any interpretations, but it does always exist as interpreting. To flee the anxiety caused by the nothingness, dasein inauthentically embraces the interpretations that already exist within the lifeworld into which it is 'thrown'. If dasein does not embrace these pre-existing interpretations, it experiences anxiety, which Heidegger describes as the fear of nothingness (Heidegger, 1962; Gadamer, 1996). "Anxiety is intimately connected with an oppressive sense of constriction, with sudden exposure to the vastness and strangeness of the world" (Gadamer, 1996, p.153). Dasein desires to be free of anxiety, and therefore flees into inauthenticity, to a place of familiarity, a place where it feels at home, that is 'within-the-world' or within the lifeworld. However, dasein is cognisant that the interpretations 'within-the-world' are not true to dasein's essential nature; in consequence, dasein also becomes anxious because it does not feel at ease with these interpretations.

Thus, *dasein* experiences anxiety in a state of nothingness, and also when accepting the interpretations present in the lifeworld. Therefore, one of the fundamental ontological modes of *dasein* is anxiousness (Heidegger, 1962; Inwood, 1997; Inwood, 1999). As a result, when anxiety emerges, *dasein's* 'thrownness' and existence are disclosed (Heidegger, 1962; Inwood, 1997).

Anxiety is always anxiety about something (Gadamer, 1995). The women who heeded the call to explore their anxiousness questioned their existence as women, and mothers, as they examined the roles the world stated they should adhere to. The women in this study also experienced inauthentic anxiety, in the form of fear about their ability or inability to breastfeed their infants. When dasein dwells in this inauthentic 'within-the-world' mode, dasein becomes mindful of failing to live up to the perfect ideal, and therefore experiences guilt. Terms such as emotionally crippled were employed by the participants to describe their anxiousness and guilt in relation to the issue of breastfeeding.

Fear is fear of something 'within-the-world', within the realm of our thrownness; fear is always an intra-world experience (Heidegger, 1962; Inwood, 1999). As fear stems from something that resides 'within-the-lifeworld', Heidegger refers to fear as the inauthentic counterpart of anxiety (Heidegger, 1962; Inwood, 1997).

"Fear is not just an inner feeling; it opens up a world of potential threats" (Inwood, 1999, p. 16); dasein fears consequences. For the women in this study, fear was produced in anticipation of being unable to breastfeed.

8.8.1 Anxiety and the Fear of Failing

In New Zealand society, breastfeeding holds a privileged position above other forms of infant feeding, with breastmilk portrayed by 'the they'27 as easily produced and delivered. Therefore, fear of failure is experienced when faced with not being able to produce breastmilk. As the infant is most important to a mother, not being able to satisfy the infant's needs is interpreted as failure. Women interpret failing at breastfeeding to mean they are not good mothers, and as a result their infants will no longer need or want them.

For Cathy, breastfeeding was an integral part of being a mother, but she only managed to breastfeed her son for nine weeks. She attributed her difficulty breastfeeding her son to the early introduction of a bottle, following which her son always struggled with latching onto her breast. After giving birth to her second child, a daughter, the attending midwife had managed to facilitate a successful breastfeed, which left Cathy greatly relieved and hoping for a more successful breastfeeding experience. However, the day after that first feed, Cathy was in pain from feeding, her daughter would not settle, and the midwife suggested giving her a bottle. Because of her previous experience with her son, Cathy was hesitant to follow the midwife's advice. Cathy faced a difficult decision as her fear was not pain from breastfeeding; rather, it was fear that her daughter would reject her breastmilk, and reject her as a mother, if she was given a bottle.

I had heard her [my daughter] screaming for hours, she had hardly slept, and I was already bleeding [from my nipples] after that first one and a half days because she was feeding so much and there probably wasn't that much [breastmilk] there for her. Also she had [managed to] latch on [to my breast]; just that fear that if she had a bottle maybe she wouldn't want me [my breast] again, she wouldn't want me either. (Cathy. Int. 18. Para. 41).

²⁷ Refer to Chapter Six for a description of 'the they'.

Eva questioned her very existence when her daughter was three weeks old and she was struggling with the decision whether or not to continue breastfeeding. She spoke of how her extreme fatigue prevented her from thinking clearly, and thus she was more susceptible to feelings of failure and guilt.

When I was having that crisis about whether to stop breastfeeding and go onto the bottle or not, I was in tears at the idea of having to give up because I so badly wanted to do it. I also went through a time when I thought if I can't breastfeed her I'm not being a very good mother, I remember that quite clearly, and feeling really terrible about that. In fact it is not true, and I realise now, there's heaps of things I can do for her as a mother and no one's going to replace me whether I breastfeed or not. But, at the time feelings were heightened so much, and I was exhausted and a bit sick as well so it was like 'what's the point in my being here if I can't breastfeed?'

(Eva. Int. 19. Para. 73).

For Sam, fear emerged in the experience of being at home alone for the first time with her newborn son following their discharge from hospital. Due to a family history of milk allergies, she did not believe that formula was an option for her infant; however, she struggled to breastfeed, and in her desperation sought help from sources, such as La Leche League, that she considered she would not normally have resorted to.

So we were on our own for the first time completely, it was so daunting. I think it got to about 8.30 at night and we [my husband and I] couldn't get him [our son] on [to my breast] and I just didn't know what to do. At that stage we hadn't used any formula because I had a history of milk allergy in my family and so I was scared to even try formula in case he reacted badly to that... I thought 'My god, I can't feed my baby at all, what's going to happen?' I didn't know what to do, who to ring - I was looking in the phone book under the La Leche League, and things like that, to try to get someone to help.

(Sam. Int. 12. Para. 16).

Mel clearly articulated how she felt during the time leading up to the transition from breastfeeding to formula feeding her daughter. She was desperate in her attempt to continue breastfeeding because she was afraid that otherwise her daughter might die from cot death, likening having to stop breastfeeding to a train coming towards her in a tunnel. She spoke of how she just wanted someone to let her know her baby would be safe on formula. Three years after giving birth, she still recalled vividly how it was to be in her metaphorical tunnel.

In your mind you really want to breastfeed, you really want to breastfeed. It's like the light at the end of the tunnel, when you can see that you might not be able to do it you're panicking - you try everything, you're grasping at straws, and you get quite emotional. It's like you can see the light coming towards you and you know that you're going to have to stop. And people are saying "try this, don't give up, we will support you either way." But, oh god why don't you just tell me "it is alright, it is ok to go to the bottle." You hear people say "cot death" and "the baby's a lot better if you breastfeed them", all this stuff is wiggling around in your head. It would be nice if someone said "look don't worry your baby will be fine, there's nothing wrong with formula, it's perfect," nobody told me that. I remember I was so emotional even now I can feel quite hyped because I can feel all the emotions that whiz through you.

With my daughter I wanted to keep going, I'd feed her in the morning and then give her a top up or I wouldn't give her a top up [of formula] because I wanted to have fed her properly myself. I'd know that she might be waking a bit earlier and I kept trying to do it myself, I really wanted to keep going [with breastfeeding], but it was overwhelming. I knew that I'd have to give up, I really wanted to keep going but in the back of your mind is someone saying 'it's not working, it's not working.' It's horrible.

(Mel. Int. 17. Para. 16 & 52).

These narratives demonstrate women's fear that they will fail to breastfeed their infants. Some of the participants managed to overcome their initial breastfeeding difficulties, leaving their fear behind; but for some women, their fear of being unable to continue breastfeeding their infant became a reality. For these women, breastfeeding became an experience of guilt because they could no longer provide their infants with what they believed was best.

8.8.2 **Guilt**

Just as dasein exists as essentially anxious, dasein also exists as essentially guilty, meaning that dasein is unable to achieve infinite possibilities of being; therefore, because of guilt, dasein exists in a deficient state (Heidegger, 1962). To escape the unsettledness of essential guilt, dasein withdraws into the inauthenticity of everyday guilt (Dreyfus, 1991; Heidegger, 1962). Heidegger defines everyday guilt as a "lacking in some way... this kind of lacking is a failure to satisfy some requirement which applies to one's existent Being-with-Others" (Heidegger, 1962, p. 328). Inauthentic dasein falls under 'the they's' influence, adopts its interpretations, and thus experiences guilt in accordance with what dasein is told to feel guilt about (Heidegger, 1962; Inwood, 1999). This is analogous to dasein fleeing essential anxiety and submerging itself under the influence of 'the they', leading to inauthentic anxiety. Those women who participated in this study and were unable to continue breastfeeding experienced everyday guilt because they were influenced by 'the they', which advocates breastfeeding as the infant feeding method of choice. Their inability to breastfeed was interpreted as a failure to live up to 'the they's' expectations, leading to guilt.

Mel struggled through prolonged engorgement and continuous mastitis during the six weeks she tried to breastfeed her son. During that time in her life, she had similar experiences to other mothers who struggled with the decision of whether or not to continue breastfeeding. Mel questioned her ability to be a good mother if she ceased to breastfeed, but in the end the decision was made for her, as she lay in a hospital bed receiving intravenous antibiotics for severe mastitis.

People kept saying "just keep going [with breastfeeding] and it will go OK" but we should have stopped heaps earlier. We shouldn't have got to such a state where I couldn't stand up, I couldn't eat - I couldn't do anything. I was useless to everyone including my son. I felt terribly, terribly guilty about not being able to breastfeed. And you know you get all this stuff shoved at you all the time about how breastfeeding is best and you almost feel like you're not a good parent if you can't. There's all that baggage and I kept thinking he's doing this to me and I'm trying my best, and we're not, we're not getting it.

(Mel. Int. 13. Para. 8).

Mel went on to describe her embodied experience of guilt at having stopped breastfeeding. She spoke of the great difficulty she had had with her son, and how she did not like him because of the disturbance he had caused in her life, which caused her to suffer from additional guilt. Meanwhile, she was so worn out from her experience that she was unable to think clearly about the situation, and she concluded that she was not a good mother.

My midwife was great; she sent me along to the Plunket Family Centre and they were fantastic in their support. Like they talked a lot about bonding and I told them that I felt guilty. I was very emotional, incredibly tearful all the time about feeling inadequate. And so they [the Family Centre staff] were really good setting me up with baby massage classes earlier than I probably would normally have gone. I could go to the Family Centre at anytime. My son was very unsettled as well [because of being on antibiotics, which didn't help me. So they would take him for the day, probably once a fortnight, for quite some time and that was great. But the guilt lasted a long time probably about six months. I also felt quilty that I didn't like him [my son] that much, because I really didn't. He caused a lot of hassle, my husband was off work for six weeks and there was always a drama. I was so tired and I just couldn't cope. I was used to being someone completely in control and it was out of control, and this thing cries all the time and I can't comfort it. And then you're not a good mother... Your perception is that you're not a good mother on top of that.

(Mel. Int. 13. Para. 24).

Sue spoke of how she had let her son feed at the breast for long periods of time, and yet despite this she did not have any evidence that he was receiving any breastmilk. She did not understand why the health care professionals did not heed her concerns about her infant's health; therefore, her ability to listen to her embodied knowledge, and use it to nurture her infant, were undermined. Out of concern for her infant, she finally fed him formula, but subsequently suffered from feelings of guilt. Because she had provided her infant food that did not come from her body, she felt that she was not providing the 'best'. This feeling of guilt for inadequately providing for her son was reinforced by the health care professionals, who she believed were supposed to be there supporting her in her mothering role. So, to avoid being exposed to additional emotional 'hammering', she chose to not disclose the use of formula.

All day Friday he was feeding, feeding, feeding and I could hear his stomach rumbling. He didn't have a wet nappy all that day, and when I rang someone and said "this is really strange" they said "oh no that's fine". But I thought 'no, I don't think it is, he's dehydrating, he's not having any fluids go through him.' So I fed him formula on the Saturday. I had the guilts for a week about the formula. I was still breastfeeding him because I had the midwives coming in and they kept on saying "no everything's fine" and trying to get me to wean him off the formula but I just felt that he needed it.

They [the midwives] would come in and you almost felt like you had to hide the bottles and the formula, because they would ask how you were feeding him [my son] and you'd had it drummed into you that 'breast is best'. Then they'd watch you breastfeed, and everything was going well, he was latching on very well, but the moment you suggested that you were topping him up with formula they were sort of like: 'Yes well that's alright' but each time they came in they were like 'you're feeding him less formula aren't you?' That was practically their first question. So in the end I found it easier to say "yeah, yeah we've got him off the formula totally," which wasn't the truth but it was just easier. At home with a newborn baby, first time around and the last thing that you need is to be hammered from that quarter as well.

(Sue. Int. 6. Para. 20 & 22).

After deciding to stop breastfeeding her infant, Debbie's breasts, although permanently scarred, were able to heal. However, as a result of her decision to wean her son, she traded her physical suffering for the suffering of guilt. The cause of this guilt was the inability to provide her infant with milk that, in her perception, meant thriving and health. Debbie acknowledged that another foundational cause of the guilt she experienced was the concern that her son would no longer love her.

I'm quite happy not to be breastfeeding now, for a long time I felt really guilty about it because everyone else was. I felt really guilty for quite a while. For probably about another two months after I stopped I felt really guilty that I wasn't breastfeeding, but now I've seen that he can go so well on the formula that I'm OK with it. Whereas I thought 'he's going to get all these health problems and he's not going to thrive as well' and

he's been fine... You do feel guilty for a long time after you stop. Which you shouldn't do. If you can't do it it's OK, it's not bad to stop, your baby's not going to love you any less or anything...
(Debbie. Int. 15. Para. 119 & 123).

In the narratives discussed in this, and the previous sub-section, the participants expressed both fear and guilt when describing their breastfeeding experience. The inauthentic modes of anxiety and guilt were expressed in the form of a fear of failing their infants if they were unable to breastfeed, and guilt if they either failed to breastfeed, prematurely weaned their infants, or introduced artificial formula feeds. If they were unable to resolve the unsettledness resulting from their experience with breastfeeding, they felt emotionally crippled.

8.8.3 Emotionally Crippling

Some of the women who were subject to the embodied experiences of fear, failure and guilt in relation to their experience with breastfeeding were no longer able to cope with the strength of their emotions. This led them to employ such terms as *emotionally crippling* to ultimately describe their embodied experience of breastfeeding.

As previously mentioned, Mel's daughter suffered from severe reflux, which caused Mel to struggle to breastfeed for the first five weeks after giving birth, at which point she started to feed her infant expressed breastmilk. Due to a reducing breastmilk supply, Mel could no longer sustain her daughter's nutritional requirements by expressing, and by the time her daughter was ten weeks of age she was being fully formula fed. Mel's effort to continue breastfeeding was fuelled by a strong desire to provide her daughter with the ultimate source of nutrition, and she spoke of the emotionally crippling experience she had during the weaning process; of how her thoughts and decision making were influenced by others, and how she longed desperately for support from those around her. During the interview, Mel returned to this novella many times.

She [the baby] was on the Zantac for six months and Prepulsid till seven months and that was fine, perfect, we never had any feeding problems after that. She would sit up and she'd drink her bottle cause I could feed

her that way and that was it, wonderful. I don't know why somebody didn't tell me in the beginning to give up. It's really horrible because people are saving to you "breastfeeding is the best thing for your baby". but it's emotionally crippling for you to have to give it up. Or with the Plunket nurse saying 'I'll support you no matter what', that's really nice but I didn't want that, I wanted someone to say "it's not working, go to formula it's a lot easier, don't berate yourself over this and just don't worry about it", but they don't. They try and convince you to do otherwise. It's really hard, it's hard because you're trying to do it and you need their... they're supportive but they're on the edge, they're not in there helping you with your problem, not helping you deal with it, not telling you "well look if it's not working then give it up." Being a new mother you've got so many people giving you assistance and advice, without anyone saying 'look at it this way'. But I suppose they don't want to, they don't want you to give up so they have to let it be your choice. But it was crappy with my first daughter... When I went to the Family Centre for help they said to me "it won't make any difference if you give your child a bottle than if you breastfeed her." Why were they getting her to sit on my knees and breastfeed her upwards? To me that's an unnatural position. They should have been telling me that "you might have to give up breastfeeding", getting me prepared for the worst. You don't want to give up but you have to. If someone had prepared me maybe for having to give it up it would have been a lot easier, I wouldn't have persevered so much with express breastfeeding.

There are heaps of people out there with worse problems but you're absorbed in yourself and it was tough. I know it didn't last very long but those five weeks were definitely a long time in my life...

(Mel. Int. 17. Para. 2, 4, 18).

Sam's experience of breastfeeding her son was extremely distressing. She dreaded every feed and doubted her ability to resolve the breastfeeding difficulties she was encountering. Like many other mothers, she desperately wanted to continue for the sake of her son, but struggled to rise above her tiredness and find the strength she needed. Although her partner provided crucial support, she battled to overcome feelings of inadequacy in relation to her ability to mother because she could not fully breastfeed her son.

It [breastfeeding] was so hard and so stressful. Like every time he'd wake for a feed I'd think 'oh, here we go again'. There was nothing, nothing pleasurable about it at all. I wanted to make it work for all the reasons that you're taught; that you know, that breastfeeding is good for you and for your baby primarily. So I wanted to try and make it work, but as I said, there were several times when I was willing to flag it away, but I have a really supportive partner, he just was great. If it wasn't for him I think we would've given it away at two weeks... Even now at ten weeks if we get slightly out of that routine, if I try to do too much, then it becomes really difficult again. About once a week, if I can't get my son to settle I still give him 50 mls of formula just to get him down. I still kindof have an issue with that. I don't know if it's a threat to your sense of womanliness or your ability as a mother because you like to be able to feed your baby yourself completely. Sometimes it still does upset me if I can't, if I need to give him a top up with formula. But on the other hand I get worried if I don't think he's gaining enough weight. So it's kind-of a no-win situation really. (Participant has tears in her eyes while talking). (Sam. Int. 12. Para. 12 & 20).

Throughout the interview, Sam repeatedly returned to her inability to resolve her feelings of inadequacy as a mother because of the difficulties she had in breastfeeding. She talked of the constant worry about being unable to satisfy her young son's nutritional needs. However, Sam demonstrated how her definition of successful breastfeeding changed with every positive breakthrough she achieved, and thus she constantly doubted her embodied capabilities as a breastfeeding woman.

Sometimes I wonder if it [breastfeeding] was really worth it. (Participant was crying). I know that it is in terms of the nutritional value and all of that kind of thing, but the stress that I put myself through, I don't know whether it was worth it really... With the constant nagging, worry of whether my milk supply is adequate and whether he's doing OK, but it's fine. (Participant is crying). The more we progress though, the older he gets and the better his weight becomes the more secure I get in the knowledge that things are OK. We used to be giving him a comp [of formula] about three or four times a week, and now we're giving it to him once a week. Sometimes I think that I'll feel like I've finally succeeded when we get through say two weeks without having to give him a comp.

I don't know why that will be a milestone for me but it probably will, and that will be when I want to start (laugh) giving him a bottle anyway I suppose.

(Sam. Int. 12. Para. 56 & 58).

Emma vigorously questioned her ability to mother during the first month after she gave birth to her son. Despite her previous experience with young infants, she believed she was unable to care for her own. She laboured to breastfeed her son, letting him feed for hours on end, but believed he was only receiving insignificant amounts. For Emma, concerns arose with breastfeeding when she recognised her infant son was not gaining weight. She spoke to her midwife about her concerns only to have them dismissed and, therefore, silenced. Over a period of four weeks Emma had the growing realisation that all was not well, and despite attempting to get her concerns validated, became increasingly alarmed by her son's distress. Her son required constant vigilance because of his distress, which, combined with concerns that were not validated by health care professionals, resulted in Emma self-diagnosing depression. This depression resulted in her rejecting her son at four weeks postpartum, which she described as the worst month of her life.

I didn't have the problems with latching [my son] on or anything, we latched on perfectly not a problem. It was just the whole amount of time that he was on me. I didn't know that I wasn't producing enough milk for him. He would be on me for like three hours then off for half an hour screaming then back on again for three hours continuously. No body pointed out to me "hey he's not getting enough." He had blood in his nappy and the midwife said "Oh no its OK, keep going" but I know now not to listen and to do what I feel is right. I actually said to her "I'm not producing enough am I?" and she was like "yes you are, yes you are just keep going."... He was losing a lot [of weight]. I actually resent the midwife for letting him go through that for so long and not doing anything about it. [It occurred over] four weeks. [My son] he would just scream continuously, I didn't have a break, I was just a mess by the end of it. Absolutely beside myself... [I was thinking] 'What the hell have I done?' I used to work at a kindergarten, in a day care centre, and I had heaps to do with babies so I was really confident about having him. I thought I knew what I was doing but when I couldn't stop him crying and I did not know why, I absolutely lost it. [I was thinking:] Why have I done this? I

shouldn't have brought a baby into the world if I didn't know how to look after it.' But, it's a lot better now because he sleeps and feeds properly. But no, that first month was just horrible. The worst experience of my life at that point, the first month.

(Emma. Int. 14. Para. 71-77, 81 & 83).

Jan suffered with a tremendous amount of guilt associated with weaning her son at three weeks, following her hospitalisation and surgery for severe mastitis. She spoke of how she was burdened with feelings of failure because of her inability to breastfeed, comfort, or be what she perceived to be a good mother. She questioned, four years after the experience, whether she was suffering from postnatal depression because of feelings of complete despair brought on by her experience with breastfeeding.

I was so tired and I just couldn't cope. I was used to being someone completely in control and it [the baby] was out of control, and this thing cries all the time and I can't comfort it. And then you're not a good mother... Your perception is that you're not a good mother on top of that... I may have been going to have PND (postnatal depression) anyway but I was never treated for it, but I had this... complete despair about breastfeeding.

(Jan. Int. 13. Para. 24 & 28).

The first time a woman breastfeeds her way of 'being-in-the-world' is transformed, because she begins a process of being-in-the-world in a different, as-yet un-experienced way. According to Gadamer (1975, p.111)

"transformation means that something is suddenly and as a whole something else, that this other transformed thing that it has become is its true being, in comparison with which its earlier being is nil. When we find someone transformed, we mean precisely this, that he [or she] has become another person as it were".

Following this transformation, previously-held meanings provide women with neither the interpretive framework nor the understandings they need to navigate their way through the embodied experience of breastfeeding. Thus, women find themselves in a place of unsettledness, and it is in this state that they encounter authentic anxiousness. In an attempt to be at ease 'within-the-world' in which

they are 'thrown', women flee to the world of 'the they'; to provide themselves with an interpretive framework, women choose to accept 'the they's' interpretations of what it is to be a mother, and the edict that all mothers can, and must, breastfeed²⁸. In accepting 'the they's' interpretations, women embrace the very ideals that produce the fear of failure, and feelings of guilt, and thus they become 'emotionally crippled'. This closes down on the possibilities that 'being-in-the-world' holds for a breastfeeding women.

Merleau-Ponty (1981, p. 441) points out that "our attitude towards the world, when it has received frequent confirmation, acquires a favoured status for us" (Merleau-Ponty, 1981, p. 441). The women in this study were free to adopt a certain embodied way of being in the lifeworld. They had a choice to either develop their own interpretations or accept 'the they's' interpretations of what it is to be a breastfeeding woman. The consequences of accepting the interpretations as offered by 'the they' meant they experienced inauthentic anxiety in relation to their breastfeeding experience.

The way in which a woman chooses to exist 'within-the-world' will come to the fore when she breastfeeds. Either she will adhere to inauthenticity and remain involved in the public world, or she will free herself from their interpretations and understandings, and attend to dasein's authentic anxiety. In the mode of authentic anxiousness, *dasein* sees that the meanings provided by 'the they' are just an illusion, and so the rules the world espouses are no longer seen as essential because they become meaningless (Dreyfus, 1991). Thus, one is returned to *dasein's* authentic ownmost, and the interpretative possibilities of being and associated embodiment are freed to emerge (Heidegger, 1962; Dreyfus, 1991).

8.9 Conclusion

When women initially discussed their experiences of breastfeeding they employed language embedded in notions of Cartesian dualism, an orientation inherited from centuries of approaching the body as an object. However, as

²⁸ See Chapter Six for a full discussion about participants' understandings of 'the they's' influence over breastfeeding interpretation.

they continued to relay their experience with breastfeeding, the language embedded in the subject-object discourse lost its relevance and they turned instead to a language that spoke of an embodied way of 'being-in-the-world'. This embodied experience of breastfeeding was a hidden experience.

For the women who participated in this study, breastfeeding was an experience with embarrassment, pain and fatigue. It was also about feeling inauthentic anxiousness in the form of fear of failing with breastfeeding. If a woman did fail to adhere to the average breastfeeding norm, as espoused by 'the they', she experienced guilt. If a woman was unable to resolve her feelings of guilt, she became emotionally crippled, and thus the possibilities of being were closed down upon. However, if women attend to their authentic embodied experiences, they gain access to a multitude of interpretive possibilities through which being is opened up. The opening up of being occurs because women respond to the call to care through breastfeeding, and it this experience that will be explored in the following chapter. This opening up of being through breastfeeding is explored in the following chapter, within the context of women's experience of solicitude.

Chapter Nine

Breastfeeding as Concernful Solicitude

9.1 Introduction

This chapter explores how women are called to care for their infants, and begins by attending to an exploration of Heidegger's concept of care. It goes on to highlight how women decide to breastfeed because they care deeply about their infants, eventually coming to more fully know and understand their infant's needs through breastfeeding. This chapter demonstrates that, as the relationship between a woman and her infant matures, women come to recognise that breastfeeding offers their infants comfort. The women's narratives also reveal that the relationship formed as a result of breastfeeding is close, exclusive and intimate; therefore, this relationship is privileged. This privileged relationship offers women insight into their embodied capabilities, whilst providing health and well-being for their infants. Ultimately, the relationship offered through breastfeeding opens up women's potential for being.

9.2 Ontological Understanding of Care

Dasein exists in the world 'with-other', and one of the fundamental or essential modes of being-with-other is that of care (Heidegger, 1962). Care can be defined as 'concern-for', or 'care-for'. The things or 'others' that call out to dasein capture its concern or care. Dasein's attention is focused on that which it is concerned with, and as a result, dasein comes to understand both 'other' and being, through the everyday things that capture its focus or care (Heidegger, 1962; Inwood, 1999).

Because of the importance of care in *dasein's* ability to come to know and understand the world, Heidegger considers that the primordial being of *dasein* is itself care (Heidegger, 1962); therefore, care is *dasein's* basic mode of 'being-

in-the-world'. "Care, as a primordial structural totality, lies before every factical²⁹ 'attitude' and 'situation' of Dasein, and it does so existentially *a priori*' (Heidegger, 1962, p. 238). Care provides the ontological basis for approaching entities, such as objects and 'others', within-the-world. Because care is an essential part of being, it is not possible to exist without care, and according to Inwood (1999, p.2) "no one is wholly carefree, careless or uncaring". Everyone is fundamentally careful and navigates their way in the world accordingly (Heidegger, 1962; Inwood, 1999)³⁰. It is possible to encounter 'happiness', 'sadness' and 'anger' only because *dasein* exists as care (Dreyfus, 1991; Heidegger, 1962). In the everydayness of life we experience emotions, such as 'gladness' or 'dismay', only because we care; therefore, being can be made visible through the comportment that is care (Heidegger, 1962).

Dasein's structure of care provides the basis for everyday connection with entities 'within-the-world'. There are two types of care that are employed by dasein, one mode of care for objects and another one for 'others'. Heidegger employs the term concern to describe the care that is exhibited when one attends to objects that are encountered 'within-the-world'. Concern is that which manipulates items and puts them to use (Heidegger, 1962). However, this type of care does not accurately portray the care with which we attend to others; instead, Heidegger (1962) employs the word solicitude³¹ to describe this type of care. Therefore, care 'within-the-world' is always exhibited through either concern or solicitude (Heidegger, 1962).

Care or solicitude is expressed towards 'others' through two distinct modes (Heidegger, 1962). One can be described as a deficient mode of care, which typically characterises the "everyday, average Being-with-one-another" with whom we have no familiar relationship (Heidegger 1962, p. 158). Heidegger (1962, p. 158) describes this as "being for, against, or without one another, passing one another by, not 'mattering' to one another". Therefore, care

²⁹ According to Inwood (1997, p. 124) "factical, facticity; similar to factual, factuality except that they are applied only to *dasein*, e.g. the sheer fact that one exists."

³⁰ It is important to note that care does not describe the ontic ways of 'being-in-the-world' but purely the ontological-existential manner (Dreyfus, 1991; Heidegger, 1962).

³¹ In this context, solicitude is an attempt to translate the German word *fursorge*, defined by the translators of Heidegger's (1962) work, Macquarrie and Robinson, as "the kind of care which we find in 'prenatal care' or 'taking care of the children'" (p. 157).

towards other can be deficient, in which case 'inconsiderateness', 'indifference' and 'perfunctoriness' result (Heidegger, 1962). The other distinct mode of care or solicitude is described as proper, or authentic care, and is expressed by 'considerateness' and 'forbearance'.

Authentic care is exhibited by either 'leaping in' or 'leaping-ahead' (Heidegger, 1962). Dasein can exercise care by leaping in for the other and taking away care, which describes how dasein essentially 'takes over' another's care. Heidegger (1962), however, warns that this type of care can result in the domination and dependence of the individual whose care is taken away. An example of this as it appears for a breastfeeding woman might occur when a well-meaning health care professional 'leaps in' and latches an infant on to a woman's breast. In this case, the health care professional has taken away the care of the breastfeeding woman, because she has prevented the women from gaining the skills she needs, resulting in the woman's prolonged dependence on an outsider for assistance with breastfeeding.

The second type of authentic solicitude is the type that *leaps ahead* of the 'other' "not in order to take away his 'care' but rather to give it back to him authentically as such for the first time" (Heidegger, 1962, p. 159). Solicitude in this mode desires to facilitate independence, not dependence (Heidegger, 1962; Inwood, 1999). In doing so, the other becomes 'free' to care for himself or herself. Heidegger (1926) talks about how, when two individuals devote themselves to the same task, each taking hold of their *dasein*, they become authentically bound together, thus freeing each other into their ownmost Being. Therefore, through authentic solicitude, *dasein* can be brought into its ownmost being (Heidegger, 1962).

9.3 Responding to the Call to Care: Coming to Know and Understand Other

As one dwells 'with-Other' through the authentic mode of care or solicitude, one comes to understand the Other (Heidegger, 1962). Heidegger (1962) claims that we understand others because of things or entities that concern us within-the-world. From this it follows that we can engage in concernful interactions while being-with another on an everyday basis, thus forming the basis of our

understandings of others. The women who participated in this study expressed concernful solicitude for their infants. This was evidenced by narratives that expressed how they had come to care whilst pregnant; the call to care became even louder following the birth of their infants, facilitating the women coming to know, and ultimately to understand, their infants.

9.3.1 The Call to Care

The care that is expressed towards 'other' is that of solicitude; however in everydayness the mode of solicitude that is most commonly expressed is deficient because it fails to attend to other with authentic solicitude. To express an authentic mode of solicitude it is important to be open to 'being-with-one-another'. The example of a woman who has just given birth provides the backdrop for an interesting interpretation, and explanation, of this idea.

When one is first introduced to a new individual it is customary to greet and converse: two people become both linguistically and sensorally acquainted, and both verbal and non-verbal communication is assessed. This pattern of communication normally occurs prior to the development of a trusting and intimate relationship. However, for a woman who has just given birth, current practice is to introduce the newly arrived infant to the breast as soon as possible. Breastfeeding requires that a mother be close to her child, that their bodies touch, and that a mother embraces her infant to her. Some women initially struggled with coming-to-know their infant in such an intimate manner prior to becoming 'acquainted', and as a result a woman could come to 'mistrust' or confuse her infant's signals.

Emma provided a narrative that relays how she wanted to know the child she had just given birth to before putting him to her breast to feed. To be forced to be 'close' to someone whom she had just met was a difficult experience for her.

You're still trying to get over what you're holding, having the baby there, then to have to be launched into feeding him so soon - I would have preferred to get used to having him and be able to do it [breastfeed]. I know you can't because they're hungry but it would have been a lot nicer... During the night, that first night, he was just shocking. I read that babies sometimes can go without a breastfeed for 24 hours when they're

first born but he was just on me continuously that whole night and it was horrible... I just wanted to look at him not have him on me. (Emma. Int. 14. Para. 13-19).

Newborn infants communicate primarily through non-verbal communication and crying, while a mother expresses solicitude towards her infant by striving to understand these unfamiliar communications. A crying, inconsolable infant speaks to a mother of a distressed baby. Equally, an infant that smiles in response to his or her mother's care portrays a satisfied, fulfilled child. It is important to remember that infants typically do not smile for the first six weeks (Muscari, 2001); however, the participants in this study engaged in a process of coming-to-know their newborn infants, and actively sought positive responses from their infants, as such responses provided a visible measurement of the effectiveness of their caring. At times, their interpretations of their infant's responses to the breast and breastfeeding influenced their decisions whether or not to continue to breastfeed.

Sue had completely weaned her son when he was three months old. She believed she had insufficient milk to satisfy her son's needs despite letting him feed for hours on end. For Sue, the most difficult aspect of her breastfeeding experience was having her young son inconsolable despite frequent feeding. As she deeply cared for her infant, watching him suffer was devastating for her.

[My son] he wasn't sleeping. He was a newborn baby and, particularly during the day, even from one or two days old we couldn't get him to settle during the day at all. He just wasn't contented at all - that was the most soul-destroying thing. You could see this poor little so-and-so was just really unhappy and there was nothing I could do. I thought I was doing the right thing by trying to breastfeed him and it still didn't settle him because he wasn't getting anything as we found out later. It was awful.

(Sue. Int. 6. Para. 34).

Lee questioned whether she should stop breastfeeding her son as she was in pain due to cracked nipples. Her doubt was reinforced by the perception that her son was not enjoying breastfeeding as he would constantly come on and off the nipple during a breastfeed. However, once her son was about six weeks old, he displayed behaviours that indicated to Lee that he was also enjoying the

experience of breastfeeding, and Lee's decision to continue to breastfeed was validated. Lee breastfed her son until he was 15 months of age.

I thought I could really easily have given up [breastfeeding] because it was such a hassle and so painful and my son didn't even seem to enjoy it that much in the first six weeks. But then he definitely settled to enjoying it and being happy... He probably wasn't not enjoying it, it's just that a baby under six weeks old is less responsive. I remember, after he was six weeks old he used to do this really cute thing. That you'd fed him and he was full, he would pull his lips away from the nipple and give you a huge grin, [he would] look up at you and "ooh" (laugh) and sort of smile and then he'd return to feeding. That was really sweet. (Lee. Int. 10. Para. 30 & 32).

Debbie decided to wean her son due to the severity of the pain associated with cracked nipples. Her perception was that she was being harmed by breastfeeding her son, and her resentment towards him for harming her negatively affected their relationship. In her narrative, Debbie provides an example of how difficult it can be to express authentic solicitude when 'other' is causing you pain. For Debbie, the pain resulted in the end of the breastfeeding relationship. She believed that if she had not made the decision to wean her son their relationship would not have improved.

Because of the pain that I was having from [breastfeeding], because I was cracked and bleeding, it was just horrible. I resented my son a lot for that. If I hadn't made the decision to bottle-feed we still wouldn't be getting along.

(Debbie. Int. 15. Para. 50).

Jan expressed how she dealt with her son with solicitude, sharing that, through her relationship with her son, she came to understand breastfeeding as painful. She provides an example of a woman who felt forced to care through breastfeeding, believing she had no choice. The call to care through breastfeeding was strong; however, Jan's experience of breastfeeding pain drowned out the call and provided her with an overwhelming message she could not ignore. With graphic language she images how it was to breastfeed her son, describing the effect that her experience had on her expectations for breastfeeding her second child.

[My son] he was really aggressive so it wasn't even that it was a nice experience once he got on [the breast]. He was chomping even though he had no teeth, he was like a barracuda it was awful. He was really horrible. So when I got pregnant again with my second child I was terrified of breastfeeding.

(Jan. Int. 13. Para. 8).

The experience of breastfeeding her second child, a daughter, was a very positive one for Jan. She spoke in depth about how breastfeeding facilitated authentic solicitude, leading to a close connection with her daughter. Jan was also cognisant of how her relationship with her daughter was adversely affected by her first breastfeeding experience.

[Breastfeeding] it's made a big difference [with my daughter]. Which is pretty hard for my son, but our relationship is fine now, it just got off to a rough start. I'm so pleased it's changed, I was so relieved because for a long time I thought I was never going to like him. That sounds awful doesn't it? It's very true, very honest when I say that I didn't like him at all. I can remember sitting in the bathroom, cause I used to have to use a breast pump initially too and that didn't work either, thinking if something happened to him I would never have had another child. That kind of whole thing like: 'if you were taken away tomorrow, I really wouldn't be that upset.' That is awful. It's a horrible, awful, awful feeling and thing to think. But thank goodness I only thought that once. (laugh). I would never have harmed him, it was not a harming thing, just if I was given the opportunity I would never go through this again.

(Jan. Int. 13. Para. 52).

Jan spoke of how she had to resolve her feelings of guilt for not continuing to breastfeed and for not enjoying her son before she could have another child. In the following narrative, she describes how she came to the position of wanting to have another baby.

I think I knew that I had a choice and that I was right over that guilt thing. That I knew that if we had another baby that I didn't have to breastfeed, that I had a choice and that if things could actually go OK and I could get on top of it. If I had to bottle-feed from day one then that's what I'd do and they'll still be a healthy and happy child, and that they wouldn't be

deprived or be so negatively affected. I wouldn't feel so completely guilty about that.

(Jan. Int. 13. Para. 54).

According to Inwood (1997, p. 52) to "will, wish or strive for anything whatsoever one must in advance already be care" (Inwood, 1997, p. 52). Women approached their infants with a stance of concernful care, and as a result wished and strove to breastfeed them. Through breastfeeding, women expressed their care; however, an important part of caring for a new infant is learning what it is that they require to be happy. This presents a significant challenge for most women. Some of the participants decided that breastfeeding did not satisfy their infants or work for them, but those women who persevered discovered that breastfeeding was the best way to comfort their baby.

9.3.2 Understanding Breastfeeding as Comforting

According to Gadamer (1996, p. 112), a comforted person is one who is "open to new things, ready to embark on new enterprises and, forgetful of ourselves, scarcely notices the demands and strains which are put on us... This is what health is". The participants in this study understood that an aspect of coming to know and understand a new infant is being able to console and offer them comfort when they are crying. In this context, comfort meant more than just a lack of crying; it was the ability to offer their infants a passage by which they could be readied to embark on life, and give themselves over to all that life had to offer. For the women it represented health for their infants.

For these women, breastfeeding became the tool with which they could sooth and comfort their infants. The relationships they forged through breastfeeding developed because of the solicitude a mother feels for her infant, was facilitated by time spent, being constantly present, and working together. As the relationship matured, the women came to understand that breastfeeding meant comfort, and ultimately, a sense of health for their infants, which in turn meant it facilitated their own sense of well-being.

In the following narrative, Kay shares how breastfeeding her son comforted him like nothing else would, offering him a sense of safety, security and reassurance

that he would be taken care of. Breastfeeding represented more than nutritional health — it also enabled Kay to offer her son embodied well-being.

It just sort of happens, if my son wants a feed he gets a feed.

Sometimes at night he might want an extra one before he goes to bed, he might've had a bad dream or something - it comforts him as well.

Two nights ago he woke up crying for some reason, I think he must have had a bad dream or the wind was howling, it could have rattled the windows, and he wanted a comfort feed. It wasn't a feed, more of a comfort thing for him. There's no way of comforting him like a comfort feed. I used to do that when he was little too, just feed and feed but not really feeding.

(Kay. Int. 8. Para. 82).

Jan had the opportunity to bottle feed her son and to breastfeed her daughter, and she noticed marked differences in her ability to comfort her two children. In the following narratives, she contrasts her experiences in relation to comforting her son and daughter. Breastfeeding became the ultimate source of consolation for Jan's daughter when she was upset, unlike her brother, who was considerably more difficult to console. Jan discussed why she believed that being at the breast furnished her daughter with a safe, secure haven, noting that breastfeeding seemed to "take away" the pain. The pain, which held her daughter back from engaging actively in life, was removed through breastfeeding. She spoke of how fulfilling it was, as a mother, to be able to offer her daughter life through comfort, and satisfy her child so completely.

I liked bottle-feeding him [my son], I didn't think there was much difference. But I would say that when he was beside himself, you know if he got himself really upset about something or you know, when he'd had his immunizations, or something like that, to breastfeed and to really snuggle with them, and to give them a breastfeed, they're almost completely satisfied. You know you can put them completely at ease and they're much happier. The bottle's not quite the same in my perception. I can snuggle close and give him his bottle but we're not quite as close and all encompassing as breastfeeding in those situations.

Sometimes if she's [my daughter's] been distraught she's not drinking at all. She's just, you know, snuggling. She pacifies herself with my breast really, rather than having something to drink. It's not the milk, it's the being close. With my son there were a couple of occasions when I had wished I could, I would even bring out my breast and think 'Well even if you can just get close to it would that make you feel better?' But he was long gone past that stage then. Whereas, [my daughter] she'll always come to it but then she could just be a breasty girl, you know she just might be that kind of baby anyway... Initially with my daughter I felt like a bit of a milking machine, you know. I didn't enjoy it at all, it was just a function. Now that she has two bottles a day it is more of a comfort than having a lot of milk and it's nice. I quite like it. It's quite snuggly. It's usually first thing in the morning, you know very early in the morning, and it's just her and I and it's very quiet and she's having a lovely snuggle and something to eat and that's very nice. It's not just a function, it's a nice cuddly time.

I can almost feel it mentally as well as physically, I can feel my daughter wanting to be close. It's almost like when you snuggle in with your husband or partner, you know, and you can feel that real closeness. It's like having a really nice hug... Sometimes you snuggle that little bit closer and it just feels really nice. It feels as if you're cocooned. And, in that crying, I'm hurt or frightened or something that's what she does, she snuggles right in. She doesn't want her head out at all, she wants your arms right around her, she wants to be cocooned. And my son doesn't do that at all and he never has snuggled like that. He wants to be over your shoulder, held that way... [To be able to comfort like that is] satisfying, really satisfying, almost like you're taking away all the pain or the fright or the hurt and you can do it so quickly. You can soothe fears and all kinds of things almost immediately... It's great, fantastic. You can sometimes do it just by holding, but she responds so much faster and relaxes with breastfeeding, often they're tense when they've hurt themselves, and she just relaxes straight away.

(Jan. Int. 13. Para. 46, 48-50 & 84-92).

One participant, Eva, spoke of how she experienced a sense of well-being because she could completely comfort her infant through breastfeeding. It was a method of comforting that Eva, due to the positive solicitude she felt for her

daughter, was pleased to be able to provide. Breastfeeding supplied comfort and facilitated the health of both mother and child, releasing them both to engage in new enterprises and give themselves over to what life had to offer.

It's a really lovely feeling because you have the baby in your arms, which is a lovely feeling in itself, and you're satisfying her needs using your own body, which is a very satisfying feeling as well. If comfort is defined by having a sense of well being while doing it then yes, absolutely. There definitely has been times when the only way we could calm her down, if she's worked herself into a frenzy, is to pop her on my breast and immediately she'll stop crying and she'll calm down completely whether she's hungry or not, she receives comfort, and I'm really happy to provide that.

(Eva. Int. 19. Para. 66).

Breastfeeding also provided Sally with a great deal of pleasure. She especially enjoyed breastfeeding at night. Not being subjected to external pressures such as outside employment afforded Sally the freedom to focus on her relationship with her son, and a part of this relationship was breastfeeding. She spoke of how breastfeeding also offered her health by providing her with energy for all that life holds.

I really love it [breastfeeding]. I really love it. It's just great. Especially the middle of the night feeds. I wasn't pressured to go to work, I didn't have outside commitments. If I was really tired I'd sit down for an hour and have a feed and we're both going to get revived. I got a really strong letdown, but you also get an emotional response with that and I really did. I don't so much anymore, it still relaxes me and I enjoy it, but in the early days I used to really get a feeling of total contentment within a minute of him latching on. There is a hormone response to my understanding but it was all tied up with that.

(Sally. Int. 11. Para. 32).

As the women who participated in this study came to understand the ways in which their individual infants communicated, they came to understand that breastfeeding offers the ultimate source of comfort. The comfort and health that breastfeeding offered meant that both the women and infants were able to actively return to exploring and engaging with their existence. Breastfeeding

offered comfort, born through relational closeness, which enabled the return to life.

9.4 Being Close

Dasein exists as 'being-with-other'; knowing oneself is possible as a result of 'being-with-others', as dasein understands 'other' in relation to understanding itself (Heidegger, 1962). Heidegger (1962, p. 161) claims that "solicitous concern is understood in terms of what we are concerned with, along with our understanding of it". To express solicitude for 'other' we must come to understand and know ourselves. Heidegger (1962) also describes how we normally express the deficient modes of solicitude, such as indifference or passing each other by. In these deficient modes, it's impossible to come close to others, or know oneself, as one must come close to the 'other' to understand oneself. If dasein remains 'aloof' and 'hides away', it cannot come to know others, or even itself.

According to Heidegger (1962, p. 161), "being-with-one-another must follow special routes of its own in order to come close to 'others". However, if one dwells closely and chooses to open up to 'other', authentic solicitude can grow out of this closeness. As one dwells with-other in a 'proximal' or close manner, he or she comes to understand the 'other', and also understands how to 'be' with 'other'. Being in close proximity makes 'being-toward-other' and thus, closeness, possible (Heidegger, 1962, p. 161).

Breastfeeding requires that a woman and her infant be physically close; it also requires emotional openness to the infant, otherwise a woman will feel "tense, unhappy and uncomfortable" (Eva. Int. 19. Para. 12). By dwelling closely with their infants through breastfeeding, the women in this study were 'being-toward' their infants, which allowed them to understand their infants; thus, authentic solicitude for their infants grew. Those women who persevered with breastfeeding spoke of how it facilitated closeness between them, and through this closeness they came to understand and know themselves, as well as their infant.

Sally spoke of how she expressed solicitude in breastfeeding her son. She enjoyed breastfeeding because it was an act that excluded others, even those who were relationally close to her son, such as his father and grandparents. She relayed how she did not want to lose this privileged relationship, as it was through being close that Sally came to understand fulfilment in a different and special way.

[Breastfeeding] it's just something else that no one else can share with my son; my husband can't and my parents can't. I think it just makes us really, really close to each other because it's something that only ever the two of us can share... I really enjoy the fact that he needs me and I really don't want to give that up. That's part of the reason that I'm breastfeeding, you know a small part of it but it is definitely there, because as soon as I stop he's not going to need me the same... I don't want to lose that.

(Sally. Int. 11. Para. 88 & 90).

Breastfeeding facilitated a time where Lee's son focused solely on her, not on any other. Breastfeeding provided the time to be relationally involved and thereby solidify their attachment through being close.

[Breastfeeding] it's one of the times when you have a really nice time with your baby just responding to you, not to everything else as well.

And it's a lovely time to talk to them, to hold them and cuddle them and all that kind of thing.

(Lee. Int. 10. Para. 60).

The understanding that Emma came to was that breastfeeding her infant gave her a sense of purpose and a belief that she was needed. The 'always-nearness' of her infant meant that there was a depth to their relationship, facilitated by solicitude, that was expressed through breastfeeding. This close relationship made Emma feel special, and weaning her child off breastmilk onto the bottle meant that she would no longer have exclusivity within their relationship. Emma's relationship with her baby was valued, unique and distinct from other relationships because of the authentic solicitude and closeness that breastfeeding created.

When you're pregnant you know that you're the one that's keeping them alive, and keeping them well and breastfeeding them and you've got them all the time, but once they're on the bottle they can be separated from you... You're no longer special. (Emma. Int. 14. Para. 105).

Abby spoke of the closeness that she felt for her children as a result of breastfeeding. She had breastfed her first child, a daughter, for five months before conceiving her second child, at which point she could no longer maintain her milk supply and subsequently weaned her daughter. She went on to breastfeed her second child for 18 months. At the time of the interview, Abby was breastfeeding her third child, a boy, who was five and a half months of age. She spoke about the differences in her relationships with her daughter and her two sons, attributing these to the closeness of breastfeeding. Solicitude allowed Abby to build a close relationship with her sons. The closeness experienced through this relationship taught Abby's sons about authentic solicitude, which was expressed through their reciprocal closeness and learned ability to express solicitude for Abby.

I fed my son longer and I'm closer to him than I am to my daughter who I didn't feed for very long at all... I think it's because you've spent so much time together. They're latching onto you, you're sharing a little bit more, more than you would just sticking a bottle into their mouth. My other son is quite clingy, this one [the youngest child] is fine but the other one gets quite jealous. He's forgotten what breastfeeding was but he knew that closeness.

(Abby. Int. 2. Para. 38 & 41).

For Kate, breastfeeding meant a physical closeness to her young son. She talked about how she enjoyed the quiet and stillness of feeding at night, and enjoyed having the time to focus on her son away from the busy-ness of the day's routine. She discussed how seven months after her son's birth she was still amazed by the miracle of his life and how he was growing and developing as an individual. Breastfeeding him facilitated the relationship between mother and son and also provided Kate with time to learn about his personality.

I get lots more cuddles because that's all part of the breastfeeding. That tummy on tummy stuff and in the middle of the night it's just you and him

and it's like you and the baby are the only ones in the whole world that are up at that time of the morning. I'm sure if you were to phone around there are a hundred thousand other women doing the same thing... We just had a light on in the room and the radio playing quietly in his room for a while and it's a really calm sort of experience. I love to sleep so I was really thinking it was going to be torture, but I found you're a zombie, but when you see them it's far more pleasant than I thought it would be... I suppose because you're getting to know this little person. [My partner and I] we sit here at night and go 'Oh my god - we made him', that whole new parent cliché where you are stunned that you've done something like this ... you set this thing in action when you conceived and it's happening to you and all you've got to do is eat properly through the pregnancy and the rest takes over and then just one thing after they're born, you're doing it.

It's a bond. [The baby,] it's now a separate entity, he's not a part of you anymore, he's not reliant on you, he's becoming more so as his personality is developing. There's a real feeling of doing that bonding... And I think breastfeeding helped all of that. I still love it sometimes if he sleeps too long I want to go and get him up. You actually miss him because there's almost an unnatural bond I suppose. I'm just so pleased that breastfeeding worked because I don't have to boil bottles and mix formula and the cost of it. All we've had to do is look after me. (Kate. Int. 3. Para. 14, 16, 18 & 20).

Eva shared that mothering a child irrevocably changes the way in which a woman understands herself. A woman who has mothered will always have an enduring link with her offspring. In part, this link is due to the responsibility or solicitude a woman feels towards her child. Eva recognised the unique nature of this relationship and acknowledged that her relationship with her daughter is different to any other relationship she will ever have.

There's something really special about being a mother and no matter what happens to your children you're always going to feel a special bond. It's tied into a quote I heard 'husbands and partners can actually come and go but if you don't have children you'll always be childless, but if you have children you'll always be a mother'. You're always going to have your children... It's special, special, they're your full responsibility.

(Eva. Int. 19. Para. 79).

Breastfeeding had opened up the opportunity for Eva to experience closeness with her daughter. She spoke about how the relationship and closeness between them existed in part because they had both worked at it together.

Certainly breastfeeding has made me feel closer to my baby. I'm pleased that we've been able to make it happen together. I would emphasize that it's a two-way thing, you have to learn your part but the baby also has to learn to do her part, the fact that we've managed to do that together I think makes me feel closer to her. I think I would, especially in the first few months, have felt that I'd let her down if I hadn't been able to do it.

(Eva. Int. 19. Para. 116).

Eva had experienced a great deal of difficulty during the process of establishing breastfeeding, and had to persevere through a significant amount of pain to continue. At the time of the interview, she was still breastfeeding her daughter, who was three months of age. Eva understood the embodied experience of breastfeeding pain; however, she referred to breastfeeding as 'easy', and stated that "it just happens". She expressed a stance of openness toward her daughter. This led to relational closeness with her daughter, which gave her the strength to stay-the-course required of her to breastfeed, thereby overcoming her embodied experience of pain. During the interview, Eva breastfed her daughter, and in the following narrative, which emerged at the end of the breastfeed, Eva expressed authentic solicitude:

(Laugh). You just had a lovely time didn't you? (Comment is directed to the baby). I must say this is the best moment when she comes off and her face is all pink and she can hardly hold her head up and she looks all pleased, and fat cheeked... It's not deliberate, it just happens, you feel such wonderful feelings towards her and then she's so cute and it's just easy to do it. It just happens. You could say I don't have a choice, because I'm her sole caregiver but there's ways you could look after your baby if you didn't want to. It's just such a natural, easy thing. And, she's just so cute I can't keep my hands off her (laugh). (Eva. Int. 19. Para. 140-142).

For Amy, breastfeeding facilitated mutual growth for both mother and son. Amy points out that she had to make the commitment to be there through breastfeeding, requiring her to dwell with openness with other. Initially, their relational growth developed because of the consistency of contact associated with breastfeeding a young infant. As time went on, the frequency of breastfeeds diminished, but the foundational bond formed through their constant contact paved the way for a secure relationship. Openness to being close through breastfeeding facilitated authentic solicitude, and thus their relationship flourished.

It was just a process, a continuum, something that grew. Initially when my son was born you connected to him through the consistency of feeding him. You knew that you were going to have to feed him in a couple of hours and that was it, you were just constantly feeding him. It never really felt like it at the time and gradually as he gets older it becomes more of a pattern, there's more of a routine to it. You grow with it and he grows with it and to me it became more and more a part of the lifestyle. As he grew I was growing too. We both grew physically and mentally.

In terms of my relationship with my son I believe it was truly enhanced by breastfeeding. For his first year of life and even now I suppose I have spent so much time with him and have been there for him through breastfeeding. I've had to be there for him and making a commitment to breastfeed, I've had to be there for him. In that way our relationship is now so strong and so solid because of the foundation of that first year, and even now has been built on that knowing that I'm always going to be there for him, to feed him when he needs it. In terms of our relationship I believe that's really grown and been really strong as a result of that. (Amy. Int. 1. Para. 32 & 34).

Kay also spoke about the closeness that exists between a mother and her baby when engaged in a relationship that incorporates breastfeeding. Kay was a solo mother who did not share custody with her son's father. She relayed an incident in which her young son responded to her, and discussed how breastfeeding was more than just physically providing nutrition. Breastfeeding provides many cherished intimate moments for a mother and her infant. The relationship a mother has with her child when breastfeeding is reciprocal and

facilitates the development of authentic solicitude. It is this mode of solicitude that cements the relationship between them.

Not that long ago I was sitting on the couch there feeding him and my girlfriend was sitting in here, and my son was feeding then he stopped and he was looking up at me and I was just looking at him and he went like this with his hand to my face and just touched me. My friend said 'Oh look at that that's so neat'. It was just a neat feeling, like a real bond. I guess that's what you can see when you're feeding them all the time. It's a special moment, I guess that's the only way I can really explain it was through that moment. It's like the way he just looks up at you and gives a smile and fouches your face like that, it's really neat. It's really hard to explain. Breastmilk is the best thing for the baby but it's also something else as well, not just feeding the baby, it's a bond.

Sometimes I look at my son and I think 'I haven't only just given birth to him, but I'm making him into who he is now. You're giving him the best possible start to life. They say if you breastfeed it offers the best start in life. Sometimes you feel so much love for him when you're feeding him, I don't know why. Sometimes it makes you really happy. Sometimes you get tears in your eyes, it's happiness, not sadness. One day he's going to grow too big for this (laugh).

It's a special time while I'm breastfeeding my son. And at night before he goes to bed he knows, he sometimes looks at you and grins from ear to ear. It's for me only. No one else can share in that. Which may sound a bit selfish, but I carried him for nine months, I gave birth to him, I've looked after him and no one has helped me right from day one. Right from the day I was pregnant I've had no help. So the way I see it I deserve that (laugh).

(Kay. Int. 8. Para. 58, 64 & 119).

Women come to understand other, as well as themselves, through breastfeeding. They also come to experience authentic care in different ways because of their experience of breastfeeding. As previously mentioned, Heidegger (1962) highlighted that authentic solicitude is exhibited through two positive, but different, modes, described as 'leaping-in' and 'leaping-ahead'.

Both examples of authentic solicitude provide insight into the being of a woman that breastfeeds.

9.4.1 'Leaping-In'

Authentic solicitude is exhibited when one 'leaps-in', to take-away, or take-up, the care of another (Heidegger (1962). This type of solicitude is typified by the dependency of 'other', and can result in 'other' being dominated. When considering the relationship of a woman and her infant, the infant is by necessity dependent upon his or her mother, which offers the opportunity to explore Heidegger's concept of solicitude as expressed when 'leaping-in'.

The women who participated in this study expressed authentic solicitude through the mode of 'leaping-in'; however, it was not because of a wish to dominate, but rather because breastfeeding meant that their infants were dependent on them. Mel shared her enthusiastic perceptions of what the breastfeeding relationship meant for her, reveling in the idea of being able to completely and exclusively care for her baby.

How wonderful is breastfeeding – you put your baby on and you're looking after your baby completely. There doesn't have to be anybody around at all, and it's just you and your baby. As long as you eat and drink you're taking care of every single one of its needs and that to me was wonderful.

(Mel. Int. 17. Para. 24).

Sally expressed authentic solicitude for her infant through 'leaping-in' and providing for her infant, which meant she took away her infant's care or concern. She also spoke of how she believed she was closer to her son because of breastfeeding him. She spoke about how her companions could leave their infants at home, but to her that was inconceivable because of the dependent nature of breastfeeding, and the resultant bond this relationship created.

One mother at coffee group the other day said her husband had taken her child away for the weekend to see his parents and it was the first weekend without her child and I just couldn't even comprehend that. I'm breastfeeding so physically I couldn't do it but emotionally I couldn't

imagine being away from my son for that period of time. I mean I go to work for 3 1/2 hours, and the first thing I do when I come in is go and check him in his cot and make sure he's alright because it seems like so long since I've seen him... I mean my son needs me, like really needs me, whereas perhaps these other babies don't need their mothers quite the same, yea it's tricky. It's really hard not being able to talk about it cause I can't say "I think you're not as good a mother" cause it's not that they're not good mothers it's just... the closeness. (Sally. Int. 11. Para. 50).

Sally's narrative highlights how solicitude is 'leaping-in', and within this context means accepting the dependent nature of the relationship between mother and infant, which is especially evident when a woman chooses to breastfeed. The other type of solicitude the participants expressed was exhibited through the mode of 'leaping-ahead'.

9.4.2 Leaping-Ahead

Authentic solicitude also 'leaps ahead' and desires to provide the other with authentic care and independence (Heidegger, 1962; Inwood, 1999). Liz provided an excellent example of solicitude that 'leaps ahead'. She had struggled to establish breastfeeding, and it had taken commitment to reach the point where she enjoyed breastfeeding her daughter. The solicitude she felt for her daughter was expressed in Liz's desire to be very close to her daughter. During this time of being close, Liz was able to silently watch her daughter and dream about her daughter's future. For Liz, this type of solicitude was expressed in "doing the right things", which meant that she could ensure a positive future for her daughter. Liz was actively anticipating the day when her young daughter would secure her own independence.

It's amazing watching my daughter and trying to imagine what she's thinking because often she'll be lying there and her eyes are flicking side-to-side looking or looking at my chest (laugh). I just often sit there and stare at her looking at me. But I also do watch BBC World News in the middle of the night (laugh). So I'm not always just sitting gazing totally at my daughter but... (laugh). You're looking at this beautiful child that you've nurtured and that you're continuing to nurture and my

mind often wanders. You wonder what is she going to be like when she grows up and, you know, are you doing the right things - yes I am, obviously she's happy - that lovely sense of being very close to your baby.

(Liz. Int. 9. Para. 62).

Kate offered authentic solicitude as expressed by 'leaping-ahead' to facilitate her son's health and wellness. For Kate, breastfeeding was more than providing physical nourishment, it was a way to gift her son the best she was capable of giving. Through the provision of breastmilk, Kate endowed security, comfort, health and love on her young son.

Very early on when he was newborn I really felt like I could visualize, I could just imagine him taking all that from me as well as the milk, all the love, nurturing, security, all the things you want your baby to have. I wanted him to feel like he was in this huge big cocoon of love and security so that he would always feel content and never feel all that sort of stuff. And you can actually visualize that, leaving your body with the milk and going into him. And you'd sit there at night and that's what I could do and I can still do it sometimes. Even today when I'm sick, I visualize all these little antibodies so that he doesn't get it from me. (Kate. Int. 3. Para. 66).

A woman who breastfeeds must be open to being close to her infant, to achieve which her solicitude must be authentic. Authentic solicitude is needed because a woman must possess the willingness to alter her life and embrace her infant in such a close and intimate manner. If her solicitude is authentic, then through this closeness she is able to concernfully nurture her infant. Authentic solicitude also offers a woman the opportunity to come to know herself in a new and increasingly authentic manner as "knowing oneself is grounded in 'being-with'" (Heidegger, 1962, p. 161). As women learn more about themselves and their infants through 'being-with' in a close manner, the breastfeeding relationship becomes privileged.

9.5 Breastfeeding as a Privileged Relationship

When two individuals dedicate themselves to the same activity, "they become authentically bound together... which frees the other in his freedom for himself" (Heidegger, 1962, p. 159). The type of authenticity with which one is released into their ownmost possibilities for being is the solicitude that 'leaps-ahead' of the other to promote independence. A woman breastfeeds because she desires that her infant have the best she can offer, by way of the physical benefits breastfeeding offers, as well as the relationship that is developed through such intimate nurturing. The desired result of this authentic solicitude is an infant that will eventually gain his or her independence and take hold of that which is his or her ownmost potentiality for being. The breastfeeding relationship that is forged through authentic solicitude facilitates an exclusive, close, and intimate connection seldom experienced in other relationships; therefore, the breastfeeding women in this study considered that the relationships they shared with their infants were privileged.

For some women, breastfeeding is intrinsically linked to the mothering role. Abby shared how she had come to understand the relationship that breastfeeding fosters between a woman and her infant. Abby was a young Maori mother who had three children, and believed that breastfeeding was an integral aspect of mothering. She spoke of how breastmilk was a gift a mother could provide for her baby, claiming that, despite the pain a woman might go through to breastfeed, breastfeeding was special and the best food source for her baby.

I think it's something special for the mother to go through to breastfeed her baby. The fact that you know it is your child and you're giving it what you can. If you really want to be a mother I guess that's part of it. (Abby. Int. 2. Para. 116).

Abby believed it was important to facilitate the relationship with her infant through breastfeeding, by being focused on breastfeeding when it was time to feed her babies. She believed that to breastfeed she needed to relinquish her previously-held sense of self, and embrace the mutual dependency required to breastfeed, a process that could only occur if she was open to expressing authentic solicitude. She believed it was imperative to relax and think about the baby, otherwise the baby would perceive this and fail to settle and feed. Abby

recognised that breastfeeding represented more than nutrition; it was about a relationship that required being 'wholly' present. If she was not focused on their relationship then breastfeeding would not work.

Just sitting and relaxing - that's what you're really doing, is thinking about breastfeeding the baby. If you're thinking of other things you can't be feeding the baby. If you're tense and feeding your baby the baby feels it and he's going to carry on crying, no matter what he's not going to eat... Bottle-feeding you just stick it in their mouth and they're quiet. No matter what kind of mood you're in as long as the bottle's in their mouth they're not worried about it...

(Abby. Int. 2. Para. 94 & 96).

Abby was disappointed to wean her first daughter at five months of age. Because she became pregnant with her second child, her milk supply was greatly diminished and she could no longer continue. She mourned the loss of breastfeeding her daughter and spoke about how it affected the closeness of their relationship. She spoke of how breastfeeding facilitated closeness and trust between a mother and her child, thus making the relationship a privileged one.

Since I've been able to experience both [breastfeeding and bottle-feeding]. That's where you are bonding with your child is different. I've found that with my daughter it's definitely different. As far as closeness is concerned she's not as close to me as my sons are... With communication it is time and because you're breastfeeding them they know that you're giving them time and they know that you will have time for them... I've always been told and read quite a bit about it, that the touching thing is a bonding thing so I guess that that would be part of it that skin-to-skin contact... A lot of times when he wants to be breastfed its not because he's hungry, it's just because he wants his time. (Abby. Int. 2. Para. 98-100 & 104-106).

For Abby, breastfeeding involved both her own mother and her mother-in-law. Abby's extended family valued the relationship forged through breastfeeding and supported Abby to breastfeed, thereby assisting her to develop the relationship with her children. By providing certain foods to nourish Abby's milk supply, the grandmothers provided their grandchildren with a gift for their life

and nourishment. In the following narrative, Abby explains how her mother-inlaw supported the privileged relationship that Abby had with her children through breastfeeding.

My mother-in-law had a few children herself and she's looked after all my nieces and nephews, they've got quite a big extended family. She helped out with my breastfeeding. When my babies were first born she would always bring over a meal for me at lunchtime and dinnertime so that she knew that her grandchildren were having something decent to eat. Not just feeding me but more for the babies. (Abby. Int. 2. Para. 74).

Breastfeeding afforded the opportunity for Kate to be close to her young son and thus she was coming to know and appreciate him in a very special manner. She rejoiced in the way his life had started and reflected on how he was growing, developing, and responding to her as he aged. Breastfeeding meant being close, and establishing a warm nurturing relationship together. Kate and her son had come to know each other through breastfeeding, experiencing and expressing this relational closeness by giving themselves over to being playful and delighting in the process. As a result, their breastfeeding relationship was privileged.

My son's head has just been of fascination to me since the time he was born and he's had, he's got that scar on the side - that's a forceps scar so he was sort of wrenched into this world unceremoniously in the end and his head has just fascinated me ever since. I don't know if it's to do with that or it's just to do with the fact you always love babies, you know yourself that, when you hold babies the smell of their head is always lovely. The top of their head and their hair. They're so incredibly warm and soft and his heartbeat too, you know with the fontanelle. He often now holds hands with me. If he's not tired and he's just feeding he'll put his foot up on my shoulder or he does some real boy things like his feet are going the whole time. It's a lovely time. It's very seldom that I've done it but it's really nice if I feed him when I'm just in a dressing-gown, to have tummy-on-tummy is really lovely as well. There's lots of handholding at the moment, as he's got older. There used to be lots of touching his head - I think that's what wore all the hair off for a while. I'm a fiddler, if he's got any sort of thing, you're doing the ear touching,

touching his hair, give him the once over while he's still for five minutes as well. See whether his fingernails need cutting and all that sort of thing. It's a relaxing time now. If I'm hyped he picks it up really easily now. If there's someone else here or something better that he might want to look at. He'll do the flirty thing now when he's on the boob too, look over backwards and you know, even with his dad. So it's obviously a place where he feels really secure too. I've had such a positive experience with it. I wish a lot more people did. (Kate. Int. 3. Para. 74).

Eva offered a poignant example of a woman who authentically cared for her daughter. In her solicitude, Eva desired that her daughter be healthy, happy and thriving, and theoretically 'leapt ahead', doing her best to ensure that her daughter would achieve all she hoped. Therefore, Eva chose to breastfeed despite cracked nipples and severe mastitis. She spoke passionately of how thrilled she was to see her little daughter thriving as a result of her persistence with breastfeeding. Eva posited that watching her daughter growing and developing successfully gave breastfeeding a purpose beyond the pain. She believed that being able to succeed through the pain and continue breastfeeding was one of her greatest achievements. It was through this achievement that Eva felt privileged: not only had she succeeded at breastfeeding, she was also providing her daughter with everything she needed for life. As a result of this privileged relationship Eva gained insight, as her possibilities of, and potential for, 'being' were revealed.

It's the best feeling in the world. People will say how healthy and beautiful she looks and I think I'm actually growing her myself. She's thrived on breastmilk and those times of discomfort you don't even remember them anymore. I'm not, you know, delighted with the pain but knowing that there's a higher purpose.

I actually feel, personally, that it's one of my greatest achievements to have got through that really difficult time and to have breastfed the baby thus far. I'm really proud of myself (laugh)... I felt special that I was a breastfeeding mother. Even when I was walking around without the baby there was something special about me because I was breastfeeding. I'm actually someone else's food source. I felt superior in a way... but maybe I'm a bit more special because I'm a breastfeeding

mum. I know that that will actually go away once I'm not breastfeeding anymore and I'll move into a different stage of life. At the moment I'm certainly enjoying it. I enjoy the feeling that I have achieved that. (Eva. Int. 19. Para. 11,19 & 27).

As one comes to know another in an authentically-caring manner, one also comes to know himself or herself, and thus one's ownmost potentiality for Being is realised (Heidegger, 1962; Paley, 2000). Eva discussed how privileging the breastfeeding relationship she had with her young daughter had altered her perspective and changed her as a person. She spoke about the deep sadness she felt at possibly failing her infant because she was not coping with the pain she was experiencing in relation to breastfeeding. Eva provided another example of solicitude that 'leaps in', conceding that being responsible for providing another life with the nourishment needed was a daunting task; however, what emerged as significant from leaping-in and taking away the care of another was what she had personally gained from having privileged the relationship. For Eva, succeeding at this important role had changed her perspective on life significantly.

The idea of having a person fully dependent on me to provide them with milk from my own body, has certainly changed me because I've found that concept pretty daunting, especially with my nipples being so sore and having mastitis. I really wanted to rise to that challenge, but I struggled physically to do that and that was awful, I got very depressed because I wasn't doing it very well. The idea that I was the milk source for someone else did change me quite a lot, it's again hard to describe exactly why or in what aspects. It probably softened me, and made me calm down and made me look after myself a little bit more because my daughter is dependent on me to be healthy so that she could get what she needed.

(Eva. Int. 19. Para. 118).

Kate spoke of how privileging the breastfeeding relationship with her son meant having her ownmost potential opened up to her, and as a result she considered herself enriched. She was able to extend her horizons through further understanding the cyclical nature of life. In the following narrative, Kate explains how she came to this extended understanding, relaying how coming-to-understand her embodied ability had been an awakening experience, and

how she reveled in this new knowledge. Through her privileged journey with breastfeeding, she had learned to celebrate the life-giving power women possess.

My Nana died a couple of weeks ago. They [the hospice staff] thought she was going to pass away and the hospice [staff] said we could stay up there for the night if we wanted to so my husband stayed here [at home] with the baby and I went into the hospice and I stayed there till about two o'clock in the morning and mum said to me you've got a little boy to feed so go home. I was just going to go straight home in the morning to feed him but the whole time that you're there you're aware that - what's my milk going to be like in the morning? I'd been up all night and it was a fairly sort of emotional time so I came home about two am and then got up to feed him at seven, but driving home was so neat, to be coming home from that situation, it was that whole cyclical thing where someone's dying but you're going home to this new life. Breastfeeding's such a positive - it's a new life thing, as opposed to the end of a life. It's one of the few times that you'll ever do anything like that I suppose. So it was really good to come out of that atmosphere and come home to something quite positive. It's still with you, you're coming home to this little dude that needed you to be there for him. That was a really positive time as well to happen. It certainly brought out a part of me that I probably didn't know that I had. The whole function, when I was pregnant it was probably the same. I used to have this analogy that I used to tell everybody, that it was like when you're born you're given, 34 years ago I got given this Porsche, and I've just been driving it around town and all of a sudden by getting pregnant I'm actually taking it out on the open-road and seeing exactly what it can do. Like I've not been using my body to anywhere near its capabilities and all of sudden I've got this prestige car that I've been driving as a supermarket trolley just around town, and all of a sudden it's out on the open-road and it's amazing what we can do. We're awesome. Women are amazing and I think any man that goes through it with his wife or lady would be in awe of them. Even right up to the breastfeeding and including that. It's just a fact, we're machines designed to, and it's not saying that that's what your whole life's all about but it's bloody amazing to know that that's what your body can do.

(Kate. Int. 3. Para. 70).

At the close of her interview, Kay offered the following narrative as a summative reflection on what she had gained through her breastfeeding experience. She concurred with Kate's understanding that breastfeeding had opened up her possibilities for being, which she attributed to privileging the relationship she had developed with her son through breastfeeding. She relays how her relationship with her son had completely altered her life; she speaks of a relationship 'with-other' that supports and strengthens life, a relationship that released both her son and herself to achieve their ownmost potential.

I haven't changed my life for my son, it's [that] he and I are building a life together. We're building a new life. He's changed my life but we're changing it together, for the better, so I don't feel tied down. (Kay. Int. 8. Para. 156).

A breastfeeding relationship is often forged through hard work, pain and perseverance. However, it is also through this privileged relationship that a woman comes to know her child and establishes a trusting relationship. Paley (2000) asserts that, if *dasein* maintains itself primarily in deficient modes of solicitude, then successful attainment of positive modes of solicitude "may count as an attainment, a successful realisation of what one truly is" (p. 69). A successful breastfeeding relationship offers an example of the successful attainment of a positive mode of solicitude. It also highlights how women can come to know who they truly are, and what they are truly capable of, as reflected in Kate's (Int. 3. Para. 70) narrative about the embodied knowledge that she had gained through her experience of breastfeeding.

A woman who provides her infant with authentic solicitude is concerned with the care of her infant. This infant is not influenced by, or familiar with, their ways of 'being-in-the-world'; therefore, the relationship a woman has with her infant is a relationship that is isolated from the tensions that exist in being with 'the they'. Instead, the infant chooses to communicate in ways that are not sullied by such things as the call to averageness. If a woman can authentically attend to her infant, then the relationship is freed to be what it is, and all that matters is what is passing between her and her infant. This unique and privileged relationship provides a woman with enhanced understanding of what it is to be a woman in a way she would not otherwise have access to. She may possess fulfilment

because she has experienced this relationship; she may also fathom in greater depth the purpose of life. As a result of these enhanced understandings of womanhood, she is released into her ownmost potential for being.

The mode of authentic solicitude expressed by a mother through breastfeeding is the solicitude that both 'leaps-in' and 'leaps-ahead'; it is solicitude that aims to open up the path by which the 'other' can be freed to care. The relationship built through breastfeeding, through 'being-with' in the mode of authentic solicitude, may be the closest relationship that infants have in their first year of life. When a woman is willing to express authentic solicitude through a privileging of the breastfeeding relationship, her infant will experience closeness 'with-other'. In doing so, the infant will also be exposed to authentic solicitude, and thus, the infant will learn about both care and themselves through 'being-with-other', and ultimately be freed to secure their future independence.

9.6 Summary

The primordial mode of Being is that of care – we care for 'others' with solicitude. A woman expresses solicitude for her infant when she chooses to breastfeed. Initially it can be difficult to understand the cues that a new infant offers, and out of her concern, along with difficulties associated with the establishment of breastfeeding, a mother may decide to wean her infant. However, if she continues to breastfeed her infant, then she comes to understand that breastfeeding provides comfort and closeness. In consequence, breastfeeding becomes a privileged relationship through which a woman provides her infant with authentic solicitude, by both 'leaping-in' and 'leaping-ahead', thus providing a way for her infant to ultimately gain independence through health and happiness. Through this experience with authentic solicitude, both mother and infant are freed to attain their ownmost potentiality of being: "Authenticity involves taking responsibility for who I am and who I will become, recognising that at every moment I am in the process of building a life - a life that is 'my' life, not merely in the sense that I am the one living it, but in the sense that I have created it" (Paley, 2000, p. 73).

This marks the end of a sequence of chapters that have explored the findings of this research, focussed on the rich narratives provided by the women who participated in this study. The following chapter offers an interpretive discussion of these findings, from the viewpoint of the researcher rather than the participants.

Chapter Ten

Discussion

10.1 Introduction

Statistics published by the Ministry of Health (1999b) indicate that, within New Zealand, breastfeeding initiation and duration rates have remained static for some time, despite national strategies to achieve the WHO's (2002) goal that women exclusively breastfeed their infants for the first six months and continue to breastfeed for two years (Ministry of Health, 1999b; Ministry of Health, 2001; Ministry of Health, 2002). I have argued that, unless researchers and policy authors pay attention to the experiences of breastfeeding women, current breastfeeding rates are unlikely to change. With that in mind, this research study has explored women's experience of breastfeeding in New Zealand. Its findings reveal that breastfeeding occurs within the context of historicallymediated societal interpretations, and breastfeeding is understood by women to be an embodied practice that ultimately facilitates their authentic care of their infant. To fully explore the meaning of the participants' experiences of breastfeeding, this chapter offers the researcher's interpretations of these findings and explores the implications for health care professionals. The limitations of this study, along with the areas that would benefit from additional research, are also discussed.

10.2 Revisiting the Aim and Methodology of the Study

The aim of this study as articulated in Chapter One has been to explore women's experience of breastfeeding in New Zealand. The thesis that emerged from adherence to this aim is that breastfeeding women experience society, embodiment and concernful solicitude in ways that were previously hidden from them, and would have remained so without the experience of breastfeeding. These new experiences provide a woman with a wealth of knowledge about intuitive embodied listening, as well as societal interpretations of breastfeeding.

This study has employed a reflective lifeworld research methodology, based on the philosophical work of Heidegger (1962), Gadamer (1989) and Merleau-Ponty (1981), which has guided me to approach my research with a stance of openness, whilst being aware of the impact of historicity on current understandings of breastfeeding. It also provided the research's foundation on the hermeneutic circle, which is integral to calling forth of meaning from the participants' narratives.

Breastfeeding is an issue that has been written about at great length (see Chapters Two and Three for a review of the relevant literature); however, commonly-held understandings based on this literature frequently ignore the meaning that breastfeeding has for women, and the literature that does focus on meaning has little sway in the public health policy arena. Addressing the issue required a methodology that enabled a return to a well-researched phenomenon to re-examine it for the meaning that breastfeeding has for women in New Zealand. Reflective lifeworld research allowed me to re-examine what it means to be a breastfeeding woman in New Zealand, unconstrained by previously-developed theoretical frameworks.

The need to respond to the participants' narratives, valuing them for their inherent meaning, was of particular importance to me. It was my desire not to infuse their account with the language that currently exists around breastfeeding, but rather to enable these women to use language that reflected their experience of the lifeworld as it appeared to them. In doing so, women can take ownership of their knowledge and facilitate a language that supports, rather than undermines, their experience. Reflective lifeworld research allowed me to embrace women's language, rather than excluding or altering it to fit a predetermined framework. To that end, I have included extensive passages from the women themselves.

The following discussion renders my interpretations of the narratives offered by the breastfeeding women who participated in this study. The various sections highlight specific issues that have emerged, and that need to be explored and resolved if health care professionals are to provide support to breastfeeding women in accordance with their needs.

10.3 Coming-to-Know: Creating Breastfeeding Knowledge

Giving birth to, and breastfeeding, children is the domain of women. These acts are understood to belong intrinsically to women, and they represent and constitute aspects that are integral to mothering. Women who become mothers may have the opportunity to use their bodies to engage in the process of nurturing and growing infants and children, and in doing so, embrace a significant amount of responsibility and accept a challenging task.

The women who participated in this study articulated a strong desire to breastfeed their infants, because it represented to them the ultimate source of nutrition. Other research studies also highlight that women choose to breastfeed because of a deep desire to provide their children the very best they can offer (Arora et al., 2000; Earle, 2002). This desire provides strong motivation to initiate breastfeeding, deeply affecting the decision-making process of women who breastfeed.

First-time mothers who choose to breastfeed engage in a process of 'coming to know self' as women capable of breastfeeding, a process that for some begins at a young age when girls become aware that women feed their infants at the breast. The culture women have grown up in provides them with a backdrop from which to interpret breastfeeding, which affects their own decisions about infant feeding and ultimately mediates their approach to breastfeeding. However, before a woman begins to breastfeed she must know that her body is capable of producing milk and feeding an infant. The knowledge that an infant can thrive on the milk she provides requires confidence and awareness of her body's ability.

Ryan (1999a) demonstrated that women have the ability to embrace a variety of differing perspectives that enable them to interpret breastfeeding and their experiences with it. Prior to ever bringing a baby to her breast, a woman will have gathered a wealth of information that will form the basis from which to interpret her experience. This groundwork comes to the fore as a woman contemplates being a woman who breastfeeds, and the decisions she makes prior to breastfeeding pave the way for the experience she will have of breastfeeding. For example, a woman may establish a desire to be discrete

when feeding her infant around others, or she may embrace the concept of demand feeding, which will affect her experience once breastfeeding begins. By making these determinations, a woman defines who she will be as a breastfeeding mother. This process of coming to know oneself as capable of breastfeeding takes time, and will occur even if a woman does not engage in a premeditated examination of her beliefs.

First-time mothers must come to know their bodies in a different way. Beginning with pregnancy, they must navigate their way through new territory, coming to understand their breasts as functional and capable of sustaining another life. This process is intrinsically linked to coming to know that their infants are enjoying breastfeeding (see Chapter Nine, Section 9.3.1). Each infant will respond differently at the breast, and learning to interpret their infant's cues and behaviour provides women with the confidence that they can care for their baby's needs through breastfeeding. This is evidenced by the participants who discussed at length how they came to understand breastfeeding as the ultimate source of comfort for their infants. These issues have also been highlighted in another New Zealand study (Dignam, 2001).

Understanding the breast as functional necessitates an understanding of the 'always-thereness' of breastmilk, and requires that women learn to manage the unpredictability of breasts that make milk. Women by necessity must embrace and accept their altered sense of self, or they will continually be uneasy with the re-configuration of their bodily boundaries and purpose. The inability to reconcile their bodily breastfeeding experiences can ultimately mean the end of breastfeeding for some women, especially in a culture that values independence and freedom (Beasley & Heritage, 1998/1999).

Unachievable ideals of motherhood, including breastfeeding practices, result in women interpreting themselves as failures (Maushart, 1999; Rubenstein, 1998; Swigart, 1998). This study has clearly demonstrated how women strongly desire to mirror the perfect breastfeeding mother, defined by having a thriving healthy infant who is settled in a routine. Such ideals can shroud the reality of breastfeeding, which follows a complex path of coming-to-know self as well as the infant. A lack of understanding of the process of establishing and maintaining breastfeeding can lead women to believe that they have failed at breastfeeding, either wholly or in part. Unfortunately, this study highlights that

many women find themselves in this circumstance, due in large part to the silencing of the more difficult aspects of breastfeeding.

10.4 The Shrouding of Breastfeeding in Silence

Once a woman conceives, she will at some point birth a baby, even if the infant has not fully matured – she has no control over this. And although a woman's body will initially produce milk for a newly birthed infant, women do have a choice whether to initiate and continue to breastfeed their babies. The reasons women give for not continuing to breastfeed have been well documented (Gunn, 1984; Mozingo et al., 2000; Scott et al., 2001; Vogel & Mitchell, 1998b). Despite this, the women who participated in this study discussed how a realistic picture of breastfeeding was unavailable from their social networks, antenatal education, or the various types of media. While the silence that exists around giving birth, and the overwhelming nature of becoming a mother, is well recognised in the literature (Maushart, 1999; Rubenstein, 1998; Swigart, 1998), the silence that exists around the realities of breastfeeding is not.

This silence undermines women's confidence when facing unexpected breastfeeding difficulties, and may lead to feelings of isolation in women struggling to establish breastfeeding, especially when peers seem to have 'perfect' infants. The belief that they are not measuring up to the ideal mother further silences them, the end result being an inability to seek the help they need. The participants in this study who struggled to breastfeed continued to do so in silence, and shared how they did not want to discuss their experience with others in case they were perceived to be failures, or somehow contributed to breastfeeding difficulties for other women who would at some point in the future initiate breastfeeding.

Breastfeeding is both a process and an art – it is a skill that takes time for both women and infants to learn, requiring both technical and embodied knowledge to become a transparent aspect of mothering. To gain this knowledge a woman must spend time 'doing' breastfeeding, which means navigating the pitfalls, a process that can be greatly facilitated by the sharing of experience through communities that are knowledgeable and supportive of breastfeeding.

The silencing of knowledge about the realities of establishing breastfeeding creates a vicious circle. If a woman struggles to breastfeed and subsequently weans earlier than planned, it is very likely that, to avoid being perceived as a failure, she will not discuss her embodied experience with others. Thus the embodied knowledge she has gained about the difficulties of establishing breastfeeding is effectively lost, being of no benefit to others because of her silence. By hiding the reality that breastfeeding is often difficult, the silence that shrouds breastfeeding fuels the propagation and reinforcement of unrealistic ideals of mothering. It is important therefore, to examine women's embodied experience of breastfeeding, regardless of its duration. This examination begins through addressing how our society has come to understand the breast as an object.

10.5 Re-conceptualising the Objectification of Breasts and Embodiment

Despite being strongly critiqued by philosophers such as Kant and Wittgenstein, Descartes' concept of the mind-matter split has endured in Western thinking (Kenny, 1998). We have come to accept and understand our bodies as objects, which has significance for breastfeeding women because they have inherited a language to describe breasts and breastfeeding that objectifies the experience. This dominant language silences the language that offers embodied understandings of breastfeeding, affecting women's ability to articulate their experiences of breastfeeding. The following discussion highlights the impact that objectification has on breastfeeding, beginning with the influence of biomedical technology.

It has been argued that breastfeeding does not comply with scientific technology, as breastfeeding can never be completely understood in reductionist terms, unlike artificial formula supplied to an infant in a bottle (Van Esterik, 1994). The inability to quantify breastmilk does not fit comfortably with objectified measurement, which complies with the ordered, and time-structured manner valued in our society (McBride, 1999). This need to quantify affects the manner in which health care practitioners, who work alongside women with young infants, approach breastfeeding; historically, artificial formula has been

the preferred means of infant feeding as it complies with the values of the medical community (Ryan & Grace, 2001).

These understandings also inform women's interpretations of breastmilk, and can negatively impact upon their confidence if their infants are unsettled, or do not grow in a way that mirrors the norm. If women are insecure about their infant's intake, they begin to doubt their embodied ability to respond to their infant's need. It takes time for women to feel comfortable with breastfeeding, as their previously-held values about objective measurement undermine their ability to trust themselves. Sam (Int. 12) provided a poignant example of a woman who struggled to trust her embodied capabilities. She concernfully monitored her infant's weight and provided complementary artificial formula, constantly questioning her ability to breastfeed her son fully. She struggled with overwhelming feelings of doubt and guilt about her ability to mother. The immeasurable nature of breastfeeding is meaningful for women who are struggling to breastfeed, and can ultimately undermine its establishment.

Another consequence of the objectification of bodies and breasts is that women have come to understand their bodies as something to be managed and tamed, something they have control over, and that the object-body can be manipulated into behaving in a manner acceptable to them. It is interesting to note that women tend to objectify their breasts when they are struggling to breastfeed (Schmied & Barclay, 1999). The meaning that this context holds is that breasts are something to be fought and managed with tools such as pumps, shields, and medications.

The interpretation of breasts as objects and the subsequent impact on the breastfeeding process silences the strong messages that come from embodied knowledge. For breastfeeding to continue over time, women must force themselves to pay attention to their embodied knowledge and respond to it. This study has highlighted how a woman's embodied knowledge rises to the fore as she dwells with breastfeeding. It is this embodied knowledge that is meaningful for breastfeeding women, rather than knowledge based on the breast as object, and therefore it holds sway over women's decisions about breastfeeding.

During the interview process women began to peel away the layers of language that described breastfeeding as a reductionist, objectified, medicalised process

that represented singly a source of nutrition, turning instead to a language that described breastfeeding as an embodied act. The knowledge that they gained through breastfeeding was turned into a language that saw the emergence of rich narratives to describe what it was, for them, to breastfeed.

What emerged from the participants' narratives about the embodied act of breastfeeding was that, for the majority of them, it was painful. This is consistent with the literature, which documents that breastfeeding pain is common (Blair et al., 2003; Blyth et al., 2002; Chapman et al., 1985; Hill et al., 1994; Page et al., 2003). Breastfeeding pain disrupts the establishment of a satisfying breastfeeding relationship. Some of the participants discussed how the pain of breastfeeding caused them to resent their infants, undermining their confidence in their ability to mother effectively. Women discussed how the pain resulted in a failure to form positive attachments with their infants; this was in itself a different source of embodied pain, which sometimes meant that women regretted those first weeks, months, and occasionally years spent with their children.

Despite experiencing breastfeeding pain, it became evident that a woman's interpretation of her embodied experiences can hinder or enhance her breastfeeding. For example, Eva encountered numerous difficulties that resulted in a great deal of pain for months, but because she believed breastfeeding difficulties would in time be overcome, she had the perseverance to continue (see Chapter 7 page 38, Eva. Int. 19. Para. 6,7,9&12). In contrast, Jan interpreted breastfeeding, and the pain she experienced as a result, as an entity over which she had no control, and she struggled to accept her body as breastfeeding (see Chapter Seven page 23 Int. 13 para. 65).

Ultimately, this research demonstrates that women are capable of generating their own embodied knowledge. Bartlett (2002) postulated that women's breasts are knowledgeable and supply women with invaluable information about breastfeeding; however, women need to learn to listen to this knowledge, which comes initially through just being aware of the messages being sent. As a woman dwells listening to her embodied knowledge, she can learn to interpret the messages from her breasts. This research has highlighted that, when a woman listens to her embodied knowledge, she is able to interpret, respond, and alter her breastfeeding practices accordingly, facilitating a positive

breastfeeding experience. However, women need the ability and resolve to persevere with the establishment of breastfeeding.

10.6 Breastfeeding Requires Perseverance

Many of the participants were not prepared for how challenging their breastfeeding experience was. For some, their belief that breastfeeding is an intrinsic part of mothering, and that it represents more than simply a source of nutrition, provided the motivation to persevere despite the difficulties they encountered. The manner in which women spoke about this was reminiscent of how Bottoroff (1990) described women's motivation to breastfeed as a test of perseverance. However, it was evident that for some of the participants in this study, breastfeeding was challenging because it was a direct affront to their conceptions of bodily boundaries. For these women, enfolding another being intimately required courage and a great deal of commitment, without which they could not have successfully established breastfeeding. This was especially true if they were struggling with feelings of embarrassment or pain.

Chapter Seven focused on how women choose a certain way of 'being-in-the-world' which, according to Merleau-Ponty (1981), will be favoured above other ways of being. This way of 'being-in-the-world' dictates how women will respond to breastfeeding. A woman may not have breastfeed an infant before, but her understandings of breasts, breastfeeding and artificial formula, developed over a lifetime, will direct the path she will take as a breastfeeding mother. A definitive example of this emerges when considering the issue of perseverance in the face of breastfeeding difficulty. What makes one woman continue to breastfeed when it's painful, while another would stop? It's my contention that women choose how much they are willing to go through to breastfeed, a choice that is directly influenced by their way of 'being-in-the-world'.

This research has offered a number of examples of women who have chosen to continue with breastfeeding despite difficulties. These women all discussed similar experiences with pain associated with breastfeeding; however, women's interpretations of their experience were directly related to the meaning that breastfeeding held for them. Women for whom breastfeeding was extremely

painful, and whose way of 'being-in-the-world' allowed them to persevere, found that the relationship with their infant that breastfeeding was facilitating made the pain worthwhile. Women need to believe that the benefits of breastfeeding far outweigh the difficulties experienced whilst breastfeeding if they are to persevere and prolong the duration of the breastfeeding relationship.

10.7 Prolonging Breastfeeding: Breastfeeding as Relationship

Breastfeeding does not occur in a vacuum; it must, by necessity, involve at least two people, typically a mother and her baby. Breastfeeding is a relationship. The participants in this study discussed how a part of forging a breastfeeding relationship was learning how to respond to their infant's cues and behaviours, facilitating a positive reaction from their infants. These initial interactions provided the basis for the establishment of their breastfeeding relationship.

This is particularly evident during the initial postnatal period when women dwell closely with their infants. A newborn consumes so much of a woman's time, and the focus of her thought and attention is directed accordingly. This 'beingwith' an infant enables women to be concerned with, and focused on, their infant. Being-with means that the two come to know each other, defining a dynamic relationship between mother and child, which is essential if breastfeeding is to continue.

During the initial breastfeeding period, women were typically concerned with the technical function of breastfeeding; however, those women who were able to persevere with breastfeeding came to know breastfeeding as comfort for their infants. They described this comfort as 'all encompassing', illustrating the potential power of the breastfeeding act, especially in a society that values settled infants (see: Beasley & Heritage, 1998/1999; Beasley, 2002; McBride, 1999). As women breastfeed over time, it becomes less about the technical act or function of breastfeeding, and increasingly about the relationship between themselves and their infants. Ultimately, the physical act of breastfeeding becomes transparent, as women came to know and understand breastfeeding as relationship. This is reminiscent of Heidegger's (1962) description of a

skilled carpenter using a hammer; the hammer ceases to exist in his hand and the 'becoming' of the wood is what emerges as significant.

Benner, a noted nursing researcher, has focused on understanding how nurses develop their nursing skills, which has implications when examining the issue of women gaining breastfeeding knowledge. She describes how acquiring the skills to nurse involves "learning to inhabit a familiar differentiated world through the development of concerns, perceptive skills and skilful comportment" (Benner & Wrubel, 1992, p. 8). Nurses progress through stages from novice to expert as they gain the experience and skill base needed to become proficient at nursing. Benner's understanding also helps to explain how women acquire the skills to breastfeed. Women need to learn to inhabit a differentiated world, and this is achieved through the development of concerns along with intuitive knowledge, which is developed by 'being-with' their infants through breastfeeding.

The women in this study described how they experienced a closeness with their infants as a result of breastfeeding them. It was a closeness born from mutual physical dependence on each other – the mother for the physical comfort of releasing milk, and the infant to have their hunger satisfied. Breastfeeding closeness came from spending significant amounts of time together, which facilitated coming to know in an intimate manner. This feeling in turn strengthened the relationship between them, and the women in this study believed that the work of breastfeeding assisted in forging a relationship whose closeness extended beyond the duration of the breastfeeding period. This relationship had a transforming effect on these women. Through breastfeeding, they developed embodied knowledge that increased their confidence in their ability to mother, and experienced relational closeness in a unique way.

As women dwell with breastfeeding they come to form their own interpretations of their experience. However, the influence of 'the they' on women's interpretations affects both their understanding of breastfeeding and how they experience their culture whilst breastfeeding.

10.8 'The They's' Influence on Breastfeeding Behaviour

Women's breastfeeding behaviour is mediated by history and current social and cultural understandings of breastfeeding. This concept has been demonstrated through the narratives that have emerged in this study, and support the findings of Ryan's (1999a) study into women's narratives in post-World War II New Zealand. These influences inescapably inform women's understandings and interpretations of their breastfeeding experience.

Prior to breastfeeding, women understand their breasts in accordance with their culture, and in New Zealand culture the breast is primarily interpreted as sexual (see Chapter Six, Section 6.2.3). Women feel pressured to conform to externally-prescribed body types, central to which is the concept of the breast as sexual rather than functional; therefore, women struggle with the changes that breastfeeding imposes on their bodies. The sexualized breast heightens women's anxiety in relation to breastfeeding. Women must come to understand their breasts as being more than sexual, and that they are capable of growing a thriving infant. This is a significant adjustment that can result in an uncomfortable struggle for women to reconcile their altered sense of self. Ultimately, breastfeeding must conform to standards of social behaviour. Unfortunately, these understandings can interfere with women's breastfeeding, interrupting a seamless transition to being a breastfeeding woman.

Another of 'the they's' major influences on breastfeeding is the conceptualisation of breasts as belonging to the private sphere; similarly, breastfeeding as a derivative is relegated to the same private domain. In consequence, women do not see other women breastfeeding. If they do, breastfeeding etiquette dictates that the breastfeeding woman will make it look effortless, discrete and ultimately an easy activity; if a woman cannot achieve this, she will not breastfeed around others (Kennedy, 2000; McLeod et al., 1998; Sheeshka et al., 2001). Thus 'the they' imposes silence on the experience of establishing breastfeeding, hiding it from the public domain and precluding women from understanding ahead of time that breastfeeding can be difficult.

This study has determined that women strongly desire to mirror the perfect breastfeeding mother as espoused by 'the they', which can be extremely difficult if they are struggling with feeding their infants. However, because of

breastfeeding's relegation to the private sphere, women must navigate society's rules about breastfeeding through engaging in the act of breastfeeding, and learning to reconcile the responses of others. Research has indicated that women are particularly receptive to the responses of those around them when breastfeeding (Benson, 2000; Dignam, 1998; Heath et al., 2002; McLeod et al., 1998; Murphy, 1999). This taking up of meaning occurs only whilst breastfeeding, and significantly influences decisions about how and when a woman will breastfeed.

The influence of 'the they' on breastfeeding also emerged from the women's narratives in discussions about prolonged breastfeeding being unacceptable. This unwritten rule was enforced by 'the they' through disapproving looks, and women generally understanding that it was not acceptable. Morse (1987) discussed the coercive nature of society in relation to prolonged breastfeeding, noting that previously-supportive networks began to undermine breastfeeding once the infant reached one year of age; as a result, women would either withdraw from publicly breastfeeding or wean their infant. The women participating in this study also echoed the narratives that emerged from both Morse's (1987) and Murphy's (1999) study into socially-prescribed weaning practices.

It is interesting to note that, when this research was presented in 2003 to postgraduate midwifery students enrolled in a paper on breastfeeding, the narratives on breastfeeding as care (see Chapter Nine) prompted much discussion about prolonged breastfeeding. Although "prolonged" was not defined for the students, the longest period for which a woman participating in this study breastfed was two years. In response to the shared narratives, the overwhelming message from the midwives was that breastfeeding that long was deviant. They espoused that women who engage in such prolonged breastfeeding are potentially abusive and needy co-dependents, who continue to breastfeed to their child's detriment. That health care professionals, who are supportive of breastfeeding, openly conceive of women and their breastfeeding relationships with their children in this manner, demonstrates the strength of the bias that exists around breastfeeding in New Zealand culture.

It is important to remember that 'the they' is neither good nor bad — it simply provides a framework by which being is interpreted. 'The they' provide prescribed, everyday ways of being-in-the-world, which offer particular

interpretations of being a woman who breastfeeds. Ultimately, this study has shown that the possibilities of being as breastfeeding women are closed down upon through being with 'the they'. However, if new societal understandings, and interpretations, of breastfeeding were to emerge, women would be offered different types of language through which they could interpret their breastfeeding experiences.

10.9 Alternative Interpretations for Women's Breastfeeding Experiences

Once women have given birth they become privy to a new world of knowledge and understandings about birthing (Maushart, 1999), and this is also true of breastfeeding. The women who participated in this study discussed how, prior to breastfeeding, they understood it to be an effortless and natural process that takes place between a mother and her infant. Once they had attempted to breastfeed, they came to understand it as problematic. This view is becoming increasingly common amongst women who are mothers, as evidenced by popular press articles discussing how difficult women are finding breastfeeding (see Sarney, 2003/2004). However, such narratives tend not to discuss breastfeeding relationships that last longer than three months. While acknowledging that breastfeeding requires considerable time and perseverance to become comfortable (Blair et al., 2003; Page et al., 2003), there needs to be room for interpretations of breastfeeding as pleasurable, not just problematic. If women's interpretations of breastfeeding experiences are mediated by the narratives available to them, as demonstrated in this study and in Ryan (1999a), there is a need for access to an interpretation that accepts the pleasurable aspects of breastfeeding.

The participants in this study who espoused an interpretation of breastfeeding as pleasurable talked about the exclusive nature of their breastfeeding relationship with their children. It is a relationship they did not wish to share. Dignam (2001) discusses how, for intimacy to occur between a woman and her baby during breastfeeding, a significant amount of time needs to be invested in the relationship. She also reports that women who experience intimacy with their infants value the exclusive nature of the relationship. It is something no one else shares. The women in this study chose to use the word 'closeness' to

describe their relationship with their baby; however, there were distinct parallels between Dignam's work on intimacy and breastfeeding and the narratives that emerged in this study. Ultimately, it takes time and commitment to the exclusivity of the relationship for a woman to come to know her baby in a close or intimate way. The closeness that some women experience in their breastfeeding relationship does not occur in a hurried manner. Breastfeeding closeness emerges through the maturing breastfeeding relationship.

The concept that both a woman and her baby can flourish because of their relationship has been discussed by Traina (2000). An infant's need for affection can also be enjoyed by its mother, which in turn facilitates the flourishing of being. Within the context of this research topic, it emerged that women flourish because of their relationship with their infants. Through breastfeeding, both women and their infants are enabled to progress towards their 'ownmost' potential; however, breastfeeding success requires that women exhibit authentic care for their infant through their commitment to breastfeeding. This commitment is exemplified by those participants who breastfed their infants longer than normal³², and felt that they were perceived as deviant by society because of it. Because of the strength and confidence these women gained from the closeness of their relationships with their infants, they continued to breastfeed despite intense social pressure to wean.

Women who participated in this study described breastfeeding their infants using the following language: it's all about the closeness, touching, cocooning, all-encompassing, exclusivity. Over time, women who breastfeed come to understand breastfeeding as more than nutrition, and the participants employed language that reflected this; it's about a relationship, a physical relationship that provides for both the women's and infant's wellbeing. The relationship is, therefore, privileged.

There is also a need to address issues of maternal sensuality and sexuality when discussing breastfeeding. Breastfeeding as an intrinsic part of motherhood is unavoidably sensual; unfortunately, the interpretation of it being

³² In New Zealand, 19-21 percent of European-descent, 13-14 percent of Maori, and 17-18 percent of Pacific Island babies are fully breastfed at six months (Ministry of Health, 2002). Fully breastfed means that the infant is having no other liquids or solids except water or medicine.

a sensual experience is silenced in the public domain because it is considered to be a deviant aspect of womanly experience. However, the language used by those women in this study who successfully established breastfeeding reflects that breastfeeding is sensual and can be linked to women's experience of sexuality. Some of these women would fall into the 'significant minority' referred to by Traina (2002, p. 373), who asserts that "a significant minority adhere to the more sensual description of motherhood because it makes more sense of their embodied experience."

This research reveals that women can transcend 'the they's' critiques of breastfeeding as singularly functional. Indeed, some participants opted for an interpretation of breastfeeding that embraced it as a sensual act. Society needs to accept that women can and do derive sensual pleasure from breastfeeding – it is a reward for offering themselves to their infants unreservedly, and should not be understood as deviant. Access to these understandings may encourage women to persevere with breastfeeding.

Women who believed that breastfeeding offered them a positive experience reveled in their embodied ability to nurture life. Breastfeeding was tangible evidence to them of the powerful position they held, even if it was unseen by others. This concept is related to Dennis's (1999) research into breastfeeding and confidence, which demonstrates that, prior to giving birth, if women are confident in their ability to breastfeed they are more likely to succeed at breastfeeding. Conversely, this study indicates that women gain confidence while breastfeeding through a gradual process of coming to understand that their bodies are amazing at responding to infant's needs (see Chapter Nine, Section 9.5), both physically and emotionally. Through breastfeeding, women are enabled to experience an otherwise unknown aspect of their embodied capabilities – physically growing another life while at the same time providing for their own wellbeing.

Breastfeeding is about life, about women living and opting to use their embodied abilities to sustain life. It may not be easily accomplished, but in doing so women creatively respond to their infant's needs and experience an aspect of life otherwise not available to them. Women in this study who engaged in breastfeeding over longer periods than most reported experiencing a relationship and closeness that no other experience had offered to them. Breastfeeding provides women with a privileged understanding of embodied

living and giving of life. It is important that women support each other, sharing the knowledge they have gained through the embodied act of breastfeeding.

10.10 Rebuilding Breastfeeding Support

As women dwell together, supporting each other and sharing breastfeeding narratives, they gain knowledge and encourage each other to persevere in the face of difficulties. This research has captured narratives of women as they navigate their way through the interpretations of breastfeeding offered by 'the they.' Women who persevere with breastfeeding despite encountering difficulties are powerful and strong women; they have managed to flee 'the they's' interpretations and prescribed ways of being. Women who choose to engage with their infants in ways that allow them to respond to their embodied knowledge, refusing to feel guilty over something they have no control over, exhibit strength. These women draw closer to their 'ownmost' potential as a result of their breastfeeding experience. However, the reality is that women are frequently not in a position to unpack and examine the force that is exerted by 'the they'. This research demonstrates that a woman's longing to offer her infant the best is a powerful motivator, and it is this desire that holds women captive to the compulsion to reflect the perfect breastfeeding woman and mother.

Women need to critically examine idealised notions of breastfeeding that confine their interpretations to that of being a "good" or "bad" mother. Women need to question their beliefs and understandings of what breastfeeding is, and what it will mean to them in the context of their relationship with their infant. This is a challenging task in our society, with its ready-to-hand frameworks by which we can measure our performance. Instead, women need to create their own individual framework by which their experience of breastfeeding can be measured. One such interpretation might be the one suggested by Hunter (1998), who promotes the use of the phrase 'good enough breastfeeding' to describe women's experiences. Such an interpretation would provide women the opportunity to understand their experiences through a framework that reflects the unpredictable nature of breastfeeding.

The loss of a breastfeeding culture has significantly undermined women's access to breastfeeding support networks. This loss can be attributed to a number of factors, most notably the demise of extended family structures, the relegation of breastfeeding to the private sphere, and the medicalisation of breastfeeding. It is also worth noting that community organisations that facilitate women-to-women breastfeeding support are often viewed as extremist. This interpretation emerges in light of the difficulties many women have experienced with breastfeeding, which makes it difficult for them to resolve personal experiences with the ideals publicly espoused by such groups. However, through the sharing of experiences and embodied knowledge gained through breastfeeding, women can find the support they need to initiate and continue to breastfeed.

As long as breastfeeding remains a silenced aspect of women's embodied experience, nothing will change. Language has the ability to create. This study is an attempt to speak the unspoken and give rise to alternative understandings about breastfeeding in New Zealand that enable women to fully experience their embodied capabilities.

10.11 Implications for Health Care Professionals Working Alongside Breastfeeding Women

This research offers health care professionals meaningful insight into what it is to be a breastfeeding woman in New Zealand. Traditionally, research that has explored the issue of breastfeeding has focused on demographic trends, biomedical topics, and interventions by which the breastfeeding process can be managed. In contrast, this study set out to explore the meaning that breastfeeding has for women who are actively engaged in breastfeeding. It has focused on their narratives to facilitate an understanding of their lifeworld as it emerges in accordance with their 'thrownness', as opposed to that of health care professionals.

What has emerged as paramount is the need for health care professionals to take time to listen to women's embodied experiences of breastfeeding. Any planned interventions need to take into account the culture, community and social supports that form the basis for women's previously-held understandings

about breastfeeding. Failure to do this will make any suggested interventions ineffective, as the context in which women reside will always dictate their interpretive frameworks. Ignoring a woman's embodied experience when assisting her to establish and continue breastfeeding will also undermine her ability to listen to her embodied knowledge – a skill that is crucial to the longevity of breastfeeding.

The vantage point offered to health care professionals who listen is preeminent, resulting in interventions that will support and nurture a woman who
seeks to prolong breastfeeding. As described above, the breastfeeding process
requires that women learn to listen to their embodied knowledge. This can be
facilitated through health care professionals' awareness that each breastfeeding
dyad is unique and will require certain adaptations for breastfeeding to work for
both the mother and her infant. Health care professionals need to foster
women's confidence in their ability to listen to their embodied knowledge and
adjust breastfeeding practices accordingly. In so doing, they will enable
breastfeeding women to speak the previously unspoken, increasing their
confidence in their embodied ability to breastfeed, and assisting them in
developing language that facilitates breastfeeding longevity.

Another important implication of this research is the key role that interpretations of embodied experiences play in women's view of their performance as breastfeeding mothers. Offering new and different ways of interpreting breastfeeding experience will alter the language they have at their disposal for describing their breastfeeding experience. For example, if a woman considers that she has 'failed' at breastfeeding one of her children, she may then interpret herself as a failure at breastfeeding, undermining future breastfeeding experiences; however, if she is enabled through language to gain an interpretation of each breastfeeding experience as unique, this may provide her with confidence that she is capable of breastfeeding subsequent offspring.

It is important for health care professionals to be aware that breastfeeding offers women a unique and new embodied experience. In particular, the experiences of a first time breastfeeding mother are completely new: her embodied sensations, interpretations of 'the they' and strong urge to care for her infant are all things that she has never experienced before. Her perceptions of them will be influenced by many factors, and may be positive or negative. A crucial part of the health care professional's role is to assist her to acknowledge and resolve

her experiences with breastfeeding if they have been negative. Women need to be supported in their role as mothers, and the resolution of an 'emotionally crippling' breastfeeding experience is paramount. This requires a high level of courage, commitment and empathy on the part of the health care professional; unfortunately, this research attests that we often contribute to negative breastfeeding experiences, rather than being of help.

10.12 Limitations of this Research

Unstructured interviews acquire validity and reliability through audit paths and decision trails, which are made evident when the researcher is documenting the analysis of the transcripts (Field & Morse, 1985; Omery, 1983). The validity and reliability of this research has been addressed by incorporating the participants' conceptualisations of issues raised during the interviews (Munhall & Boyd, 1993). This was achieved by constantly returning to the women's narratives, allowing their original intent to be continually called forth. Being mindful of these narratives throughout the process enabled the research to remain firmly established in the participants' reports of their experiences of the lifeworld.

It is widely acknowledged that research methods that employ unstructured interviews have limited applicability when attempting to extrapolate to wider populations; however, qualitative methods do provide in-depth perspectives on phenomena (Baker et al., 1992), thus increasing the reader's understanding of the participants' experience. By engaging with reflective lifeworld research methods, health care professionals can extend their personal horizons, allowing them to approach individuals in innovative and responsive ways. In the context of this study, reflective lifeworld research offers meaningful insights for health care professionals working alongside breastfeeding women in New Zealand, and demonstrates the need for contextualised, individual breastfeeding support.

This study allowed participants to self-select themselves for participation. This may have resulted in a disproportionate number of participants who had either unusually positive or negative breastfeeding experiences, creating a bias toward narratives that are strongly positioned and not necessarily reflective of the typical breastfeeding experience. However, the research findings have been presented in a number of forums, and the feedback that has been received

indicates that many women experience a resonance with the participants' narratives.

A limitation of the use of a reflective lifeworld research methodology is that it's intended to be interpretive, not emancipatory. I have found it very difficult to lay aside my desire, as a feminist, to examine the power structures that influence breastfeeding women's world and offer an emancipatory discourse by way of explanation. In trying to avoid laying a particular interpretive framework, such as feminism, over the women's narratives, I have attempted to limit my interpretations of women's breastfeeding experiences from Chapters Six through Nine, focusing only on the narratives as they have emerged from the transcripts. Instead, I have offered my own interpretive account in the present chapter, and have noted in Chapter One that I position myself as a feminist, and am unable to extricate myself from my 'thrownness'.

It should also be noted that, as a researcher, I am held sway by the scientific tradition, as evidenced by the traditional approach taken in setting out the research in this document. Despite its traditional format, however, the document has not been written in the traditional order. Writing began by outlining my 'thrownness', historicity and horizon, the understanding of which assisted me in examining the women's narratives. This approach allowed me to take into account those things that held meaning for me, and lay them aside accordingly. The methodology chapter and method chapter were completed next, followed by the findings chapters. The literature review chapters were developed after Chapters Six through Nine were completed, and the Discussion Chapter was written last. This atypical thesis development reflects my understanding of reflective lifeworld research as a methodology and method, and allowed me to focus on the women's narratives to the exclusion of what was previously written about the experience of breastfeeding.

10.13 Future Directions for Research

This study did not specifically target particular ethnic groups, and the participants were predominantly New Zealanders of European descent.

Additional in-depth qualitative research that is expressly and appropriately directed towards women from differing ethnic backgrounds, such as Maori and

the Pacific Island and Asian nations, by researchers immersed in their specific world view, is needed. This would enable health care professionals to direct interventions that are specifically appropriate for women from these differing people groups.

Research that examines breastfeeding women's partners' experiences of breastfeeding would add to the knowledge base of New Zealand's breastfeeding culture, as would engaging women who had not breastfeed in conversations about breastfeeding. An examination that specifically targeted breastfeeding women's experience of health care professionals would also be beneficial. Women who participated in this study highlighted significant concerns about the professional support they received whilst they were attempting to establish breastfeeding. These concerns are documented in this study; however, a more extensive examination is warranted so that health care professionals can critically examine the manner in which they engage with breastfeeding women.

This research has highlighted that many women struggle in the initial months of breastfeeding. Its findings could serve as the basis for a larger study into women's initial breastfeeding experiences, exploring which interventions assisted them to continue with breastfeeding, and which contributed to early weaning of their infants.

The use of technology to assist with the establishment of breastfeeding is another area of interest for future research. The women participating in this study spoke many times about the different pieces of technology they employed to assist them to breastfeed. For some participants, breastfeeding began to work the moment they threw away these devices and returned to baby-to-nipple and skin-to-skin contact. A theoretical understanding of how technology contributes to breastfeeding may allow health care professionals to unpack the concepts underpinning commonly-utilised implements, and their usefulness for breastfeeding dyads.

The impact of formalised breastfeeding support groups on women's breastfeeding experience in New Zealand also warrants additional research. If such groups were demonstrated to be beneficial, women might be more readily encouraged to form communities that support rather than undermine their breastfeeding experience.

Research into women's experience of maternal sensuality and sexuality in relation to breastfeeding is also warranted. Some of the women who participated in this study hinted that this was an aspect of their breastfeeding experience; however, cultural understandings of motherhood prevented an open discussion of this issue. If women were aware that breastfeeding can contribute to their emotional well being in this way, it may encourage them to persevere with the initiation phase of breastfeeding.

Finally, the use of other theoretical perspectives to explore other interpretations of women's experience of breastfeeding may offer additional insight. In particular, the use of a feminist methodology that examines the hegemonic discourses for how they support a patriarchal society at the expense of breastfeeding women would be useful. A researcher embracing a post-structuralist stance would offer a differing perspective. Women's narratives on breastfeeding are rich and complex, reflecting multiple contexts, and the use of a variety of lenses to interpret their stories can only enhance that which is known about breastfeeding.

10.14 Conclusion

This chapter has explored how reflective lifeworld research has offered an innovative perspective from which to examine issues surrounding breastfeeding in New Zealand. This research methodology has enabled the emergence of new knowledge that is meaningful for health care professionals working alongside women who are breastfeeding within New Zealand. Through breastfeeding, women gain access to unique embodied experiences that would have otherwise remained undiscovered. As women gain experience at breastfeeding, they come to understand it as a complex process that takes place between themselves and their infant. Breastfeeding longevity is dependent on a woman receiving knowledge and support from the community in which she finds herself; however, knowledge about the realities of breastfeeding is often hidden from women.

Finally, this chapter has highlighted potential limitations of this study and how future research can further add to our knowledge of breastfeeding. The next

chapter offers a summative account of this research and the thesis that has emerged from this study into women's experience of breastfeeding.

Chapter Eleven

The Conclusion

11.1 Introduction

This chapter offers a final account of the central thesis and major findings of this research study into the breastfeeding women's experiences of breastfeeding in New Zealand. A summative account of the written report is also offered.

11.2 Revisiting the Thesis

This research holds central the thesis that breastfeeding offers women experience that would otherwise remain hidden to them. Breastfeeding provides women with understandings of their embodiment, 'the they', and care for another, in a manner that they had previously not been able to experience. This thesis encompasses a number of sub-themes, which further illuminate the nature of breastfeeding as an opening to experience, including: breastfeeding as silenced, breastfeeding as embodied, 'the they's' interpretative framework, and breastfeeding as relational.

A key finding that emerges from the women's narratives in this study is that the realities of what it is to be a breastfeeding woman are hidden from those who have not yet breastfed, undermining their ability to establish breastfeeding when they become mothers. Women will always interpret their breastfeeding experiences in accordance with their thrownness. The culture in which they reside provides the language through which they understand breasts and breastfeeding. In New Zealand culture, women interpret their performance as breastfeeding women in relation to the espoused image of the perfect breastfeeding mother.

Motivated by a strong desire to provide their infant with the best, and influenced by societal perceptions through 'the they', women hide from public scrutiny if

they are unable to mirror the prescribed image. Not only does this leave them open to feelings of failure, guilt and sometimes depression, it prevents them from accessing support and hides their knowledge from other women, perpetuating a circle of silence. If knowledge of the realities of breastfeeding is made accessible to all women, the image of the perfect breastfeeding mother will be altered, effectively breaking the circle. Unveiling this knowledge, and establishing it as a normative part of preparing to mother, will enable women to persevere through the difficulties associated with initiating breastfeeding.

Breastfeeding is unavoidably an embodied experience. Despite this, we have inherited a language that constructs breasts and breastfeeding as objects that need to be controlled and managed with technology. This means that, during the initial stages of breastfeeding, women interpret their personal breastfeeding experiences in accordance with this body-as-object framework. However, this research has determined that, as women continue to breastfeed, they come to understand breastfeeding as an embodied act and create a language to reflect that. For some women, their embodied experience of breastfeeding is one of pain, tiredness and grief; for others it becomes a self-defining accomplishment.

Participants who were able to successfully establish and continue to breastfeed their infants spoke of breastfeeding as relational. The relationship between these women and their infants was nurtured through the time spent together breastfeeding, a process that involved coming to know their infants in a close manner. They valued this closeness for its unique, exclusive nature, and held the breastfeeding relationship privileged above all others. The breastfeeding relationship held open the possibility that women could achieve their ownmost potential, as they experienced authentic care for their infant. Ultimately, their own well-being was significantly enhanced through breastfeeding their infants.

11.3 A Review of the Chapters

Chapter One introduced this study and provided a justification for its pursuit. It outlined the context in which the study took place and introduced reflective lifeworld research as the methodology. This chapter also allowed me to position myself in relation to the research process, outlining my historicity and horizon when approaching the study.

Chapter Two outlined the history of infant feeding in New Zealand, examining this country's heritage and how it has influenced the development of breastfeeding knowledge. It also explored the influence of global policy documents on New Zealand's national strategies for increasing breastfeeding rates.

The literature that explores the experience of breastfeeding was examined in Chapter Three. This review highlights how breastfeeding falls into the domain of health care professionals at the expense of women's own knowledge about breastfeeding. Researchers' interests are served by their research, which elevates them to the position of experts who can redeem breastfeeding women from their breastfeeding problems. It is from this position that I have argued for a research approach that values women's embodied knowledge.

Chapter Four began with an overview of the development of Western philosophy, demonstrating the philosophical foundations for the chosen methodology of reflective lifeworld research. The principles inherent in this methodology were explored for their applicability to the phenomenon of breastfeeding women's experience of breastfeeding in New Zealand.

Chapter Five explored how reflective lifeworld research provides a method by which one can explore the phenomenon of being a breastfeeding woman. It outlined the research process in detail, highlighting how ethical concerns involved in research of this type were considered. Finally, issues related to trustworthiness were explored within the context of this research.

Chapters Six through Nine relayed the participants' narratives. Chapter Six focused on how breastfeeding women experience 'the they', capturing how women believe they must adhere to externally-prescribed images if they are to be considered a good breastfeeding mother. This chapter highlights how women struggling with breastfeeding suffer in silence because they do not want to be perceived as not being good enough. Ultimately, women's breastfeeding experiences are most often interpreted through 'the they's' interpretive framework, closing down on women's potential for being.

Chapter Seven explored how women have inherited a language that interprets their breasts as objects. Heidegger's account of objects is used as a lens to interpret these narratives; however, it is demonstrated that this

conceptualisation has limited value for understanding breastfeeding. Instead it is proposed that a different language is needed to more fully understand the phenomenon of being a breastfeeding woman. A language that articulates the embodied experience of breastfeeding was, therefore, explored in Chapter Eight.

A differing approach to interpreting women's breastfeeding experiences is offered in Chapter Eight. Examining the participants' narratives from the vantage point of Merleau-Ponty's (1981) exploration of embodiment, it emerged that the embodied breastfeeding experience is powerful and transformative in both positive and negative ways.

The falling away of breastfeeding as function, and the emergence of breastfeeding as relationship, was examined in Chapter Nine. This chapter discussed how the participant's expressed authentic solicitude for their infants, and for those who continued to breastfeed, the breastfeeding relationship was privileged because it enabled both the women and their infants to move towards their ownmost potential.

Finally, Chapter Ten explored the implications this research has for health care professionals who work alongside breastfeeding women. The major findings of this chapter were discussed in relation to the available literature, and the researcher's interpretations of the findings were also offered. It also noted the limitations of this work and proposed directions for further research.

11.4 Conclusion

Reflective lifeworld research has offered a methodology and method for examining the lifeworld of a breastfeeding woman. It has facilitated the emergence of knowledge that values the transformative effect that breastfeeding holds for women. Through its guiding principles of openness, historicity and the hermeneutic circle, it has emerged that breastfeeding offers women a unique embodied experience that allows them to experience 'the they' and authentic care for another.

Despite the undeniable power of the breastfeeding experience in a woman's life, the realities of breastfeeding remain hidden to those who have not breastfed. To establish breastfeeding, women need to come to understand their embodied breastfeeding knowledge, and have the language that enables them to interpret this knowledge. The availability of an embodied breastfeeding language would unveil the realities of breastfeeding, and offer an alternative perception of "the perfect breastfeeding mother"; this may offer an avenue for increasing breastfeeding rates in New Zealand.

This research also offers fresh understandings for health care professionals working alongside breastfeeding women. It is important that professionals examine their own historicity and prejudices surrounding breastfeeding, and listen to individual breastfeeding women's experiences, taking into account the context women are in when planning interventions. The valuing of embodied knowledge and its associated language will encourage women to articulate their breastfeeding experiences, and increase their confidence in their embodied abilities and knowledge.

References

- Abel, S., Park, J., Tipene-Leach, D., Finau, S., & Lennan, M. (2001). Infant care practices in New Zealand: A cross-cultural qualitative study. *Social Sciences and Medicine*, *53*, 1135-1148.
- Aldridge, D., & Stevenson, C. (2000). Social poetics as research and practice: Living in and learning from the process of research. *Nursing Inquiry, 8*, 19-27.
- Alm, B., Wennergren, G., Norvenius, S., Skjaerven, R., Lagercrantz, H., Helweg-Larson, K., & Irgens, L. (2002). Breast feeding and the sudden infant death syndrome in Scandinavia, 1992-1995. Archives of Disease in Childhood, 86, 400-402.
- American Academy of Pediatrics. (1997). Breastfeeding and the use of human milk. *Pediatrics*, *100*, 1035-1039.
- Annells, M. (1996). Hermeneutic phenomenology: Philosophical perspectives and current use in nursing research. *Journal of Advanced Nursing*, 23, 705-713.
- Anyon, C. (1976). Normal haemoglobin values in urban Polynesian infants: The possible deleterious influence of artificial feeding. *The New Zealand Medical Journal*, *84*, 474-476.
- Apple, R. (1994). The medicalization of infant feeding in the United States and New Zealand: Two countries, one experience. *Journal of Human Lactation*, *10*(1), 31-37.
- Arora, S., McJunkin, C., Wehrer, J., & Kuhn, P. (2000). Major factors influencing breastfeeding rates: Mother's perception of father's attitude and milk supply. *Pediatrics*, 106(5), 67e-72.
- Auerbach, K. (1984). Employed breastfeeding mothers: Problems they encounter. *Birth*, *11*(1), 61-64.

- Auerbach, K. (1988). Beyond the issue of accuracy: Evaluating patient education materials for breastfeeding mothers. *Journal of Human Lactation*, *4*(3), 108-110.
- Auerbach, K. (1994). Qualifying breastfeeding. *Journal of Human Lactation*, 10(2), 69-70.
- Australian College of Paediatrics. (1998). Policy statement on breastfeeding. Journal of Pediatric and Child Health, 34, 412-413.
- Avery, M., Duckett, L., & Frantzich, C. (2000). The experience of sexuality during breastfeeding among primiparous women. *Journal of Midwifery and Women's Health*, 45, 227-236.
- Bachrach, V., Schwarz, E., & Bachrach, L. (2003). Breastfeeding and the risk of hospitalization for respiratory disease in infancy: A meta-analysis. Archives of Pediatric and Adolescent Medicine, 157, 237-243.
- Baker, C., Wuest, J., & Stern, P. (1992). Methods slurring: The grounded theory phenomenological example. *Journal of Advanced Nursing*, 17, 1355-1360.
- Bartlett, A. (2002). Breastfeeding as headwork: Corporeal feminism and meanings for breastfeeding. *Women's Studies International Forum*, 25, 373-382.
- Basire, K., Pullon, S., & McLeod, D. (1997). Baby feeding: The thoughts behind the statistics. *New Zealand Medical Journal*, 110, 184-187.
- Baumslag, N., & Michels, D. (1995). *Milk, money, and madness: The culture and politics of breastfeeding.* London: Bergin & Garvey.
- Beasley, A. (1991). Breastfeeding studies: culture, biomedicine, and methodology. *Journal of Human Lactation*, 7(1), 7-14.
- Beasley, A. (1996a). Breastfeeding for the first time: A critical-Interpretative perspective on experience and the body politic (Vol. 1). Palmerston North: Massey University.

- Beasley, A. (1996b). The medicalisation of breastfeeding an example of biomedical hegemony. *Sites*, *32*, 50-62.
- Beasley, A. (2002). Call for action on breastfeeding promotion. Public Health Association. Retrieved March 1, 2004, from http:///www.pha.org.nz/phadocs/breastfeedingjuly2002.doc
- Beasley, A., Chick, N., Pybus, M., Weber, J., Mackenzie, D., & Dignam, D. (1998). 'I was scared I had run out of milk!': Breastfeeding and the perception of insufficient milk among Manawatu mothers. In A. Beasley, & A. Trlin, (Eds.), *Breastfeeding in New Zealand: Practice, problems and policy* (pp. 57-74). Palmerston North: The Dunmore Press.
- Beasley, A., & Gordon, R. (1998/1999). Insufficient milk syndrome. Breastfeeding Communiqué, 9, 19-21.
- Beasley, A., & Trlin, A. (Eds.). (1998). *Breastfeeding in New Zealand: Practice, problems and policy*. Palmerston North, New Zealand: The Dunmore Press Limited.
- Beasley, B., & Heritage, A. (1998/1999). Breastfeeding is a cultural practice. Breastfeeding Communiqué, 9, 15-17.
- Becker, G. (1992). Breastfeeding knowledge of hospital staff in rural maternity units in Ireland. *Journal of Human Lactation*, *8*(3), 137-142.
- Beckman, D. (1977). The mechanical baby: A popular history of the theory and practice of child raising. Westport, Connecticut.: Lawrence Hill & Company.
- Benn, C., Budge, C., Anderson, A., & Wright, B. (2002). Breastfeeding help and support in hospital: A survey of women's experiences. *Breastfeeding Communiqué*, 13, 7-18.
- Benner, P. (1994). Interpretive phenomenology. London: Sage Publications.
- Benner, P. (2000). The roles of embodiment, emotion and lifeworld for rationality and agency in nursing practice. *Nursing Philosophy, 1*, 5-19.

- Benner, P., & Wrubel, J. (1992). Gaining a differentiated clinical world in critical care nursing. *Advances in Nursing Science*, *14*(3), 13-28.
- Benson, S. (2000). Adolescent mothers' experience of parenting and breastfeeding: A descriptive study. *Breastfeeding Communiqué*, *11*, 19-27.
- Bentovim, A. (1976). Shame and other anxieties associated with breast-feeding:

 A systems theory and psychodynamic. Oxford: Elsevier Excerpta

 Medica.
- Bergum, V. (1991). Being a phenomenological researcher. In J. Morse, (Ed.), Qualitative nursing research: A contemporary dialogue (revised ed., pp. 54-71). New York: Sage Publications.
- Bhandari, N., Bahl, R., Mazumdar, S., Martines, J., Black, R., & Bhan, M. (2003). Effect of community-based promotion of exclusive breastfeeding on diarrhoeal illness and growth: A cluster randomised controlled trial. *The Lancet*, *361*, 1418-1423.
- Blair, A., Cadwell, K., Turner-Maffei, C., & Brimdyr, K. (2003). The relationship between positioning, the breastfeeding dynamic, the latching process and pain in breastfeeding mothers with sore nipples. *Breastfeeding Communiqué*, 14, 64-72.
- Blum, L. (1993). Mothers, babies, and breastfeeding in late capitalist America: The shifting contexts of feminist theory. *Feminist Studies*, *19*, 291-311.
- Blyth, R., Creedy, D., Dennis, C., Moyle, W., Pratt, J., & De Vries, S. (2002).

 Effect of maternal confidence on breastfeeding duration: An application of breastfeeding self-efficacy theory. *Birth*, 29(4), 278-284.
- Bottorff, J. (1990). Persistence in breastfeeding: A phenomenological investigation. *Journal of Advanced Nursing*, *15*, 201-209.
- Bottorff, J., & Morse, J. (1990). Mothers' perceptions of breastmilk. *JOGNN: Journal of Obstetric Gynaecological and Neonatal Nursing, 19*, 518-527.

- Bozoky, I., & Corwin, E. (2001). Fatigue as a predictor of postpartum depression. *JOGNN: Journal of Obstetric, Gynaecological and Neonatal Nursing*, *31*, 436-443.
- Britton, C. (1997). 'Letting it go, letting it flow': Women's experiential accounts of the letdown reflex. *Social Sciences in Health, 3*(3), 176-187.
- Bryant Merrill, E. (1987). Learning how to mother: An ethnographic investigation of an urban breastfeeding group. *Anthropology and Education Quarterly*, 18, 220-240.
- Burnham, D., & Fieser, J. (2001). *Rene Descartes*. Internet Encyclopaedia of Philosophy. Retrieved November 06, 2001, from http://www.utm.edu/research/iep/d/descartes.htm
- Callister, P., Podmore, V., Galtry, J., & Sawicka, T. (1995). Striking a balance: Families, work, and early childhood education. Wellington: New Zealand Council for Educational Research.
- Carlson Gielen, A., Faden, R., O'Campo, P., Brown, H., & Paige, D. (1991).

 Maternal employment during the early postpartum period: Effects on initiation and continuation of breast-feeding. *Pediatrics*, 87, 298-305.
- Carter, P. (1995). Feminism, breasts and breastfeeding. London: Macmillan Press Ltd.
- Cavendish, A. (1964). Early Greek Philosophy. In D. O'Connor, (Ed.), *A Critical History of Western Philosophy* (pp. 1-13). New York: The Free Press.
- Chalmers, B., Ransome, O., & Herman, A. (1987). Psychosocial factors related to infant feeding patterns. *Journal of Reproductive and Infant Psychology*, *5*, 153-164.
- Chapman, J., Macey, M., Keegan, M., Borum, P., & Bennett, S. (1985).
 Concerns of breast-feeding mothers from birth to four months. *Nursing Research*, 34, 374-377.

- Chezem, J., Montgomery, P., & Fortman, T. (1997). Maternal feelings after cessation of breastfeeding: Influence of factors related to employment and duration. *Journal of Perinatal and Neonatal Nursing*, 11(2), 61-70.
- Chulada, P., Arbes, S., Dunson, D., & Zeldin, D. (2003). Breastfeeding and the prevalence of asthma and wheeze in children: Analyses from the Third National Health and Nutrition Examination Survey, 1988-1994. *Journal of Allergy & Clinical Immunology*, 111, 328-336.
- Cody, J. (1953). A man of two worlds: A biography of Sir Maui Pomare. Wellington: A.H. & A.W. Reed.
- Collaborative Group on Hormonal Factors in Breast Cancer. (2002). Breast cancer and breastfeeding: Collaborative re-analysis of individual data from 47 epidemiological studies in 30 countries, including 50 302 women with breast cancer and 96 973 women without the disease. *Lancet*, 360, 187-195.
- Cooper, D. (1999). *Existentialism: A Reconstruction*. Oxford: Blackwell Publishers.
- Corrao, G., Tragnone, A., Caprilli, R., Papi, C., Andreoli, A., Paolo, M., Riegler, G., Rigo, G., Ferrau, O., Mansi, C., Ingrosso, M., & Valpiani, D. (1998). Risk of inflammatory bowel disease attributable to smoking, oral contraception and breastfeeding in Italy: A nationwide case-control study. *International Journal of Epidemiology*, 27, 297-404.
- Coubrough, L. (1999). Breastfeeding definitions for monitoring the national health outcome targets in New Zealand: Review of the evidence and recommendations. Final Report: Prepared for Ministry of Health.

 Wellington: Ministry of Health.
- Cronin, C. (2003). First-time mothers identifying their needs, perceptions and experiences. *Journal of Clinical Nursing*, 12, 260-267.
- Cumming, R., & Klinebergm, R. (1993). Breastfeeding and other reproductive factors and the risk of hip fractures in elderly women. *International Journal of Epidemiology*, 22, 684-691.

- Dahlberg, K., & Nancy, D. (1997). A lifeworld paradigm for nursing research. *Journal of Holistic Nursing*, *15*, 303-317.
- Dahlberg, K., Drew, N., & Nystrom, M. (2001). *Reflective Lifeworld Research*. Sweden: Studentlitteratur.
- Davidson, F. (1984). Infant feeding in New Zealand: A review of infant feeding practices from the 1890s to the present day. *Proceedings of the Nutrition Society of New Zealand*, 9, 6-17.
- Davies, R. (1989). Breast feeding: Are Dunedin mothers choosing to continue? Nursing Praxis in New Zealand, 5(1), 4-10.
- Davis, M. (1998). Review of the evidence for an association between infant feeding and childhood cancer. *International Journal of Cancer*, 78, 29-33.
- De Carvalho, M., Robertson, S., Friedman, A., & Klaus, M. (1983). Effect of frequent breast-feeding on early milk production and infant weight gain. *Pediatrics* 72, 307-311.
- Dennis, C. (1999). Theoretical underpinnings of breastfeeding confidence: A self-efficacy framework. *Journal of Human Lactation*, *15*(3), 195-201.
- Dennis, C. (2002). Breastfeeding peer support: Maternal and volunteer perceptions from a randomized controlled trial. *Birth*, *29*(3), 169-176.
- Dewey, K., Cohen, R., Brown, K., & Landa Rivera, L. (2001). Effects of exclusive breastfeeding for four versus six months on maternal nutritional status and infant motor development: Results of two randomized trials in Honduras. *The Journal of Nutrition*, 131, 262-267.
- Dignam, D. (1995). Understanding intimacy as experienced by breastfeeding women. *Health Care For Women International*, *16*, 477-485.
- Dignam, D. (1998). Breastfeeding, intimacy and reconceptualising the breast. In A. Beasley, & A. Trlin, (Eds.), *Breastfeeding in New Zealand: Practice*,

- problems and policy (pp. 75-93). Palmerston North: The Dunmore Press
- Dignam, D. (2001). 'Linking as One': An intimate breastfeeding moment.

 Unpublished doctoral dissertation, Massey University, Palmerston North.
- Dodd, J. (1999). Breastfeeding: A cross cultural perspective. *Baby Feeding News: Infant Nutrition Issues for Health Professionals*, 20, 4-5.
- Dow, D. (1999). *Maori health and government policy 1840-1940*. Wellington: Victoria University Press.
- Dreyfus, H. (1991). Being-in-the-world: A commentary of Heidegger's being and time. New Baskerville: The Massachusetts Institute of Technology Press.
- Duncan, B., Holberg, C., Wright, A., Martinez, F., & Taussig, L. (1993). Exclusive breast-feeding for at least four months protects against otitis media. *Pediatrics*, *91*, 867-872.
- Durie, M. (1998). Whaiora: Maori health development (2nd ed.). Auckland: Oxford University Press.
- Durie, M. (2001). *Mauri ora: The dynamics of Maori health*. Auckland: Oxford University Press.
- Earle, S. (2000). Why some women do not breast feed: Bottle feeding and fathers' role. *Midwifery*, *16*, 323-330.
- Earle, S. (2002). Factors affecting the initiation of breastfeeding: Implications for breastfeeding promotion. *Health Promotion International*, 17, 205-214.
- Ellis, D. (1992). The impact of agency policies and protocols on breastfeeding.

 NAACOG's: Clinical Issues in Perinatal & Women's Health Nursing 3,
 553-559.

- Ellison-Loschmann, L. (1997). *Maori women's experiences of breastfeeding*.

 Unpublished master of arts in midwifery dissertation, Victoria University of Wellington, Wellington.
- Ellison-Loschmann, L. (1998). The breastfeeding experience of four first-time Maori mothers. In A. Measley, & A. Trlin, (Eds.), *Breastfeeding in New Zealand: Practice, problems and policy* (pp. 37-74). Palmerston North: The Dunmore Printing Company Ltd.
- Essex, C., Smale, P., & Geddis, D. (1995). Breastfeeding rates in New Zealand in the first six months and the reasons for stopping. *New Zealand Medical Journal*, 8, 355-357.
- Fellow, C. (1994). Is breastfeeding a feminist issue? *New Zealand College of Midwives Journal, October*, 22-25.
- Ferguson, A. (1998). Bottle babies: A New Zealand guide to guilt-free bottle feeding. Auckland, New Zealand: Celebrity Books.
- Fergusson, D., Horwood, L., Shannon, F., & Taylor, B. (1978). Infant health and breast-feeding during the first 16 weeks of life. *Australian Paediatrics Journal*, *14*, 254-258.
- Field, P., & Morse, J. (1985). *Nursing Research: The application of qualitative approaches*. London: Croom Helm.
- Fjelland, R., & Gjengedal, E. (1994). A theoretical foundation for nursing as a science. In P. Benner (Ed.), *Interpretive phenomenology: Embodiment, caring, and ethics in health and illness* (pp. 3-25). London: Sage Publications.
- Flight, R., & Adam, D. (1986). Breastfeeding patterns in west Auckland in 1985. New Zealand Medical Journal, 99, 511-512.
- Ford, K., & Labbok, M. (1990). Who is breast-feeding? Implications of associated social and biomedical variables for research on the consequences of method of infant feeding. *American Journal of Clinical Nutrition*, 52, 451-456.

- Ford, R., Schluter, P., & Wild, C. (1996). Breastfeeding in Canterbury over three decades. *New Zealand Medical Journal*, 109(1029), 343-345.
- Ford, R., Wild, C., Mitchell, E., & Tuohy, P. (1995). Infant feeding patterns in Canterbury. *New Zealand Medical Journal*, 108, 59-61.
- Gadamer, H. (1994). *Heidegger's ways*. (J. Stanley, Trans.). Albany: State University of New York Press. (Original work published 1994).
- Gadamer, H.-G. (1989). *Truth and method* (2nd revised ed.). (J. Weinsheimer, & D. Marshall, Trans.). New York: The Continuum Publishing Company. (Original work published 1960).
- Gadamer, H. G. (1996). *The enigma of health*. (J. Gaiger, & N. Walker, Trans.). Stanford: Stanford University Press. (Original work published 1993).
- Gadow, S. (2000). Philosophy as falling: Aiming for grace. *Nursing Philosophy*, 1, 89-97.
- Galtry, J. (1997). Suckling and silence in the USA: The cost and benefits of breastfeeding. *Feminist Economics*, 3(3), 1-24.
- Galtry, J. (1998). Breastfeeding and paid employment: The experiences of Sweden, the United States and New Zealand. In A. Beasley, & A. Trlin, (Eds.), *Breastfeeding in New Zealand: Practice, problems and policy* (pp. 141-168). Palmerston North: The Dunmore Press.
- Geanellos, R. (1998). Hermeneutic philosophy. Part I: Implications of its use as methodology in interpretive nursing research. *Nursing Inquiry*, *5*, 154-163.
- Gigliotti, E. (1995). When women decide not to breastfeed. *American Journal of Maternal Child Nursing*, 20, 315-321.
- Gillman, M. (2002). Breast-feeding and obesity. *The Journal of Pediatrics*, 141, 749-750.

- Glover, J., & Sandilands, M. (1990). Supplementation of breastfeeding infants and weight loss in hospital. *Journal of Human Lactation*, 6(4), 163-166.
- Gordon, R. (1998). The role of La Leche League in the promotion and support of breastfeeding. In A. Beasley, & A. Trlin, (Eds.), *Breastfeeding in New Zealand: Practice, problems, and policy* (pp. 127-139). Palmerston North: The Dunmore Printing Company.
- Greer, G. (1999). The whole woman. London: Transworld Publishers.
- Guignon, C. (1983). *Heidegger and the problem of knowledge*. Indianapolis: Hackett Publishing Company.
- Gunn, T. (1984). The incidence of breast feeding and reasons for weaning. *New Zealand Medical Journal*, 97, 360-363.
- Hanson, L., Korotkova, M., Haversen, L., Mattsby-Baltzer, I., Hahn-Zoric, M., Strandvik, B., & Telemo, E. (2002). Breast-feeding, a complex support system for the offspring. *Pediatrics International*, 44, 347-352.
- Harrison, M., Morse, J., & Prowse, M. (1985). Successful breast feeding: The mother's dilemma. *Journal of Advanced Nursing*, *10*, 261-269.
- Hauck, Y., Langton, D., & Coyle, K. (2002). The path of determination: Exploring the lived experience of breastfeeding difficulties. *Breastfeeding Review*, 10(2), 50-62.
- Heath, A., Reeves Tuutle, C., Simons, M., Cleghorn, C., & Parnell, W. (2002). A longitudinal study of breastfeeding and weaning practices during the first year of life in Dunedin, New Zealand. *Journal of the American Dietetic* Association, 102, 937-943.
- Heidegger, M. (1962). *Being and time* (7th ed.). (J. Macquarrie, & E. Robinson, Trans.). Oxford: Blackwell Publishers. (Original work published 1927).
- Heikkila, M., & Saris, P. (2003). Inhibition of *Staphylococcus aureus* by the commensal bacteria of human milk. *Journal of Applied Microbiology*, 95, 471-483.

- Henderson, A., Stamp, G., & Pincombe, J. (2001). Postpartum positioning and attachment education for increasing breastfeeding: A randomized trial. *Birth*, 28, 236-242.
- Heritage, A. (2003). Baby Friendly Hospital Initiative a progress report. *Aroha*, 5(6), 10.
- Hill, P. (1992). Insufficient Milk Supply Syndrome. *NAACOG'S: Clinical Issues in Perinatal & Women's Health Nursing*, *3*, 605-612.
- Hill, P., & Humenick, S. (1989). Insufficient milk supply. *Image: Journal of Nursing Scholarship*, 21(3), 145-148.
- Hill, P., & Humenick, S. (1996). Development of the H & H Lactation Scale.

 Nursing Research, 45(3), 136-140.
- Hill, P., Humenick, S., & West, B. (1994). Concerns of breastfeeding mothers: The first six weeks postpartum. *Journal of Perinatal Education*, *3*(4), 47-58.
- Hillervik-Lindquist, C. (1992). Studies on perceived breast milk insufficiency:

 Relation to attitude and practice. *Journal of Biosocial Science*, *24*, 413-425.
- Hills-Bonczyk, S., Avery, M., Savik, K., Potter, S., & Duckett, L. (1993).

 Women's experiences with combining breast-feeding and employment. *Journal of Nurse-Midwifery*, 38(5), 257-266.
- Hills-Bonczyk, S., Tromiczak, K., Avery, M., Potter, S., Savik, K., & Duckett, L. (1994). Women's experiences with breastfeeding longer than 12 months. *Birth*, 21(4), 206-212.
- Hochhausen, A. (1991). *Guilt: The pervasive agenda in the motherhood test.*Unpublished doctoral dissertation, University of Wisconsin, Madison.
- Hood, L., Faed, J., Silva, P., & Buckfield, P. (1978). Breast feeding and some reasons for electing to wean the infant: A report from the Dunedin

- multidisciplinary child development study. *The New Zealand Medical Journal*, 88, 273-276.
- Huby, P. (1964). Socrates and Plato. In D. O'Connor, (Ed.), *A critical history of western philosophy* (pp. 14-35). New York: The Free Press.
- Humphreys, A., Thompson, N., & Miner, K. (1998). Intention to breastfeed in low-income pregnant women: The role of social support and previous experience. *Birth*, *25*, 169-174.
- Hunter, H. (1998). Good enough breastfeeding? *MIDIRS Midwifery Digest*, 8(4), 34-35.
- Husserl, E. (1970). *Logical investigations* (Vol. 1). (J. Findlay, Trans.). London: Routledge & Kegan Paul. (Original work published 1900).
- Husserl, E. (1967). *Cartesian meditations: An introduction to phenomenology.*(D. Cairns, Trans.). The Hague: Martinus Nijoff. (Original work published 1929).
- Inch, S., & Fisher, C. (2000). Breastfeeding: Early problems. *The Practicing Midwife*, *3*, 12-15.
- Infante-Rivard, C., Fortier, I., & Olsen, E. (2000). Markers of infection, breast-feeding and childhood acute lymphoblastic leukaemia. *British Journal of Cancer*, *83*, 1559-1564.
- Ingram, J., Johnson, D., & Greenwood, R. (2002). Breastfeeding in Bristol: Teaching good positioning, and support from fathers and families. *Midwifery*, 18, 81-101.
- Inwood, M. (1997). Heidegger. Oxford: Oxford University Press.
- Inwood, M. (1999). A Heidegger dictionary. Oxford: Blackwell.
- Ironside, P. (1997). *Preserving, reading, writing, thinking and dialogue:*Rethinking doctoral education in nursing. Unpublished doctoral dissertation, University of Wisconsin, Madison.

- Isabella, P., & Isabella, R. (1994). Correlates of successful breastfeeding: A study of social and personal factors. *Journal of Human Lactation, 10,* 257-264.
- Ivarsson, A., Hernell, O., Stenlund, H., & Ake Persson, L. (2002). Breast-feeding protects against celiac disease. *The American Journal of Clinical Nutrition*, 75, 914-921.
- Jones, C. (2001). Maternal transmission of infectious pathogens in breast milk. Journal of Paediatrics and Child Health, 37, 576-582.
- Kant, I. (1999). Critique of pure reason. (P. Guyer, & A. Wood, Trans.).
 Cambridge: Cambridge University Press. (Original work published 1781).
- Kedgley, S. (1996). *Mum's the word: The untold story of motherhood in New Zealand*. Auckland: Random House New Zealand Ltd.
- Keefe, M. (1986). Comparison of neonatal night time sleep-wake patterns in nursery versus rooming-in environments. *Nurse Research*, 36(3), 140-144.
- Kennedy, A. (2000). 'Doing what's best at the time': First time mothers learning to breastfeed away from home: A grounded theory study. Unpublished master of philosophy dissertation, Massey University, Palmerston North.
- Kenny, A. (1998). A brief history of western philosophy. Oxford: Blackwell Publishers.
- Kerr, A. (1981). Lower respiratory tract illness in Polynesian infants. *The New Zealand Medical Journal*, 93, 333-335.
- Kessler, L., Gielen, A., Diener-West, M., & Paign, D. (1995). The effect of a woman's significant other on her breastfeeding decision. *International Lactation Consultant Association*, 11(2), 103-108.
- Koch, T. (1994). Interpretive approaches in nursing research: The influence of Husserl and Heidegger. *Journal of Advance Nursing*, *21*, 827-836.

- Krasner, D. (1997). Carrying on despite the pain: Living with painful venous ulcers a Heideggerian hermeneutic analysis. Unpublished doctoral dissertation, University of Maryland, Baltimore.
- Kull, I., Wickman, M., Lilja, G., Nordvall, S., & Pershagen, G. (2002).
 Breastfeeding and allergic diseases in infants: A prospective birth cohort study. Archives of Diseases in Childhood, 87, 478-481.
- Kuriniji, N., & Shiono, P. (1991). Early formula supplementation of breastfeeding. *Pediatrics*, *88*, 745-750.
- Kyenkya-Isabirye, M., & Malgalheas, R. (1990). Teen mothers' support group role in the health care system. *International Journal of Genecology and Obstetrics*, *31*, 85-90.
- Labbok. (2000). What is the definition of breastfeeding. *Breastfeeding Abstracts*, 19(3), 19-21.
- Labbok, M., & Krasovec, K. (1990). Toward consistency in breastfeeding definitions. *Studies in Family Planning*, *21*, 226-230.
- Laufer, A. (1990). Breastfeeding: Toward resolution of the unsatisfying birth experience. *Journal of Nurse-Midwifery*, *35*(1), 42-45.
- Law, J. (2000). The politics of breastfeeding: Assessing risk, dividing labor. Signs: Journal of Women in Culture and Society, 25, 407-450.
- Lawrence, R., & Lawrence, R. (1999). *Breastfeeding: A guide for the medical profession.* (5th ed.). St. Louis: Mosby.
- Leder, D. (1984). Medicine and paradigms of embodiment. *The Journal of Medicine and Philosophy*, 9, 29-43.
- Leff, E., Gagne, M., & Jefferis, S. (1994). Maternal perceptions of successful breastfeeding. *Journal of Human Lactation*, *10*(2), 99-104.

- Leipert, B. (1996). The value of community health nursing: A phenomenological study of the perceptions of community health nurses. *Public Health Nursing*, *13*, 50-57.
- Leonard, V. (1994). A Heideggerian phenomenological perspective on the concept of person. In P. Benner, (Ed.), *Interpretive phenomenology: Embodiment, caring, and ethics in health and illness* (pp. 43-63). London: Sage Publications.
- Lethbridge, D., McClurg, V., Henrikson, M., & Wall, G. (1992). Validation of the nursing diagnosis of ineffective breastfeeding. *JOGNN: Journal of Obstetrics, Gynecologic and Neonatal Nursing*, 22, 57-63.
- Long, D. (2003). Breast sharing: Cross-feeding among Australian women. Health Sociology Review, 12, 103-110.
- Lothian, J. (1994). The decision to breastfeed. *Journal of Perinatal Education*, 3(4), 39-45.
- MacIntyre, A. (1964). Existentialism. In D. O'Connor, (Ed.), A critical history of western philosophy (pp. 509-529). New York: The Free Press.
- Maclean, H. (1990). Women's experience of breastfeeding. Toronto: University of Toronto Press.
- MacLennan, C. (2001). The intimacy of breast feeding. *New Zealand Nursing Review, August*, 11-13.
- Magee, B. (Ed.). (1987). The great philosophers: An introduction to western philosophy. New York: Oxford University Press.
- Makhlouf Obermeyer, C., & Castle, S. (1997). Back to nature? Historical and cross-cultural perspectives on barriers to optimal breastfeeding. *Medical Anthropology*, 17, 39-63.
- Markus, R. (1964). Augustine. In D. O'Connor, (Ed.), *A critical history of western philosophy* (pp. 79-97). New York: The Free Press.

- Matthews, M. (1991). Mothers' satisfaction with their neonates' breastfeeding behaviours. *JOGNN: Journal of Obstetrics, Gynecologic and Neonatal Nursing*, 20, 49-55.
- Matulonis Williams, K., & Morse, J. (1989). Weaning patterns of first-time mothers. *Maternal Child Nursing*, *14*(3), 188-192.
- Maushart, S. (1999). The mask of motherhood: How becoming a mother changes everything and why we pretend it doesn't. New York: The New Press.
- McBride, K. (1999). Breastfeeding women's perceptions of the importance of infant care routines. Unpublished master of nursing dissertation, Massey University, Albany.
- McIntosh, J. (1985). Barriers to breastfeeding: Choice of feeding method in a sample of working class primiparae. *Midwifery*, 1, 213-224.
- McLeod, D., Pullon, S., & Basire, K. (1998). Factors affecting baby feeding:
 Reflections and perceptions of Hutt Valley mothers. In A. Beasley, & A.
 Trlin, (Eds.), Breastfeeding in New Zealand: Practice, problems and policy (pp. 15-35). Palmerston North: Dunmore Press.
- Meighan, M., Davis, M., Thomas, S., & Droppleman, P. (1999). Living with postpartum depression: The father's experience. *MCN: American Journal of Maternal Child Nursing*, *24*(4), 202-208.
- Mein Smith, P. (1986). *Maternity in dispute: New Zealand 1920-1939*. Wellington: Government Printing Office.
- Merleau-Ponty, M. (1981). *Phenomenology of perception*. (C. Smith, Trans.). London: Routledge & Kegan Paul. (Original work published 1945).
- Meyer, D., & de Oliveira, D. (2003). Breastfeeding policies and the production of motherhood: A historical-cultural approach. *Nursing Inquiry, 10*, 11-18.
- Ministry of Health. (1997). Food and nutrition guidelines for healthy children: A background paper. Wellington: Ministry of Health.

- Ministry of Health. (1998). Progress on health outcome targets: The state of public health in New Zealand. Wellington: Ministry of Health.
- Ministry of Health. (1999a). Food and nutrition guidelines for healthy infants and toddlers (Aged 0-2): A Background Paper. Wellington: Ministry of Health.
- Ministry of Health. (1999b). *Progress on health outcome targets*. Wellington: Ministry of Health.
- Ministry of Health. (2001). *DHB toolkit: To improve nutrition*. Wellington: Ministry of Health.
- Ministry of Health. (2002). *Breastfeeding: A guide to action*. Wellington: Ministry of Health.
- Morse, J. (1985). The cultural context of infant feeding in Fiji. In L. Marshall, (Ed.), *Infant care and feeding in the South Pacific: Food and nutrition in history and anthropology* (Vol. 3, pp. 255-268). New York: Gordon and Breach.
- Morse, J. (1989). "Euch, those are for your husband!": Examination of cultural values and assumptions associated with breastfeeding. *Health Care for Women International*, 11, 223-232.
- Morse, J., Bottorff, J., & Boman, J. (1989). Patterns of breastfeeding and work: The Canadian experience. *Canadian Journal of Public Health, 80*(3), 182-188.
- Morse, J., Ewing, G., Gamble, D., & Donahue, P. (1992). The effect of maternal fluid intake on breast milk supply: A pilot study. *Canadian Journal of Public Health*, 83(3), 213-216.
- Morse, J., & Harrison, M. (1987). Social coercion for weaning. *Journal of Nurse-Midwife*, 32(4), 205-210.
- Mortensen, E., Michaelsen, K., Sanders, S., & Reinisch, M. (2002). The association between duration of breastfeeding and adult intelligence. *Journal of the American Medical Association*, 287, 2365-2371.

- Moustakas, C. (1994). Phenomenological research methods. London: Sage.
- Mozingo, J., Davis, M., Droppleman, P., & Merideth, A. (2000). "It wasn't working": Women's experiences with short-term breastfeeding. *MCN: Maternal Child Nursing*, *25*(3), 120-126.
- Msuya, J., Harding, W., Robinson, M., & McKenzie-Parnell, J. (1990). The extent of breast feeding in Dunedin 1974-1983. *New Zealand Medical Journal*, 103, 68-70.
- Munhall, P., & Boyd, O. (Eds.). (1993). *Nursing research: A qualitative perspective*. New York: National League for Nursing Press.
- Murphy, E. (1999). 'Breast is best': Infant feeding decisions and maternal deviance. *Sociology of Health and Illness, 21*, 187-208.
- Muscari, M. Advanced pediatric clinical assessment: Skills and procedures.

 New York: Lippincott Williams & Wilkins.
- Needham, C. (1999). Finding the ethical standard of medical science in the age of the sciences. *Journal of Evaluation in Clinical Practice*, *5*, 41-46.
- Newman, J. (1990). Breastfeeding problems associated with the early introduction of bottles and pacifiers. *Journal of Human Lactation*, 6(2), 59-63.
- New Zealand Breastfeeding Authority. (2000). New Zealand Breastfeeding Authority newsletter. New Zealand Breastfeeding Authority, 1, 1-18.
- New Zealand College of Midwives. (1992). *Protecting, promoting, and*supporting breastfeeding: The breastfeeding handbook. Dunedin: Right
 Words New Zealand.
- Nidditch, P. (1964). Spinoza. In D. O'Connor, (Ed.), *A critical history of western philosophy* (pp. 187-203). New York: The Free Press.
- O'Connor, D. (Ed.). (1964). *A critical history of western philosophy*. London: Collier Macmillan Publishers.

- Olafson, F. (1987). *Heidegger and the philosophy of mind*. New Haven: Yale University Press.
- Omery, A. (1983). Phenomenology: A method for nursing research. *Nursing Science*, *5*(2), 49-63.
- Ozturk Ertem, I., Kaynak, G., Kaynak, C., Ulukol, B., & Baskan Gulnar, S. (2001). Attitudes and practices of breastfeeding mothers regarding fasting in Ramadan. *Child: Care, Health and Development, 27*, 545-554.
- Page, T., Lockwood, C., & Guest, K. (2003). Management of nipple pain and/or trauma associated with breastfeeding. *Joanna Briggs Institute Reports*, 1(4), 127-147.
- Paine, P., & Dorea, J. (2001). Gender role attitudes and other determinants of breast feeding intentions in Brazilian women. *Child: Care, Health and Development*, 27, 61-72.
- Paley, J. (2000). Heidegger and the ethics of care. *Nursing Philosophy*, 1, 64-75.
- Palmer, G. (1993). The politics of breastfeeding. London: Pandora Press.
- Palmer, R. (1969). Hermeneutics: Interpretation theory in Schleiermacher,

 Dilthey, Heidegger and Gadamer. Evanston: Northwestern University

 Press.
- Parahoo, K. (1997). *Nursing research: Principles, process and issues*. Houndmills: Macmillan.
- Parker, J., & Gibbs, M. (1998). Truth, virtue and beauty: Midwifery and philosophy. *Nursing Inquiry*, *5*, 146-153.
- Paton, L., Alexandra, J., Nowson, C., Margerison, C., Frame, M., Kaymakei, B., & Wark, J. (2003). Pregnancy and lactation have no long-term deleterious effect on measure of bone mineral in healthy women: A twin study. *The American Journal of Clinical Nutrition*, 77, 707-714.

- Perry, P., & Trlin, A. (1985). Socio-demographic factors in relation to breast feeding duration among Manawatu women. *New Zealand Population Review*, *11*, 94-110.
- Plager, K. (1994). Hermeneutic phenomenology: A methodology for family health and health promotion study in nursing. In P. Benner, (Ed.), *Interpretive phenomenology: Embodiment, caring, and ethics in health and illness* (pp. 65-83). London: Sage Publications.
- Plancoulaine, S., Charles, M., Lafay, L., Tauber, M., Thibult, N., Borys, J., & Eschwege, E. (2000). Infant-feeding patterns are related to blood cholesterol concentration in prepubertal children aged 5-11 years: The Fleurbaiz-Laventie Ville Sante study. *European Journal of Clinical Nutrition*, *54*, 114-119.
- Polatti, F., Capuzzo, E., Viazzo, F., Colleoni, R., & Klersy, C. (1999). Bone mineral changes during and after lactation. *Obstetrics and Gynaecology*, 94, 52-56.
- Polit, D., & Hungler, B. (1995). *Nursing research: Principles and methods*. Philadelphia: Lippincott.
- Pool, I. (1991). *Te lwi Maori: A New Zealand population past, present and projected.* Auckland: Auckland University Press.
- Pool, I., & Bedford, R. (1996). *Macro social change in New Zealand: Historical and international context*. Unpublished Discussion Paper, University of Waikato, Hamilton, New Zealand.
- Pownall, B. (2000). Baby friendly hospital initiative: A survey study for the fulfilment of master of arts. Unpublished master of arts dissertation, Massey University, Albany, New Zealand.
- Priest, S. (2000). Merleau-Ponty's concept of the body-subject. *Nursing Philosophy*, 1, 173-174.

- Public Health Commission. (1994). Sudden infant death syndrome: The Public Health Commission's advice to the Minister of Health 1993-1994.

 Wellington: Public Health Commission.
- Public Health Commission. (1995a). Food and nutrition guidelines for healthy breastfeeding women: A background paper. Wellington: Public Health Commission.
- Public Health Commission. (1995b). Guidelines for healthy infants and toddlers (0-2 years). Wellington: Public Health Commission.
- Ramsden, I. (2001). Improving practice through research. *Kai Tiaki Nursing New Zealand, February*, 23-28.
- Ravelli, A., van der Meulen, J., Osmond, C., Barker, D., & Bleker, O. (2000).

 Infant feeding and adult glucose tolerance, lipid profile, blood pressure and obesity. *Archives of Disease in Childhood*, 82, 248-252.
- Reinharz, S. (1983). Phenomenology as a dynamic process. *Phenomenology* and *Pedagogy*, 1, 77-79.
- Renfrew, M., Woolridge, M., & Ross-McGill, H. (2000). Enabling women to breastfeed: A review of practices which promote or inhibit breastfeeding with evidence-based guidance for practice. London: The Stationery Office.
- Rentschler, D. (1991). Correlates of successful breastfeeding. *IMAGE: Journal of Nursing Scholarship*, 23, 151-154.
- Righard, L. (1998). Are breastfeeding problems related to incorrect breastfeeding technique and the use of pacifiers and bottles? *Birth*, *25*, 40-44.
- Rimene, C., Hassan, C., & Broughton, J. (1998). *Ukaipo: The place of nurturing: Maori women and childbirth.* Dunedin: The Ngai Tahu Maori Health

 Research Unit, Department of Preventative and Social Medicine.

- Riordan, J., & Auerbach, K. (1993). *Breastfeeding and human lactation*. Boston: Jones and Bartlett.
- Roberts, D. (1980). Growth of breast fed and bottle fed infants. *New Zealand Medical Journal*, 92, 45-46.
- Robertson-Malt, S. (1999). Listening to them and reading me: A hermeneutic approach to understanding the experience of illness. *Journal of Advanced Nursing*, 29, 290-297.
- Rosenblatt, K., & Thomas, D. (1993). Lactation and the risk of epithelial ovarian cancer: The WHO collaborative study of neoplasia and steroid contraceptives. *International Journal of Epidemiology*, 22, 192-197.
- Royal New Zealand Plunket Society. (1995). *Percentage of breastfed babies* 1922-1995. Dunedin: Royal New Zealand Plunket Society.
- Royal New Zealand Plunket Society. (1996). *Breastfeeding policy: January* 1996. Dunedin: Royal New Zealand Plunket Society.
- Rubenstein, C. (1998). *The sacrificial mother: Escaping the trap of self-denial.*New York: Hyperion.
- Ruowei, L., & Grummer-Strawn, L. (2002). Racial and ethnic disparities in breastfeeding among United States infants: Third national health and nutrition examination survey, 1988-1994. *Birth*, 29, 251-257.
- Russell, B. (1996). History of western philosophy and its connection with political and social circumstances from the earliest times to the present day (2nd ed.). London: Routledge.
- Rutty, J. (1998). The nature of philosophy of science, theory and knowledge relating to nursing and professionalism. *Journal of Advanced Nursing*, 28, 243-250.
- Ryan, A., & Martinez, G. (1989). Breast-feeding and the working mother: A profile. *Pediatrics*, 83, 524-531.

- Ryan, K. (1999a). Women's narratives of infant feeding: The politics of knowledges and practices in post W.W.II New Zealand. Unpublished doctoral dissertation, University of Otago, Dunedin.
- Ryan, K. (1999b). Women's narrative of infant feeding: The politics of knowledges and practices in post W.W.II New Zealand. Research report of a study undertaken for a doctoral dissertation, University of Otago, Dunedin, New Zealand.
- Ryan, K., & Grace, V. (2001). Medicalization and women's knowledge: The construction of understandings of infant feeding experiences in post-WW II New Zealand. *Health Care for Women International*, 22, 483-500.
- Salmon, M. (1994). The cultural significance of breastfeeding and infant care in early modern England and America. *Journal of Social History, Winter*, 249-269.
- Sarney, E. (2003/2004). Breastfeeding blues. Little Treasures, 100, 16-23.
- Schafer, E., Vogel, M., Viegas, S., & Hausafus, C. (1998). Volunteer peer counsellors increase breastfeeding duration among rural low-income women. *Birth*, *25*(2), 101-106.
- Schmied, V., & Barclay, L. (1999). Connection and pleasure, disruption and distress: Women's experience of breastfeeding. *Journal of Human Lactation*, *15*, 325-334.
- Schmied, V., & Lupton, D. (2001). Blurring the boundaries: Breastfeeding and maternal subjectivity. Sociology of Health and Illness, 23, 234-250.
- Scott, J., Landers, M., Hughes, R., & Binns, C. (2001). Psychosocial factors associated with the abandonment of breastfeeding prior to hospital discharge. *Journal of Human Lactation*, 17, 24-30.
- Sears, M., Greene, J., Willan, A., Taylor, D., Flannery, E., Cowan, J., Herbison, P., & Poulton, R. (2002). Long-term relation between breastfeeding and development of atopy and asthma in children and young adults: A longitudinal study. *The Lancet*, 360, 901-907.

- Sheeshka, J., Potter, B., Norrie, E., Valaitis, R., Adams, G., & Kuczynski, L. (2001). Women's experiences of breastfeeding in public places. *Journal of Human Lactation*, 17, 31-38.
- Shepherd, C., Power, K., & Carter, H. (2000). Examining the correspondence of breastfeeding and bottle-feeding couples' infant feeding attitudes. *Journal of Advanced Nursing*, 31, 651-660.
- Silfverdal, S., Bodin, L., Ulanova, M., Hahn-Zoric, M., Hanson, L., & Olcen, P. (2002). Long-term enhancement of the lgG2 antibody response to Haemophilus influenzae type B by breastfeeding. *Pediatric Infectious Disease Journal*, 21, 816-21.
- Simondon, K., Delaunay, V., Diallo, A., Elguero, E., & Simondon, F. (2003). Lactational amenorrhoea is associated with child age at time of introduction of complementary food: A prospective cohort study in rural Senegal, West Africa. American Journal of Clinical Nutrition, 78, 154-161.
- Slavin, S., & Harvey, D. (1981). Unlimited suckling time improves breastfeeding. *The Lancet, i,* 392-393.
- Smythe, E. (1998). 'Being safe' in childbirth: A hermeneutic interpretation of the narratives of women and practitioners. Unpublished doctoral dissertation, Massey University, Palmerston North.
- Sorrell, T. (1987). *Descartes: A very short introduction*. Oxford: Oxford University Press.
- Spiegelberg, H. (1969). *The phenomenological movement: A historical introduction*. (2nd ed., Vol. 1). The Hague: Martinus Nijhoff.
- Stufkins, J. (2000). Baby Friendly Hospital Initiative. *Breastfeeding Communiqué*, *11*, 13-15.
- Swearingen, J. (1977). Reflexivity in Tristram Shandy: An essay in phenomenological criticism. New Haven: Yale University Press.

- Swigart, J. (1998). *The myth of the perfect mother: Parenting without guilt.*Lincolnwood: Contemporary Books.
- Tarkka, M., Paunonen, M., & Laippala, P. (1998). What contributes to breastfeeding success after childbirth in a maternity ward in Finland? Birth, 25(3), 175-181.
- Thomson, J. (1964). Berkeley. In D. O'Connor, (Ed.), *A critical history of western philosophy* (pp. 236-252). New York: The Free Press.
- Thurtle, V. (1997). New Zealand mothers and attitudes towards breastfeeding. *Health Visitor*, 70, 263-264.
- Toschke, A., Vignerova, J., Lhotska, L., Osancova, K., Koletzko, B., & von Kries, R. (2002). Overweight and obesity in 6 to 14 year-old Czech children in 1991: Protective effect of breastfeeding. *Journal of Pediatrics*, *141*, 764-769.
- Traina, C. (2000). Maternal experience and the boundaries of Christian sexual ethics. Signs: Journal of Women in Culture and Society, 25, 369-405.
- Trlin, A., & Perry, P. (1982). Breast feeding trends among Manawatu women: A cohort approach. *New Zealand Medical Journal*, *95*, 573-577.
- UNICEF. (2002). Current status of Baby-Friendly Hospital Initiative. Retrieved September 9, 2003, from www.unicef.org/programme/breastfeeding/assets/statusbfhi.pdf
- Van Esterik, P. (1989). The politics of breastfeeding: An advocacy perspective. In P. Stuart-Macadam, & K. Dettwyler, (Eds.), *Breastfeeding bicultural perspectives* (pp. 145-165). New York: Aldine de Gruyter.
- Van Esterik, P. (1994). Lessons from our lives: Breastfeeding in a personal context. *Journal of Human Lactation*, 10(2), 71-74.
- van Odijk, J., Kull, I., Borres, M., Brandtzaeg, P., Edberg, U., Hanson, L., Host, A., Kuitunen, M., Olsen, S., Skerfving, S., Sundell, J., & Wille, S. (2003). Breastfeeding and allergic disease: A multidisciplinary review of the

- literature (1966-2001) on the mode of early feeding in infancy and its impact on later atopic manifestations. *Allergy*, *58*, 833-839.
- Vares, T. (1992). Feminist women talk about breastfeeding. *Women's Studies Journal*, 8(2), 25-41.
- Vogel, A., & Mitchell, E. (1998a). The establishment and duration of breastfeeding. Part one: Hospital influences. *Breastfeeding Review*, 6, 5-9.
- Vogel, A., & Mitchell, E. (1998b). The establishment and duration of breastfeeding. Part two: Community influences. *Breastfeeding Review*, 6, 11-16.
- Vogel, A., & Mitchell, E. (1998c). The Baby Friendly Hospital Initiative: Evidence and implementation. In A. Beasley, & A. Trlin (Eds.), *Breastfeeding in New Zealand: Practice, problems and policy* (pp. 169-192). Palmerston North: The Dunmore Press Limited.
- Wagner, C., Hulsey, T., Southgate, M., & Annibale, D. (2002). Breastfeeding rates at an urban medical university after initiation of an educational program. *Southern Medical Journal*, *95*, 909-913.
- Walters, A. J. (1995). The phenomenological movement: Implications for nursing research. *Journal of Advanced Nursing*, 22, 791-799.
- Waring, M. (2000). The economics of breastfeeding. Paper presented at the Australian Lactation Consultants' Association Biennial Conference, October 13, 2000. Adelaide, Australia.
- Watson, K. (2001). Culture, breastfeeding, and bottles. Vision, 7(13), 19-23.
- World Health Organisation, & UNICEF. (1981). The WHO/UNICEF international code of marketing of breastmilk substitutes. Geneva, Switzerland: WHO/UNICEF.

- World Health Organisation, & UNICEF. (1989). *Protecting, promotion and supporting breast-feeding: The special role of maternity services*.

 Geneva: World Health Organisation.
- World Health Organisation, & UNICEF. (1990). The Innocenti Declaration:

 Breastfeeding in the 1990s: A Global Treatise. Geneva: World Health
 Organisation.
- World Health Organisation. (2002). *Infant and young child nutrition: Global strategy on infant and young child nutrition*. Geneva: World Health Organisation.
- Williams, I. (2000). The WHO Code 19 years on. A quick look at the rest of the world. Nursing Mothers' Association of Australia. Retrieved December 22, 2000, from http://home.vicnet.net.au/nmaa/who.html
- Williams, N. (1997). Maternal psychological issues in the experience of breastfeeding. *Journal of Human Lactation*, *13*, 57-60.
- Williams, P., Innis, S., & Vogel, A. (1996). Breastfeeding and weaning practices in Vancouver. *Canadian Journal of Public Health*, 87(4), 231-236.
- Wood, P., & Giddings, L. (2001). Phenomenology telling a story: An interview with Liz Smythe. *Nursing Praxis in New Zealand*, *17*(3), 15-26.
- Yalom, M. (1997). A history of the breast. New York: Ballantine.
- Young, T., Martens, P., Taback, S., Sellers, E., Dean, H., Cheang, M., & Flett, B. (2002). Type 2 diabetes mellitus in children: Prenatal and early infancy risk factors among Native Canadians. Archives of Pediatric and Adolescent Medicine, 156, 651-655.

Appendix I

Information Sheet

My name is Karen McBride. I am a Registered Nurse who is a doctoral student at Massey University. I would like to invite you to be a part of a research project that explores what it is like for women in New Zealand to breastfeed.

The benefits of breastfeeding are well known and the World Health Organisation says that all infants should be breastfed for the first four to six months.

However, women are choosing to stop breastfeeding their babies before they are four to six months old.

This study is being done in order to gain an understanding of what it is like for women to breastfeed infants in this country. It is hoped that this study will help health professionals who work with new mothers to better understand what breastfeeding is like for women here in New Zealand.

Even if you are not currently breastfeeding or chose to not continue to breastfeed I am interested in your personal story. You have been invited to join this study because you have had previous experience with breastfeeding in New Zealand.

The study will consist of one audio-taped interview. If you do not want the interview taped I can take notes on what we talk about instead. Another way of telling your story may be to write down your experience of breastfeeding.

The interview will last no longer than 90 minutes. The interview will take place at a time and place that you choose. The interview will be one on one. I will be the only interviewer. During the interview you will be asked to tell your story about breastfeeding in New Zealand. At the interview if you have any item(s) (for example a poem or a picture) you wish to bring that you feel helps to describe what breastfeeding is like for you, bring it with you to the interview. The goal is to describe your experience as fully as possible.

During the meeting with the researcher you will have the opportunity to select a name, other than your true name, by which you will be known while you take part in the study.

I will be talking to women about their experience of breastfeeding over the next year. After this year is over I will start analysing the interviews. The analysis of the transcripts will take some time and will continue over approximately three years.

You might wonder if there are any risks for you being involved in this research. Your experience of breastfeeding may not have been a positive one, and you may find it hard to talk about. It may also be that sharing your story, good or bad, may be helpful to you and other women who have had similar experiences. You may gain little personally from participating in this study, however, it is hoped that this study will help women in the whole of New Zealand.

A trained typist will transcribe the audiotapes produced from the interviews. The typist will sign a confidentiality agreement. These audiotapes will then be kept in locked storage at the School of Health Sciences at Massey University – Albany, in a different location to the consent form signed by you. Unless you want the tape of your interview returned, all tapes will be destroyed at the finish of this research project.

All transcripts will be identified by number codes, or by the false name you have chosen. The only other people who will have access to these items will be the research supervisors. No research participants' names will appear on any material produced as a result of this research without the permission of the individual involved. I want to let you know that every step will be taken to ensure your identity is not disclosed.

The transcripts of all the interviews will be returned to you so you can confirm that the transcripts are a true copy of our talk together. After you have read the transcript you can ask for any information to be changed or withdrawn. After you have approved the transcript, the analysis of the transcripts will start and your story will become a part of many stories.

Massey University Human Ethics Committee and The Royal New Zealand Plunket Society of New Zealand Ethics Committee have approved this research. If you have concerns about ethical conduct involved with the research project please contact either of the following agencies:

The Chairman

Health Advocates Trust

Massey University Human

97 Manukau Road

Ethics Committee

Epsom

Massey University - Albany

Phone: (09) 623 5798

Private Bag 102904

North Shore Mail Centre

Phone: (09) 443 9700.

Feel free to ask any questions you might have about the project before, during and, after the interview process. Taking part in this research is entirely voluntary.

The supervisors for this research are:

Dr. Gillian White

Dr. Cheryl Benn

Albany Group Co-ordinator

Associate Professor

Midwifery Programmes Co-ordinator

School of Health Sciences

School of Health Sciences

Massey University,

Massey University, Albany

Palmerston North

Yours faithfully,

Karen McBride

Appendix II

Consent Form

- I have read and understand the information sheet provided on the research to be conducted.
- I agree to participate in this study exploring women's experience of breastfeeding.
- I am aware that I am entitled to ask any questions I might have about the research at any time.
- I understand I have the right to withdraw from the study, until such time that
 the analysis of the interview has commenced, without providing any reason
 for this action.
- I understand that all precautions will be taken to secure my anonymity. No personal details will at any time appear on the interview transcripts. This form and any audiotapes will be kept in separate locations under lock and key in the locked archives at the School of Health Science at Massey University Albany. I have the right to request the audiotape of my interview at the conclusion of the project. If I do not request the audiotape it will be destroyed upon the completion of this project. This research project should be completed within the next four years.
- I understand that unidentifiable transcripts may be discussed with other health
 - care professionals so a more full interpretation of the text can be developed.
- I understand that I have the right to have a copy of my interview transcripts.
- All information about my identity will be destroyed upon the completion of the research.

•	I understand that a copy of the research report will be made available to me,			
	at my request, upon its completion.	The final p	roject report will be made	
	available to me through contact with	the resear	cher.	
Sig	nature:		Date:	
Ad	ldress:		Telephone:	

Appendix III

Typist's Confidentiality Contract

1
Have accepted the job of transcribing the taped interviews of the research data collected by Karen McBride as a part of her doctoral research at Massey University.
I understand that the information gathered for the research is confidential.
I will undertake all necessary steps to ensure any material pertaining to, or concerning the research data will be treated as confidential. Only I will hear the audiotapes. The transcriptions will be done in private and stored in a safe place. Whilst in my possession all transcripts and audiotapes will be kept in a locked storage facility. I will be the only person with access keys to the storage facility. No copies of any transcripts will be retained, either on disk, paper, or hard drive, by myself after the transcriptions are complete.
Signed: Date: